Performance Limits and Design Issues in Wireless Networks

Thesis by

Amir Farajidana

In Partial Fulfillment of the Requirements

for the Degree of

Doctor of Philosophy



California Institute of Technology Pasadena, California

2007

(Defended September 5, 2006)

© 2007

Amir Farajidana All Rights Reserved To my family Baba, Maman, Sara, and Helia

Acknowledgements

I thank my advisor, Professor Babak Hassibi, for his guidance and support during the past five years. His understanding, flexibility, and vision have been truly inspiring. Professor Hassibi allowed me to work independently and explore different ideas, but at the same time he was always available for discussing problems and providing me with valuable insights.

I would also like to thank Professors Robert J. McEliece, P.P. Vaidyanathan, Shuki Bruck, and Tracey Ho for being on my committee and for providing me with valuable suggestions during my study.

My appreciation goes to Professor Amin Shokrollahi at Ecole Polytechnique Fédérale de Lausanne (EPFL) for providing me with the opportunity of working in his lab for the summer of 2005. I enjoyed our technical conversations during this period very much.

My warmest thanks go to my colleagues and my friends at Caltech and at the ALGO lab at EPFL. In particular, I would like to convey my gratitude to my groupmates Ali Vakili, Chaitanya Kumar Rao, Daniel Thai, Devdutt Marathe, Frédérique Oggier, Haris Vikalo, Maralle Fakhereddin, Masoud Sharif, Michela Muñoz-Fernández, Mihailo Stojnic, Radhika Gowaikar, Sormeh Shadbakht, Tareq Al-Naffouri, Vijay Gupta, Weiyu Xu, and Yindi Jing; and to my lab-mates, Cédric Florens, Edwin Soedarmadji, Jeremy Thorpe, Mostafa El-Khamy, Ravi Palanki, and Sarah Fogal. I will deeply miss the long nontechnical and technical conversations in the lab during the past five years with Chaitanya, Radhika, Masoud, and Vijay. My gratitude goes to Mehdi Molkaraie, Payam Pakzad, Mahdi Cheraghchi, and Soheil Mohajer for the great times that I spent with them in Lausanne.

I am deeply grateful to Shirley Beatty for her professional assistance on administrative matters and her kindness and care.

I would like to thank my collaborators, Radhika Gowaikar, Vijay Gupta, Masoud Sharif, Ali Vakili, Ravi Palanki, and Professors Michelle Effros and João Hespanha, without whom much of this thesis was not possible.

I am indebted to many of my friends for their constant support and the joyful moments that have brought to my life. In particular, my special thanks go to Behnam and Hossein. Behnam and Hossein have been real friends and great support from the moment I arrived at United States. I am indebted to both of them for their kindness. I would like to thank my friends Siamak and Maryam. Siamak's abstract and thorough perspective on problems and Maryam's enthusiasm and energy have always been inspiring for me.

And at the end, I would like to give my deepest thanks to my parents, my sister, and my wife for their unconditional love, support, and their belief in me. Certainly, this thesis would have not been possible without their support. I am grateful to my parents, Ahmad Faraji Dana and Maryam Pilehvari, for their support and their sacrifice that has helped me reach to this stage. My father's perfectionism and seek for the truth and my mother's energy and encouragement has been always a great motivation for me. I thank my sister, Sara, for her unlimited kindness. Helia, the love of my life, and my true friend, has brought ultimate joy to my life and has supported me in every step of my study. I thank her for being full of energy and surprises, for sharing her unique and delicate view of the world with me. I would also like to thank my late grandmother, my first and my coolest teacher ever, Zahra Hodjat. I dedicate this thesis to my family as an inadequate but sincere expression of appreciation and love.

Abstract

The increasing utilization of networks, especially wireless networks, for different applications and in different aspects of modern life, has directed a great deal of attention towards the analysis and optimal design of networks. Distinguishing features of the wireless environment and the distributed nature of the network setup have raised many important challenges in finding the performance limits of different tasks such as communication, control, and computation over networks. There are also many design issues concerning the complexity and the robustness of wireless systems that should be addressed for a thorough understanding and an efficient operation of wireless networked systems. This thesis deals with a few of the challenges associated with the fundamental performance limits and optimal design of wireless networks.

In the first part, we analyze performance limits of two applications for a special class of wireless networks called wireless erasure networks. These networks incorporate some of the essential features of the wireless environment. We look at the performance limits of two applications over these networks. The first application is data transmission with two different traffic patterns, namely multicast and broadcast. The capacity region and the optimal coding scheme for the multicast scenario are found, and outer and inner bounds on the capacity region for the broadcast scenario are provided. The second application considered in this thesis is estimation and control of a dynamical process at a remote location connected through a wireless erasure network to a sensor observing the process. In this case, we characterize the minimum steady-state error and its dependency on the parameters of the network. The final problem considered in the first part of the thesis concerns power consumption (as a performance measure) in wireless networks. We propose and analyze a simple scheme based on the idea of distributed beamforming that saves us in terms of power consumption for dense sensor and ad-hoc networks. We quantify this gain compared to the case when nodes have isolated communications without participating in the network.

The second part of the thesis deals with two design issues in the downlink of cellular wireless networks. The first issue is related to quality of service provisioning in the downlink scenario. We investigate the problem of differentiated rate scheduling in which different users demand different sets of rates. We obtain explicit and practical scheduling schemes to achieve the rate constraints and at the same time maximize the throughput. These schemes are based on the idea of opportunistic beamforming, are simple, and require little amount of feedback to the transmitter. We further show that the throughput loss due to imposing the rate constraints is negligible for large systems.

The next issue considered in this thesis is the robustness of the capacity region of multiple antenna Gaussian broadcast channels to the channel estimation error at the transmitter and the users. These channels are mathematical models for the downlink of cellular systems. We provide an inner bound on the capacity region of these channels and show that this inner bound is equivalent to the capacity region of a dual multiple access channel with a noise covariance that depends on the transmit powers. This duality is explored to show the effect of the estimation error on the sum-rate for a large number of users and in the large power regime. Finally, a training-based scheme for the block fading multiple antenna broadcast channels is proposed.

Thesis Supervisor: Babak Hassibi

Title: Associate Professor of Electrical Engineering

Contents

	Dedication				
	Acknowledgements				
Abstract					
	1 Int	roduction	1		
	1.1	Sources of Challenges in Wireless Networks	2		
	1.2	Performance Limits In Wireless Networks	6		
	1.3	Design Issues in Wireless Cellular Networks	7		
	1.4	Contributions of the Thesis	9		
	2 Caj	pacity of Wireless Erasure Networks	15		
	2.1	Introduction	15		
	2.2	Preliminaries	19		
		2.2.1 Notation	19		
		2.2.2 Definitions for Directed Graphs	20		
	2.3	Network Model	23		
	2.4	Problem Statement	28		
	2.5	Main Results	30		
	2.6	Proof of Theorems	33		
		2.6.1 Proof of Theorems 1 and 2	33		
		2.6.1.1 Converse	33		

		2.6.1.2 Achievability \ldots \ldots \ldots \ldots \ldots	34
		2.6.1.3 Probability of Error	36
		2.6.2 Proof of Theorem 3	42
	2.7	Linear Encoding	42
		2.7.1 Achievability Result for Linear Encoding	44
	2.8	Discussion	46
		2.8.1 Packet Size and Cycles	46
		2.8.2 Side-information at Destination Nodes	46
		2.8.3 Achievable Rates Without Side-information	48
	2.9	Conclusion	56
3	Bro	dcast Problems over Wireless Networks	58
	3.1	Introduction	58
	3.2	Problem Statement	60
	3.3	Broadcast Problems Over WE Networks	62
		3.3.1 Time-sharing Scheme for WE networks	62
	3.4	Erasure Broadcast Channel	65
		3.4.1 Degraded (m, n) -Erasure Broadcast Channels	67
		3.4.2 Non-degraded (m, n) -Erasure Broadcast Channel	71
		3.4.3 Discussion	78
		3.4.3.1 Correlated Erasures	78
		3.4.3.2 Generalizing the Proof Technique	83
	3.5	Γighter Outer bounds for WE networks	85
	3.6	Conclusion	91
4	A F	actical Scheme for Wireless Networks	93
	4.1	Introduction	93
	4.2	Γwo Wireless Network Models	95

	4.3	Optim	izing Over Sub-networks Does Not Work	97
	4.4	A Poss	sible Set of Network Operations	102
	4.5	Proble	m Statement	104
	4.6	Determ	mining the Rate at a Node – $R_D(i)$	106
		4.6.1	Finding the Rate in Gaussian Wireless Networks	106
		4.6.2	Finding Rate in Erasure Wireless Networks	108
	4.7	Algori	thm to Find Optimum Policy	110
	4.8	Analys	sis of the Algorithm	111
		4.8.1	Proof of Optimality	112
	4.9	Examp	bles	115
		4.9.1	Multistage Erasure Relay Networks	115
		4.9.2	Multistage Gaussian Relay Networks	117
		4.9.3	Erasure Network with Four Relay Nodes	119
		4.9.4	Gaussian Network with Three Relay Nodes	119
		4.9.5	Gaussian Network with Four Relay Nodes	120
	4.10	A Dist	ributed Algorithm for the Optimal Policy	121
	4.11	Upperbounds On the Maximum Rate		
		4.11.1	Upperbound for Gaussian Networks	124
		4.11.2	Upperbound for Erasure Networks	125
	4.12	Conclu	isions	126
5	\mathbf{Esti}	matior	n over Wireless Erasure Networks	128
	5.1	Introd	uction	128
	5.2	Proble	m Setup	133
	5.3	Prelim	inary Results	137
	5.4	Optim	al Encoding at Each Node	142
		5.4.1	Presence of Delays	147

		5.4.2	Channel Between the Controller and the Actuator	147
	5.5	Stabil	ity Analysis	148
		5.5.1	Network with Links in Parallel	152
		5.5.2	Necessary Condition for Stability in Arbitrary Networks	152
		5.5.3	Network with Links in Series	154
		5.5.4	Sufficient Condition for Arbitrary Networks	155
	5.6	Perfor	mance Analysis	160
		5.6.1	Networks with Links in Series	163
		5.6.2	Network of Parallel Links	165
		5.6.3	Arbitrary Network of Parallel and Serial Links	166
		5.6.4	Networks with Arbitrary Topology	168
			5.6.4.1 Lower Bound	169
			5.6.4.2 Upper Bound	170
	5.7	Exam	ples	170
	5.8	Gener	alizations	175
		5.8.1	Correlated Erasure Events	175
			5.8.1.1 Markov Events	176
			5.8.1.2 Spatially Correlated Events	176
		5.8.2	Synthesis of a Network	178
		5.8.3	Unicast Networks	179
	5.9	Concl	usions	180
6	Pow	ver Eff	iciency of Sensor and Ad-Hoc Networks	182
	6.1	Introd	luction	182
	6.2	Notat	ion and System Model	187
		6.2.1	Notation and Definitions	187
		6.2.2	System Model and Problem Statement	188

	6.3	An Ex	ample: Multi-antenna Systems	193
	6.4	Sensor	ry networks	195
		6.4.1	"Listen and Transmit" Protocol	195
		6.4.2	Finding a Lower Bound	196
		6.4.3	Finding an Upper Bound	200
		6.4.4	Main Result: Sensory Case	202
		6.4.5	Discussion on Synchronicity	203
	6.5	Ad-ho	c Networks	205
		6.5.1	"Listen and Transmit" Protocol	206
		6.5.2	Finding Upper and Lower Bounds	209
		6.5.3	Power Allocation	213
		6.5.4	Main Result: Ad-hoc Case	218
		6.5.5	A Further Result	222
		6.5.6	Discussion on Synchronicity	225
		6.5.7	Complete Knowledge of the Channel	226
	6.6	Conclu	usion	232
7	Diff	erentia	ated Rate Scheduling for Cellular Systems	233
	7.1	Introd	uction	233
	7.2	Proble	em Formulation	237
	7.3	Prelim	ninary Results for MIMO GBC	241
		7.3.1	The Capacity Region of MIMO GBC	241
		7.3.2	Opportunistic Beamforming	243
		7.3.3	Tighter Scaling Laws	245
	7.4	Chann	nels with a Small Number of Users	248
		7.4.1	Case 1: Two-User Channels	249
		7.4.2	Case 2: Low SNR Regime	250

		7.4.3	Throughput Loss	252
	7.5	Chann	els with Many Users	253
		7.5.1	Optimal Differentiated Opportunistic Beamforming Scheme .	254
		7.5.2	Time-Division Opportunistic (TO) Beamforming	257
		7.5.3	Weighted Opportunistic (WO) Beamforming	261
		7.5.4	Superposition Coding for Single Antenna Broadcast Channels	268
	7.6	Simula	ation Results	276
	7.7	Conclu	usion	279
8	MI	MO Ga	aussian Broadcast Channel with Estimation Error	284
	8.1	Introd	uction	284
	8.2	System	n Model	287
	8.3	Inner	Bound on the Capacity Region	289
	8.4	Optim	al Power Allocation	295
	8.5	Scalin	g Laws of the Achievable Sum-rate	298
	8.6	Traini	ng	300
	8.7	Conclu	usion	306
9	Fut	ure W	ork	307
Bi	Bibliography 31			

List of Figures

1.1	Sources of challenges in analysis and design of wireless networks	3
1.2	A simple network with one relay component, one source and one desti-	
	nation	5
1.3	A cellular system with multiple antennas in the transmitter (base station).	9
2.1	A directed acyclic graph with four nodes and five edges. The cut-set	
	$\{(3,4),(3,2),(1,2)\}$ is shown by the dashed line.	22
2.2	(i): An erasure wireless network with the graph representation of exam-	
	ple 2.1. Probability of erasure on link (i, j) is ϵ_{ij} . Each node (e.g., node	
	3) transmits the same signal (X_3) across its outgoing channels. Since	
	the network is interference-free, node 4 receives both signals Y_{24} and Y_{34}	
	completely. (ii): In this network, cut-capacity for s-set $\mathcal{V}_s = \{1,3\}$ is	
	$C(\mathcal{V}_s) = 1 - \epsilon_{12} + 1 - \epsilon_{32}\epsilon_{34} \dots \dots$	25
2.3	For the cut-set specified by the s-set $\mathcal{V}_s = \{1, 3, 4\}$, the cut-capacity is	
	$C(\mathcal{V}_s) = 1 - \epsilon_{12} + 1 - \epsilon_{46} + 1 - \epsilon_{35}\epsilon_{32}$.	28
2.4	A wireless erasure network with two sources, $S = \{1, 2\}$, and one desti-	
	nation, $\mathcal{D} = \{3\}$.	33
2.5	A wireless erasure line network	48
2.6	A simple network.	52
2.7	Performance of different schemes for four scenarios	55

3.1	(a): A wireless erasure network with one source and two destination	
	nodes. (b): Different achievable regions for the network in part (a)	64
3.2	(a):An (m, n) -erasure broadcast channel (b): An $(1, n-1)$ -erasure broad-	
	cast channel	66
3.3	Illustrating the proof technique	77
3.4	Capacity region of the degraded $(2, 2)$ -erasure broadcast channel of Ex-	
	ample 3.2	84
3.5	(a): Erasure broadcast channel derived from $\mathcal{E}' = \{(1,2), (1,3), (1,4)\}$	
	for the network in Figure 3.1. (b): Erasure broadcast channel con-	
	structed from $\mathcal{E}' = \{(1,2), (1,4), (4,3)\}$ for the same network as in (a).	87
3.6	(a): A wireless erasure network with two destinations. (b): Erasure	
	broadcast channel derived form $\mathcal{E}' = \{(1,3), (2,3), (2,5), (4,3), (4,6)\}$	
	for the network of part (a). \ldots \ldots \ldots \ldots \ldots \ldots	88
3.7	Time-sharing region and different outer bounds for Example 3.6	91
4.1	Example of a network	96
4.2	Modified Erasure Channel	97
4.3	Proof of Theorem 4.1.	100
4.4	Multistage relay network.	117
4.5	Erasure network with four relay nodes	119
4.6	Gaussian network with three relay nodes.	120
4.7	Gaussian network with four relay nodes	121
4.8	Rates for the Gaussian network of Figure 4.7.	121
5.1	The set-up of the control across communication networks problem. For	
	most of the discussion in the chapter, we will ignore the network between	
	the controller and the actuator. See, however, Section 5.4.2. \ldots	133
5.2	Example of a network of combination of parallel and serial links	168

Simulated and theoretical results for a line network	171
Simulated and theoretical results for a parallel network	172
Bridge network and the networks used for calculating lower and upper	
bounds	172
Simulated values and theoretical bounds for the bridge network	173
Simulated difference in performance between algorithm in which no en-	
coding is done and our optimal algorithm for series connection of n links	.174
Loss in performance as a function of packet drop probability for n links	
in series	175
Sensory and Ad-hoc wireless networks	183
Condition on the channel coefficients	190
Single-hop versus Multi-hop	191
Power efficiency for interference suppression case	231
A MISO broadcast channels with differentiated rate users	238
β versus μ_1 for a channel with $n = 2. \ldots \ldots \ldots \ldots \ldots$	251
Ratio of the throughput with rate constraints over the sum-rate capacity	
versus β for a channel with $n = 2$	253
The decision region for power allocation in the superposition coding in	
two group case: If $(x, y) \in \mathcal{R}_1$, all the power is allocated to the best user	
of group one. If $(x, y) \in \mathcal{R}_2$, all the power is allocated to the best user	
of group two. If $(x, y) \in \mathcal{R}$, then power is split between the best users	
of the two groups as in (7.50)	271
The sum of the transmitted rates for WO, TO, and SC, as well as the	
sum-rate capacity of the single antenna broadcast channel as a function	
of the number of users for a system with $K = 2$ and $\beta = 2$	277
	Simulated and theoretical results for a line network Simulated and theoretical results for a parallel network

7.6	The ratio of the rates transmitted to the two groups of users as a function	
	of the number of users for WO and SC for a system with $\beta=2.$	277
7.7	The sum of the transmitted rates for WO, TO, and SC, as well as the	
	sum-rate capacity of the single antenna broadcast channel as a function	
	of the number of users for a system with $K = 2$ and $\beta = 4$	278
7.8	The ratio of the rates transmitted to the two groups of users as a function	
	of the number of users for WO and SC for a system with $\beta=4.\ .$	278
7.9	The sum of the transmitted rates for WO, TO, as well as the oppor-	
	tunistic for a broadcast channel with $M = 2, K = 2, \frac{\alpha_2}{\alpha_1} = 2$, and $\frac{\beta_1}{\beta_2} = \frac{1}{2}$	
	as a function of the number of users. \ldots \ldots \ldots \ldots \ldots \ldots	279
7.10	The ratio of the rates transmitted to the two groups of users as a function	
	of the number of users for WO and TO for a system with $M = 2, K = 2$,	
	$\frac{\alpha_2}{\alpha_1} = 2$, and $\frac{\beta_1}{\beta_2} = \frac{1}{2}$	280
7.11	A closer look at the ratio of the rates transmitted to users in different	
	groups for the example in Figure 7.10	280
7.12	The sum of the transmitted rates for WO, TO, as well as the opportunis-	
	tic for a broadcast channel with $M = 2, K = 3$, and $\beta_1 = 1, \beta_2 = 2$, and	
	$\beta_3 = 3$ as a function of the number of users	281
7.13	The ratio of the rates transmitted to the two groups of users as a function	
	of the number of users for WO and TO for a system with $M = 2, K = 3$,	
	and $\beta_1 = 1, \beta_2 = 2$, and $\beta_3 = 3.$	281
8.1	Multi-antenna Gaussian broadcast channel with channel estimation error	. 288
8.2	Inner bound on the capacity region with estimation error	290
8.3	A dual representation of the inner bound of the capacity region with	
	estimation error.	296

Chapter 1 Introduction

The range of applications of wireless technology is no longer limited to transmission of voice in cellular networks. Numerous types of services, e.g., images, file transfer, and streaming video, with different Quality of Service (QoS) requirements, are available in today's highly heterogeneous cellular networks.

Furthermore, cellular networks are no longer the only type of wireless networks in use. Wireless local and metropolitan area networks (LAN, MAN) offer connectivity in the office, at home, or among buildings. Wireless networks, such as ad-hoc and sensor networks, are also deployed for various purposes such as environmental monitoring, industrial, transportation, and home systems automation, control of distributed embedded systems (such as robots or Unmanned Aerial Vehicles), and even medical services [66, 67, 68]. These networks are interconnected systems of devices that are capable of communication, computation, data storage, and adaptation in a distributed fashion. Unlike cellular networks, these networks typically operate without any predetermined infrastructure.

The increasing utilization of networks, especially wireless networks, in different aspects of modern life has redirected the attention from the point-to-point setup towards the network (or multi-user) setup. There are many features of the wireless environment and the distributed nature of a network setup that make the analysis and design of wireless networks a very challenging problem. It is already known that the theory and protocols developed for point-to-point communications channels do not necessarily lead to satisfactory performance in a network setup. Therefore, new techniques and methodologies, designed specifically for networks, are required that capture the features of wireless networks and provide optimal performance in terms of rate, delay, and robustness. Also, because of the limited amount of available resources, such as power, bandwidth, antenna, and memory, a new look at the efficient usage of these resources is essential.

In essence, we are seeking the best strategy in the network that combats (and in many cases exploits [102]) the features of the wireless medium, uses the resources available as efficiently as possible, and provides the users with different types of demands. The key point is that in a network setup these tasks should be performed in a *distributed* fashion through the cooperation of the users and based on the *local* information available to each user.

The work presented in this thesis can be thought of as ways to find these optimal strategies or to bound the optimal performance for special classes of wireless networks. Figure 1.1 demonstrates the different sources of challenges in the design and analysis of wireless networks. In the following section we look at each source in more detail.

1.1 Sources of Challenges in Wireless Networks

Features of the Wireless Medium

There are many features of the wireless medium that distinguish it from other media. The wireless medium is a *shared* medium. This means that unlike wireline systems, where there exists dedicated physical connections between users, every user can essentially receive an attenuated version what other users are transmitting. In such a system, the manner of transmission is *broadcast* of the signal and there is *interference* in reception of a signal. Another property of a wireless channel is its random time-



Figure 1.1: Sources of challenges in analysis and design of wireless networks.

varying behavior due to the mobility of users and other objects, as well as obstacles in the environment. More specifically, the channel to a given user might have poor conditions at some times and favorable conditions at other times. This is called the *fading* behavior of the channel [1]. In many situations, multiple copies of the transmitted signal may be received with different delays and different strengths. This is referred to as "multipath fading" and can severely deteriorate the performance when the transmitted signals have shorter duration (e.g., broadband transmission). Conventionally, the goal is to combat the randomness introduced by the environment. However, in recent years, there has been another view and that is to exploit the inherent randomness in the environment to increase the performance [102, 127, 121]. For instance, the multi-user diversity gain in the downlink of cellular systems is based on this idea, i.e., in a system of many users with random quality of reception (fading), there exists one user with good quality of reception with very high probability.

3

4

Efficient Use of Resources

Today's wireless systems are faced with an ever-growing demand for higher rates and quality of services. However, the available resources such as bandwidth, power, and number of antennas is limited. Therefore, efficient usage and allocation of these resources is more important than ever. In many scenarios, from mobile users in cellular networks to sensor nodes deployed in a remote area, the components of the network have limited power supplies. In these networks, efficient use of the available energy (power) supply is a critical issue. Bandwidth is another valuable resource in wireless systems, especially in high data rate broadband communications. Also, there has been a great interest in exploiting the spatial degrees of freedom in wireless systems by deploying multiple antennas at the transmitter and the receiver [109, 150, 148, 105]. The possibility of using multiple antennas in a network (multi-user) setup have been recently explored [117, 133].

Demands and Services

As mentioned earlier, wireless systems have become highly heterogenous. Different types of applications such as voice, internet, and video-on-demand, are provided over wireless networks. Depending on the application, the main performance measure will vary. For instance, video-on-demand applications not only require high rates but also are sensitive to delay. For many detection schemes, in ad-hoc networks reliability is the main concern. In many scenarios, resource allocation in wireless networks aims at optimizing over two conflicting performance measures at the same time, such as reliability and rate or delay and rate [154, 155]. Finding strategies that provide these different demands in an efficient manner is a challenging task.



Figure 1.2: A simple network with one relay component, one source and one destination.

Operation In a Network Setup

Finding the optimal strategy for the nodes of a network in order to optimally perform a given task is very much an open problem. Consider the simple network, with only three nodes, shown in Figure 1.2. The desired task is reliable communication from the source to the destination with the aid of the relay node. The relay node is connected to both the source and destination through communications channels. Even for this simple network, finding the optimal operation at each node for maximizing the rate of reliable communication is unsolved [53]. The main difficulty in a network setup is the distributed nature of the information in the network. Each user has only access to local information and has to cooperate with other nodes in a distributed fashion to maximize the performance.

The above sources have raised many important and interesting challenges regarding the performance limits of different tasks such as communications and computation over networks. In addition, there are many design issues concerning the complexity and the robustness of the systems that should be addressed for a thorough understanding and efficient operation of wireless networked systems.

This thesis deals with a few of the challenges associated with wireless networks. In the first part of the thesis, Chapters 2 through 6, we analyze the capabilities of wireless networks for different tasks. More specifically, we look at the limits of rate, power efficiency, and estimation error for the tasks of reliable data-transmission and estimation across wireless networks.

In the second part of the thesis, Chapters 7 and 8, we look at a few design issues for wireless cellular networks, namely differentiated QoS provisioning and the robustness issues in downlink scheduling of cellular networks.

1.2 Performance Limits In Wireless Networks

In his ground breaking paper "A mathematical theory of communication," [2], Claude E. Shannon revealed the fundamental performance limit of reliable communications, i.e., capacity, between a transmitter and receiver pair. This work started the new field of information theory and laid down a foundation for modern digital communications. Since then, one of the main goals in communications theory has been to devise practical methods to approach the capacity of point-to-point channels.

As mentioned earlier, the increase in utilization of wireless networks that involves multiple users rather than a single pair of users for different applications has spurred a great deal of research in various areas of wireless networks. Multi-terminal information theory is an example of these areas that looks at the fundamental performance limits of reliable communication in a multi-user setup [53]. Furthermore, since tasks like distributed estimation, computation, and control across communication networks (channels) are emerging as (primary) applications for many wireless networks, there has been a great deal of research aiming at finding the fundamental limits of performance for these tasks as well [3, 4, 5].

However, unlike single-user information theory, we do not have a complete understanding of the optimal performance for these applications. This is mainly because of the challenges discussed in Section 1.1. Multi-terminal information theory is still in its early stages. Many problems in multi-terminal information theory are still unsolved. As mentioned in the previous section, even for the simple network of Figure 1.2, the capacity region and the optimal strategy are not known [14]. Finding the performance limits of different applications is a fundamental task for the following reasons:

- It gives us a measure of the capabilities of the network and sets a standard for what we should aim for.
- It provides us with intuition on the optimal solution and guides practical strategies for achieving the limits.
- Identifying the limits of performance and the main parameters of the network affecting those limits gives insights for the optimal design of networks and even convergence of different applications on the same network.

In the first part of this thesis, we mainly look at the performance limits of different applications over wireless networks. In Chapters 2 through 6, we consider a simple model that captures some of the essential features of wireless networks. We look at two applications, namely data transmission and control and estimation, over these networks and find the performance limits and optimal operation for these applications.

1.3 Design Issues in Wireless Cellular Networks

The downlink of cellular systems is known to be a major bottleneck for future broadband wireless communications. The down-link scenario refers to the case in which the base station (possibly equipped with multiple antennas) simultaneously provides service to multiple mobile subscribers in a cell (as shown in Figure 1.3). From an information-theoretic perspective, broadcast channels [13], and in particular the Gaussian broadcast channel (GBC), can be used to model the downlink in cellular systems. There exists an abundance of information-theoretic results describing the limits of the achievable rates to the users in single-input single-output (SISO) Gaussian broadcast channels (see e.g., [115, 116]). More recently, there has been growing interest in the use of multiple antennas (at the transmitter, receivers, or both) for wireless communication networks. This has led to an interest in the multiple-input multiple-output (MIMO) Gaussian broadcast channel, where the transmitter and the various users may be equipped with multiple transmit and receive antennas, respectively. For these channels, the entire capacity region [118] is shown to be achieved by an interference cancelation scheme referred to as *dirty paper coding* (DPC) [119].

While dirty paper coding is the optimal transmission scheme, it is computationally expensive and requires the transmitter to have perfect knowledge of the channel state information (CSI) for all the users. More importantly, unlike point-to-point multiantenna channels, the multi-user capacity depends heavily on whether the transmitter knows the channel coefficients to each user [121, 117, 118]. For instance, in a Gaussian broadcast channel with M transmit antennas and n single antenna users, the sum rate capacity scales like $M \log \log n$ for large n if perfect channel state information (CSI) is available at the transmitter, yet only logarithmically with M if it is not. Therefore, it seems that the performance of the broadcast channels is very sensitive to the channel knowledge. However, in practice the transmitter and the users do not have access to the exact channel realization and have only an estimate of it.

Moreover, in today's cellular systems, users might request different applications such as voice, internet, or video-on-demand. This implies that the base station has to provide differentiated services to different users, yet at the same time maximize the throughput. Therefore, the best operating point on the capacity region will no longer be the sum-rate capacity point. While achieving the sum-rate capacity point has been studied before, devising simple and practical schemes that come close to other points on the capacity region has not been studied in depth.

The issues mentioned above raise the following questions:

• Do there exist simple schemes that require little amount of feedback to the transmitter and yet maximize the throughput while providing different users



Figure 1.3: A cellular system with multiple antennas in the transmitter (base station).

with different demands (rates)?

• What is the effect of channel estimation errors at the transmitter and/or users on the capacity region of MIMO Gaussian broadcast channels?

These questions are mainly dealt with in Chapters 7 and 8. First, we propose simple scheduling schemes that require little CSI and provide users with different rates. Then, we look at the performance of broadcast channels with channel estimation error at the transmitter and the receivers and propose a training-based scheme for these channels.

1.4 Contributions of the Thesis

In this section, we review the contents of each chapter and mention the main contributions. The thesis is organized in such a way that different chapters can be more or less read independently.

Chapter 2 through Chapter 6 deal with finding performance limits of different applications over wireless networks.

In Chapter 2 we introduce a simple model for wireless networks called *wireless* erasure networks. This model incorporates the broadcast nature of transmission in wireless environment, but assumes that there is no interference. This model is reasonable for wireless networks where information is packetized and some interferenceavoidance scheme is in effect. Examples of this scenario can be Software Defined Radios (SDR) operating based on Orthogonal Frequency Division Multiplexing (OFDM) techniques [19]. We look at a specific traffic pattern called the *multicast problem*. In this problem, a subset of users, denoted by the "destination set," are interested in the same message from a user, denoted by the "source node." The main contribution of this chapter is to find the multicast capacity of wireless erasure networks with arbitrary topology and show that it has an interesting interpretation related to the concept of minimum-cut capacity in flow graphs. However, in our case, the definition of the cut-capacity should be modified so that it takes into account the broadcast feature of the network. It is further shown that random linear encoding of received packets at each node suffices to achieve the optimal performance.

In Chapter 3 we look at another traffic pattern, referred to as the *broadcast* problem, over wireless erasure networks. In this problem, a subset of users, denoted by the "destination set," demand different types of information from the same user, called the "source node." The goal is to find the capacity region, i.e., set of achievable rates of communication between the source node and the destinations. First, we find the capacity region of broadcast problems for a subclass of wireless erasure networks called *erasure broadcast channels*. We show that the capacity region of these channels is achieved by time-sharing between the destinations at each node in the network. The proof technique used relies on optimization theory and may be extended to more general networks. We use the result for erasure broadcast channels to find an outer bound for the capacity region of the broadcast problems for general wireless erasure networks.

In Chapter 4 we look back at the multicast problem for wireless networks with *limited* operation allowed at each node. More specifically, unlike Chapter 2, we assume that the nodes can only perform one of two operations. They can either *forward* the signal they have received without any attempt to recover the content of the signal, or they can *decode and re-encode* the received signal, recover its content, and re-encode and send it across the network. The second operation is basically what is known as "routing" (or switching) in today's conventional network protocols. We first show that because of the distinguishing features of the wireless medium, routing data across the network does not necessarily give satisfactory results in a wireless setup. For instance, a multi-hop approach, in which every relay node *decodes* the received message, is not always the best paradigm for wireless networks. We then propose a greedy algorithm that finds the best operation among forwarding and decoding and re-encoding at each node in order ro maximize the achievable rate in the network.

In Chapter 5 we turn our attention from data communications to estimation and control across wireless networks. We consider the problem of estimation and control of a dynamical process at a remote destination node. The destination node is connected to the sensor that observes the process through a wireless erasure network model introduced in Chapter 2. The performance measure in this problem is the steady-state estimation error covariance at the destination node. We find the condition for stability of the system and the exact performance measure in terms of the parameters of the process and the network. It is shown that the stability condition resembles a max-flow min-cut condition; however, the definition of the cut-value is different from the one introduced in Chapter 2. In addition, we find that the effect of the network on the performance is mainly through the *latency* introduced by the network in receiving recent measurements at the destination. The results of this chapter and Chapter 2 provide guidelines for the design of networks that are optimal for both communication

and estimation.

Chapter 6 looks at another measure of performance in wireless networks, namely power consumption. We are interested in the power efficiency of large and random wireless sensor and ad-hoc networks. The fundamental question that we answer is, "Is it beneficial in terms of power consumption for nodes to operate in a network setup?" We show that densely deployed wireless sensors networks can gain in terms of power consumption by operating in a network and using a simple protocol. The protocol is essentially based on performing *distributed beamforming* at different nodes of the network. We quantify this gain by showing that for large random wireless ad-hoc network with n users and $r \leq \sqrt{n}$ simultaneous source/destination pairs located in a domain of fixed area, one can support the same rate as a single point-to-point system, but by expending only $\frac{1}{\sqrt{n}}$ the energy. This result shows the value of cooperation among nodes in terms of power consumption.

Having looked at the performance limits of wireless networks in previous chapters, we look at two design issues in cellular networks in Chapters 7 and 8.

In Chapter 7 we consider the downlink of wireless cellular systems where users have different rate demands. We model this scenario as a multiple input multiple output (MIMO) Gaussian broadcast channel. We assume n homogenous users divided into K groups. Users in the same group require the same rate. We further assume that the ratio of the groups' rates are given. The objective is to design strategies that are simple, require limited amount of information about the channel at the transmitter, and maximize the throughput (sum of the rates to all users) while maintaining the rational rate constraints. In general, this problem appears to be computationally intractable since the ergodic capacity region is described as the convex hull of (an infinite) set of rates. We focus on the asymptotic regime of large n where explicit results can be found. In particular, we propose three scheduling schemes to provide the rational rate constraints, namely the weighted opportunistic beamforming (WO),

time division opportunistic beamforming (TO), and superposition coding (SC) for the single antenna case. In WO, we transmit each beam to the user that has the largest "weighted" signal to interference plus noise ratio (SINR). In TO, each group has its own time slot during which the transmitter chooses the user with the best SINR from the corresponding group. Superposition coding is the scheme that achieves the information-theoretic capacity region for the single antenna case. For each scheduling we give an explicit scheme to guarantee the rational rate constraints. We also analyze the throughput loss due to the rate constraints for all three different schemes. In particular, we show that the throughput loss compared to the maximum throughput (i.e., the sum rate capacity without any rate constraints) tends to zero for large n. Thus, there is not much of a penalty in providing different levels of service to different users. We also analyze the convergence rate of all the schemes and provide simulations supporting the theoretical analysis.

In Chapter 8 we look at robustness issues for the downlink scheduling in cellular networks. More specifically, we consider a MIMO Gaussian broadcast channel with channel estimation error at the transmitter and the receivers. We propose an achievable region based on the dirty paper coding scheme. We further show that the achievable region is equivalent to the capacity region of a dual Gaussian multiple access channel with noise covariance that depends on the transmit powers. We look at the achievable sum-rate for large broadcast channels with estimation error and show that as long as the estimation error does not vary with the number of users, the scaling behavior of the sum-rate is similar to the case with no estimation error. At the end of this chapter, we analyze the performance of training-based methods for broadcast channels with estimation error. We find the optimal energy and time that should be allocated for training. Based on the training scheme, it is shown that in order to achieve a linear increase in the sum-rate in terms of the number of transmit antennas in the high SNR regime (also known as the multiplexing gain), a linear fraction of energy should be allocated for the training phase. In other words, to achieve the multiplexing gain one needs a high fidelity description of the channel at the transmitter.

Finally in Chapter 9, we discuss a few interesting open problems that have been brought up by the research undertaken in this thesis.

Chapter 2

Capacity of Wireless Erasure Networks

2.1 Introduction

Determining the capacity region for general multi-terminal networks has been a longstanding open problem. An outer bound for the capacity region is proposed in [53]. This outer bound has a nice min-cut interpretation: The rate of flow of information across any cut (a cut is a partition of the network into two parts) is less than the corresponding cut-capacity. The cut-capacity is defined as the maximum rate that can be achieved if the nodes on each side of the cut can fully cooperate and also use their inputs as side-information.

The difficulty in multi-terminal information theory is that this outer bound is not necessarily tight. For instance, for the single relay channels introduced in [8], no known scheme achieves the min-cut outer bound of [53].

However, for a class of network problems called multicast problems in *wireline* networks, it is shown that the max-flow min-cut outer bound can be achieved [9, 10, 54]. A multicast problem comprises one or more source nodes (at which information is generated), several destinations (that demand all information available at the source nodes), relay nodes, and directed communication channels between some nodes. It is assumed that each channel is statistically independent of all other channels. Also,

as the name suggests, the communication between different nodes is done through physically separated channels (wires). This means that the communication between two particular nodes does not affect the communication between others. In this setup, the maximum achievable rate is given by the minimum cut-capacity over all cuts separating the source nodes and a destination node. Because of the special structure of wireline networks, the cut-capacity for any cut is equal to the sum of the capacities of the channels crossing the cut.

This remarkable result for wireline networks is proved by performing separate channel and network coding in the network. First, we perform channel coding on each link of the network, so as to make it operate error-free at any rate below its capacity. This way, the problem is transformed into a flow problem in a graph where the capacity of each edge is equal to the information-theoretic capacity of the corresponding channel in the original network. If there is only one destination node, standard routing algorithms for finding the max-flow (min-cut) in graphs [52] achieve the capacity. However, when the number of destinations is more than one, these algorithms can fail. The key idea in [9] is to perform coding at the relay nodes. By [10, 54], linear codes are sufficient to achieve the capacity in multicast problems. These ideas are formulated in an algebraic framework and generalized to some other special network problems in [54]. Since then, there has been a great deal of research on the benefits of coding over traditional routing schemes in networks from different viewpoints, such as network management, security, etc. [11, 12].

In a wireless setup, however, the problem of finding the capacity region is more complicated. The main reason is that unlike wireline networks, in which communication between different nodes is done using separated media, in a wireless system the communication medium is shared. Hence, all transmissions across a wireless network are *broadcast*. Also, any communication between two users can cause *interference* to the communication of other nodes. These two features, broadcast and interference, present new issues and challenges for performance analysis and system design. The capacity regions of many information-theoretic channels that capture these effects are not known. For instance, the capacity region for general broadcast channels is an unsolved problem [13]. The capacity of general relay channels is not known. However, there are some achievable results based on block Markov encoding and random binning[14]. These ideas have been generalized and applied to a multiple relay setup in [15] and [16].

In this chapter we look at a special class of wireless networks which only incorporates the broadcast feature of wireless networks.¹We model each communication channel in the network as a memoryless erasure channel. We will often assume that the erasure channels are independent; however, we show that the results also hold when the various erasure channels are correlated. We require that each node sends out the same signal on each outgoing link. However, for reception we use a multiple access model without interference, i.e., messages coming into a node from different incoming links do not interfere. In general, this is not true for a wireless system. However, this can be realized through some time, frequency, or code division multiple access scheme. ² This simplification is important in making solution of the problem tractable. Even the capacity of a single relay channel is not known.

Finally, we assume that complete side-information regarding erasure locations on each link is available to the destination (but not to the relay) nodes. If we assume that the erasure network operates on long packets, i.e., packets are either erased or received exactly on each link, then this assumption can be justified by using headers in the packets to convey erasure locations or by sending a number of extra packets containing this information. By making the packets very long, the overhead of transmitting the

 $^{^{1}}$ [17] and [18] have considered applications of network coding at the network layer for cost (energy) minimization in lossless wireless ad-hoc networks. In this chapter, we look at wireless features of the network in the physical layer.

²Emerging technologies like Software-Defined Radio (SDR) based on OFDM techniques [19] also have these properties.

erasure locations can be made negligible compared to the packet length. We should remark that provided that the side-information is available to the destinations, all results in this work hold for any packet length.

We should mention that our model is appropriate for wireless networks where all information transmission is packetized and where some form of interference-avoidance is already in place. Channel coding within each packet can be used to make each link behave as a packet erasure channel. Although our model does not incorporate interference (primarily because it is not clear what interference means for erasure channels) one way, perhaps, to account for interference is to allow the erasure channels coming into any particular node to be correlated (something that is permitted in our model).

The main result is that a max-flow min-cut type of result holds for multicast problems in wireless erasure networks under the assumptions mentioned above. The definition of cut-capacity in these networks is such that it incorporates the broadcast nature of the network. We further show that similar to the wireline case, for multicast problems over wireless erasure networks, linear encoding at nodes achieves all the points in the capacity region. Working with linear encoding functions reduces the complexity of encoding and decoding. Building on the results of this work and using ideas from LT coding [20], it is shown in [21] that it is possible to reduce the delay incurred in the network. In their scheme, instead of using linear *block* codes, which is what we do here, the nodes send random linear combinations of their previously received signals at each time. This way nodes do not need to wait for receiving a full block before transmitting, which reduces the delay.

We once more need to emphasize the importance of the side-information on the erasure locations (or any other mechanism that provides the destination with the mapping from the source nodes to their incoming signals) for our result to hold. Interestingly, all the cut capacities of the network remain unchanged by making the above described side-information available to the receiver nodes. Thus, in some sense, what is shown here is that with appropriate side information made available to the receivers, the min-cut upper bound on capacity can be made tight. It would therefore be of further interest to see whether for other classes of networks it is possible to come up with the appropriate side-information to make the min-cut bounds tight.

This chapter is organized as follows. Section 2.2 defines notation used in this chapter and reviews some graph theoretic definitions of importance. We introduce the network model in Section 2.3 and the problem setup in Section 2.4. Section 2.5 states the main result for multicast problems over wireless erasure networks with side-information available at destinations. Section 2.6 includes proofs of these results. Section 2.7 demonstrates the optimality of linear encoding. Section 2.8 includes a discussion of our network assumptions. Also, the performance of different coding schemes when side-information is not available is analyzed and compared. We mention future directions of our work and conclude in Section 2.9.

2.2 Preliminaries

2.2.1 Notation

Upper case letters (e.g., X, Y, Z) usually denote random variables, and lower case letters (e.g., x, y, z) denote the values they take. Underlined letters (e.g., \underline{x}) are used to denote vectors. Sets are denoted by calligraphic alphabet (e.g., $\mathcal{A}, \mathcal{B}, \mathcal{C}$). The complement of a set \mathcal{A} is shown by \mathcal{A}^c . The transpose of matrix \underline{x} is shown by \underline{x}^{\dagger} . $\exp(x)$ is used to denote 2^x .

Subscripts specify nodes, edges, inputs, outputs, and time. For instance, v_2 and X_2 could denote node number two and the output of node number two in the network, respectively. Unless otherwise mentioned, commas are used to separate time subscripts from other subscripts. Superscripts are also used to refer to different sources.
\mathcal{V}	node set
${\mathcal E}$	edge set
S	the set of source nodes
\mathcal{D}	the set of destination nodes
$[\mathcal{V}_x,\mathcal{V}_y]$	$x - y$ cut-set described by x-set \mathcal{V}_x
X_i	symbol transmitted from node i
X_i^n	a transmitted block of n symbols from node i
Y_{ij}	channel output of edge (i, j)
Y_i	symbols received at node i from all incoming channels
$w^{(s)}$	message transmitted from source s
$\mathcal{W}^{(s)}$	message index set at source node s
$\hat{w}_{d_i}^{(s)}$	estimate at destination d_i of the message transmitted from s
$P_{d_i}^{(n)(s)}$	prob. of error in decoding source s at destination d_i

Table 2.1: Some important notation in this chapter

For example, $w^{(s)}$ could denote the message sent by node s.

Consider a sequence of numbers x_1, x_2, x_3, \ldots . We use notation x^n to denote the sequence x_1, x_2, \ldots, x_n . We also use notation $(x_i, i \in \mathcal{I})$ to denote the ordered tuple specified by index set \mathcal{I} . Finally, $|\mathcal{X}|$ is the cardinality of set \mathcal{X} , and $2^{\mathcal{X}}$ is the set of all the subsets of \mathcal{X} . Table 2.1 summarizes our notation.

2.2.2 Definitions for Directed Graphs

In this Section, we briefly review the concepts and definitions from graph theory used in this work[22].

A directed graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ has vertex set \mathcal{V} and directed edge set $\mathcal{E} \subset \mathcal{V} \times \mathcal{V}$. Without loss of generality, let

$$\mathcal{V} = \{1, 2, \dots, |\mathcal{V}|\}$$

We assume that the graph is finite, i.e., $|\mathcal{V}| < \infty$. For each node, $v \in \mathcal{V}$, $\mathcal{N}_O(v)$ and $\mathcal{N}_I(v)$ are the set of edges leaving from and the set of edges going into v, respectively.

Formally,

$$\mathcal{N}_O(v) = \{(v, u) | (v, u) \in \mathcal{E}\} \text{ and}$$
$$\mathcal{N}_I(v) = \{(u', v) | (u', v) \in \mathcal{E}\}.$$
(2.1)

The out-degree $d_O(v)$ and in-degree $d_I(v)$ of v are defined as $d_O(v) = |\mathcal{N}_O(v)|$ and $d_I(v) = |\mathcal{N}_I(v)|$. A sequence of nodes v_0, v_1, \ldots, v_n such that $(v_0, v_1), (v_1, v_2), \ldots, (v_n, v_0)$ are all in \mathcal{E} is called a cycle. An acyclic graph is a directed graph with no cycles.

An x - y cut for $x, y \in \mathcal{V}$ is a partition of \mathcal{V} into two subsets, \mathcal{V}_x and $\mathcal{V}_y = \mathcal{V}_x^c$, such that $x \in \mathcal{V}_x$ and $y \in \mathcal{V}_y$. The *x*-set \mathcal{V}_x (or *y*-set \mathcal{V}_y) determines the cut uniquely. For the x - y cut given by \mathcal{V}_x , the cut-set $[\mathcal{V}_x, \mathcal{V}_y]$ is the set of edges going from the *x*-set to *y*-set, i.e.,

$$[\mathcal{V}_x, \mathcal{V}_y] = \{(u, v) | (u, v) \in \mathcal{E}, u \in \mathcal{V}_x, v \in \mathcal{V}_y\}.$$

We also define \mathcal{V}_x^* as

$$\mathcal{V}_x^* = \{ v | \exists u \text{ s.t. } (v, u) \in [\mathcal{V}_x, \mathcal{V}_y] \}.$$

 \mathcal{V}_x^* is the set of nodes in the *x*-set that has at least one of its outgoing edges in the cut-set.

Example 2.1. Consider the acyclic directed graph shown in Figure 2.1. $\mathcal{V} = \{1, 2, 3, 4\}$ is the set of nodes, and $\mathcal{E} = \{(1, 2), (3, 2), (1, 3), (3, 4), (2, 4)\}$ is the set of edges. The source and destination nodes are s = 1 and d = 4, respectively. The *out-degree* of node 3 is 2, i.e., $d_O(3) = 2$. Looking at the s - d cut specified by *s-set* $\mathcal{V}_s = \{1, 3\}$, the cut-set $[\mathcal{V}_s, \mathcal{V}_d]$ is the set $\{(3, 4), (3, 2), (1, 2)\}$, and $\mathcal{V}_s^* = \{1, 3\}$.

At the end of this section we define the notion of partial ordering of the nodes in the graph. Consider two distinct nodes i and j of the network. Exactly one of the following three will occur:



Figure 2.1: A directed acyclic graph with four nodes and five edges. The cut-set $\{(3,4), (3,2), (1,2)\}$ is shown by the dashed line.

- 1. There is a directed path from i to j. In this case we will say that $i \prec j$.
- 2. There is a directed path from node j to node i. In this case we will say that $j \prec i$.
- 3. There is no directed path from node i to node j or from node j to node i. In this case we will say that j and i are incomparable.

Note that since we assume acyclic networks, we cannot have directed paths both from i to j and from j to i. Thus, we have a partial ordering for nodes in the network. For example, in Figure 2.3 we have $3 \prec 4$, but 2 and 5 are incomparable. Note that the partial ordering gives us a (non-unique) sequence of nodes starting with s such that for every node i, all the nodes j that satisfy $j \prec i$ are before it in the sequence [58]. Call such a sequence \mathcal{T} . A possible sequence \mathcal{T} for Figure 2.3 is (s, 3, 2, 5, 4, d).

2.3 Network Model

Wireless Packet Erasure Networks

We model the wireless packet³ erasure network by a directed acyclic graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$. Each edge $(i, j) \in \mathcal{E}$ represents a memoryless packet erasure channel from node *i* to node *j*. For most of this chapter, we assume that erasure events across different links are independent. However, as described later, the results go through for correlated erasure events. For independent erasure events, a packet sent across link (i, j) is either erased with probability of erasure ϵ_{ij} or received without error. We denote the input alphabet (the set of possible packets) of the erasure channel by \mathcal{X} .⁴

Let $Z_{ij,t}$ be a random variable indicating erasure occurrence across channel (i, j)at time t. For independent erasure events, $Z_{ij,t}$ has a Bernoulli distribution with parameter $1 - \epsilon_{ij}$. If an erasure occurs on link $(i, j) \in \mathcal{E}$ at time t, the value of $Z_{ij,t}$ will be zero; otherwise $Z_{ij,t}$ will be one. Note that the behavior of the network can be fully determined by the values of $Z_{ij,t}$ for all links and all times and the operation performed at each node.

We assume that transmissions on each channel experience one unit of time delay. The input of all the channels originating from node i is denoted by X_i chosen from input alphabet \mathcal{X} . Note that with this definition we have required that each node transmit the same symbol on all its outgoing edges, i.e., all channels corresponding to edges in $\mathcal{N}_O(i)$ have the (same) input X_i (see Figure 2.2.) This constraint incorporates broadcast in our network model. The output of the communication channel corresponding to edge $(i, j) \in \mathcal{E}$ is denoted by Y_{ij} ; Y_{ij} lies in output alphabet $\mathcal{Y} = \mathcal{X} \cup \{e\}$, where e denotes the erasure symbol. We also assume that the outputs

³Throughout this chapter a packet can be of any length. When the length of packets is one, the channel is a binary erasure channel.

⁴For simplicity and without loss of generality we consider $\mathcal{X} = \{0, 1\}$ in our analysis and proofs. However, we should remark that all the results and analysis hold for input alphabet of arbitrary length.

of all channels corresponding to edges in $\mathcal{N}_{I}(i)$ are available at node *i*. This condition is equivalent to having no interference in receptions in the network. Having this, let $Y_{i} = (Y_{ji}, (j, i) \in \mathcal{N}_{I}(i))$ be the symbols that are received at node *i* from all its incoming channels. We have $Y_{i} \in \prod_{j:(j,i)\in\mathcal{E}} \mathcal{Y}$. The relation between the Y_{i} s and X_{i} s defines a coding scheme for the network.

Based on the properties of the network mentioned above, if we consider the inputs and outputs up to time t, then the conditional probability function of the outputs of all the channels (edges) up to time t, given all the inputs of all the channels up to time t and all the previous outputs, can be written as follows for all t

$$\Pr\left(\left(y_{ij,t}, (i,j) \in \mathcal{E}\right) \middle| (x_l^t, l \in \mathcal{V}), (y_{ij}^{t-1}, (i,j) \in \mathcal{E})\right)$$
$$= \Pr\left(\left(Y_{ij} = y_{ij,t}, (i,j) \in \mathcal{E}\right) \middle| (X_l = x_{l,t}, l \in \mathcal{V})\right).$$

For independent erasure events, we further have

$$\Pr\left(\left(y_{ij,t}, (i,j) \in \mathcal{E}\right) \middle| (x_l^t, l \in \mathcal{V}), (y_{ij}^{t-1}, (i,j) \in \mathcal{E})\right)$$

$$= \prod_{i \in \mathcal{V}} \prod_{j: (i,j) \in \mathcal{N}_O(i)} \Pr\left(Y_{ij} = y_{ij,t} \middle| X_i = x_{i,t}\right).$$

$$(2.2)$$

Network Problems

Any network problem is characterized by a collection of information sources, a collection of source nodes at which one or more information sources are available, and a collection of destination nodes. Each destination node demands a subset of information sources. More specifically, a network problem is a quintuple $\mathcal{P} = (\mathcal{M}, \mathcal{S}, \mathcal{D}, \Upsilon_g, \Upsilon_r)$, where



Figure 2.2: (i): An erasure wireless network with the graph representation of example 2.1. Probability of erasure on link (i, j) is ϵ_{ij} . Each node (e.g., node 3) transmits the same signal (X_3) across its outgoing channels. Since the network is interference-free, node 4 receives both signals Y_{24} and Y_{34} completely. (ii): In this network, cut-capacity for s-set $\mathcal{V}_s = \{1, 3\}$ is $C(\mathcal{V}_s) = 1 - \epsilon_{12} + 1 - \epsilon_{32}\epsilon_{34}$.

- \$\mathcal{M}\$ = \$\{m_1, \ldots, m_{|\mathcal{M}|}\}\$ denotes the set of information messages (sources).⁵ We assume that each of the information messages is an i.i.d uniformly distribution random process. Different information messages are assumed to be independent.
- $\mathcal{S} = \{s_1, s_2, \dots, s_{|\mathcal{S}|}\} \subset \mathcal{V}$ denotes the source nodes.
- $\Upsilon_g: \mathcal{S} \to 2^{\mathcal{M}}$, where $\Upsilon_g(s_i)$ is the subset of messages generated at source node s_i .
- $\mathcal{D} = \{d_1, d_2, \dots, d_{|D|}\} \subset \mathcal{V}$ denotes the set of destination nodes.
- $\Upsilon_r : \mathcal{D} \to 2^{\mathcal{M}}$, where $\Upsilon_r(d_j)$ is the subset of message demanded at destination node d_j .

We should remark that in the above definition $S \cap D$ may not be empty, i.e., a node can be a destination node for one information source and a source node for another. Also, destination nodes can act as relay nodes for other destination nodes in the network.

⁵Throughout this thesis, \mathcal{M} and \mathcal{W} are used interchangeably to denote the set of information messages.

The class of network problems that we consider in this chapter is the multiple source/multiple destination *multicast* problem, where each of the destinations demands all of the information sources and source nodes generate disjoint messages. Mathematically, a multicast problem, $\mathcal{P}_{mc} = (\mathcal{M}, \mathcal{S}, \mathcal{D}, \Upsilon_g, \Upsilon_r)$, is a network problem where $\Upsilon_g(\cdot)$ is a partition of \mathcal{M} and $\Upsilon_r(d_i) = \mathcal{M}$ for all destination nodes. In Chapter 3, we look at another type of network problems referred to as *broadcast problem*.

Side-information at Destinations

In most parts of the chapter we assume that each destination node $d \in \mathcal{D}$ has complete knowledge of the erasure locations on each link of the network that is on a path from the source set to d. In other words, d knows values of the $z_{ij,t}$, for all $(i, j) \in \mathcal{E}$ and all times t, for which (i, j) is on at least one path from one of the sources to d. This serves as channel side-information provided to the destinations from across the network. In the case when we consider large packets (alphabet), this side-information can be provided using negligible overhead. More discussion of this model appears in Section 2.8.

Cut-capacity Definition

Consider an s - d cut given by s-set \mathcal{V}_s as defined in Section 2.2.2. We define $X(\mathcal{V}_s)$ and $Y(\mathcal{V}_s)$ as

$$X(\mathcal{V}_s) = \{X_i | i \in \mathcal{V}_s^*\} \text{ and}$$

$$Y(\mathcal{V}_s) = \{Y_{ij} | (i, j) \in [\mathcal{V}_s, \mathcal{V}_s^c]\}.$$
(2.3)

At the end of this section, we define the cut-capacity for wireless erasure networks. In wireline networks, the value of the cut-capacity is the sum of the capacities of the edges in the cut-set [54]. Such a definition of cut-capacity in wireline networks makes sense because the nodes can send out different signals across their outgoing edges. However, this is not the case for wireless erasure networks where broadcast transmissions are required. The following definition of cut-capacity is different from that in the wireline network settings, and it incorporates the broadcast nature of transmission in our network.

Definition 2.1. Consider an erasure wireless network represented by $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ and probabilities of erasure ϵ_{ij} as described in Section 2.3. Let s and d_l be the source and destination nodes, respectively. The cut-capacity corresponding to any $s - d_l$ cut represented by s-set \mathcal{V}_s is denoted by $C(\mathcal{V}_s)$ and is equal to

$$C(\mathcal{V}_s) = \sum_{i \in \mathcal{V}_s^*} \left(1 - \prod_{j: (i,j) \in [\mathcal{V}_s, \mathcal{V}_{d_l}]} \epsilon_{ij} \right).$$
(2.4)

Example 2.2. Consider the network represented by the directed graph of example 2.1. (See Figure 2.2.) For the s - d cut specified by the s-set $\mathcal{V}_s = \{1, 3\}$, the cut-capacity is

$$C(\mathcal{V}_s) = 1 - \epsilon_{12} + 1 - \epsilon_{32}\epsilon_{34}.$$

Looking at this example, we see that all edges in the cut-set that originate from a common node, i.e., edges (3, 2) and (3, 4), together contribute a value of one minus the product of their erasure probabilities, i.e., $1 - \epsilon_{32}\epsilon_{34}$ to the cut-capacity. This observation holds in general for wireless erasure networks.

Example 2.3. As another example, consider the network shown in Figure 2.3 with one source s = 1 and one destination d = 6. The cut-capacity corresponding to the s - d cut specified by $\mathcal{V}_s = \{1, 3, 4\}$ is $C(\mathcal{V}_s) = 1 - \epsilon_{12} + 1 - \epsilon_{46} + 1 - \epsilon_{35}\epsilon_{32}$.



Figure 2.3: For the cut-set specified by the s-set $\mathcal{V}_s = \{1, 3, 4\}$, the cut-capacity is $C(\mathcal{V}_s) = 1 - \epsilon_{12} + 1 - \epsilon_{46} + 1 - \epsilon_{35}\epsilon_{32}$.

2.4 Problem Statement

We next define the class of block codes considered. A $(\lceil 2^{nR_1} \rceil, \dots, \lceil 2^{nR_{|S|}} \rceil, n)$ code for the multicast problem in a wireless erasure network described in the previous sections consists of the following components:

- A set of integers W^(si) = {1, 2, ..., [2^{nRi}]} for each source node s_i ∈ S. W^(si) represents the set of message indices corresponding to node s_i. w^(s) denotes the message of source s ∈ S. We assume that the messages are equally likely and independent.
- A set of encoding functions $\{f_{i,t}\}_{t=1}^n$ for each node $i \in \mathcal{V}$, where

$$x_{i,t} = f_{i,t}(w^{(i)}, y_i^{t-1})$$

is the signal transmitted by node i at time t. Note that $x_{i,t}$ is a function of the message $w^{(i)}$ that node $i \in \mathcal{V}$ wants to transmit in the current block ⁶ and all symbols received so far by node i from its incoming channels. If i is not a source node, we set $w^{(i)} = 0$ for all blocks and all times.

⁶The value of $w^{(i)}$ does not change in one block.

• A decoding function g_{d_i} at destination node $d_i \in \mathcal{D}$,

$$g_{d_i}: \mathcal{W}^{(d_i)} \times \mathcal{Y}^n_{d_i} \times \{0, 1\}^{n|\mathcal{E}|} \to \prod_{s \in \mathcal{S}} \mathcal{W}^{(s)}$$

such that

$$\underline{\hat{w}}_{d_i} = (\hat{w}_{d_i}^{(s)}, \quad s \in \mathcal{S}) = g_{d_i}(w^{(d_i)}, y_{d_i}^n, (z_{ij,t}, (i,j) \in \mathcal{E}, 1 \le t \le n)),$$
(2.5)

where $\hat{w}_{d_i}^{(s)}$ is the estimate of the message sent from source $s \in \mathcal{S}$ based on received signals at d_i , information source available at d_i^{7} , $w^{(d_i)}$, and also the erasure occurrences on all the links of the network in the current block.

Note that X_i, Y_{ij} , and Y_i all depend on the message vector $\underline{w} = (w^{(s)}, s \in S)$ that is being transmitted. Therefore, we will write them as $X_i(\underline{w}), Y_{ij}(\underline{w})$, and $Y_i(\underline{w})$ to specify what specific set of messages is transmitted.

Associated with every destination node $d \in \mathcal{D}$ and every information source $s \in \mathcal{S}$ is a probability that the message will not be decoded correctly:⁸

$$P_d^{(n)(s)} = \Pr\left(\hat{W}_d^{(s)} \neq W^{(s)}\right),\tag{2.6}$$

where $P_d^{(n)(s)}$ is defined under the assumption that all the messages are independent and are uniformly distributed over $\mathcal{W}^{(s)}, s \in \mathcal{S}$. The set of rates $(R_s, s \in \mathcal{S})$ is said to be achievable if there exist a sequence of $(\lceil 2^{nR_1} \rceil, \ldots, \lceil 2^{nR_{|\mathcal{S}|}} \rceil, n)$ codes such that $P_d^{(n)(s)} \to 0$ as $n \to \infty$ for all $s \in \mathcal{S}$ and $d \in \mathcal{D}$. The capacity region is the closure of the set of achievable rates.

⁷If $d_i \notin \mathcal{S}$ without loss of generality, we set $w^{(d_i)} = 0$ and $\mathcal{W}^{(d_i)} = \{0\}$ for all blocks.

⁸Note that if d is a source node, we assume without loss of generality that $P_d^{(n)(d)} = 0$.

2.5 Main Results

In this section we present the main results of this chapter.

Theorem 2.1. Consider a single source/single destination wireless erasure network described by the directed acyclic graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ and the assumptions of Section 2.3. Let $s \in \mathcal{V}$ and $d \in \mathcal{V}$ denote the network's source and destination, respectively. Then the capacity of the network with side-information at the destination is given by the value of the minimum value s-d cut. More precisely, we have

$$C = \min_{\mathcal{V}_s: \mathcal{V}_s \text{an } s - d \text{ cut}} C(\mathcal{V}_s).$$
(2.7)

Remark 2.1. The results presented above are stated for erasure wireless networks with broadcast property (and no interference). However, based on these results, it is possible to derive the capacity of multicast problems over error-free networks (with the broadcast property and without interference), with or without capacitated links.

Remark 2.2. Although we have assumed that the erasure events across the network are independent, the capacity results also hold for the case when the erasure events are correlated, i.e. Z_{ij} , $(i, j) \in \mathcal{E}$ are dependent on each other. In that case the definition of the cut capacity should be modified to

$$C(\mathcal{V}_s) = \sum_{i \in \mathcal{V}_s^*} \left(1 - \Pr\left(Z_{ij} = 0, \ j : (i, j) \in [\mathcal{V}_s, \mathcal{V}_s^c]\right) \right).$$
(2.8)

Example 2.4. Recall the single source / single destination network of example 2.2. (See Figure 2.2.) By Theorem 2.1, the capacity of this network is

$$C = \min\{1 - \epsilon_{12} + 1 - \epsilon_{32}\epsilon_{34}, 1 - \epsilon_{34} + 1 - \epsilon_{24}, 1 - \epsilon_{12}\epsilon_{13}, 1 - \epsilon_{13} + 1 - \epsilon_{24}\}.$$

The following theorems generalize the single source/ single destination result to

general multicast problems.

Theorem 2.2. Consider a multiple source/single destination wireless erasure network described by directed acyclic graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ and the assumptions of Section 2.3. Suppose that the destination requests all of the information from all of the sources. Let $\mathcal{S} \subset \mathcal{V}$ and $d \in \mathcal{V}$ denote the set of source nodes and the destination node, respectively. The capacity region of the network with side-information provided at the destination is given by

$$C(\mathcal{G}, \mathcal{S}, d) \stackrel{\triangle}{=} \left\{ (R_s, s \in \mathcal{S}) \middle| 0 \le \sum_{s \in \mathcal{V}' \cap \mathcal{S}} R_s \le C(\mathcal{V}') \qquad \forall \ \mathcal{V}' \subset \mathcal{V} - \{d\} \right\}.$$
(2.9)

In other words, the total rate of information transmission to d across any cut $[\mathcal{V}', \mathcal{V}_d]$ should not exceed the cut-capacity of that cut.

Example 2.5. Consider the network shown in Figure 2.4 with two sources $\{1, 2\}$ and one destination $\{3\}$. Then according to Theorem 2.2, the capacity region is

$$\{(R_1, R_2) \in \mathbb{R}^+ \times \mathbb{R}^+ | R_1 \le 1 - \epsilon_{12}\epsilon_{13}, R_2 \le 1 - \epsilon_{23}, R_1 + R_2 \le 1 - \epsilon_{23} + 1 - \epsilon_{13}\}.$$

Theorem 2.3. Consider a multicast problem with multiple sources and multiple destinations. Let $S, D \subset V$ denote the set of source nodes and destination nodes, respectively. The capacity region of the network with side-information is given by the intersection of the capacity regions of the multicast problem between the sources and each of the destinations, i.e,

$$C(\mathcal{G}, \mathcal{S}, \mathcal{D}) = \bigcap_{d \in \mathcal{D}} C(\mathcal{G}, \mathcal{S}, d).$$
(2.10)

Corollary 2.3. Consider a multicast problem with one source denoted by s and multiple destinations denoted by $d_1, \ldots, d_{|\mathcal{D}|}$. The capacity of the network is given by the minimum value of the cuts between the source node and any of the destinations, i.e.,

$$C = \min_{d_i \in \mathcal{D}} \min_{\mathcal{V}_s: s - d_i \, cut} C(\mathcal{V}_s).$$

Example 2.6. Consider the network shown in Figure 2.2. Suppose that we are decoding at node 2 and 4, i.e., $\mathcal{D} = \{2, 4\}$. Based on Corollary 2.3, the capacity of this network is

$$C = \min\{1 - \epsilon_{12} + 1 - \epsilon_{32}, 1 - \epsilon_{34} + 1 - \epsilon_{24}, 1 - \epsilon_{12}\epsilon_{13}, 1 - \epsilon_{13} + 1 - \epsilon_{24}\}.$$

The above results show that the capacity region for multicast problems over wireless erasure networks has a max-flow min-cut interpretation. This result is similar to multicast problems in wireline networks [9]; however the definition of the cut-capacity is different. Recall from [9] that in wireline networks, the cut-capacity is the sum of the capacities of the links in the cut-set. Since wireless erasure networks incorporate broadcast, the cut-capacity is the sum of the capacities of each broadcast system that operates across the cut.

The next theorem states that linear network coding is sufficient for achieving the capacity region.

Theorem 2.4. Consider a multicast problem with multiple sources and multiple destinations. Then, any rate vector in the capacity region $C(\mathcal{G}, \mathcal{S}, \mathcal{D})$ of the network defined in Theorem 2.3 is achievable with linear block coding.

In the next section, we prove Theorems 2.1, 2.2, and 2.3. In Section 2.7, we look at the performance of the network using random linear coding and prove Theorem 2.4.



Figure 2.4: A wireless erasure network with two sources, $S = \{1, 2\}$, and one destination, $D = \{3\}$.

2.6 Proof of Theorems

2.6.1 Proof of Theorems 1 and 2

In this section we prove the results stated for multi-source/ single destination network problems. We start by proving the converse.

2.6.1.1 Converse

We prove the converse part by considering perfect cooperation among subsets of nodes. Consider the cut specified by d-set \mathcal{V}_d . Let all of the nodes in \mathcal{V}_d and all of the nodes in \mathcal{V}_d^c cooperate perfectly, i.e., each node has access to all of the information known to nodes in its set. In this case, we have a multiple input, multiple output point-to-point erasure channel. Consider all source nodes in \mathcal{V}_d^c . Then, clearly the sum-rate of these source nodes must be less than the capacity of the multiple input multiple output point-to-point erasure channel. The capacity of this point-to-point communication channel is

$$C_{col} = \max_{P(x_i, i \in \mathcal{V}_d^{c^*})} I((X_i, i \in \mathcal{V}_d^{c^*}); (Y_{ij}, (i, j) \in [\mathcal{V}_d^c, \mathcal{V}_d])).$$

Since the channels are independent and memoryless, the mutual information is maximized when the different X_i s are i.i.d. and uniform on the input alphabet \mathcal{X} . In this case, the above mutual information equals the cut-capacity corresponding to the cut-set $[\mathcal{V}_d^c, \mathcal{V}_d]$, i.e.,

$$C_{col} = C(\mathcal{V}_d^c).$$

Therefore, for any cut-set $[\mathcal{V}_d^c, \mathcal{V}_d]$, the sum-rate of the information sources in set \mathcal{V}_d^c satisfies

$$\sum_{s \in \mathcal{S} \cap \mathcal{V}_d^c} R_s \le C_{col} = C(\mathcal{V}_d^c).$$

The complete analysis appears in [57]. The proof follows the same lines as the maxflow min-cut upper bound of Cover and Thomas for multi-terminal networks [53, Sec. 14.10].

2.6.1.2 Achievability

In this section we prove that all of the rates arbitrarily close to rates in the capacity regions given in Theorems 2.1 and 2.2 are achievable for a multiple sources/ single destination multicast problem. We next use random coding techniques to show this result.

We employ random block codes in the network. Each node transmits the next block of n symbols only after it has received all n symbols corresponding to the present block from each of its incoming channels. Let L_{max} denote the length of the longest path from a source to the destination in the network. Since each transmission introduces one unit of time delay, the maximal delay between the transmission of a message from one source and to its receipt at the destination using block codes of length n is nL_{max} . We do not use any information from previously decoded blocks to decode the current set of messages. Also note that since our model assumes that the reception is interference-free, there is no confusion up among different blocks at any node. Therefore, if the network operates for nB units of time (i.e., B blocks of length-n symbols), then the destination has received all of the information required for decoding the $B - L_{\text{max}}$ first messages transmitted from each source $s \in S$, i.e., $w_b^{(s)}, b = 1, \ldots, B - L_{\text{max}}$. Since the network size is finite, as $B \to \infty$, for fixed n, the rate $R_s \frac{B-L_{\text{max}}}{B}$ approaches R_s .⁹

The same codebook and encoding and decoding functions are used for all the blocks. We explain the coding scheme for transmitting one set of messages from the sources to the destination. Below we describe the encoding and decoding processes.

• Codebook Generation and Encoding: For each node $i \in \mathcal{V}$, the encoding function

$$f_i: \mathcal{W}^{(i)} \times \mathcal{Y}^n_i \to \mathcal{X}^n$$

is generated randomly as follows. For each $y^n \in \mathcal{Y}_i^n$ and for each $w^{(i)} \in \mathcal{W}^{(i)}$ we draw the symbols of $f_i(w^{(i)}, y^n) \in \mathcal{X}^n$ randomly and independently according to a binary Bernoulli distribution with parameter 1/2. Thus, the channel input at node $i \in \mathcal{V}$ is $X_i^n = f_i(w^{(i)}, y^n)$ when the message at node i is $w^{(i)}$ and the incoming sequence is $y^n \in \mathcal{Y}_i^n$. The destination has perfect knowledge of all the encoding functions $f_i(.), i \in \mathcal{V}$ thus generated.¹⁰

• **Decoding:** The destination "simulates" the network to decode the messages. Suppose that message vector $\underline{w}_0 = (w_0^{(s)}, s \in S)$ is transmitted and $y_d^n(\underline{w}_0)$ is received at destination d. By assumption, the receiver knows the erasure locations on all the links of the network, i.e., $(z_{ij}^n, (i, j) \in \mathcal{E})$. Having all of the

⁹We could also consider the case when different sources transmit different numbers of messages in *B* block uses. In that case, if L_s denotes the longest path from $s \in S$ to the destination, we could transmit $B - L_s$ messages from information source *s* to the destination. However, for simplicity of notation and analysis we assume that all of the nodes send the same number of messages in a synchronized fashion.

¹⁰Note that the encoding functions thus constructed satisfy a causality condition that is more strict than what is defined in Section 2.4. Here, each transmitted block is only a function of immediately previous block of received symbols. In Section 2.3, each transmitted symbol could be a function of all previous symbols.

erasure locations and all of the encoding functions applied at different nodes in the network,¹¹ the destination can compute the values of $X_i^n(\underline{w}), Y_{ij}^n(\underline{w})$ and $Y_i^n(\underline{w})$ for all nodes and edges for any $\underline{w} \in \prod_{s \in S} \mathcal{W}^{(s)}$. If there exists a unique message vector $\underline{w} \in \prod_{s \in S} \mathcal{W}^{(s)}$ such that the computed value of $Y_d^n(\underline{w})$ equals the value $y_d^n(\underline{w}_0)$ of the received signal at the destination, then \underline{w} is declared as the decoder output. Otherwise, the decoder declares an error.

Since the computed value of $Y_d^n(\underline{w}_0)$ for transmitted message \underline{w}_0 always matches the received signal at the destination, an error occurs if and only if there is another message vector $\underline{w} \neq \underline{w}_0$ for which $Y_d^n(\underline{w}) = Y_d^n(\underline{w}_0) = y_d^n(\underline{w}_0)$. In the next section we compute the probability of this event and show that for large blocks this probability can be made arbitrarily close to zero provided that the rate vector $(R_s, s \in S)$ is inside the capacity region described in Theorems 2.1 and 2.2.

2.6.1.3 Probability of Error

Let $\Pr(err)$ be the probability of error averaged over all possible functions f_i . In other words, if $P_e^{(n)}$ is the probability that $\hat{w}_0^{(s)}$, the destination's estimate of the transmitted message \underline{w}_0 , is not equal to \underline{w}_0 , then $\Pr(err)$ is the expected value of $P_e^{(n)}$ over all possible encoding functions at all nodes.¹² More precisely,

$$P_e^{(n)} = \Pr\left(\exists s \in \mathcal{S} \text{ s.t. } \hat{w}_0^{(s)} \neq w_0^{(s)}\right),$$

and $\Pr(err) = \mathbb{E} P_e^{(n)}$. Because of the symmetry of the code construction,

 $\Pr(err) = \Pr(err|\underline{W} = \underline{w}_0 \text{ is transmitted}), \qquad (2.11)$

 $^{^{11}\}mathrm{We}$ also assume that the destinations knows the topology of the network.

¹²Note that if $P_e^{(n)}$ goes to zero as n grows larger, so will $P_d^{(n)(s)}$ of (2.6) for every $s \in \mathcal{S}$.

where $\underline{W} = (W^{(s)}, \quad s \in \mathcal{S})$. Therefore we will find the average probability of error when message vector \underline{w}_0 is transmitted from the sources. Recall the notation $X_i^n(\underline{w}_0)$ and $Y_i^n(\underline{w}_0)$ and $Y_{ij}^n(\underline{w}_0)$, $(i, j) \in \mathcal{E}$. For each $\underline{w} \in \underline{\mathcal{W}} \stackrel{\triangle}{=} \prod_{s \in \mathcal{S}} \mathcal{W}^{(s)}, \ \underline{w} \neq \underline{w}_0$, define the following event:

$$E(\underline{w}) = \{Y_d^n(\underline{w}) = Y_d^n(\underline{w}_0)\}.$$
(2.12)

Let $\mathcal{A}_{\delta}^{(n)}(i)$ be the event that the erasure locations on the channels going out of node i are jointly δ -strongly typical, i.e.,

$$\mathcal{A}_{\delta}^{(n)}(i) = \{ (z_{ij}^n, j : (i, j) \in \mathcal{E}) \text{ are jointly } \delta \text{-strongly typical} \},\$$

[1, Equation (13.107)] and define

$$\mathcal{A}^{(n)}_{\delta} = \bigcap_{i=1}^{|\mathcal{V}|} \mathcal{A}^{(n)}_{\delta}(i).$$

Note that by the weak law of large numbers [53], $\Pr(\mathcal{A}^{(n)}_{\delta}(i)) \to 1 \text{ as } n \to \infty$, and hence for all $\delta > 0$,

$$\Pr\left(\mathcal{A}_{\delta}^{(n)}\right) \ge 1 - |\mathcal{V}|\delta,$$

for *n* sufficiently large. Using the definition of the above events, Pr(err) can be written as

$$\Pr(err)$$

$$= \Pr(err|\underline{W} = \underline{w}_{0})$$

$$= \Pr\left(\bigcup_{\underline{w}\in\underline{\mathcal{W}}-\{\underline{w}_{0}\}} E(\underline{w})\right)$$

$$= \Pr\left(\bigcup_{\underline{w}\in\underline{\mathcal{W}}-\{\underline{w}_{0}\}} E(\underline{w})|\mathcal{A}_{\delta}^{(n)}\right) \Pr\left(\mathcal{A}_{\delta}^{(n)}\right) + \Pr\left(\bigcup_{\underline{w}\in\underline{\mathcal{W}}-\{\underline{w}_{0}\}} E(\underline{w})|\mathcal{A}_{\delta}^{(n)^{c}}\right) \Pr\left(\mathcal{A}_{\delta}^{(n)^{c}}\right)$$

$$\leq \sum_{\underline{w}\in\underline{\mathcal{W}}-\{\underline{w}_{0}\}} \Pr\left(E(\underline{w})|\mathcal{A}_{\delta}^{(n)}\right) + |\mathcal{V}|\delta.$$
(2.13)

Therefore, using strong typicality and the union bound on the probability of events, we only look at network instantiations that are "strongly typical." We next bound the conditional probability of $E(\underline{w})$ given $\mathcal{A}_{\delta}^{(n)}$.

Corresponding to each cut in the network represented by d-set $\mathcal{V}_d \ni d$, define the following event:

$$B[\mathcal{V}_d] = \bigg(\bigcap_{i \in \mathcal{V}_d} \{Y_i^n(\underline{w}) = Y_i^n(\underline{w}_0)\}\bigg) \bigcap \bigg(\bigcap_{i \in \mathcal{V}_d^c} \{Y_i^n(\underline{w}) \neq Y_i^n(\underline{w}_0)\}\bigg).$$
(2.14)

The interpretation of the above event is as follows. By definition of $E(\underline{w})$, we know that the received signal at the destination is the same for \underline{w} and \underline{w}_0 , but $\underline{w} \neq \underline{w}_0$. Therefore, we can partition the nodes of the network into two sets: the "distinguishable" and the "indistinguishable" set. The "distinguishable" set contains all nodes for which the signal received at those nodes when \underline{w} is transmitted differs from the signal received when \underline{w}_0 is transmitted. All the other nodes for which the received signals for \underline{w} and \underline{w}_0 are the same are in the "indistinguishable" set. Clearly, these two sets define a cut. Event $B[\mathcal{V}_d]$ corresponds to the case when the "indistinguishable" set (containing d) is equal to $\mathcal{V}_d \subset \mathcal{V}$. Note that these events are all disjoint and also $E(\underline{w}) = \bigcup_{\mathcal{V}_d:d-set} B[\mathcal{V}_d]$.

Define

$$\mathcal{K}(\underline{w}) = \{s | s \in \mathcal{S}, w_0^{(s)} \neq w^{(s)}\}$$

$$(2.15)$$

to be the subset of source nodes for which the corresponding messages in \underline{w} and \underline{w}_0 are different. Set $\mathcal{K}(\underline{w})$ is not empty since $\underline{w}_0 \neq \underline{w}$ by assumption. In what follows we bound the probability of event $B[\mathcal{V}_d]$ by considering the edges in the cut-set $[\mathcal{V}_x, \mathcal{V}_x^c]$, where $\mathcal{V}_x \stackrel{\triangle}{=} \mathcal{V}_d^c \cup \mathcal{K}(\underline{w})$. Note that \mathcal{V}_x^c is a *d*-set since if the destination is a source of information, it is aware of the message it has transmitted, and so $d \notin \mathcal{K}(\underline{w})$.

Consider any edge $(i, j) \in [\mathcal{V}_x, \mathcal{V}_x^c]$. We know that the transmitted signal $X_i^n = f_i(W^{(i)}, Y_i^n)$ from node *i* is a function of the message it wants to transmit, $W^{(i)}$, and

the received signal at its incoming edges, Y_i^n . For any node i in $\mathcal{V}_d^c \cup \mathcal{K}(\underline{w})$, either the received signal Y_i^n or message $w^{(i)}$ is different for message vectors \underline{w} and \underline{w}_0 . Thus, for a randomly designed code, the transmitted signal by node i for message vector \underline{w} is independent of the corresponding X_i^n for message vector \underline{w}_0 . Using this observation we next bound the probability of the event $E(\underline{w})$ conditioned on $\mathcal{A}_{\delta}^{(n)}$.

$$\Pr\left(E(\underline{w})|\mathcal{A}_{\delta}^{(n)}\right) = \Pr\left(\bigcup_{\mathcal{V}_{d}:d\text{-}set} B[\mathcal{V}_{d}]|\mathcal{A}_{\delta}^{(n)}\right) = \sum_{\mathcal{V}_{d}:d\text{-}set} \Pr\left(B[\mathcal{V}_{d}]|\mathcal{A}_{\delta}^{(n)}\right)$$

$$= \sum_{\mathcal{V}_{d}:d\text{-}set} \Pr\left(\left(\bigcap_{j\in\mathcal{V}_{d}} \{Y_{j}^{n}(\underline{w}) = Y_{j}^{n}(\underline{w}_{0})\}\right) \bigcap\left(\bigcap_{i\in\mathcal{V}_{d}} \{Y_{i}^{n}(\underline{w}) \neq Y_{i}^{n}(\underline{w}_{0})\}\right)|\mathcal{A}_{\delta}^{(n)}\right)$$

$$\stackrel{(a)}{\leq} \sum_{\mathcal{V}_{x}:\mathcal{K}(\underline{w})\subset\mathcal{V}_{x}} \Pr\left(\bigcap_{i,j:(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]} \{(w^{(i)},Y_{i}^{n}(\underline{w})) \neq (w_{0}^{(i)},Y_{i}^{n}(\underline{w}_{0})),Y_{ij}^{n}(\underline{w}) = Y_{ij}^{n}(\underline{w}_{0})\}|\mathcal{A}_{\delta}^{(n)})$$

$$\stackrel{(b)}{=} \sum_{\mathcal{V}_{x}:\mathcal{K}(\underline{w})\subset\mathcal{V}_{x}} \Pr\left(\bigcap_{i\in\mathcal{V}_{x}}\bigcap_{j:(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]} \{(w^{(i)},Y_{i}^{n}(\underline{w})) \neq (w_{0}^{(i)},Y_{i}^{n}(\underline{w}_{0})),Y_{ij}^{n}(\underline{w}) = Y_{ij}^{n}(\underline{w}_{0})\}|\mathcal{A}_{\delta}^{(n)})$$

Here (a) follows since $\Pr(A, B) \leq \Pr(A)$ for any events A and B. Instead of looking at equalities on every edge and every node of the network, we are looking at the nodes having an edge from $[\mathcal{V}_x, \mathcal{V}_x^c]$ connected to them, where $\mathcal{V}_x = \mathcal{V}_d^c \cup \mathcal{K}(\underline{w})$. (b) is clear from the definition of \mathcal{V}_x^* . Simplifying the probability of error further, we have

$$\begin{aligned} \Pr\left(E(\underline{w})|\mathcal{A}_{\delta}^{(n)}\right) & (2.16) \\ \stackrel{(c)}{=} & \sum_{\substack{\mathcal{V}_{x}\\\mathcal{K}(\underline{w})\subset\mathcal{V}_{x}}} \Pr\left(\bigcap_{i\in\mathcal{V}_{x}^{*}}(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]}\left\{Y_{ij}^{n}(\underline{w})=Y_{ij}^{n}(\underline{w}_{0})\right\}\middle|\mathcal{A}_{\delta}^{(n)},\bigcap_{i\in\mathcal{V}_{x}^{*}}\left\{(w^{(i)},Y_{i}^{n}(\underline{w}))\neq(w_{0}^{(i)},Y_{i}^{n}(\underline{w}_{0}))\right\}\right) \\ & \cdot\Pr\left((w^{(i)},Y_{i}^{n}(\underline{w}))\neq(w_{0}^{(i)},Y_{i}^{n}(\underline{w}_{0})),\forall i\in\mathcal{V}_{s}^{*}\right) \\ \leq & \sum_{\substack{\mathcal{V}_{x}\\\mathcal{K}(\underline{w})\subset\mathcal{V}_{x}}}\Pr\left(\bigcap_{i\in\mathcal{V}_{x}^{*}}(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]}\left\{Y_{ij}^{n}(\underline{w})=Y_{ij}^{n}(\underline{w}_{0})\right\}\middle|\mathcal{A}_{\delta}^{(n)},\bigcap_{i\in\mathcal{V}_{x}^{*}}\left\{(w^{(i)},Y_{i}^{n}(\underline{w}))\neq(w_{0}^{(i)},Y_{i}^{n}(\underline{w}_{0}))\right\}\right) \\ \stackrel{(d)}{=} & \sum_{\substack{\mathcal{V}_{x}:\mathcal{K}(\underline{w})\subset\mathcal{V}_{x}}}\prod_{i\in\mathcal{V}_{x}^{*}}\Pr\left(\bigcap_{j:(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]}\left\{Y_{ij}^{n}(\underline{w})=Y_{ij}^{n}(\underline{w}_{0})\right\}\middle|\mathcal{A}_{\delta}^{(n)},\left\{(w^{(i)},Y_{i}^{n}(\underline{w}))\neq(w_{0}^{(i)},Y_{i}^{n}(\underline{w}_{0}))\right\}\right),\end{aligned}$$

where (c) is clear from the definition of conditional probability and (d) follows from fact that averaged over all possible functions f_i , the conditional events shown in the equation are independent for different is in \mathcal{V}_x^* .

Now we bound the expression given in (2.16) for any node $i \in \mathcal{V}_x^*$. Note that since $(w^{(i)}, Y_i^n(\underline{w})) \neq (w_0^{(i)}, Y_i^n(\underline{w}_0))$ at node $i, X_i^n(\underline{w}) = f_i(w^{(i)}, Y_i^n(\underline{w}))$ and $X_i^n(\underline{w}_0) = f_i(w_0^{(i)}, Y_i^n(\underline{w}_0))$ are chosen independently and uniformly from $\{0, 1\}^n$. Therefore, the probability that they are the same in at least α_i specific locations is at most $2^{-\alpha_i}$. Looking at a fixed node $i, Y_{ij}^n(\underline{w}) = Y_{ij}^n(\underline{w}_0)$ for all j such that $(i, j) \in [\mathcal{V}_x, \mathcal{V}_x^c]$ only if all the locations where $X_i^n(\underline{w})$ and $X_i^n(\underline{w}_0)$ differ get erased on all these edges. Because of the δ -strong typicality of the erasure locations on edges $(i, j) \in [\mathcal{V}_x, \mathcal{V}_x^c]$, the number of locations at which erasure occurs on all the edges of interest, say $\alpha_i(\mathcal{V}_x)$, satisfies

$$\left|\frac{1}{n}\alpha_i(\mathcal{V}_x) - \Pr\left(Z_{ij} = 0, \ j: (i,j) \in [\mathcal{V}_x, \mathcal{V}_x^c]\right)\right| \le \frac{\delta}{2^{|\{j:(i,j)\in[\mathcal{V}_x, \mathcal{V}_x^c]\}|}} \le \delta.$$

Therefore, $X_i^n(\underline{w})$ and $X_i^n(\underline{w}_0)$ cannot differ in more than $n(\Pr(Z_{ij} = 0, j : (i, j) \in [\mathcal{V}_x, \mathcal{V}_x^c]) + \delta)$ locations, and the probability of this event is no more than

$$\exp\left(-n(1 - \Pr\left(Z_{ij} = 0, \ j: (i, j) \in [\mathcal{V}_x, \mathcal{V}_x^c]\right) - \delta\right)\right) = \exp\left(-n(1 - \prod_{j:(i, j) \in [\mathcal{V}_x, \mathcal{V}_x^c]} \epsilon_{ij} - \delta)\right).$$
(2.17)

Combining this with the last equation of (2.16) gives ¹³

$$\Pr\left(E(\underline{w})|\mathcal{A}_{\delta}^{(n)}\right) \leq \sum_{\mathcal{V}_{x}:\mathcal{K}(\underline{w})\subset\mathcal{V}_{x}}\prod_{i\in\mathcal{V}_{x}}\exp\left(-n(1-\prod_{j:(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]}\epsilon_{ij}-\delta)\right)$$

$$=\sum_{\mathcal{V}_{x}:\mathcal{K}(\underline{w})\in\mathcal{V}_{x},d\notin\mathcal{V}_{x}}2^{n|\mathcal{V}_{x}^{*}|\delta}\cdot\exp\left(-n\sum_{i\in\mathcal{V}_{x}^{*}}(1-\prod_{j:(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]}\epsilon_{ij})\right)$$

$$\leq 2^{n|\mathcal{V}|\delta}\sum_{\mathcal{V}_{x}:\mathcal{K}(\underline{w})\subset\mathcal{V}_{x},d\notin\mathcal{V}_{x}}2^{-nC(\mathcal{V}_{x})}.$$
(2.18)

¹³Using (2.17) it can be easily verified that the arguments that follow will exactly go through for correlated erasure events with cut-capacity, $C(\mathcal{V}_x)$, defined as in 2.8.

Combining (2.13) and (2.18) together gives

Р

$$\begin{aligned} \mathbf{r} (err) &\leq |\mathcal{V}|\delta + 2^{n|\mathcal{V}|\delta} \sum_{\underline{w} \in \underline{\mathcal{W}} - \{\underline{w}_0\}} \sum_{\substack{\nu_x: \mathcal{K}(\underline{w}) \subset \mathcal{V}_x \\ d \notin \mathcal{V}_x}} 2^{-nC(\mathcal{V}_x)} \\ &= |\mathcal{V}|\delta + 2^{n|\mathcal{V}|\delta} \sum_{\mathcal{K} \subset \mathcal{S}} \sum_{\substack{w \in \underline{\mathcal{W}} - \{\underline{w}_0\} \\ \mathcal{K}(\underline{w}) = \mathcal{K}}} \sum_{\substack{v_x: \mathcal{K}(\underline{w}) \subset \mathcal{V}_x \\ d \notin \mathcal{V}_x}} 2^{-nC(\mathcal{V}_x)} \\ &= |\mathcal{V}|\delta + 2^{n|\mathcal{V}|\delta} \sum_{\mathcal{K} \subset \mathcal{S}} \sum_{\substack{v_x: \mathcal{K} \subset \mathcal{V}_x \\ d \notin \mathcal{V}_x}} \sum_{\substack{w \in \underline{\mathcal{W}} - \{\underline{w}_0\} \\ \mathcal{K}(\underline{w}) = \mathcal{K}}} 2^{-nC(\mathcal{V}_x)} \\ &= |\mathcal{V}|\delta + 2^{n|\mathcal{V}|\delta} \sum_{\mathcal{K} \subset \mathcal{S}} \sum_{\substack{v_x: \mathcal{K} \subset \mathcal{V}_x \\ d \notin \mathcal{V}_x}} \prod_{\substack{w \in \underline{\mathcal{W}} - \{\underline{w}_0\} \\ \mathcal{K}(\underline{w}) = \mathcal{K}}} 2^{-nC(\mathcal{V}_x)} \\ &\stackrel{(a)}{\leq} |\mathcal{V}|\delta + 2^{n|\mathcal{V}|\delta} \sum_{\mathcal{K} \subset \mathcal{S}} \sum_{\substack{v_x: \mathcal{K} \subset \mathcal{V}_x \\ d \notin \mathcal{V}_x}} 2^{-n(C(\mathcal{V}_x) - \sum_{s \in \mathcal{K}} R_s)} \\ &\stackrel{(b)}{=} |\mathcal{V}|\delta + 2^{n|\mathcal{V}|\delta} \sum_{\mathcal{V}_x: \mathcal{V}_x \subset \mathcal{V} - \{d\}} \sum_{\mathcal{K} \subset \mathcal{S} \cap \mathcal{V}_x} 2^{-n(C(\mathcal{V}_x) - \sum_{s \in \mathcal{K}} R_s)} \\ &\stackrel{(c)}{\leq} |\mathcal{V}|\delta + 2^{n|\mathcal{V}|\delta} \sum_{\mathcal{V}_x: \mathcal{V}_x \subset \mathcal{V} - \{d\}} 2^{|\mathcal{V}_x \cap \mathcal{S}|} 2^{-n(C(\mathcal{V}_x) - \sum_{s \in \mathcal{S} \cap \mathcal{V}_x} R_s)}, \quad (2.19) \end{aligned}$$

where we have used the inequality $\lceil 2^{nR_s} \rceil - 1 \leq 2^{nR_s}$ in (a). Also, (b) is derived by changing the order of summation, and (c) follows from bounding $\sum_{s \in \mathcal{K}} R_s$ by $\sum_{s \in \mathcal{V}_x \cap \mathcal{S}} R_s$ in (c). Now, by assumption the rate vector $(R_s, s \in \mathcal{S})$ is inside the capacity region given in Theorem 2.2. Therefore, for any partition of the nodes into \mathcal{V}_x and $\mathcal{V}_x^c \ni d$ we have $C(\mathcal{V}_x) - \sum_{s \in \mathcal{S} \cap \mathcal{V}_x} R_s > 0$. Therefore the exponent in the last term of the above summation is negative. The above result holds for any $\delta > 0$ and n sufficiently large. By letting $n \to \infty$ and $\delta \to 0$, we can make the upper bound on the probability of error arbitrarily close to zero. Now, by standard coding arguments we conclude that there exists some deterministic choice of encoding functions that has an arbitrarily small probability of error for the rates in the achievable rate region $C(\mathcal{G}, \mathcal{S}, d)$.

42

2.6.2 Proof of Theorem 3

In this section we outline the proof of Theorem 2.3. The analysis is very similar to Theorem 2.1. The converse part is straightforward. We know that the sources can be recovered at all the destinations; therefore we have the same argument as the converse part of Theorem 1 for the sources and any of the destinations. In particular, for any destination d_i , $i \in \mathcal{D}$, we have $(R_s, s \in \mathcal{S}) \in C(\mathcal{G}, \mathcal{S}, d_i)$. Therefore, any achievable rate vector should be in the intersection of these capacity regions, i.e,

$$(R_s, s \in \mathcal{S}) \in \cap_{d_i \in \mathcal{D}} C(\mathcal{G}, \mathcal{S}, d_i) = C(\mathcal{G}, \mathcal{S}, \mathcal{D}).$$

Hence, the converse part is done.

In order to prove the achievability of the above rates, we can use the random coding argument of Section 1. Note that averaged over all the codebooks and functions, the probability of error for each destination goes to zero. Therefore, using the union bound on probability of events, the probability of having an error in at least one destination (averaged over all the functions and codebooks) goes to zero. Using standard arguments, there exists some deterministic choice of codebooks and functions for which the probability of error in the network become arbitrarily small and that shows the achievability of the rates in $C(\mathcal{G}, \mathcal{S}, \mathcal{D})$ of Theorem 2.3 for the multiple destination case.

2.7 Linear Encoding

In Section 2.6.1.2 we showed the achievability of the capacity region as defined in Theorem 2.2 by using general random coding functions at different nodes of the network. In this section we restrict our attention to linear encoding schemes. The advantage of using linear encoding scheme is that the decoding process becomes much easier. In this case, the equivalent transfer function of the network from any source to any destination, having the erasure locations at that destination, is linear. Hence, decoding at the destination is simply forming and solving a linear system of equations.

In this section we show that linear encoders achieve capacity. Let us first define the linear block coding scheme with block length of n:

Recall that $\mathcal{W}^{(s)} = \{1, 2, \dots, \lceil 2^{nR_s} \rceil\}$ is the message set for information source $s \in \mathcal{S}$. We assume that different messages are equiprobable and independent of each other. For any $w^{(s)} \in \mathcal{W}^{(s)}$, let $\underline{b}(w^{(s)})$ be the length- nR_s binary expansion of $w^{(s)} - 1$.

The encoding operation is as follows:

Each node $i \in \mathcal{V}$ transmits *n* linear combinations of the non-erased symbols received from its incoming edges and the binary representation of the message it wants to transmit across the network. More precisely, node *i* generates a random binary matrix B_i of size $n \times n(d_I(i) + R_i)$, where $d_I(i)$ is the in-degree of node *i* and R_i is the rate of the codebook used at node *i* (in the case where *i* is not a source of information $R_i = 0$). Each element of B_i is drawn i.i.d. Bernoulli(1/2). For a given sequence *y*, let \tilde{y} be a sequence derived by replacing every *e* with 0. Note that \tilde{y} and *y* have the same length.¹⁴ If node *i* receives $Y_i^n = y_i^n$ on its incoming edges and wants to transmit message $w^{(i)}$, then it sends out $x_i = B_i \cdot [\underline{b}(w^{(i)}), \tilde{y}_i^n]^{\dagger}$. (Since the input-output relation at each node is linear, setting the erased symbols equal to zero is the same as finding linear combinations of only the non-erased bits.)

Each destination d knows all the matrices B_i and also the erasure locations Z^n on all the links across the network, since each received and transmitted symbol at any node is a linear combination of the elements of vector $b(\underline{w}) \stackrel{\triangle}{=} (\underline{b}(\underline{w}^{(s)}), s \in S)$. Therefore, each destination receives a collection of linear combinations of elements of $b(\underline{w})$. Using $\{B_i\}_{i\in\mathcal{V}}$ and Z^n , destination node d can construct the matrix that corresponds to the linear input-output relation of the network. We denote this matrix

¹⁴The corresponding mapping from alphabet $GF(q) \cup \{e\}$ to GF(q) again replaces e with 0. This variation is useful for packet erasure networks.

by $F(\{B_i\}, Z^n)$, giving $\widetilde{Y_d^n}(\underline{w}) = F(\{B_i\}, Z^n) \cdot b(\underline{w})^{\dagger}$. Note that matrix F is a function of different nodes' encoding matrices $\{B_i\}$ and Z^n .

Now, upon receiving $Y_d^n = y \in \{0, 1, e\}^{nd_I(d)}$, the destination node d looks (solves) for the message vector $\underline{w} \in \underline{\mathcal{W}} \stackrel{\triangle}{=} \prod_{s \in \mathcal{S}} \mathcal{W}^{(s)}$ such that $F(\{M_i\}, Z^n) \cdot b(\underline{w})^{\dagger} = \tilde{y}$. If there is a unique \underline{w} with this property, node d declares it as the transmitted message vector; otherwise it declares an error. Note that the actual transmitted message vector, say $\underline{w}_0 \in \underline{\mathcal{W}}$, always satisfies the above property. Therefore, an error occurs only if there is another message vector $\underline{w} \neq \underline{w}_0$ such that $Y_d^n(\underline{w}) = Y_d^n(\underline{w}_0) = y$.

2.7.1 Achievability Result for Linear Encoding

Looking at the achievability proof and probability of error analysis for general random coding in Sections 2.6.1.2 and 2.6.1.3, it can be easily verified that the linear case requires the same error events (2.12). Since the erasure vector Z^n is available at the destination, there is no difference between \tilde{Y}_i and Y_i , and we can determine one from the other. By expanding the conditional error event $E(\underline{w})$ given $A_{\delta}^{(n)}$ for different cuts in the network, all of the relations up to step (d) of equation (2.16) go through for the linear case. In fact, the relations up to step (d) only require the independence of encoding functions for different nodes of the network, which holds for the linear case. Now, we look at the following probability in (2.16):

$$P_{i} \stackrel{\triangle}{=} \Pr\left(\bigcap_{j:(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]} \{Y_{ij}^{n}(\underline{w}) = Y_{ij}^{n}(\underline{w}_{0})\} \middle| \mathcal{A}_{\delta}^{(n)}, \{(w^{(i)},Y_{i}^{n}(\underline{w})) \neq (w_{0}^{(i)},Y_{i}^{n}(\underline{w}_{0}))\}\right).$$

$$(2.20)$$

As in the general random coding argument, for a fixed i we have $Y_{ij}^n(\underline{w}) = Y_{ij}^n(\underline{w}_0)$ for all j such that $(i, j) \in [\mathcal{V}_x, \mathcal{V}_x^c]$, only if $X_i^n(\underline{w})$ and $X_i^n(\underline{w}_0)$ differ in locations where an erasure occurs on all the edges of the interest. Because of strong typicality, the number of these location is at most $n(\prod_{j: (i,j)\in[\mathcal{V}_x,\mathcal{V}_x^c]}\epsilon_{ij}+\delta)$. Therefore $X_i^n(\underline{w})$ and $X_i^n(\underline{w}_0)$ should be the same in at least $n(1-\prod_{j: (i,j)\in[\mathcal{V}_s,\mathcal{V}_s^c]}\epsilon_{ij}-\delta)$ locations. But by our encoding scheme this means that

$$B_i \cdot \underbrace{\left([w^{(i)}, Y_i^n(\underline{w})]^{\dagger} - [w_0^{(i)}, Y_i^n(\underline{w}_0)]^{\dagger} \right)}_{z}$$

should be zero in at least $n(1 - \prod_{j: (i,j) \in [\mathcal{V}_s, \mathcal{V}_s^c]} \epsilon_{ij} + \delta)$ specific locations. Also note that since $(w^{(i)}, Y_i^n(\underline{w})) \neq (w_0^{(i)}, Y_i^n(\underline{w}_0)), \underline{z}$ is a non-zero vector. From the above argument we have

$$P_{i} \leq \Pr\left(B_{i} \cdot \underline{z} \text{ be } 0 \text{ in at least } n\alpha_{i} \text{ specific locations } \middle| \underline{z} \neq 0 \right)$$

$$\stackrel{(a)}{\leq} 2^{-n\alpha_{i}} = 2^{-n(1-\prod_{j:(i,j)\in[\mathcal{V}_{x},\mathcal{V}_{x}^{c}]}\epsilon_{ij}-\delta)}, \qquad (2.21)$$

where $\alpha_i = 1 - \prod_{j:(i,j) \in [\mathcal{V}_x, \mathcal{V}_x^c]} \epsilon_{ij} - \delta$, and (a) follows from the following Lemma and its Corollary. Proof of this lemma is provided in [57].

Lemma 2.4. Let X be a non-zero vector of size $n \times 1$ from some finite field GF(q). Suppose that A is a random matrix of size $m \times n$ with i.i.d. components distributed uniformly over GF(q). Then, the coordinates of $Y = A \cdot X$ are i.i.d. uniform random variables over GF(q).

Corollary 2.5. The probability that $Y = A \cdot X$ is zero in k specific coordinates equals q^{-k} .

Now note that by replacing P_i in (2.20) and (2.16) with its bound from (2.21), we get the same bound as (2.18) for random linear codes. Therefore, linear operations are sufficient for achieving the capacity.

2.8 Discussion

2.8.1 Packet Size and Cycles

We considered binary erasure networks in this work. However, as mentioned earlier, the obtained results hold for any packet length (or more generally for any input alphabet size). The theorems stated in Section 2.5 give the maximum achievable rate per packet. Therefore, if one is interested in the maximum achievable rate per bit, assuming that the size of each packet is L bits (resp. the alphabet size is M) across the network, the capacity will be L (respectively $\log_2 M$) times the capacity as stated in the theorems.

Also, we assume that the graph representation of the wireless erasure network is acyclic. However, the upper bound derived in [57] does not rely on this assumption. By an approach similar to [9], [54], [23] (Section 11.5.2), it can be shown that the upper bound is still achievable, and therefore the capacity theorems holds. We do not get into this problem in detail here.

2.8.2 Side-information at Destination Nodes

The results stated up to now are based on the perfect knowledge of the erasure locations for each link of the network to be available at destination nodes.

Erasure channels are usually used in modeling networks for which there exists a mechanism by which the receiver (destination) can be informed of a packet dropping. Usually, this side-information is provided by using sequencing numbers in the packet header to detect lost packets. However, if we do not provide the destination with this side-information, even for the simplest case of point to point communication, the capacity is not known. In this case, the communication system is usually modeled by the *deletion channel*. This channel has been studied by some researchers, and lower and upper bounds on its capacity are found in [24, 25, 26].

Looking back at our network, for each block there are $n|\mathcal{E}|$ transmissions of packets across the network. Therefore, the erasure locations on the links of the network can be represented by $n|\mathcal{E}|$ bits. These $n|\mathcal{E}|$ should be provided to the destination through some mechanism. One approach is to use part of each packet as a header to transmit this information.

If the size of each packet is L bits, then based on our result we are able to send nCL bits across the network in a block of length n, where C is the minimum cut-capacity of the network. If the size of the packet, L, is large compared to the size of the network, or if the network is small, i.e., $|\mathcal{E}|$ is small, the amount of side-information required is negligible compared to the amount of information sent across the network. We should remark that if one is trying to map a real network to our model, the packet length and the probability of erasure are closely related, and there will be some trade-off between them.

A number of techniques can be used to reduce the required overhead for providing side-information at the destination(s). For instance, consider a wireless erasure network with one destination d. Let C denote the minimum cut-capacity for this node. Based on Theorem 2.1, this is the maximum achievable rate at d. Now consider Q to be the subset of nodes for which the minimum cut-capacity is greater than C. If we decode the messages completely and then re-encode them using random code-books, we can still achieve the capacity at destination d. However, doing this may reduce the amount of overhead required at the destination. As an example, let's look at a line network (Figure 2.5) from this point of view. The source node is the leftmost node, and the destination is the rightmost node. The minimum cut-capacity for the destination is less than or equal to the minimum cut-capacity for every other node. Therefore, the intermediate nodes can decode without degrading the performance at the destination d. Further, this approach decreases the amount of overhead required. In fact, for this special case, by decoding at every node no side-information from



Figure 2.5: A wireless erasure line network.

previous links is required at the destination. Another technique is scheduling among the nodes to minimize the average header size needed for sending across the required side-information. In this scheduling, for any link in the network we determine the nodes that should include the erasure locations on that link as a header in their transmitted packets.

Remark: A closer look at the achievability proof of Section 2.6.1.2 reveals that all that the destination nodes need to know is the mapping from the source nodes to their incoming signals for every instantiation of the network. (In other words, for every instantiation of the network, the destinations should be able to unambiguously compute their output for any given input to the network.) Any mechanism that provides destination nodes with the knowledge of this mapping will work. Providing the erasure locations for each link of the network is one possible mechanism. In the subsequent work of [21], another kind of side-information is considered for the linear encoding scenario. There, the side information is the global encoding vectors, which also allow for the input output mapping to be determined.

2.8.3 Achievable Rates Without Side-information

In this section we look at the achievable rate for single source / single destination wireless erasure networks under a number of coding schemes when the side-information is not available at the destination.

"Forward" and "Decode" Scheme

In this scheme we analyze the performance of wireless erasure networks when limited operations are allowed at each node. Consider a codebook C of rate R and block size n. This codebook is available at all the nodes and is used for encoding the information message. The source node uses this codebook to "encode" the information. We assume that all the other nodes are allowed to perform one of the following operations:

- Forward: A node operating at this mode forwards received strings unchanged.¹⁵
- Decode and re-encode: In this case, the node first decodes the message transmitted from the source node based on what it has received and the codebook C. Then, it sends out the codeword corresponding to that message in C across its outgoing links. In this way, each relay node acts as a "source" of information for other nodes in the network. The need for successful decoding at intermediate nodes may reduce R the rate of the codebook used.

The main observation is that since the source message is intended to be decoded only at destination nodes, decoding at one relay node may be sub-optimal. In fact, as it will be shown in Chapter 4, the distinguishing features of wireless media imply that decoding at every relay node and operating below the capacity of each link in the network can result in severe degradation in the achievable rate in the network (for more examples See [27, 55, 56]). For instance, for the erasure wireless networks considered here, the maximum rate when all the nodes are decoding the source message using codebook C is given by the minimum capacity of the links in the network, i.e.,

$$R_{AD} = 1 - \max_{(i,j)\in\mathcal{E}} \epsilon_{ij}, \qquad (2.22)$$

¹⁵In this scheme we consider a modified erasure channel between any two nodes in which the nodes can forward the erasure symbol without error. In other words, although similar to previous sections; however the nodes are not allowed to perform coding on erasure symbol, they can inform their local neighbors of erasure of packets.

where subscript AD refers to the all decoding case.

In the "forward" and "decode" scheme, instead of requiring all the nodes to decode the source message, we allow for another operation: "forward" ing. The objective is to find the optimal operation at every node so as to maximize the achievable rate, i.e., the rate of the codebook, in this network. In Chapter 4 we propose an efficient algorithm that finds the optimal rate for this scheme. We use R_{FD} to refer to the optimal rate using the "forward" and "decode" scheme.

Block Markov Superposition Coding

Cover and El Gamal [14] developed a coding strategy for general single relay channels based on block Markov coding and random partitioning. This strategy is generalized and used in a multiple relay setup in [15],[16].

Let $\pi(.)$ be a permutation on \mathcal{V} that fixes the source and destination nodes. This permutation determines the order in which the nodes decode and encode the information. Using this coding strategy, it is shown in [15] that we can achieve

$$R_{BM} = \max_{\pi(.)} \min_{1 \le t \le |\mathcal{V}| - 1} I(X_{\pi(1:t)}; Y_{\pi(t+1)} | X_{\pi(t+1:|\mathcal{V}| - 1)}),$$

where subscript BM is used to refer to block Markov coding, and $\pi(i : j) = {\pi(i), \pi(i+1), \ldots, \pi(j)}$. We should remark that one can choose any distribution on $(X_1, X_2, \ldots, X_{|\mathcal{V}|})$ in the above result. Applying this result to our case, we show in the following lemma that the maximum rate is achieved when the X_i s are independent and uniformly distributed.

Lemma 2.6. The maximum achievable rate using block Markov coding for wireless erasure networks is given by

$$R_{BM} = \min_{j \in \mathcal{V}} \sum_{i: (i,j) \in \mathcal{E}} (1 - \epsilon_{ij}).$$
(2.23)

Furthermore, this rate is achieved by applying independent coding at different nodes of the network.

Proof. As mentioned earlier, the achievable rate using block Markov coding is given by

$$R_{BM} = \max_{\pi(.)} \min_{1 \le t \le |\mathcal{V}| - 1} I(X_{\pi(1:t)}; Y_{\pi(t+1)} | X_{\pi(t+1:|\mathcal{V}| - 1)}),$$

where the maximization is over the joint distribution of the $(X_i, i \in \mathcal{V})$ and $\pi(.)$ permutations of the nodes that keep source and destination fixed. Now, for any mutual information term in the above equation we have

$$I(X_{\pi(1:t)}; Y_{\pi(t+1)} | X_{\pi(t+1:|\mathcal{V}|-1)})$$

$$= H(Y_{\pi(t+1)} | X_{\pi(t+1:|\mathcal{V}|-1)}) - H(Y_{\pi(t+1)} | X_{\pi(t+1:|\mathcal{V}|-1)}, X_{\pi(1:t)})$$

$$\leq H(Y_{\pi(t+1)}) - H(Y_{\pi(t+1)} | X_{i}, \quad i \in \mathcal{V})$$

$$\stackrel{(a)}{=} H(Y_{\pi(t+1)}) - H(Y_{\pi(t+1)} | X_{i}, \quad (i, \pi(t+1)) \in \mathcal{E})$$

$$\stackrel{(b)}{\leq} \sum_{i: (i, \pi(t+1)) \in \mathcal{E}} H(Y_{i\pi(t+1)}) - H(Y_{i\pi(t+1)} | X_{i})$$

$$= \sum_{i: (i, \pi(t+1)) \in \mathcal{E}} I(Y_{i\pi(t+1)}; X_{i})$$

$$\stackrel{(c)}{\leq} \sum_{i: (i, \pi(t+1)) \in \mathcal{E}} 1 - \epsilon_{i\pi(t+1)} \qquad (2.24)$$

where

- (a),(b) follows from the fact the network is memoryless; therefore given X_i (i, j) ∈
 E, Y_{ij}s are independent from each other and also output of other nodes.
- (c) follows from the fact that the capacity of an erasure channel with probability of erasure ϵ_{ij} is $1 \epsilon_{ij}$.

Using (2.24), we have

$$R_{BM} \le \min_{j \in \mathcal{V}} \sum_{i: (i,j) \in \mathcal{E}} 1 - \epsilon_{ij}.$$



Figure 2.6: A simple network.

Now we can easily verify that by choosing X_i s independent and uniformly distributed and by considering a permutation that is faithful to partial ordering of the nodes defined in Section 2.2.2, we can achieve the right-hand side of the above equation.

The above lemma (and in particular (2.23)) suggests that the achievable rate in block Markov scheme is constrained by the minimum of the sum-capacities of incoming edges to any node in the network. This constraint is less severe than the all decoding case in (2.22). Here, instead of requiring all the links to be error-free, we only require that the relay nodes be able to decode the information. However, the achievable rate is still constrained by the capability of each relay node to decode the information. The example that follows demonstrates that block Markov coding is not always efficient in our network.

There are other strategies such as compress-forward and also partial decoding at the relay nodes that can be implemented in the network [14, Theorem 6],[28],[16]. Finding the achievable rates for these schemes explicitly and in terms of parameters of the network is usually intractable since the expressions for these schemes are not simple and involve a number of auxiliary random variables. In the rest of this section we compare the performance of these schemes for a very simple but interesting erasure broadcast network. The capacity of this simple network is not known to our knowledge.

A Simple Example

The network under study here has a graph representation shown in Figure 2.6. It has two relay nodes and one destination. We assume that the relay nodes are identical in the sense that their connections to the source (respectively destination) have the same probability of erasure equal to ϵ_0 (respectively ϵ_1). The capacity of the network given side-information at the destination is $C_{SI} = \min\{1 - \epsilon_0^2, 2 - 2\epsilon_1, 2 - \epsilon_1 - \epsilon_0\}$. Using the "forward" and "decode" strategy, the maximum achievable rate is given by [27]

$$R_{FD} = \max\{1 - \max\{\epsilon_1, \epsilon_0\}, 1 - (1 - (1 - \epsilon_0)(1 - \epsilon_1))^2\}.$$

Using (2.23), the block Markov coding scheme achieves rates up to

$$R_{BM} = \min\{1 - \epsilon_0, 2(1 - \epsilon_1)\}$$

Another strategy that can be used is for the relay nodes to encode and compress their received signals, Y_i , with rate R_i and send them to the destination node reliably. Since the received signals at the relay nodes are correlated, the Slepian-Wolf encoding scheme can be used [29]. However, this scheme works only if the capacity of the channel between relay node *i* and the destination is larger than R_i . Combining this with the Slepian-Wolf rate region [29], we should have

$$H(Y_2|Y_3) \le 1 - \epsilon_1$$
 and $H(Y_2, Y_3) \le 2(1 - \epsilon_1)$.

If the above conditions are satisfied, the destination will have access to both observations Y_2 and Y_3 , and therefore we can achieve a rate of $I(X_1; Y_2, Y_3)$ in the network.

Now suppose that the distribution on the input signal, X_1 , is given by vector <u>p</u>.

It can be verified that $H(Y_2|Y_3) = H(\epsilon_0) + \epsilon_0(1-\epsilon_0)H(\underline{p}), H(Y_3, Y_2) = (1-\epsilon_0^2)H(\underline{p}) + 2H(\epsilon_0)$, and $I(X_1; Y_2, Y_3) = (1-\epsilon_0^2)H(\underline{p})$. By choosing the probability distribution appropriately, we can achieve

$$R_{SW} = \min\{1 - \epsilon_0^2, 2\{1 - \epsilon_1 - H(\epsilon_0)\}^+\}$$

where $\{x\}^+ = \max\{0, x\}$, and the subscript SW is used to refer to the Slepian-Wolf coding scheme. Note that this scheme does not work if the quality of the channels from the relay nodes to the destinations is low, i.e., if ϵ_1 is large. In this case the second term in the above formula becomes zero, and therefore R_{SW} equals zero.

We have plotted the performance of the above-mentioned strategies for four different scenarios in Figure 2.7. In Figure 2.7.(a), we plot the performance for $\epsilon_0 =$ $\epsilon_1 \in [0,1]$. For small values of ϵ_0 , the Slepian-Wolf strategy achieves capacity. Unfortunately, this approach performs poorly for large values of ϵ_0 since the quality of the channel between relay nodes and the destination is not good enough to pass the compressed data reliably. Figure 2.7.(b) shows the results for $\epsilon_1 = \epsilon_0^3$. This choice corresponds to the case when the quality of the channels from the relay nodes to the destination is better than the source-to-relay connections. Note that the performance of the block Markov scheme is not good because of the rate constraint introduced by decoding at the relay nodes. The Slepian-Wolf scheme works for a larger range of ϵ_0 compared to the network considered in part (a). This is because the channels between the source and the destination are better than in the former case. In Figure 2.7.(c), we look at one extreme case when ϵ_1 is zero, giving a perfect channel between the relays and the destination. As the figure shows, the "forward" and "decode" scheme achieves the capacity in this cases. The rate of the Slepian-Wolf scheme decreases for intermediate values of ϵ_0 , but as we increase ϵ_0 again this scheme achieves the capacity. This happens because for values of ϵ around 0.5, the entropy of the re-



Figure 2.7: Performance of different schemes for four scenarios.
ceived signals at the relay nodes increases. Therefore, the minimum required rate for reliable transmission also increases. However, as we increase the value of ϵ the required rate decreases, and the compressed signals can pass across without error. Hence, we achieve capacity. In Figure 2.7.(d) we look at the case when the quality of channels between the source and relays is better than of those between the relays and destination. For this, we have chosen $\epsilon_0 = \epsilon_1^3$. In this plot, block Markov coding outperforms other schemes and achieves capacity for large values of ϵ_1 .

There are other strategies that can be used that are not analyzed here. For instance, the authors of [14] propose a compression-based scheme for general relay channels. The approach is generalized to a multiple relay setup in [16],[28]. This scheme is based on the Wyner-Ziv compression technique with side-information at the receiver [30]. One expects this scheme to be more efficient than the Slepian-Wolf scheme proposed here. However, finding an explicit formula for the achievable rate requires a seemingly intractable optimization over a number of auxiliary random variables. Designing efficient and analytically tractable coding schemes based on these ideas deserves further investigation and can be a subject of further research.

2.9 Conclusion

We have obtained the capacity for a class of wireless erasure networks with broadcast and no interference at reception. We have generalized some of the capacity results that hold for wireline networks [9],[54] to these networks. Furthermore, we have shown that linear encoding suffices to achieve the optimal performance. We see from the proof that it is not necessary to perform channel coding and network coding separately from each other. In fact, as we will see in Chapter 4, decoding at the relay nodes and operating below the capacities of each link can actually significantly reduce the achievable rate (for more examples, see [27],[55]). Therefore, unlike the wireline scenario where each link is made error free by channel coding and network coding is then employed on top of that, our scheme only requires a single encoding function. Only the destination has to decode the received signal.

Many problems related to wireless networks remain open. Generalizing the results in this work for other network problems is one possible extension of this work. It will also be interesting to see if similar results can be obtained for other types of networks, such as erasure wireless networks, in which interference is incorporated in the reception model, networks involving channels other than erasure channels, etc.

Chapter 3

Broadcast Problems over Wireless Networks

3.1 Introduction

Different traffic patterns are present in today's communication networks. In addition to pairwise communications in the network, it is possible that a collection of users are interested in the same type of information generated in the network. This scenario resembles the multicast problem that was considered in Chapter 2 for a specific model of wireless networks. Another possibility is that a collection of users demand different types of information from a specific user referred to as source node. We refer to this problem as the *broadcast problem* in the network. What is apparent from the above two scenarios is that unlike point-to-point communication systems, where the source (transmitter) and the sink (receiver) of the information are specified, in communication networks a user can be a transmitter and receiver (or even relay) for different types of information at the same time, and this gives rise to different *network problems.* At a high level of generality, a network problem over a communication network is specified by the set of information messages, the source nodes, and the destination nodes. Each source node has access to a subset of information messages, and each destination demands some subset of the information messages. In Section 2.3 of Chapter 2 we defined a network problem, \mathcal{P} by a quintuple $\mathcal{P} = (\mathcal{M}, \mathcal{S}, \mathcal{D}, \Upsilon_g, \Upsilon_r)$,

where \mathcal{M} is the set of information message, \mathcal{S} and \mathcal{D} are the set of source and destination nodes, and Υ_r (respectively Υ_g) is the function that specifies the set of messages requested (generated) at each destination (source).

Note that the above definition is independent of the communication model defined for the network. A network problem can be defined over wireline and wireless, fixed, or mobile networks. Given a communication model for the network (which specifies the connectivity of the users and their communication capabilities), the main questions regarding a network problem \mathcal{P} are

- What is the set of possible rates (for information messages) that can be supported for \mathcal{P} over the network?
- What is the optimal communication strategy to achieve those rate?

As mentioned in Chapter 2, these questions are answered for multicast problems over wireline networks in [9, 10] and a class of deterministic networks in [32]. We also looked at multicast problems over a class of wireless networks called wireless erasure networks and found the capacity region and the optimal strategy in Chapter 2. However, for a general network problem, the answer to the above questions is unknown. In [33, 23], outer and inner bounds on the capacity region of general network problems over wireline networks are obtained.

In this chapter we consider *broadcast problems*. In a broadcast problem one source has access to multiple information messages. Each of these information messages is requested by a particular destination. In accordance to our definition, a *broadcast problem* $\mathcal{P}_{bc} = (\mathcal{M}, \mathcal{S}, \mathcal{D}, \Upsilon_g, \Upsilon_r)$ is a network problem where $\mathcal{S} = \{s\}, \Upsilon_g = \mathcal{M},$ and $\Upsilon_r(.)$ is a partition of \mathcal{M} . There are many scenarios that can be modeled by broadcast problems. A well-known example is TV broadcasting stations. Downlink of cellular systems where a base station provides service to different users is another example of such problems that has received a lot of attention in the past few years. It should be noted that the capacity region of a broadcast problem for a general network is still unknown. In fact, even in the multi-user setup where there are no relays present in the network and the destinations are directly connected to the source, the broadcast problem is not completely solved. In this case the problem is referred to as *broadcast channels*, which was first introduced by Cover in [13].

In a network setup the broadcast problem is solved for wireline networks with error-free links in [54]. The capacity region in this case has a min-cut interpretation. Furthermore, it can be shown that in this case the capacity is achieved by routing, and no coding at the intermediate nodes is required.

In this chapter we look at broadcast problems over wireless erasure networks. These networks were introduced in Chapter 2. At the beginning of the chapter we give the problem statement. We find an achievable region for broadcast problems over these networks and find the exact capacity region in the multiuser setup. Using this result we will give an outer bound on the capacity region of broadcast problems over wireless erasure networks which is tighter than the outer bounds that we get from the multicast type of arguments.

3.2 Problem Statement

The communication networks considered in this chapter, namely Wireless Erasure (WE) networks, are modeled by an acyclic directed graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ with node set \mathcal{V} and link (edge) set $\mathcal{E} \subset \mathcal{V} \times \mathcal{V}$. Each link $(i, j) \in \mathcal{E}$ corresponds to a communication channel between the node *i* and node *j*. Looking back at Chapter 2, the input of all channels originating from node *i* is denoted by X_i chosen from input alphabet \mathcal{X} . The output of the communication channel corresponding to link (i, j) is denoted by Y_{ij} ; Y_{ij} lies in alphabet set \mathcal{Y}_{ij} . We assume interference-free property for all the incoming links to a node and denote the collection of received signal at node *j* by

 $Y_j = (Y_{ij}, (i, j) \in \mathcal{E}) \in \mathcal{Y}_j \stackrel{\triangle}{=} \prod_{i:(i,j)\in\mathcal{E}} \mathcal{Y}_{ij}$. Link (i, j) corresponds to an erasure channel with probability of erasure ϵ_{ij} in WE networks.

Let $\mathcal{D} = \{d_1, \ldots, d_{|\mathcal{D}|}\}$ denote the set of destination nodes and s be the source node for the broadcast problem. Next, we define the class of block codes considered in this chapter.

A $(\lceil 2^{nR_1} \rceil, \dots, \lceil 2^{nR_{|\mathcal{D}|}} \rceil, n)$ code for the broadcast problem in a wireless erasure network described in previous sections consists of the following components:

- A set of integers $\mathcal{W}^{(d_i)} = \{1, 2, \dots, \lceil 2^{nR_i} \rceil\}$ that represent the message indices corresponding to information source intended for destination node $d_i \in \mathcal{D}$. We assume that all the messages are equally likely. All the information sources are available at the source node indexed by $s \in \mathcal{V}$.
- An encoding function for the source node s: $f_s : \prod_{d \in \mathcal{D}} \mathcal{W}^{(d)} \to \mathcal{X}^n_s$.
- A set of encoding functions $\{f_{i,t}\}_{t=1}^n$ for each node $i \neq s \in \mathcal{V}$, where $x_{i,t} = f_{i,t}(y_i^{t-1})$ is the signal transmitted by node *i* at time *t*. Note that $x_{i,t}$ is a function of all the received symbols from all its incoming channels up to time t-1.
- A decoding function g_{d_i} at destination node $d_i \in \mathcal{D}$, $g_{d_i} : \mathcal{Y}_{d_i}^n \times \{0, 1\}^{n|\mathcal{E}|} \to \mathcal{W}^{(d_i)}$ such that

$$\hat{w}^{(d_i)} = g_{d_i}(y_{d_i}^n, (s_{ij,t}, (i, j) \in \mathcal{E}, 1 \le t \le n)),$$
(3.1)

where $\hat{w}^{(d_i)}$ is the estimate of the message sent from source s based on received signals at d_i and also the erasure occurrences on all the links of the network in the current block.

Note that X_i , Y_{ij} and Y_i , all depend on the message vector $\underline{w} = (w^{(d_i)}, d_i \in \mathcal{D})$ that is being transmitted. Therefore, we will write them as $X_i(\underline{w}), Y_{ij}(\underline{w})$, and $Y_i(\underline{w})$ to specify what specific set of messages is transmitted. We define the probability of error as the probability that the decoded message at one of the destinations is not equal to the transmitted message, i.e.,

$$P_{err} = \Pr\left(\exists d_i \in \mathcal{D} : \hat{W}^{(d_i)} \neq W^{(d_i)}\right).$$
(3.2)

The set of rates $(R_i, 1 \le i \le |\mathcal{D}|)$ is said to be achievable if there exist a sequence of $(\lceil 2^{nR_1} \rceil, \dots, \lceil 2^{nR_{|\mathcal{D}|}} \rceil, n)$ codes such that $P_{err} \to 0$ as $n \to \infty$. The capacity region, $\mathcal{C}(\mathcal{G}, \{\epsilon_{ij}, (i, j) \in \mathcal{E}\})$, is the set closure of the set of achievable rates.

In the remainder of this chapter we will look at the capacity region for broadcast problem in WE networks.

3.3 Broadcast Problems Over WE Networks

As we saw in Chapter 2, with an appropriate definition of the cut-capacity, the capacity of multicast problems over wireless erasure networks has a nice min-cut interpretation. In particular, for a unicast scenario (i.e., single destination) the capacity is given by the min-cut capacity over all the source destination cuts (see Theorem 2.1). In the following we propose a time-sharing scheme for broadcast problems based on the result for unicast problems in wireless erasure networks.

3.3.1 Time-sharing Scheme for WE networks

We consider a scheme where each node in the network performs time-sharing between the destinations. In other words, each node $i \in \mathcal{V}$ allocates a fraction α_{id} , $d \in \mathcal{D}$ of its block length to transmit to destination $d \in \mathcal{D}$. These fractions may not be the same for different nodes. In fact, as we will see later, in some cases the optimal fractions are unequal.

In order to analyze the set of achievable rates with the above time-sharing schemes, it is important to note that given time-sharing parameters α_{id} , $d \in \mathcal{D}$, the achievable rate for each destination is given by a min-cut formulation. The only difference to be taken into account is that the length of the block code used at node *i* for transmitting to destination *d* is of size $\lceil \alpha_{id}n \rceil$ (rather than *n* in the proof of Theorem 2.1). However, with a generalized definition of the cut-capacity, a similar min-cut result will still hold even if the block lengths are not equal across the network. We have stated this result without proof as the following lemma.

Lemma 3.1. Consider an erasure wireless network with single source and single destination d. Furthermore, suppose that node $i \in \mathcal{V}$ uses a block code of length $\lceil \alpha_{id} n \rceil$, $\alpha_{id} \leq 1$ to perform encoding. Then, the capacity of the network with side-information at the destination and under this coding scheme is given by the minimum of the cut-capacities over all the s-d cuts, where the cut-capacity of s-d cut \mathcal{V}_s is defined as

$$C(\mathcal{V}_s, \{\alpha_{id}\}_{i \in \mathcal{V}}) = \sum_{i \in \mathcal{V}_s^*} \alpha_{id} (1 - \prod_{j: (i,j) \in [\mathcal{V}_s, \mathcal{V}_s^c]} \epsilon_{ij}).$$
(3.3)

Now consider our broadcast problem in wireless erasure networks. We represent any admissible time-sharing policy by $\underline{\alpha} = (\alpha_{id}, i \in \mathcal{V}, d \in \mathcal{D})$, where, as mentioned earlier, α_{id} specifies the fraction of the block length allocated by node *i* for transmission to destination *d*. It is clear that for any admissible time-sharing $\underline{\alpha}$, we should have $\alpha_{id} \geq 0$ and $\sum_{d \in \mathcal{D}} \alpha_{id} = 1$ for any $i \in \mathcal{V}$ and $d \in \mathcal{D}$. According to the previous lemma, the achievable rate region using a fixed time-sharing scheme given by $\underline{\alpha}$ will be

$$\mathcal{R}_{TS}(\underline{\alpha}) = \{ (R_d, \quad d \in \mathcal{D}) | \forall d \in \mathcal{D}, \quad 0 \le R_d \le \min_{\mathcal{V}_s: s \neq d \text{ cut}} C(\mathcal{V}_s, \{\alpha_{id}\}_{i \in \mathcal{V}}) \}, \quad (3.4)$$

and therefore the achievable rate region using time-sharing in the network is

$$\mathcal{R}_{TS} = \bigcup_{\underline{\alpha}} R_{TS}(\underline{\alpha}), \qquad (3.5)$$



Figure 3.1: (a): A wireless erasure network with one source and two destination nodes. (b): Different achievable regions for the network in part (a).

where union is taken over all admissible $\underline{\alpha}$ s and subscript TS is used to refer to the time-sharing scheme.

Example 3.1. Consider the wireless erasure network shown in Figure 3.1, with one source, one relay node, and two destination nodes. Based on the above argument, the achievable rate region using time-sharing is

$$\mathcal{R}_{TS} = \cup_{(\alpha,\beta)} \left\{ \begin{pmatrix} R_2 \\ R_3 \end{pmatrix} \middle| \begin{array}{c} R_2 \le \min\{\alpha(1-\epsilon_{12}\epsilon_{14}), \alpha(1-\epsilon_{12}) + \beta(1-\epsilon_{42})\} \\ R_3 \le \min\{(1-\alpha)(1-\epsilon_{13}\epsilon_{14}), (1-\alpha)(1-\epsilon_{13}) + (1-\beta)(1-\epsilon_{43})\} \end{array} \right\}$$

where $0 \le \alpha \le 1$ (respectively $0 \le \beta \le 1$) is the fraction of the block length that node 1 (respectively 4) allocates to transmit to destination 2. In Figure 3.1.(b), we have plotted the achievable rate region using the time-sharing scheme described above for $(\epsilon_{12}, \epsilon_{13}, \epsilon_{14}, \epsilon_{42}, \epsilon_{43}) = (0.8, 0.2, 0.6, 0.8, 0.2)$. As it can be seen from the plot, the time-sharing region is given by

$$\mathcal{R}_{TS} = \left\{ (R_2, R_3) \left| \frac{R_3}{0.88} + \frac{R_2}{0.52} \le 1, \frac{R_2}{0.4} + \frac{R_3}{1.6} \le 1 \right\}.$$
 (3.6)

The dashed line specifies boundaries of the region achievable by time-sharing between source and one of the destination sub-networks (this corresponds to the case when $\alpha = \beta$). As we can observe, the optimal time-sharing is not achieved by equal fractions α and β . Figure 3.1.(b) also compares the time-sharing region to the naïve min-cut outer bound derived by multicast type arguments, i.e.,

$$R_2 \leq 0.4,$$

 $R_3 \leq 0.88,$ and
 $R_2 + R_3 \leq 0.904.$

Note that the outer bound provided in the example is simply an application of the outer bound developed for the multicast problem in Chapter 2. The counterpart of this outer bound in wireline networks is tight for broadcast problems [54]. However, as we will see in the following, this outer bound is not tight for wireless erasure networks. In the following section, we first look at the performance of the time-sharing scheme proposed above for a subclass of WE networks referred to as Erasure Broadcast Channels (EBC). We will show that time-sharing achieves the capacity for this subclass and then use this result to find tighter outer bounds for a general WE network.

3.4 Erasure Broadcast Channel

We consider a subclass of wireless erasure networks called erasure broadcast channels. An (m, n)-erasure broadcast channel (see Figure 3.2.(a)) with channel matrix $\underline{\epsilon}_{m \times n}$ is a broadcast channel with m inputs and n receivers (or destinations). Destination iis connected to node j with a packet erasure channel with probability of erasure ϵ_{ij} . We can think of an (m, n)-erasure broadcast channel as a wireless erasure network with n destinations and m intermediate nodes, each connected to the source node with a link with no erasure and unlimited capacity. We can also consider an (m, n)-



Figure 3.2: (a):An (m, n)-erasure broadcast channel (b): An (1, n-1)-erasure broadcast channel

erasure broadcast channel as the product of m scalar (single input) erasure broadcast channels. The capacity of the product of two broadcast channels with degraded components is considered in [37, 38].

The (m, n)-erasure broadcast channel does not in general belong to the classes of broadcast channels for which the capacity is known, e.g., "degraded," "more capable," and "less noisy" (see [35] for an overview). However, the single input erasure broadcast channel belongs to the "degraded" class, and its capacity region has been found in [42, 39]. It is shown in [42, 39] that the time-sharing scheme is actually capacity achieving. Here we look at the general case. Note that for the (m, n)-erasure broadcast channel, the time-sharing region of (3.5) simplifies to the following:

$$\mathcal{R}_T(\underline{\epsilon}) = \bigcup_{\underline{\alpha}} \{ (R_1, \dots, R_n) | 0 \le R_j < \sum_{i=1}^m \alpha_{ij} (1 - \epsilon_{ij}) \},$$
(3.7)

where the union is over all admissible time-sharing matrices $\underline{\alpha}$ that satisfy:

- (a) $\alpha_{ij} \geq 0.$
- (b) $\sum_{j=1}^{n} \alpha_{ij} = 1$ for all inputs $1 \le i \le m$.

3.4.1 Degraded (m, n)-Erasure Broadcast Channels

Here we try to find necessary and sufficient conditions so that an (m, n)-erasure broadcast channel with erasure matrix $\underline{\epsilon}$ belongs to the "degraded" class.

Definition 3.1. Channel *i* with probability transition matrix $\Pr_i(Y|X)$ is a statistically "degraded" version of channel *j* with transition matrix $\Pr_j(Z|X)$ iff there exists a channel $\Pr(Y|Z)$ such that

$$\Pr_{i}(Y|X) = \sum_{Z} \Pr_{j}(Z|X) \Pr(Y|Z).$$

In this case we write $\Pr_i(\cdot|\cdot) \preceq \Pr_i(\cdot|\cdot)$.

Now note that if channel i is a degraded version of channel j, then for any distribution on X we should have

$$I(X;Y_j) \ge I(X;Y_i).$$

Therefore, for an (m, n)-erasure broadcast channel, if the *i*-th user is a "degraded" version of *j*-th user, then we should have $I(X; Y_j) \ge I(X; Y_i)$ for any distribution on X. In particular, if we keep all the inputs except the *k*-th one constant and let X_k be an i.i.d. Bernoulli $(\frac{1}{2})$, the following condition should hold:

$$\epsilon_{kj} \le \epsilon_{kj} \qquad \forall k \in \{1, \dots, m\}$$

It can be easily checked that this condition is also sufficient for channel i being a "degraded" version of channel j. Hence,

Lemma 3.2. An (m, n)-erasure channel with erasure matrix $\underline{\epsilon}$ belongs to "degraded"

class iff there exists some permutation $\pi(\cdot)$ on $\{1, 2, \ldots, n\}$ such that

$$\epsilon_{i\pi(j)} \leq \epsilon_{i\pi(j+1)}, \quad \forall 1 \leq i \leq m, 1 \leq j \leq n.$$

Now, we show that the capacity of the degraded erasure broadcast channels is given by the time-sharing scheme described in Section 3.3.1. In doing so, we use the following lemma, the proof of which we omit.

Lemma 3.3. Suppose U, V, X, and Y are random variables with probability distribution of form

$$Pr(U, V, X, Y) = Pr(U, V, X)Pr_{\epsilon}(Y|X),$$

where $Pr_{\epsilon}(.|.)$ is the transition probability of an erasure channel with probability of erasure ϵ . Then

$$I(U;Y|V) = (1 - \epsilon)I(U;X|V).$$

Theorem 3.1. The capacity region, $C_g(\underline{\epsilon})$, of a degraded (m, n)-erasure channel with erasure matrix $\underline{\epsilon}$ is given by time-sharing between the receivers at each input, i.e.,

$$\mathcal{C}_g(\underline{\epsilon}) = \mathcal{R}_T(\underline{\epsilon}),$$

where $\mathcal{R}_T(\underline{\epsilon})$ is given in (3.7).

Proof. Without loss of generality let us assume that for i < j, receiver j is a "degraded" version of receiver i. According to [40], the capacity of the degraded broadcast channel is given by the convex hull of the closure of the (R_1, \ldots, R_n) satisfying

$$0 \le R_j \le I(U_j; Y_j | U_{j+1}, \dots, U_n)$$

for j = 1, 2, ..., n, where $U = (U_1, ..., U_n)$ and $X = (X_1, ..., X_m)$ and $U - X - Y_1 - ... - Y_n$ forms a Markov chain.

69

Now consider the following maximization problem:

$$S^{\star}(\underline{\mu}) = \max_{(R_1,\dots,R_n)\in \mathcal{C}_g(\underline{\epsilon})} \sum_{j=1}^n \mu_j R_j,$$

where $\underline{\mu} = (\mu_1, \dots, \mu_n) \ge 0$. Note that every point on the boundary of the capacity region is the maximizing solution for some $\underline{\mu} \ge 0$. Also, the maximizing solution of the above optimization problem corresponds to a boundary point of the capacity region.

Using chain rule for mutual information, we can write

$$S^{\star}(\underline{\mu}) \leq \max_{P(U,X)} \sum_{j=1}^{n} \mu_{j} I(U_{j}; Y_{j} | U_{j+1}, \dots, U_{n})$$
(3.8)
$$= \max_{P(U,X)} \sum_{j=1}^{n} \mu_{j} I(U_{j}; Y_{1j}, Y'_{j} | U_{j+1}, \dots, U_{n})$$
$$= \max_{P(U,X)} \left(\sum_{j=1}^{n} \mu_{j} I(U_{j}; Y'_{j} | U_{j+1}, \dots, U_{n}) + \sum_{j=1}^{n} \mu_{j} I(U_{j}; Y_{1j} | U_{j+1}, \dots, U_{n}, Y'_{j}) \right)$$
$$= \max_{P(U,X)} \left(\sum_{j=1}^{n} \mu_{j} I(U_{j}; Y'_{j} | U_{j+1}, \dots, U_{n}) + \sum_{j=1}^{n} \mu_{j} (1 - \epsilon_{1j}) I(U_{j}; X_{1} | U_{j+1}, \dots, U_{n}, Y'_{j}) \right),$$
(3.9)

where $Y'_j = (Y_{2j}, \ldots, Y_{mj})$ for all $1 \leq j \leq n$, and (3.9) follows from Lemma 3.3. Defining $X' = (X_2, \ldots, X_m)$, it can be easily verified that $(U_1, \ldots, U_n, X_1) - X' - Y'_1 - \cdots - Y'_n$. This Markov property implies that

$$H(X_1|U_j,\ldots,U_n,Y'_{j-1}) = H(X_1|U_j,\ldots,U_n,Y'_j) - I(Y'_{j-1};X_1|U_j,\ldots,U_n,Y'_j).$$
(3.10)

Therefore,

$$H(X_1|U_j,\ldots,U_n,Y'_{j-1}) \le H(X_1|U_j,\ldots,U_n,Y'_j).$$

Using the above inequality, one can show that

$$\sum_{j=1}^{n} I(U_j; X_1 | U_{j+1}, \dots, U_n, Y'_j) \le H(X_1).$$

Therefore, the second weighted sum of (3.9) is at most $\max_{j} \mu_{j}(1 - \epsilon_{1j})H(X_{1})$, and replacing this in (3.9), we have

$$S^{\star}(\underline{\mu}) \le \max_{P(U,X')} \sum_{j=1}^{n} \mu_j I(U_j; Y'_j | U_{j+1}, \dots, U_n) + \max_j \mu_j (1 - \epsilon_{1j}).$$
(3.11)

The first summation in the right hand side of the inequality corresponds to the maximum weighted sum rate of an (m - 1, n)-degraded broadcast channel obtained by excluding the connections from the first input of the transmitter to all the receivers. Using similar arguments for the new (m - 1, n)-degraded broadcast channel, it can be verified that

$$\max_{P(U,X')} \sum_{j=1}^{n} \mu_j I(U_j; Y'_j | U_{j+1}, \dots, U_n) \le \sum_{i=2}^{m} \max_j \mu_j (1 - \epsilon_{ij}).$$
(3.12)

Using this in (3.11), we have

$$S^{\star}(\underline{\mu}) \leq \sum_{i=1}^{m} \max_{j} \mu_{j}(1 - \epsilon_{ij}).$$
(3.13)

Now, the right hand side value can be achieved by time-sharing. For that, input *i* transmits only to the receiver with maximum $\mu_j(1 - \epsilon_{ij})$. Therefore ,each boundary point of the capacity region can be achieved by time-sharing and $C_g(\underline{\epsilon}) = \mathcal{R}_T(\underline{\epsilon})$.

3.4.2 Non-degraded (m, n)-Erasure Broadcast Channel

In this section we show that the capacity region of the general erasure broadcast channel is given by time-sharing. In [38], the capacity of the product of two reversely degraded broadcast channels is characterized. Using this result it can be easily verified that the capacity region of (2, 2)-erasure broadcast channel is given by time-sharing. However, applying and specializing the technique of [38] for the general (m, n)-erasure broadcast channel does not seem plausible. Instead, in this chapter we use another argument to show that every boundary point of the capacity region for general erasure broadcast channel is achieved by time-sharing. The argument used here is very close to the one used in [118] to prove that the capacity of the MIMO Gaussian broadcast channel is given by Dirty Paper coding (DPC) introduced in [119].

Consider an (m, n)-erasure broadcast channel with erasure matrix $\underline{\epsilon} = [\epsilon_{ij}]$. Let $C_g(\underline{\epsilon})$ denote the capacity region of this general erasure broadcast channel. For every boundary point \underline{R}' , there exist positive μ_1, \ldots, μ_n such that \underline{R}' is the optimal solution of

$$\max_{(R_1,\ldots,R_n)\in\mathcal{C}_g(\underline{\epsilon})}\sum_{j=1}^n \mu_j R_j.$$

The idea is to construct for each value of μ_1, \ldots, μ_n a degraded (m, n)-erasure broadcast channel with channel matrix $\underline{\epsilon}^*$ whose capacity region, $\mathcal{C}_g(\underline{\epsilon}^*)$, contains the capacity region of the original channel, $\mathcal{C}_g(\underline{\epsilon})$. Moreover, we require that the time-sharing region of both channels meet at some specific point(s) on the boundary.

Let us look at the (2, 2)-erasure broadcast channels first. Suppose that the channel is not degraded. Without loss of generality, assume that

$$\epsilon_{11} \leq \epsilon_{12}, \quad \epsilon_{21} \geq \epsilon_{22}.$$

Consider the following maximization problem:

$$\max_{(R_1,R_2)\in\mathcal{C}_g}\mu_1 R_1 + \mu_2 R_2. \tag{3.14}$$

We construct a (2, 2)-degraded erasure broadcast channel that contains the capacity region C_g . For this, we consider the following two cases separately.

• $\frac{\mu_1}{\mu_2} \ge 1$: In this case, consider the erasure broadcast channel with the erasure matrix constructed from $\underline{\epsilon}$ as follows:

$$\underline{\epsilon}^{\star} = \begin{bmatrix} \epsilon_{11} & \epsilon_{11} \\ \epsilon_{21} & \epsilon_{22} \end{bmatrix}.$$

First note that the above channel is degraded. Moreover, the erasure probabilities on every edge of this new channel are less than or equal to that of the corresponding edge in the original channel. Therefore, the capacity region of the degraded channel contains C_g . Now let's look back at the maximization problem in (3.14). Based on the above discussions we know that the optimal solution is less than

$$\max_{(R_1,R_2)\in \mathcal{C}_d} \mu_1 R_1 + \mu_2 R_2,$$

where C_d is the capacity region of the obtained degraded channel with erasure matrix $\underline{\epsilon}^*$. Based on Theorem 3.1, the maximum of the above problem is achieved by time sharing, and it equals

$$\max\{\mu_1(1-\epsilon_{21}),\mu_2(1-\epsilon_{22})\}+\max\{\mu_1(1-\epsilon_{11}),\mu_2(1-\epsilon_{11})\}.$$

Since $\mu_1 \ge \mu_2$, we can write the above rate as

$$\max\{\mu_1(1-\epsilon_{21}), \mu_2(1-\epsilon_{22})\} + \mu_1(1-\epsilon_{11}).$$

Looking closely at the above rate, we observe that we can achieve the above rate in the original erasure channel by time-sharing as well. Input 1 transmits only to the first receiver, and the second input sends to the maximum of the to terms appearing in the above formula. Therefore, the boundary point corresponding to μ_1, μ_2 is achieved by time-sharing.

• $\frac{\mu_1}{\mu_2} < 1$: In this case, consider the (2,2)-erasure broadcast channel with the following erasure matrix:

$$\underline{\epsilon}^{\star} = \begin{bmatrix} \epsilon_{11} & \epsilon_{12} \\ \epsilon_{22} & \epsilon_{22} \end{bmatrix}.$$

Similar to the first case, the broadcast channel represented by $\underline{\epsilon}^*$ is degraded, and its capacity region contains C_g . Also, the solution of (3.14) coincides with the solution of the same cost function over region C_d . Thus, the boundary point corresponding to μ_1 , μ_2 is achieved by time-sharing.

Based on the above results and Theorem 3.1, we have the following result.

Theorem 3.2. The capacity region of any (2,2)-erasure broadcast channel is given by time-sharing.

We can generalize the preceding arguments to general (m, n)-erasure broadcast channels by finding a degraded erasure broadcast channel that contains the capacity region of the original channel and its time-sharing region coincides with that of the original one at a specific point. We have stated the above result as a theorem:

Theorem 3.3. Consider an (m, n)-erasure broadcast channel with channel matrix $\underline{\epsilon}$. The capacity region, $C_g(\underline{\epsilon})$, of this broadcast channel is given by time-sharing between the receivers at each input, i.e.,

$$\mathcal{C}_g(\underline{\epsilon}) = \mathcal{R}_T(\underline{\epsilon}),$$

where $\mathcal{R}_T(\underline{\epsilon})$ is defined in (3.7).

Proof. Similar to previous discussions, we need to show that every boundary point of the capacity region is achieved by time-sharing. For any $\underline{\mu} \ge 0^1$, consider the following maximization problem:

$$f_g(\underline{\mu}, \underline{\epsilon}) = \max_{(R_1, \dots, R_n) \in \mathcal{C}_g(\underline{\epsilon})} \sum_{j=1}^n \mu_j R_j, \qquad (3.15)$$

where the maximization is over the capacity region of the erasure broadcast channel with channel matrix $\underline{\epsilon}$. We construct a *degraded* (m, n)-erasure broadcast channel with channel matrix denoted by $\underline{\epsilon}^*$ from the original channel such that

• its capacity region $\mathcal{C}_g(\underline{\epsilon}^*)$ contains that of the original channel and, therefore,

$$f_g(\mu, \underline{\epsilon}) \le f_g(\mu, \underline{\epsilon}^*)$$
, and (3.16)

• the following maximization problem over the time-sharing region

$$f_T(\underline{\mu}, \underline{\epsilon}) = \max_{(R_1, \dots, R_n) \in \mathcal{R}_T(\underline{\epsilon})} \sum_{j=1}^n \mu_j R, \qquad (3.17)$$

gives the same values for both channels, i.e.,

$$f_T(\mu, \underline{\epsilon}) = f_T(\mu, \underline{\epsilon}^*). \tag{3.18}$$

Combining (3.16) and (3.18) we get

$$f_g(\underline{\mu}, \underline{\epsilon}) \leq f_g(\underline{\mu}, \underline{\epsilon}) \stackrel{(a)}{=} f_T(\underline{\mu}, \underline{\epsilon}^{\star}) = f_T(\underline{\mu}, \underline{\epsilon}),$$

where (a) follows from Theorem 3.1 and the fact that $\underline{\epsilon}^{\star}$ corresponds to a degraded

¹Here, by $\underline{a} \geq 0$ we mean every component of \underline{a} should be greater than equal zero.

channel, and this proves that time-sharing is capacity-achieving.

Before we start constructing the degraded channel, we should remark that the solution to the optimization problem in (3.17) can be written explicitly as

$$f_T(\underline{\mu}, \underline{\epsilon}) = \sum_{i=1}^m \max_{1 \le j \le n} \mu_j (1 - \epsilon_{ij}).$$
(3.19)

Next, we claim that there exists at least one input i and two receivers j, k such that

$$\mu_j \leq \mu_k, \quad \epsilon_{ik} < \epsilon_{ij}.$$

If not, then it can be easily verified that for all is, $\epsilon_{ij}s$ are ordered in the reverse order that $\mu_j s$ are ordered and, therefore, the erasure matrix $\underline{\epsilon}$ satisfies the constraint of Lemma 3.2 with the permutation that sorts $\mu_j s$ in decreasing order; hence our (m, n)erasure channel is "degraded." But, in that case we already know from Theorem 3.1 that the capacity region is achieved by time-sharing. Therefore, let i^*, j^* , and k^* be such numbers. Consider a new (m, n)-erasure channel with erasure matrix $\underline{\epsilon}^{(1)}$ derived from $\underline{\epsilon}$ by replacing $\epsilon_{i^*j^*}$ with $\epsilon_{i^*k^*}$ in the i^*j^* coordinate of $\underline{\epsilon}$. In other words, each coordinate of the new matrix is as follows:

$$\epsilon_{ij}^{(1)} = \epsilon_{ij} + (\epsilon_{i^*k^*} - \epsilon_{i^*j^*})\delta(i - i^*)\delta(j - j^*),$$

where $\delta(\cdot)$ denotes the Dirac's delta function. This new channel has the following properties:

• Its capacity region, $C_g(\underline{\epsilon}^{(1)})$, contains $C_g(\underline{\epsilon})$, since a link is replaced by a link with lower probability of erasure; therefore,

$$f_g(\mu, \underline{\epsilon}) \le f_{g}(\mu, \underline{\epsilon}^{(1)}).$$

• The value of (3.19) remains unchanged for the new channel, i.e., $f_T(\underline{\mu}, \underline{\epsilon}) = f_T(\underline{\mu}, \underline{\epsilon}^{(1)})$. To see this, first notice that

$$\begin{aligned} \max_{1 \le j \le n} \mu_j (1 - \epsilon_{i^* j}^{(1)}) &= \max\{\mu_{j^*} (1 - \epsilon_{i^* j^*}^{(1)}), \max_{1 \le j \ne j^* \le n} \mu_j (1 - \epsilon_{i^* j}^{(1)})\} \\ &= \max\{\mu_{j^*} (1 - \epsilon_{i^* k^*}), \max_{1 \le j \ne j^* \le n} \mu_j (1 - \epsilon_{i^* j})\} \\ &= \max\{\mu_{j^*} (1 - \epsilon_{i^* k^*}), \mu_{k^*} (1 - \epsilon_{i^* k^*}) \max_{1 \le j \ne j^*, k^* \le n} \mu_j (1 - \epsilon_{i^* j})\} \\ &= \max\{\mu_{k^*} (1 - \epsilon_{i^* k^*}) \max_{1 \le j \ne j^*, k^* \le n} \mu_j (1 - \epsilon_{i^* j})\} \\ &= \max_{1 \le j \le n} \mu_j (1 - \epsilon_{i^* j}), \end{aligned}$$

where we have used our initial assumption that $\mu_{j^*} \leq \mu_{k^*}$.

Now we repeat the above process and obtain erasure channels $\underline{\epsilon}^{(l)}$ for $l \geq 1$, until we cannot find any input *i*, receivers *j*, *k* with $(\mu_j \leq \mu_k)$ and $(\epsilon_{ik} < \epsilon_{ij})$ at round $l = l_0.^2$ In that case, we know that channel $\underline{\epsilon}^* = \underline{\epsilon}^{(l_0)}$ is degraded. Furthermore, its capacity region $C_g(\underline{\epsilon}^*)$ contains the capacity region of all previously derived channels (in particular the original channel), and the value of (3.19) for it remains unchanged, i.e.,

$$f_T(\mu, \underline{\epsilon}^{\star}) = f_T(\mu, \underline{\epsilon}).$$

Based on Theorem 3.1, for the derived degraded channel we have

$$f_g(\mu, \underline{\epsilon}^{\star}) = f_T(\mu, \underline{\epsilon}^{\star}).$$

Putting these together, we get $f_T(\underline{\mu}, \underline{\epsilon}) = f_g(\underline{\mu}, \underline{\epsilon})$. This completes the proof.

Remark 3.4. Theorem 3.3 can be stated in another form. That is, the capacity region of a general (m, n)-erasure broadcast channel is the Minkowski sum of the capacity regions of the scalar (single input) erasure broadcast channels between each input and

²It is clear that this process comes to an end since at each step of the process the number of triplets (i, j, k) that $\mu_j \leq \mu_k, \epsilon_{ij} \geq \epsilon_{ik}$ is reduced.



Figure 3.3: Illustrating the proof technique.

the receivers, where The Minkowski sum of two set \mathcal{A} and \mathcal{B} in n-dimensional vector space is the result of adding every element of \mathcal{A} to every element of \mathcal{B} , i.e. the set

$$\mathcal{A} \oplus \mathcal{B} = \{ a + b | a \in \mathcal{A}, b \in \mathcal{B} \}.$$

In mathematical terms, $C_g(\underline{\epsilon})$ can be written as

$$\mathcal{C}_g(\underline{\epsilon}) = \oplus_{i=1}^m \mathcal{C}_g(\underline{\epsilon}_i),$$

where $\underline{\epsilon}_i$ is the *i*-th row of $\underline{\epsilon}$ and corresponds to a single input erasure channel with *n* receivers.

Remark 3.5. The above statement of Theorem 3.3 can be generalized to a broader class of channels. Using very similar techniques, it is possible to show that the capacity region of broadcast problems for the product of a number of degraded broadcast channels with arbitrary degradation order is given by the Minkowski sum of the capacity regions of the degraded broadcast channels between each input and the receivers. This is a generalization of the result reported previously in [37].

3.4.3 Discussion

In the previous section, we showed that the capacity region of erasure broadcast channels is achieved by the time-sharing scheme introduced in Section 3.3.1. This result will enable us to provide tighter outer bounds on the capacity region of broadcast problems over general wireless erasure networks. However, before we proceed to provide the outer bounds, we point out some important remarks regarding the previous result and the proof method.

3.4.3.1 Correlated Erasures

The result of Theorem 3.3 is for independent and memoryless erasure events. One might think that the result goes through for correlated erasure events across the links. However, this is not true. In fact, we can show that time-sharing is not capacity-achieving even for "degraded" channels.

Consider an (m, n)-erasure channels with correlated erasure events.³ We model the correlated event as follows. For each set $\mathcal{A} \subseteq \{1, \ldots, m\}$, $\Pr_j(\mathcal{A})$ denotes the probability that the transmitted signals from inputs $i \in \mathcal{A}^c$ are erased and signals from $i \in \mathcal{A}$ are received successfully at receiver j. In the following Proposition, we look at the necessary and sufficient conditions for "degraded"-ness.

Proposition 3.1. Considering the previous notation, receiver j is a degraded version of receiver k iff for any subset \mathcal{U} of the power set of $\{1, \ldots, m\}$, with the property that

$$\mathcal{B} \in \mathcal{U}, \ \mathcal{A} \subseteq \mathcal{B} \Longrightarrow \mathcal{A} \in \mathcal{U},$$

we have

$$\sum_{\mathcal{A}\in\mathcal{U}} \Pr_{j}(\mathcal{A}) \leq \sum_{\mathcal{A}\in\mathcal{U}} \Pr_{k}(\mathcal{A}).$$
(3.20)

³Since the capacity region of broadcast channels with average probability of error constraint only depends on the marginals [35], we assume that different users' channels are independent from each other.

Proof. It is not hard to check that $\Pr_j(\cdot)$ is a degraded version of $\Pr_k(\cdot)$ if and only if there exists a stochastic matrix $\gamma(\cdot, \cdot)$ of dimension of $2^m \times 2^m$ indexed by the subsets of $\{1, \ldots, m\}$ with $\gamma(\mathcal{S}, \mathcal{T}) = 0$ for $\mathcal{S}, \mathcal{T}, \ \mathcal{S} \notin \mathcal{T}$ such that

$$\Pr_{j}(\mathcal{A}) = \sum_{\mathcal{B} \subseteq \mathcal{A}} \gamma(\mathcal{A}, \mathcal{B}) \Pr_{k}(\mathcal{B})$$

for all \mathcal{A}, \mathcal{B} subsets of $\{1, \ldots, m\}$.

With this, the necessity of (3.20) is clear. To prove the sufficiency, we use an argument used in [44]. The idea is based on constructing a max-flow network. Apart from source s and destination t, there are two nodes $l(\mathcal{A})$ and $r(\mathcal{A})$ for each $\mathcal{A} \subseteq \{1, \ldots, m\}$. s is connected to $l(\mathcal{A})$ with a link with capacity $\Pr_k(\mathcal{A})$. $r(\mathcal{A})$ is connected to t with a link with capacity $\Pr_j(\mathcal{A})$. Finally, $l(\mathcal{A})$ and $r(\mathcal{B})$ are connected with a link of unlimited capacity if and only if $\mathcal{B} \subseteq \mathcal{A}$. Consider a cut $[\mathcal{V}_s, \mathcal{V}_t]$ of finite capacity in this network. Let $\mathcal{V}_t = \mathcal{V}_t^l \cup \mathcal{V}_t^r$ be the partition of \mathcal{V}_t to the nodes in the l side and the r side. Now, because of the finiteness of the capacity of the cut, we should have

$$r(\mathcal{B}) \in \mathcal{V}_t^r, \ \mathcal{B} \subseteq \mathcal{A} \Longrightarrow l(\mathcal{A}) \in \mathcal{V}_r^l$$

Therefore, if we define $\mathcal{V}^* = \{\mathcal{A} | \exists r(\mathcal{B}) \in \mathcal{V}_t^r \text{ s.t. } \mathcal{B} \subseteq \mathcal{A}\}$, we have

$$\mathcal{V}^r \subseteq r(\mathcal{V}^*) \quad \text{and} \quad l(\mathcal{V}^*) \subseteq \mathcal{V}_t^l.$$
 (3.21)

The cut-capacity $C(\mathcal{V}_s)$ can be written as

$$C(\mathcal{V}_{s}) = \sum_{l(\mathcal{A})\in\mathcal{V}_{t}^{l}} \Pr_{k}(\mathcal{A}) + 1 - \sum_{r(\mathcal{B})\in\mathcal{V}_{t}^{r}} \Pr_{j}(\mathcal{B})$$

$$\geq 1 - \sum_{\mathcal{B}\in\mathcal{V}^{*}} \Pr_{j}(\mathcal{B}) + \sum_{\mathcal{A}\in\mathcal{V}^{*}} \Pr_{k}(\mathcal{A})$$

$$\geq 1,$$

where the second line follows from (3.21), and the last line follows from (3.20). This suggests that the value of min-cut is equal to one, and the minimum cut is the one that isolates either s or t from the rest of nodes. In this case, using the max-flow min-cut theorem, the max-flow is also one. Therefore, the flow in each of the links between s (respectively t) and the intermediate nodes is equal to the capacity of the corresponding link. Defining the flow between $l(\mathcal{A})$ and $r(\mathcal{B})$ as $\gamma(\mathcal{B}, \mathcal{A}) \cdot \Pr_k(\mathcal{A})$, we can easily see that $\gamma(\cdot, \cdot)$ is the desired stochastic matrix.

Remark 3.6. As mentioned earlier, for the independent erasure case, different notions of "degraded," "more capable," and "less noisy" are equivalent. However, it is not clear whether the same is true for the case when erasures are correlated. In Proposition 3.1 we identified the class of "degraded" channels. However, characterizing the class of "more capable" channels is not an easy task. In order to see it, note that channel j is "more capable" than channel k if and only if for any distribution on the input $X = (X_1, \ldots, X_m)$ we have

$$I(X; Y_k) \le I(X; Y_j).$$

For (possibly correlated) erasure channels, the above condition can be written as a linear inequality involving the joint entropies of subsets of inputs, i.e.,

$$0 \leq \sum_{\mathcal{A}} \underbrace{\left(\Pr_{j}(\mathcal{A}) - \Pr_{k}(\mathcal{A})\right)}_{\alpha(\mathcal{A})} H(X_{i}, i \in \mathcal{A}),$$

where $\sum_{\mathcal{A}} \alpha(\mathcal{A}) = 0$. Therefore, to identify the class of "more capable" channels, one should be able to characterize the set of all valid linear inequalities over the entropies of subsets of m random variables. However, the later problem has proved to be a very challenging problem. Identifying the set of linear inequalities over the space of entropy vectors is still an unsolved problem for $m \geq 4$ [46, 47, 23]. Now consider a 2 by 2 degraded erasure broadcast channel with probability of erasure events as shown in table . Consider the supporting hyperplane $\underline{\mu} = (\mu_1, \mu_2)$, and let us look at the boundary point corresponding to this point.

It is not hard to check that the boundary point of the time-sharing region corresponding to this hyperplane is

$$(R_1, R_2) = \left(\max_{i=1,2} \left\{ \mu_i(\Pr_i(\{1\}) + \Pr_i(\{1,2\})) \right\}, \max_{i=1,2} \left\{ \mu_i(\Pr_i(\{2\}) + \Pr_i(\{1,2\})) \right\} \right).$$

On the other hand, using the Bergman's formula [40], we know that the rate vectors in the capacity region of this degraded channel should satisfy

$$R_{1} \leq I(Y_{1}; X|U) = \Pr_{1}(\{1, 2\})H(X_{1}, X_{2}|U) + \Pr_{1}(\{1\})H(X_{1}|U) + \Pr_{1}(\{2\})H(X_{2}|U)$$
$$R_{2} \leq I(Y_{2}; U) = \Pr_{2}(\{1, 2\})I(X_{1}, X_{2}; U) + \Pr_{2}(\{1\})I(X_{1}; U) + \Pr_{2}(\{2\})I(X_{2}; U),$$

for some joint distribution $\Pr(U, X = (X_1, X_2))$.

To find the boundary point of the capacity region corresponding to hyperplane $\underline{\mu} = (\mu_1, \mu_2)$, we have to solve the following optimization problem

$$f(\underline{\mu}) = \max_{\Pr(U,X)} \mu_1 R_1 + \mu_2 R_2.$$

One can further write

$$f(\underline{\mu}) = \max_{\Pr(U,X)} \mu_2 \Pr_2(\{1,2\}) H(X_1, X_2) + \mu_2 \Pr_2(\{1\}) H(X_1) + \mu_2 \Pr_2(\{2\}) H(X_2) + aH(X_1, X_2|U) + bH(X_1|U) + cH(X_2|U),$$

where $a = (\mu_1 \Pr_1(\{1,2\}) - \mu_2 \Pr_2(\{1,2\})), \ b = (\mu_1 \Pr_1(\{1\}) - \mu_2 \Pr_2(\{1\})), \ and$ $c = (\mu_1 \Pr_1(\{2\}) - \mu_2 \Pr_2(\{2\})).$ We can upper bound $f(\underline{\mu})$ by substituting $H(X_1)$, $H(X_2)$, and $H(X_1, X_2)$ with 1, 1, and 2 respectively. This gives

$$f(\underline{\mu}) \geq \max_{\Pr(U,X)} aH(X_1, X_2|U) + bH(X_1|U) + cH(X_2|U)$$

$$+ 2\mu_2 \Pr_2(\{1,2\}) + \mu_2 \Pr_2(\{1\}) + \mu_2 \Pr_2(\{2\}).$$
(3.22)

The optimization problem in (3.22) can be equivalently viewed as a Linear Program (LP) over the set of entropy vectors $h = (H(X_1, X_2|U), \text{ and } H(X_1|U), H(X_2|U))$, subject to the following constraints

$$H(X_1, X_2|U) \leq H(X_1|U) + H(X_2|U),$$

$$H(X_i|U) \leq H(X_1, X_2|U), \quad i = 1, 2,$$

$$0 \leq H(X_i|U) \leq 1, \quad i = 1, 2.$$

Solving this LP gives the following upper bound for $f(\mu)$:

$$f(\mu) \ge \max\{0, a+b, a+c, a+b+c, 2a+b+c\} + 2\mu_2 \Pr_2(\{1,2\}) + \mu_2 \Pr_2(\{1\}) + \mu_2 \Pr_2(\{2\}).$$

Furthermore, the following upper bound can be always achieved by considering X_1, X_2 to be i.i.d. uniform and letting U be a member of $\{\emptyset, X_1, X_2, (X_1, X_2), X_1 \oplus X_2\}$. It can be checked that the first four possibilities for U correspond to performing timesharing. However $U = X_1 \oplus X_2$ does not correspond to time-sharing. Based on the above argument we have the following result.

Proposition 3.2. The capacity region, C, of a degraded (2, 2)-erasure broadcast channel with correlated erasure events is

$$\mathcal{C} = \operatorname{conv} \left(\mathcal{R}_{TS} \cup \left\{ (1 - \Pr_1(\emptyset), \Pr_2(\{1, 2\})) \right\} \right),$$

where \mathcal{R}_{TS} denotes the time-sharing region and conv (A) represents the convex hull

4	d	(1)	(2)	
\mathcal{A}	Ø	{1}	$\{2\}$	$\{1,2\}$
$\Pr_1(\cdot)$	0	0.35	0.35	0.3
$\Pr_2(\cdot)$	0.5	0.15	0.15	0.2

83

Table 3.1: Probability of erasure events for Example 3.2.

of set \mathcal{A} .

Example 3.2. Consider a degraded (2, 2)-erasure channel with probability of erasures shown in Table 3.1. Using Proposition 3.2, we have plotted the capacity region of this channel in Figure 3.4. The time-sharing region is also shown in the same figure. Note that for this channel the capacity region is strictly greater than the time-sharing region.

3.4.3.2 Generalizing the Proof Technique

The main idea behind the proof of Theorem 3.3 is in understanding and exploiting the structure of the achievable region (here given by the time-sharing scheme). First, it is shown that for the "degraded" case, time-sharing is capacity achieving. Next, it is shown that for any point, \underline{R} , on the boundary of the time-sharing region one can find another (erasure) broadcast channel that is "degraded," its capacity region contains the capacity region of the original channel and shares the same boundary point \underline{R} with the time-sharing region of the original channel (see Figure 3.3).

Now, turning our attention to general broadcast channels, the capacity region of these channels is still unknown. The best known inner bound is given by Marton [49]. The Marton inner bound is tight for all the cases that the capacity region is known. It is possible to generalize the approach taken above is a systematic way as follows. Consider a general broadcast channel g with n users and transition probability matrix $\Pr_g(Y_1, \ldots, Y_n | X)$, and let $\mathcal{R}_{MT}(g)$ denote the Marton region introduced in [49].



Figure 3.4: Capacity region of the degraded (2, 2)-erasure broadcast channel of Example 3.2.

Consider the following optimization problem for a given $\underline{\mu} \geq 0$ vector:

$$f_{MT}(\underline{\mu}, \Pr_g(\cdot|\cdot)) = \max_{(R_1, \dots, R_n) \in \mathcal{R}_{MT}(g)} \sum_{i=1}^n \mu_i R_i.$$
(3.23)

Let \mathcal{A}_g be the set of all channels (defined over the same input and output alphabet as g) for which g is a degraded version of, i.e.,

$$\mathcal{A}_q = \{ \Pr_h(Y_1, \dots, Y_n | X) | \Pr_q(\cdot | \cdot) \preceq \Pr_h(\cdot | \cdot) \}.$$

Furthermore, let $\mathcal{B}_g(\underline{\mu})$ be a subset of \mathcal{A}_g such that for any channel h in $\mathcal{B}_g(\underline{\mu})$ we have

$$f_{MT}(\underline{\mu}, \Pr_g(\cdot|\cdot)) = f_{MT}(\underline{\mu}, \Pr_h(\cdot|\cdot)).$$

Now, if one can prove that $\mathcal{B}_g(\underline{\mu})$ is not empty, then the boundary point of the

capacity region of g corresponding to the supporting plan $\underline{\mu}$ is achievable with the Marton coding scheme [49].

However, we should remark that the challenge in the aforementioned approach is in characterizing the Marton inner bound region. This is not an easy task since the cost function appearing in (3.23) should be maximized over the distribution of the input and a number of auxiliary random variables present in the Marton region formula. In general, this optimization problem is not convex. As a matter of fact, both in the erasure broadcast channel analyzed here and the MIMO Gaussian broadcast channel considered in [118], the optimization problem of (3.23) can not be solved explicitly. Rather, in both cases one constrains the joint probability distribution of the transmitted signal and the auxiliary random variables to a particular form and perform that optimization over the smaller space of probability distributions, e.g., i.i.d. Bernoulli random variables in erasure broadcast channels and jointly Gaussian random variables in the MIMO Gaussian broadcast channel. In both cases, one can find channels in \mathcal{A}_g that share the same boundary point. The hope is that one can take the approach proposed here for other classes of broadcast channels and use the well-developed tools in optimization theory to find their capacity region.

3.5 Tighter Outer bounds for WE networks

In this section we will give tighter outer bounds for the capacity region of a general wireless erasure network. We will use the result derived in the previous section regarding the capacity region of erasure broadcast channels. The bound presented in this section is similar to the edge-cut bounds proposed in [45].

Consider a wireless erasure network represented by the acyclic directed graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$. Let $\mathcal{E}' \subseteq \mathcal{E}$ be a collection of links in the network. Let $\mathcal{D}(\mathcal{E}') \subseteq \mathcal{D}$ denote the subset of destination nodes that will be disconnected from *s* by removing the edges

in \mathcal{E}' from the network. For each destination node d_l in $d_l \in \mathcal{D}(\mathcal{E}')$, $\mathcal{J}(d_l, \mathcal{E}') \subseteq \mathcal{E}'$ is a subset of links from \mathcal{E}' with the following properties:

- Removing $\mathcal{J}(d_l, \mathcal{E}')$ from the network disconnects d_l from s.
- No two edges in $\mathcal{J}(d_l, \mathcal{E}')$ lie on the same path from s to d_l .

In a sense, $\mathcal{J}(d_l, \mathcal{E}')$ is a minimal edge-cut for d_l . We should remark that $\mathcal{J}(d_l, \mathcal{E}')$ may not be unique. It is possible that there are more that two minimal edge-cuts. In this case, considering any of them will lead to an outer bound on the capacity region of the wireless erasure network.

Now, it is possible to construct a multiple input, multiple output erasure broadcast channel corresponding to \mathcal{E}' and find a bound for the simultaneous rate of destinations in $\mathcal{D}(\mathcal{E}')$.

We construct the corresponding erasure broadcast channel as follows. Let $\mathcal{V}(\mathcal{E}') \subseteq \mathcal{V}$ denote the set of vertices that are the originating node of at least one edge in \mathcal{E}' . The corresponding erasure broadcast channel has one input for each node in $\mathcal{V}(\mathcal{E}')$ and one receiver (destination) for each node in $\mathcal{D}(\mathcal{E}')$. Consider a (the) minimal edgecut for each of the destinations in $\mathcal{D}(\mathcal{E}')$. For each edge (i, v) in $\mathcal{J}(d_l, \mathcal{E}')$, we connect input *i* to receiver d_l through a packet erasure channel with probability of erasure ϵ_{iv} . Note that if more than one link originating from the same node is in $\mathcal{J}(d, \mathcal{E}')$, without loss of generality, we can put one link in the corresponding erasure broadcast channel with the probability of erasure equal to the product of the probabilities of erasure of those links. This way we will get an erasure broadcast channel. We refer to the channel matrix of this broadcast channel as $\underline{\epsilon}(\mathcal{J}(d, \mathcal{E}'), d \in \mathcal{D}(\mathcal{E}'))$ (or $\underline{\epsilon}(\mathcal{E}')$ whenever the edge-cuts are uniquely defined.)

Example 3.3. Let us revisit the wireless erasure network shown in Figure 3.1.(a). Let $\mathcal{E}' = \{(1,2), (1,3), (1,4)\}$. In this case, $\mathcal{D}(\mathcal{E}') = \{2,3\}, \mathcal{J}(2,\mathcal{E}') = \{(1,2), (1,4)\}$, and $\mathcal{J}(3,\mathcal{E}') = \{(1,3), (1,4)\}$. The broadcast channel corresponding to this choice of \mathcal{E}' is



Figure 3.5: (a): Erasure broadcast channel derived from $\mathcal{E}' = \{(1,2), (1,3), (1,4)\}$ for the network in Figure 3.1. (b): Erasure broadcast channel constructed from $\mathcal{E}' = \{(1,2), (1,4), (4,3)\}$ for the same network as in (a).

shown in Figure 3.5.(a). For $\mathcal{E}' = \{(1,2), (1,4), (1,3), (4,3)\}$, we have $\mathcal{D}(\mathcal{E}') = \{2,3\}$. A possible edge-cut for each destination is $\mathcal{J}(2, \mathcal{E}') = \{(1,2), (1,4)\}$ and $\mathcal{J}(3, \mathcal{E}') = \{(1,3), (4,3)\}$. The broadcast channel corresponding to this choice of \mathcal{E}' is shown in Figure 3.5.(b).

Example 3.4. Consider the wireless erasure network shown in Figure 3.6.(a). In this network, node 1 is the source node, and the destination set is given as $\mathcal{D} = \{5,6\}$. Let $\mathcal{E}' = \{(1,3), (2,3), (2,5), (4,3), (4,6)\}$. In this case, $\mathcal{D}(\mathcal{E}') = \{5,6\}$ and $\mathcal{V}(\mathcal{E}') = \{1,2,3,4\}$. Furthermore, the edge-cut sets for destinations are $\mathcal{J}(6, \mathcal{E}') = \{(4,3), (4,6), (1,3), (2,3), (4,6)\}$ and $\mathcal{J}(5, \mathcal{E}') = \{(2,5), (1,3), (4,3), (2,3)\}$. The erasure broadcast channel corresponding to this choice of edge-cuts is shown in Figure 3.6.(b).

The following theorem provides an outer bound on the capacity region of broadcast problems for WE networks based on the previous construction.

Theorem 3.7. Consider a broadcast problem over wireless erasure network $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ with source node s and destination set $\mathcal{D} = \{d_1, \ldots, d_{|\mathcal{D}|}\}$, and let \mathcal{E}' be any subset of the edges. If $(R_1, R_2, \ldots, R_{|\mathcal{D}|})$ is a set of achievable rates for the broadcast problem, we should have

$$(R_d, d \in \mathcal{D}(\mathcal{E}')) \in \mathcal{C}(\underline{\epsilon}(\mathcal{J}(d, \mathcal{E}'), d \in \mathcal{D}(\mathcal{E}'))),$$
(3.24)



Figure 3.6: (a): A wireless erasure network with two destinations. (b): Erasure broadcast channel derived form $\mathcal{E}' = \{(1,3), (2,3), (2,5), (4,3), (4,6)\}$ for the network of part (a).

where $C(\underline{\epsilon}(\mathcal{J}(d, \mathcal{E}'), d \in \mathcal{D}(\mathcal{E}')))$ denotes the capacity region of the erasure broadcast channel corresponding to edge-cuts $\mathcal{J}(d, \mathcal{E}')$ for all $d \in \mathcal{D}(\mathcal{E}')$.

Proof. For each destination $d \in \mathcal{D}(\mathcal{E}')$, consider the received signals on the links in $\mathcal{J}(d, \mathcal{E}')$, i.e., $Y(\mathcal{J}(d, \mathcal{E}')) = (Y_{ij}, (i, j) \in \mathcal{J}(d, \mathcal{E}'))$. Now consider any sequence of block codes defined in Section 3.2 whose average probability of error goes to zero as the block length, n, increases. It can be shown that $W_d - Y^n(\mathcal{J}(d, \mathcal{E}')) - Y_d^n$ forms a Markov chain. This is true because the network is acyclic and $\mathcal{J}(d, \mathcal{E}')$ disconnects s and d. Hence, using Data Processing Inequality [53], we have

$$I(W_d; Y_d^n) \le I(W_d; Y^n(\mathcal{J}(d, \mathcal{E}'))).$$

Now we have

$$nR_d = H(W_d)$$

= $I(W_d; Y_d^n) + H(W_d | Y_d^n)$
 $\leq I(W_d; Y_d^n) + n\zeta_n$
 $\leq I(W_d; Y^n(\mathcal{J}(d, \mathcal{E}'))) + n\zeta_n,$ (3.25)

where we have used the Fano's inequality in the third line, and ζ_n will go to zero as the block length n goes to infinity. By noting that the capacity region of a broadcast channel depends on the marginals only, it becomes clear that the last inequality in (3.25) suggests that $(R_d, d \in \mathcal{D}(\mathcal{E}'))$ is an achievable rate vector for the broadcast channel constructed from edge-cuts $\mathcal{J}(d, \mathcal{E}')$ for all $d \in \mathcal{D}(\mathcal{E}')$, and this proves the theorem.

Remark 3.8. As we can see from the above theorem, the outer bound is in fact intersection of some time-sharing regions, each corresponding to a subset of nodes and edges in the original network. However, unlike the time-sharing scheme of Section 3.3.1, our outer bounding technique does not take into account the fact that operation of a node across different subset of nodes and edges should be the same in all these time-sharing regions. In other words, the time-sharing parameters used for each node across different subsets of nodes and edges is different from each other and, therefore, these parameters loose their operational meaning in the original network. This is the main difference between our outer bound and the achievable region given in Section 3.3.1.

The above theorem enables us to provide tighter outer bounds for the capacity region of general wireless erasure networks. We illustrate this fact in the following example.

Example 3.5. Let us consider the wireless erasure network shown in Figure 3.1.(a) again. As in Example 3.1, we set the erasure probabilities across the network to $(\epsilon_{12}, \epsilon_{13}, \epsilon_{14}, \epsilon_{42}, \epsilon_{43}) = (0.8, 0.2, 0.6, 0.8, 0.2)$. Using Theorem 3.7, we can derive the following four new bounds for any point in the capacity region of the network shown

in Figure 3.1.(a):

$$(R_{2}, R_{3}) \in \mathcal{C}_{1} \stackrel{\triangle}{=} \mathcal{C}(\underline{\epsilon}(\{(1, 2), (1, 4)\}, \{(1, 3), (1, 4)\})),$$

$$(R_{2}, R_{3}) \in \mathcal{C}_{2} \stackrel{\triangle}{=} \mathcal{C}(\underline{\epsilon}(\{(1, 2), (1, 4)\}, \{(1, 3), (4, 3)\})),$$

$$(R_{2}, R_{3}) \in \mathcal{C}_{3} \stackrel{\triangle}{=} \mathcal{C}(\underline{\epsilon}(\{(1, 2), (4, 2)\}, \{(1, 3), (4, 3)\})), \text{ and}$$

$$(R_{2}, R_{3}) \in \mathcal{C}_{4} \stackrel{\triangle}{=} \mathcal{C}(\underline{\epsilon}(\{(1, 2), (4, 2)\}, \{(1, 3), (1, 4)\})).$$

Any achievable (R_2, R_3) should be in the intersection of the capacity region of the above four broadcast channels. We can describe each of the capacity regions C_i , $1 \leq i \leq 4$, using Theorem 3.3 as follows:

$$\begin{split} \mathcal{C}_1 &= \left\{ (R_2, R_3) \left| \frac{R_3}{0.88} + \frac{R_2}{0.52} \le 1 \right\} \\ \mathcal{C}_2 &= \left\{ (R_2, R_3) \left| \left\{ \frac{R_3 - 0.8}{0.8} \right\}^+ + \frac{R_2}{0.52} \le 1 \right\} \\ \mathcal{C}_3 &= \left\{ (R_2, R_3) \left| \frac{R_2}{0.4} + \frac{R_3}{1.6} \le 1 \right\} \\ \mathcal{C}_4 &= \left\{ (R_2, R_3) \left| \left\{ \frac{R_2 - 0.2}{0.2} \right\}^+ + \frac{R_3}{0.88} \le 1 \right\}, \end{split}$$

where $\{x\}^+ = \max\{0, x\}$. Hence, the capacity region of broadcast problem for this wireless erasure network is bounded as

$$\mathcal{C}(\mathcal{G}, \{\epsilon_{ij}, (i,j) \in \mathcal{E}\}) \subseteq \bigcap_{i=1}^{4} \mathcal{C}_i = \left\{ (R_2, R_3) \left| \frac{R_3}{0.88} + \frac{R_2}{0.52} \le 1, \frac{R_2}{0.4} + \frac{R_3}{1.6} \le 1 \right\}.$$

However, this outer bound and the time-sharing achievable region of (3.6) coincide for the particular choice of ϵ_{ij} , $(i, j) \in \mathcal{E}$ considered in this example. Therefore, the capacity region of the wireless network shown in Figure 3.1.(a) is given by time-sharing scheme of Section 3.3.1.

Example 3.6. Consider the network shown in Figure 3.1.(a) again. Assume the



Figure 3.7: Time-sharing region and different outer bounds for Example 3.6.

erasure probabilities are as follows: $(\epsilon_{12}, \epsilon_{13}, \epsilon_{14}, \epsilon_{42}, \epsilon_{43}) = (0.5, 0.8, 0.1, 0.8, 0.5)$. In Figure 3.7 we have plotted the time-sharing region of Section 3.3.1, the outer bound given by Theorem 3.7, and the multicast type min-cut outer bound. As we see from the figure, the outer bound of Theorem 3.7 is tight for some regions; however, the min-cut outer bound is not tight anywhere.

3.6 Conclusion

In this chapter we considered a special class of network problems defined as *broadcast* problem over wireless erasure networks. In a broadcast problem one source has access to multiple information messages. Each of these information messages is requested by a particular destination. Downlink of cellular systems and TV broadcasting stations are examples of these problems. We proposed an achievable region based on time-sharing for these problems. We further showed that the time-sharing scheme
is capacity-achieving in the multiuser setup, where the destinations are connected directly to the source node. The proof technique in this case requires understanding the structure of the achievable (time-sharing) region and borrows insights from optimization theory.

Using the capacity result for the multiuser setup, we gave an outer bound on the capacity region of broadcast problems over wireless erasure networks. This bound is tighter than the outer bounds that we get from the multicast type of arguments.

Finding the capacity region of broadcast problems over general wireless erasure networks remains an open problem. In order to find the capacity region, we require tighter outer bounds. Towards this end, an optimization theory viewpoint of the problem seems promising.

Chapter 4

A Practical Scheme for Wireless Networks

4.1 Introduction

In a wireline network having a single source and a single destination, we can think of information flow in the same terms as fluid flow and obtain a max-flow min-cut result to get capacity. This treatment closely follows that of the Ford-Fulkerson [52] algorithm to give us a neat capacity result. This has been well understood for many years. However, until recently, similar min-cut capacity results were not known for any other class of network problems. The remarkable results of [9, 10] say that in a wireline network setting, we can indeed achieve the min-cut upper bounds for multicast problems. These problems were defined in Chapter 2. Furthermore, it is shown that *separation* of channel coding and network coding does not degrade performance. This means that one can first perform channel coding for each link in the systems so as to make it error-free. By employing network coding on the resulting error-free network we achieve the min-cut capacity.

In Chapter 2 we also looked at the capacity of multicast problems for a class of wireless networks called wireless erasure networks. We showed that the capacity has a nice min-cut interpretation. Furthermore, it was shown that linear combining of data at intermediate nodes of the network achieves capacity. We further saw that separate channel coding and network coding is not necessary for achieving the capacity.

In this chapter we will look into the effect of *separation* of channel and network coding for *wireless* networks. In the early sections of this chapter, we will present simple *wireless* networks where this principle of separation fails. Thus we will show that operating wireless networks in a multihop manner, where each relay node decodes the message it receives, is not necessarily the right approach. This observation was first made in [55, 56]. We will also suggest some schemes of operation that will outperform those that require the ability of relay nodes to decode.

We will focus attention on two specific wireless network models. The important features that characterize a wireless network are broadcast and interference. The first model has Gaussian channels as links and incorporates broadcast as well as interference. The second model is the wireless erasure network introduced in Chapter 2 and has erasure channels as links and incorporates broadcast, but not interference. For these models, we will show that making links error-free can sometimes degrade the performance. In fact, asking nodes to simply forward their data rather than decoding it is sometimes more advantageous. This tells us that wireless networks need to be understood differently from wireline networks. We will see some explanations as to why this is the case later in the chapter.

In our study of wireless networks, we propose a scheme of network operation that permits only two operations at nodes . One is decoding to get the original data and then resending the same message as the source. The other is forwarding the data as is received. Since each node has two options, we have an exponential-sized set of possible operations. We will present an algorithm that goes over each node at most once to find the optimal operation among this set of restricted operations. This will be a greedy algorithm that avoids searching over the exponential-sized set of possible operation allocations. We also present an algorithm that can approach the best rate arbitrarily closely in an iterative manner. This will be a "decentralized" algorithm in the sense that each node needs only one bit of information from the destination in every iteration and no knowledge of the rest of the network in order to determine its own operation.

The organization of this chapter is as follows. In Section 4.2 we present two wireless network models. These will be the Gaussian wireless network and the wireless erasure networks studied in Chapter 2. In Section 4.3 we show that with these wireless models, making links or sub-networks error-free can be sub-optimal. In Section 4.4 we will formally state the two operations that nodes will be permitted to perform. With this setup, we will state our problem of allocating appropriate operations in Section 4.5. In Section 4.6 we will see how rates are calculated for all nodes in the network and how asking certain nodes to decode and others to forward can affect the rate of the network. In Section 4.7 we will state our algorithm to find the optimal policy. In Section 4.8 we will prove optimality of the algorithm. We will see some examples in Section 4.9 that will show that the gap between the "all nodes decode" strategy and our method can be significant. In Section 4.10 we will discuss the decentralized algorithm. We present upperbounds on the rate achievable by our scheme in Section 4.11. Conclusions and further questions are presented in Section 4.12.

4.2 Two Wireless Network Models

In this section we formalize two wireless network models considered in this chapter. In both cases the network consists of a directed, acyclic graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ where $\mathcal{V} = \{1, \ldots, |\mathcal{V}|\}$ is the set of vertices, and \mathcal{E} is the set of directed edges where each edge is a communication channel.

In this chapter we assume that there is only one source and one destination, i.e., we have a *unicast problem*. Without loss of generality, let s = 1 be the source node and $d = |\mathcal{V}|$ be the destination.¹ The remaining nodes are the relay nodes which must aid communication between *s* and *d*. We will assume that every edge is on some directed path from *s* to *d*. If we have edges other than these, we remove them, and what remains is our graph \mathcal{G} . We will denote the message transmitted by node *i* by X_i and that received by node *j* by Y_j . Figure 4.1 represents a network with 6 vertices



Figure 4.1: Example of a network.

and 9 edges where 1 is the source s and 6 is the destination d. X_5 is the message transmitted by node 5 and Y_2 is that received by node 2.

(a) Gaussian Wireless Networks In these networks each edge (i, j) of the network is a Gaussian channel with some fixed attenuation factor $h_{i,j}$ associated with it. In a practical system, this may be some pathloss that depends on the physical distances between the nodes. We will assume $h_{i,j}$ to be a non-negative constant. We will assume that nodes broadcast messages, i.e., a node transmits the same message on all outgoing edges. Assuming that Figure 4.1 represents a Gaussian wireless network, X_5 is the message transmitted on edges (5,3) and (5,6). We will also assume interference, i.e., the received signal at node i is the sum of all the signals transmitted on edges coming in to it and additive white Gaussian noise n_i of variance σ_i^2 . Therefore, in general, we have

$$Y_i = n_i + \sum_{j:(j,i)\in\mathcal{E}} h_{j,i} X_j.$$

¹The results of this chapter can be generalized easily for a multicast problem.

All n_i s are assumed independent of each other as well as the messages. For Figure 4.1 this implies that $Y_2 = h_{1,2}X_1 + h_{4,2}X_4 + n_2$. We will assume that all transmitting nodes have a power constraint of P.

(b) Wireless Erasure Networks

These networks were introduced in Chapter 2. In Chapter 2, the multicast capacity of these networks were found. To remind ourselves, in these networks each edge (i, j) of the network is a binary erasure channel with erasure probability ϵ_{ij} . In this chapter we further assume that nodes (other than the source node) can transmit erasures e. If an edge takes e as input, the received signal on that edge is always e. In short, the channel for edge (i, j) (for $i \neq s$) is modified as in Figure 4.2 compared to the one in Chapter 2. We incorporate broadcast in the model, i.e., each transmitting node must send out the same signal on each outgoing edge. However, we assume that the is no interference.



Figure 4.2: Modified Erasure Channel.

4.3 Optimizing Over Sub-networks Does Not Work

Theorem 4.1. For the wireless networks described in Section 4.2, making sub-networks error-free can be suboptimal.

Proof. We give some examples to demonstrate this.

Gaussian Relay Networks: Consider a Gaussian parallel relay network consisting of two relay nodes and one source-destination pair. See Figure 4.3(a). All four channel coefficients are assumed to be 1. The relay nodes 2 and 3 are solely to aid communication from source to destination. We assume that the noise power at each receiver is σ², and the transmit power at each node is P. Let ρ ≜ P/σ² be the Signal to Noise Ratio (SNR).

One way to view the network is as a cascade of a broadcast channel (from s to $\{2,3\}$) and a multiple access channel (from $\{2,3\}$ to d). This is equivalent to assuming that the relays decode their messages correctly and code them again and transmit. If the relays are receiving independent information at rates R_1 and R_2 , we have $R_1 + R_2 \leq \log(1 + \rho)$ as the capacity region. These rate pairs (R_1, R_2) can be supported by the multiple access channel, and hence the maximum rate from s to d is no greater than $\log(1 + \rho)$. If the relays are receiving exactly the same information from the source, the maximum rate of this is $\log(1 + \rho)$. In this case, the multiple access channel is used for correlated information and can support rates up to $\log(1 + 4\rho)$. In either case, asking the relay nodes to decode limits the rate from s to d to $\log(1 + \rho)$. (We note also that the broadcast sub-network is the bottleneck in both cases.)

Now consider another strategy in which the relay nodes do not decode but only normalize their received signal to meet the power constraint and transmit it to the destination. In this case the received signal at the destination is

$$Y_4 = \sqrt{\frac{P}{P + \sigma^2}} (2X_1 + n_2 + n_3) + n_4,$$

where X_1, Y_4, n_2, n_3, n_4 are, respectively, the transmitted signal from the source, the received signal at the destination, and the noises introduced at nodes 2, 3 and d. Thus, the signal received by d is a scaled version of X_1 with additive Gaussian noise. The maximum achievable rate, denoted by R_f , is

$$R_f = \log\left(1 + \frac{\frac{4P^2}{P + \sigma^2}}{\sigma^2 + \frac{2P\sigma^2}{P + \sigma^2}}\right) = \log\left(1 + \frac{4\rho^2}{3\rho + 1}\right),$$

where ρ is as before. Here, the subscript f stands for *forwarding*.

Comparing R_d and R_f , we can see $\rho = 1$ is a critical value in the following sense. For $\rho > 1$, we have superior performance in the forwarding scheme, and for $\rho < 1$ we have better rate with relay nodes decoding and re-encoding. This implies that making a sub-network error-free (in this case the broadcast section, or the links (1, 2) and (1, 3)) can sometimes be sub-optimal.

We note that decoding at one of the relay nodes and forwarding at the other is always sub-optimal.

In general, if we have $k(\geq 2)$ relay nodes in parallel rather than two, it can be easily checked that

$$R_d = \log(1+\rho)$$
 and $R_f = \log\left(1 + \frac{k^2 \rho^2}{(k+1)\rho + 1}\right)$.

With this we get a critical value of $\rho = \frac{1}{k^2 - k - 1}$, below which decoding is better and above which forwarding is better. Clearly, this goes to zero for large k. Therefore, in the limit of $k \to \infty$, it is always favorable to forward.

It turns out that this fact is also true for Gaussian relay networks in the presence of fading. The work of [55] shows that for fading Gaussian relay networks with n nodes, the asymptotic capacity achievable with the relay nodes decoding (and re-encoding) scales like $O(\log \log n)$, whereas with the forward scheme it scales like $O(\log n)$.

Similar problems are considered in [60] and [91]. The former considers bounds and achievable rates for the Gaussian network with two parallel links, and the latter considers a network with a single source and destination and the other nodes acting as relays. The second result shows that the maximum rate achievable is $O(\log n)$. This is the same as that achieved by forwarding in our scheme.



Figure 4.3: Proof of Theorem 4.1.

• Erasure Relay Network: Consider, once again, the network of Figure 4.3(a), where, now, each link represents an erasure channel with erasure probability $\epsilon_{ij} = \epsilon$. Since we have broadcast, node *s* transmits the same messages to relay nodes 2 and 3. If the relay nodes decode and re-encode, the rate is bounded by the sum-rate capacity of the broadcast system, which gives

$$R_d = 1 - \epsilon.$$

If the relay nodes simply forward what they receive, it is easy to see that the destination sees an effective erasure probability of $(1 - (1 - \epsilon)^2)$. (We will spell out how to do this calculation for a general network in Section 4.6.) Forwarding erasures is possible since we are assuming the modified erasure channel of Fig-

ure 4.2. With this we have $R_f = 1 - (1 - (1 - \epsilon)^2)^2$. Comparing R_f and R_d , we can see that $\epsilon = \frac{3-\sqrt{5}}{2}$ is a critical value, above which decoding and re-encoding is better and below which forwarding is better.

Thus we see that for this network also, making the broadcast sub-network errorfree is not always optimal.

In general, if we have k relay nodes in parallel rather than two, we have

$$R_d = 1 - \epsilon$$
 and $R_f = 1 - (1 - (1 - \epsilon)^2)^k$,

and the critical value of e is as plotted in Figure 4.3(b). Below this, forwarding is better, and above this, decoding is better. In the limit of large k, it is always better to forward.

From this we see that making links or sub-networks error-free does not ensure optimal network operation. It can sometimes be provably sub-optimal. ■

In this proof a simple operation like forwarding the received data proved to be better than decoding it. We understand this as follows. Because of the broadcast present in wireless networks, the same data naturally gets passed on to the destination along many different paths. Therefore, some nodes receive better versions of the data on incoming links than other nodes and are automatically in a better position to decode. Forcing all the nodes to decode and be error-free only imposes additional bottlenecks on the rate. Therefore, it is beneficial to carefully check the quality of the effective signal that various nodes get to see and then decide whether to ask them to decode or not.

4.4 A Possible Set of Network Operations

It follows from the previous discussions that to obtain the optimum rate over wireless networks, the nodes must perform operations other than just decoding. Determining what the optimum operation at each node should be, especially for a general wireless network, appears to be a daunting task. We shall therefore simplify the problem by allowing one of only two operations at every node. One will be the *decode and re-encode* operation as before. The other is the far simpler operation of *forwarding* the received data as is. The first operation, viz., decode and re-encode is typically the only operation used in multihop networks and many wireline networks. In effect, we are attempting to attain higher rates by introducing the additional operation of forwarding.

We will assume that the network operates in blocks of length n. We assume that the source s has a set of message indices

$$\Omega = \{1, 2, \dots, 2^{\lfloor nR \rfloor}\}$$

and an encoding function

$$f: \Omega \to \mathcal{X}^n,$$

where \mathcal{X} is \mathbb{R} for the Gaussian wireless network and $\{0,1\}$ for the erasure wireless network. To transmit message $i \in \Omega$, the source transmits f(i). With this the source operates at rate R. $\{f(1), f(2), \ldots, f(2^{\lfloor nR \rfloor})\}$ is the set of codewords or possible transmitted messages. This set is called the codebook and is denoted by \mathcal{C} . We assume that all nodes have the codebook. For the Gaussian network we will assume that the codebook meets the power constraint, i.e., $E ||f(i)||^2 \leq P$.

Here we restrict the relay nodes to two operations. These have been introduced in the examples of Section 4.3, viz., "forward" and "decode and re-encode." We now state them formally.

(a) **Decode and Re-encode**: This operation implies that when node i receives message Y_i it performs ML decoding of Y_i to determine which message index was transmitted by s. Since it has the codebook, it re-encodes the message using the same codeword that the source s would have used and transmits the same codeword. In short, it should act like a copy of the source.

However, for this to happen, we need that the decoding be error-free. This implies that the rate R at which the source operates should be no greater that the maximum rate at which node i can decode. We will see the relevance of this constraint in Section 4.5.

(b) Forward: We will describe this operation separately for the two network models. In the Gaussian network, node *i* receives message Y_i given by

$$Y_i = n_i + \sum_{j:(j,i)\in\mathcal{E}} h_{j,i} X_j.$$

$$(4.1)$$

"Forwarding" implies that the node normalizes this signal to meet the power constraint and then transmits the message. Therefore, it transmits X_i given by

$$X_i = \sqrt{\frac{P}{E \|Y_i\|^2}} \quad Y_i.$$

We will assume that $E||Y_i||^2$ is known to *i*.

For the erasure network, nodes either decode without error and transmit the original codeword or "forward" the received data. Consider node i which sees data coming in on several edges in the form of n-length blocks of bits and erasures. For the b-th bit of such a block, it either sees erasures on every edge (and this sees an *effective* erasure) or gets to see the bit on at least one incoming edge. (It cannot happen that the node sees 1 on a particular edge and 0 on

another edge for the *b*-th position. This is because of our assumption that whenever an earlier node decodes it does so without error.) Therefore, in our interference-free model, every relay node sees an effective erasure channel from the source, i.e., it sees the codeword transmitted by the source with some bits erased. "Forwarding" means broadcasting this sequence of bits and erasures.

Note that the effective erasure probability seen by node i is a function of the network topology and parameters, ϵ_{ij} . We will see in Section 4.6.2 how this effective erasure probability can be calculated.

By restricting ourselves to only two operations, we have ensured that all nodes in the network see a Gaussian channel (with some effective SNR) or erasure channel (with some effective erasure probability) with respect to the transmitted codeword. Therefore, they can do ML decoding or typical set decoding if R is no greater than the rate that they can support. We will always ensure that R satisfies this constraint.

We can think of both operations as specific forms of network coding. In both networks and with both operations, all the information coming in at a node on different edges gets pooled together – this happens automatically in the Gaussian network and is done by the node itself in the erasure network. But the node has the choice of trying to decode, thus imposing a rate constraint, or can simply forward the information, hoping that some other node would have a better chance of decoding.

Having described the two operations permitted to the relay nodes in the two networks, we are now ready to formally state the problem.

4.5 **Problem Statement**

Since we allow only two operations to nodes, viz., "decode and re-encode" and "forward," and every relay node must perform one of these, it is enough to specify the set of relay nodes that "decode and re-encode" in order to completely specify the working of the network. The source and destination will always be excluded from this set.

If a set $D \subseteq \mathcal{V} - \{s, d\}$ is the set of nodes ² that "decode and re-encode," we will call D a **policy** for network operation.

Under policy D, each node of the network sees an effective (Gaussian or erasure) channel from the source. Let the effective SNR that node i sees under policy D be denoted by $\rho_D(i)$ for Gaussian networks. For erasure networks we denote the effective erasure probability seen by node i under policy D by $\epsilon_D(i)$. Therefore, the rate that node i can support under policy D is $\log(1 + \rho_D(i))$ or $(1 - \epsilon_D(i))$ for Gaussian or erasure networks, respectively. In general we will call this $R_D(i)$. Nodes in D as well as the destination must be able to perform error-free decoding. This means that the rate at which the source transmits must be no greater than the rates at which these nodes can decode. This tells us that under policy D, the rate R at which we can operate the network is constrained by

$$R \le \min_{i \in D \cup \{d\}} R_D(i). \tag{4.2}$$

We denote this minimum by R_D . So,

$$R_D = \min_{i \in D \cup \{d\}} R_D(i).$$
(4.3)

Intuitively, asking some nodes to decode means that there are more copies of the source in the network, and hence the rate which the destination can support increases. On the other hand, asking a node to decode introduces a constraint on the rate R. This is the tradeoff for any policy D. For instance, in Figure 4.1 consider nodes 2 and 4. If 4 forwards, node 2 sees an effective erasure probability of $\epsilon_{42}\epsilon_{12} + \epsilon_{14}\epsilon_{12}(1 - \epsilon_{42})$. (We will see how this has been calculated in Section 4.6.2.) On the other hand, if 4

²The set D should not be confused with the set of destinations defined in Chapter 2. Here, since we are looking at unicast problems, there exists only one destination d.

decodes, node 2 is at an advantage since it sees a lower effective erasure probability, viz., $\epsilon_{12}\epsilon_{42}$. However, asking 4 to decode puts a constraint on the rate as seen by (4.2) since the rate that 4 can support is only $(1 - \epsilon_{14})$. This constraint is $R_D \leq 1 - \epsilon_{14}$.

Our problem is to find the policy that gives the best rate, i.e., to find D such that R_D is maximized, viz.,

$$\max_{D} \min_{i \in D \cup \{d\}} R_D(i).$$

First we need to address the question of finding $R_D(i)$, i.e., of finding the rate at node *i* under policy *D*. Recall that X_i and Y_i are the transmitted and received messages at node *i*. If we are using policy *D*, we will denote these by $X_D(i)$ and $Y_D(i)$. We may drop the subscript *D* if it is clear which policy we are referring to. Note that for the source, the transmitted message is X_1 irrespective of the policy.

4.6 Determining the Rate at a Node – $R_D(i)$

In this section we describe a method to find the rate at an arbitrary node i when the set of decoding nodes is given by D. Therefore, we need to find the effective SNR or erasure probability of the received signal $Y_D(i)$. In order to do that, we need the concept of a partial ordering on the nodes introduced in Section 2.2.2.

Next we address the issue of determining the rate under a particular policy. We discuss this separately for Gaussian wireless networks and Erasure wireless networks.

4.6.1 Finding the Rate in Gaussian Wireless Networks

Recall that $Y_D(j)$ is the received signal at v_j under policy D. Once we know $Y_D(j)$, we can determine the signal power and the noise power in it. Denote these by $P_D(j)$ and $N_D(j)$, respectively. Consider node j. If it is decoding, $X_D(j) = X_1$. If it is forwarding,

$$X_D(j) = \sqrt{\frac{P}{E \|Y_D(j)\|^2}} \quad Y_D(j) = \sqrt{\frac{P}{P_D(j) + N_D(j)}} \quad Y_D(j).$$

We now outline a method for finding the rate for all the nodes by proceeding in the order given by \mathcal{T} defined in Section 2.2.2. Without loss of generality, assume that the nodes are already numbered according to a partial ordering. Therefore $\mathcal{T} = (s, 2, ..., d)$. Then, for 2 we only have an edge coming in from s, and hence

$$Y_D(2) = h_{1,2}X_1 + n_2.$$

Let our induction hypothesis be that we know $Y_D(j)$ for j = 1, ..., i - 1. For $Y_D(i)$ we now have

$$Y_{D}(i) = n_{i} + \sum_{j:(j,i)\in\mathcal{E}} h_{j,i}X_{D}(j)$$

$$= n_{i} + \sum_{j:(j,i)\in\mathcal{E},j\in D\cup\{s\}} h_{j,i}X_{1} + \sum_{j:(j,i)\in\mathcal{E},j\notin D\cup\{s\}} h_{j,i}X_{D}(j)$$

$$= n_{i} + \sum_{j:(j,i)\in\mathcal{E},j\in D\cup\{s\}} h_{j,i}X_{1} + \sum_{\substack{j:(j,i)\in\mathcal{E}\\j\notin D\cup\{s\}}} h_{j,i}\sqrt{\frac{P}{P_{D}(j) + N_{D}(j)}} Y_{D}(j).$$
(4.4)

By our hypothesis, we know all the $Y_D(j)$ that occur in the last summation; Substituting for these, we get $Y_D(i)$. Careful observation indicates that this will be a linear combination of X_1 and the noise terms n_2, \ldots, n_i .

In general, if this linear combination is given by

$$Y_D(i) = a_D X_1 + \sum_{j=2}^i a_{D,j}(i) n_j,$$

we have $P_D(i) = a_D^2 P$ and $N_D(i) = \sum_{j=2}^i a_{D,j}^2(i)\sigma_j^2$. Once these are known, the SNR

is simply $\rho_D(i) = \frac{P_D(i)}{N_D(i)}$, and the rate can be calculated as $R_D(i) = \log(1 + \rho_D(i))$. Clearly, the complexity of this procedure is $O(|\mathcal{V}|)$.

4.6.2 Finding Rate in Erasure Wireless Networks

We first put this problem in a graph theoretic setting. We are given a directed, acyclic graph where certain nodes act as sources. For us, the set $D \cup \{s\}$ is the set of source nodes. All the edges of the graph have certain probabilities of failing, i.e., of being absent. For us, these are the erasure probabilities of the channel. With this setup, for every node v in the network (excluding s, but including those in D) we need to find the probability that there exists at least one directed path from some source node to this node. This is the Network Reliability problem in one of its most general formulations [59, 61]. This is a well-studied problem and is known to be #P-hard [61]. Although no polynomial-time algorithms to solve the problem are known, efficient algorithms for special graphs are known. An overview of the network reliability problem can be found in [62]. In the rest of this section we propose two straightforward methods to compute the probabilities of connectivity that we are interested in. We will also mention some techniques that can reduce the computation involved in these methods.

Assume we have a policy D. Consider a node i of the network. To find $R_D(i)$ we need to find $\epsilon_D(i)$. A bit is erased at node i if it is erased on all incoming links. Remember from Chapter 2 that the erasure event of link (i, j) is represented by Z_{ij} . This random variable takes the value 0 when a bit is erased and the value 1 when a bit is not erased. Thus, it is a Bernoulli random variable with probability $(1 - \epsilon_{ij})$.

Consider all the directed paths from s to i. Let there be k_i paths. Denote the paths by B_1, \ldots, B_{k_i} . Let path B_j consist of l_j edges. We specify path B_j by writing in order the edges it traverses, i.e., with the sequence $((v_{j_1}, v_{j_2}), (v_{j_2}, v_{j_3}), \ldots, (v_{j_{l_j}}, v_{j_{l_j+1}}))$. We know that $s = v_{j_1}$ and $i = v_{j_{l_j+1}}$. Consider the set of vertices excluding i that are on path j, i.e., $\{v_{j_i} : i = 1, \ldots, l_j\}$. Some nodes in this set may belong to D, i.e., they are decoding nodes. In this case we know that they transmit the original codeword exactly. Let t be the largest index in this set such that v_{j_t} decodes. Therefore, i will not receive bit b along path B_j only if an erasure occurs on an edge that comes after v_{j_t} in the path. We associate with path B_j the product of the random variables that affect this, viz.,

$$Z_{j}^{\star} = Z_{j_{t}, j_{t+1}} \cdot Z_{j_{t+1}, j_{t+2}} \cdot \dots \cdot Z_{j_{l_{j}}, j_{l_{j+1}}}$$

This product is zero if one of the Z random variables takes value zero, which, in turn, means that an erasure occurred on that edge.

Now, *i* sees an erasure only when none of the paths from *s* to itself manage to transmit the bit to it. Therefore, *i* sees an erasure when $Z_j^{\star} = 0$ for all the paths $B_j, j = 1, \ldots, k_i$. Therefore we have

$$R_D(i) = 1 - \epsilon_D(i)$$

= $1 - \Pr\left(\bigcap_{j=1}^{k_i} (Z_j^* = 0)\right)$
= $\Pr\left(\bigcup_{j=1}^{k_i} (Z_j^* \neq 0)\right)$

One way to evaluate this is by checking all possible combinations of values that the Z variables can take and finding the total probability of those combinations that satisfy $\bigcup_{j=1}^{k_i} (Z_j^* \neq 0)$. This procedure has complexity $O(2^{|\mathcal{E}|})$. One observation that can make this procedure more efficient is the following – if we know that setting a certain subset of the Z variables to 1 is enough to make the event $\bigcup_{j=1}^{k_i} (Z_j^* \neq 0)$ happen, then for every superset of this subset, setting all the Z variables in that superset to 1 is also enough to make the event $\bigcup_{j=1}^{k_i} (Z_j^* \neq 0)$ happen. With this, we may have to check out fewer than the $2^{|\mathcal{E}|}$ possible combinations of values for the Z variables and reduce the complexity.

Another way to evaluate this is by using the Inclusion Exclusion Principle [58].

This gives us

$$\Pr\left(\bigcup_{j=1}^{k_i} Z_j^{\star} \neq 0\right) = \sum_{r=1}^{k_i} \sum_{1 \le j_1 < \dots < j_r \le k_i} (-1)^{r+1} \Pr\left(Z_{j_1}^{\star} \neq 0, \dots, Z_{j_r}^{\star} \neq 0\right).$$

Since we have k_i paths, the above expression has $2^{k_i} - 1$ terms. A general term of the form $\Pr(Z_{j_1}^* \neq 0, \ldots, Z_{j_r}^* \neq 0)$ can be evaluated by first listing all the Zvariables that occur in at least one of the r terms. Say these are $Z_{i_1j_1}, \cdots, Z_{i_qj_q}$. Now $\Pr(Z_{j_1}^* \neq 0, \ldots, Z_{j_r}^* \neq 0)$ is given by the product $(1 - \epsilon_{i_1j_1}) \times \cdots \times (1 - \epsilon_{i_qj_q})$. This procedure has complexity $O(|\mathcal{E}|2^k)$ where k is the max_i k_i . In this procedure, the complexity of listing all the variables in a certain set of r terms can be reduced by storing the lists that one makes for sets of (r-1) terms and simply adding on the zterms from the r-th term to the appropriate list.

4.7 Algorithm to Find Optimum Policy

In general, since we have $|\mathcal{V}| - 2$ relay nodes and each node has two options, viz., "forwarding" and "decoding and re-encoding," we have $2^{|\mathcal{V}|-2}$ policies. To find the optimum policy we can analyze the rate for each of these policies and determine the one that gives us the best rate. This strategy of exhaustive search requires us to analyze $2^{|\mathcal{V}|-2}$ policies.

Here, we propose a greedy algorithm that finds the optimum policy D which maximizes the rate. This algorithm requires us to analyze at most $|\mathcal{V}| - 2$ policies. In the next section we will give a proof of correctness for this algorithm.

(a) Set $D = \emptyset$.

- (b) Compute $R_D(i)$ for all $v_i \in \mathcal{V}$. (Use techniques of Section 4.6.) Find $R_D = \min_{i \in D \cup \{d\}} R_D(i)$.
- (c) Find $M = \{i | i \notin \{s, d\} \cup D, R_D \le R_D(i)\}.$

- (d) If $M = \emptyset$, terminate. D is the optimal strategy.
- (e) If $M \neq \emptyset$, find the largest $D' \subseteq M$ such that $\forall v \in D'$, $R_D(v) = \max_{i \in M} R_D(i)$. Let $D = D \cup D'$. Return to 2.

At each stage of the algorithm we look for nodes that are seeing a rate as good as or better than the current rate of network operation. If there are no such nodes, the algorithm terminates. If there are such nodes, we choose the best from among them. Thus, in every iteration, the nodes we add are such that they do not put additional constraints on the rate of the network. Therefore, the rate of the network can only increase in successive iterations.

Note that since we assume a finite network, this algorithm is certain to terminate. Also, since D cannot have more than $(|\mathcal{V}| - 2)$ nodes, the algorithm cycles between steps 2 to 5 at most $(|\mathcal{V}| - 2)$ times. This is significantly faster than the strategy of exhaustive search that requires us to analyze $2^{|\mathcal{V}|-2}$ policies.

The complexity of the algorithm depends on how fast the computation of $R_D(i)$ can be done. We have seen techniques for this computation in Section 4.6.

4.8 Analysis of the Algorithm

We first prove a Lemma regarding the effect of decoding at a particular node on the rates supportable at other nodes.

Lemma 4.1. When node v is added to the decoding set D, the only nodes i that may see a change in rate are $v \prec i$. This change can only be an increase in rate, i.e., $\forall i$ such that $v \prec j$ we have $R_D(i) \leq R_{D\cup\{v\}}(i)$. Every other node j is unaffected, i.e., $R_D(j) = R_{D\cup\{v\}}(j)$.

Proof. We give a proof for the Gaussian Network. We omit the proof for erasure networks since it uses the same ideas.

Gaussian Network : Recall the computation of $\rho_D(i)$ described in Section 4.6.1. The computation for $Y_D(i)$ depends only on (some of) the $Y_D(j)$ where (j,i) is an edge. Therefore, inductively, it is clear that $Y_D(v_i)$ (and hence $\rho_D(i)$) depends only on the nodes v where $v \prec i$. Therefore, the only nodes that are affected when v changes its operation (from "forwarding" to "decoding and re-encoding") are $v \prec i$. The rest are unaffected.

Consider one of the $X_D(j)$ terms in (4.4). Note that each of these are of power P of which some power is the signal power and the rest is the noise power. If node j changes its operation from forwarding to decoding, $X_D(j) = X_1$, i.e., the signal power increases to P and the noise power goes to 0. If node j is forwarding, $X_D(j)$ is only a scaled version of $Y_D(j)$. Since it is always of power P, if the SNR at node j increases, the signal power in $X_D(j)$ increases while the noise power decreases. From (4.4) we see that in both these cases, there is an increase in the signal power of $Y_D(i)$ and a decrease in the noise power. This implies an increase in the SNR.

Therefore, when v is added to D, by induction, for all nodes $v \prec i$, the SNR, if affected, can only undergo an increase. Naturally, we have the same conclusion for the rate.

This Lemma tells us that adding nodes to the set of decoding nodes can only increase the rate to other nodes. While this sounds like a good thing, it also puts a constraint on the rate as indicated by (4.2). It is this tradeoff that our algorithm seeks to resolve by finding the optimal set of decoding nodes.

4.8.1 **Proof of Optimality**

Theorem 4.2. The algorithm of Section 4.7 gives us an optimal set of decoding nodes.

Proof. Let S be an optimal set of decoding nodes. Let D be the set returned by the algorithm. We will prove that $R_D \ge R_S$. Then, since S is optimal, we will have

 $R_D = R_S.$

We prove $R_D \ge R_S$ in two steps. First we show that $R_{S\cup D} \ge R_S$. Then we show that $S \cup D - D = \emptyset$, i.e., $S \cup D = D$. This will complete the proof.

Step 1: In every iteration, the algorithm finds subsets D' and adds them to D. Denote by D_i the subset that is added to D in the *i*-th iteration. Assuming the algorithm goes through m iterations, we have $D = D_1 \cup \cdots \cup D_m$ where the union is over disjoint sets. In the algorithm, when D_i is added to D, all the nodes in it are decoding at the same rate, which is $R_{D_1 \cup \cdots \cup D_{i-1}}(v)$ for $v \in D_i$. We will call this rate $R_{\text{algo},i}$. Consider the smallest i such that $D_i \nsubseteq S$, i.e., D_i is not already entirely in S. **Claim**: Adding D_i to S does not decrease the rate, i.e., $R_{S \cup D_i} \ge R_S$.

Proof. Because of the acyclic assumption on the graph, we will have some nodes $v \in S$ such that $\forall u \neq v \in S$, we either have $v \prec u$, or v and u are incomparable. Let L be the set of all such nodes v. Note that by Lemma 4.1, node v supports a rate $R_S(v) = R_{\emptyset}(v)$. By (4.3), for every $v \in L$ we have the necessary condition

$$R_S \le R_S(v) = R_{\emptyset}(v). \tag{4.5}$$

Also note that D_1, \ldots, D_{i-1} are all in S, and by the definition of L and Lemma 4.1 we have

$$R_{\emptyset}(v) = R_{D_1 \cup \dots \cup D_{i-1}}(v). \tag{4.6}$$

We now consider two cases.

- If for some $w \in L$ we also have $w \in D_i$, then from (4.5) and (4.6) we have $R_S \leq R_S(w) = R_{\emptyset}(w) = R_{D_1 \cup \dots \cup D_{i-1}}(w) = R_{\text{algo},i}.$
- On the other hand, if none of the nodes in L are in D_i, pick any node v ∈ L.
 We have v ∉ D_i. We now consider two subcases.

- Let $v \notin D_1, \ldots, D_{i-1}$. We note from Steps 3 and 5 of the algorithm that

it picks out from the set of nodes not in D, all nodes with the best rate. Since v does not get picked, we have $R_{\text{algo},i} > R_{D_1 \cup \cdots \cup D_{i-1}}(v)$. This along with (4.5) and (4.6) gives us $R_S \leq R_{D_1 \cup \cdots \cup D_{i-1}}(v) < R_{\text{algo},i}$.

- The other possibility is that $v \in D_1 \cup \cdots \cup D_{i-1}$. Since the D_i s are disjoint, there is a unique j such that $v \in D_j$. Since $v \in L$, by Lemma 4.1, $R_{\text{algo},j} = R_{D_1 \cup \cdots \cup D_{j-1}}(v)$. With the same argument as that for (4.6), we have $R_{\emptyset}(v) = R_{D_1 \cup \cdots \cup D_{j-1}}(v)$. But since the algorithm never decreases rate from one iteration to the next, we have $R_{\text{algo},i} \geq R_{\text{algo},j}$. Putting these together we get $R_{\text{algo},i} \geq R_{\text{algo},j} = R_{D_1 \cup \cdots \cup D_{j-1}}(v) = R_{\emptyset}(v)$. With (4.5) this gives us $R_S \leq R_S(v) = R_{\emptyset}(v) \leq R_{\text{algo},i}$.

Therefore, in every case, we have shown that $R_S \leq R_{\text{algo},i}$. This implies that adding the rest of the nodes from D_i to S will not put additional constraints on R_S and, hence, cannot decrease the rate. Therefore, we have $R_{S \cup D_i} \geq R_S$.

Since S is optimal, this proves that $S \cup D_i$ also achieves optimal rate. We can now call this set S, and for the next value of i such that $D_i \nsubseteq S$, we can prove that $S \cup D_i$ has optimal rate. Continuing like this we have that $S \cup D$ is optimal, or, in other words, $R_{S \cup D} \ge R_S$.

Step 2: Next we wish to show that $S \subseteq D$, i.e., $S \cup D - D = \emptyset$. Let us assume the contrary. Let $T = S \cup D - D$. Therefore, $T \cap D = \emptyset$, but $T \subseteq S$. Thus, $D \cup S = D \cup T$, where D and T are disjoint. Consider $v \in T$ such that $\forall u \neq v \in T$, we either have $v \leq u$, or v and u are incomparable. We have $R_{D \cup T}(v) = R_{D \cup S}(v)$. By Lemma 4.1, $R_{D \cup T}(v) = R_D(v)$. Also, the constraint of (4.2) tells us that $R_{D \cup S} \leq R_{D \cup S}(v)$. Finally, note that since the algorithm terminates without adding v to D, we have $R_D > R_D(v)$. Putting these inequalities together we have $R_D > R_D(v) = R_{D \cup T}(v) = R_{D \cup T}(v) = R_{D \cup S}(v) = R_{D \cup S}(v) = R_{D \cup S}(v)$.

From Steps 1 and 2 we have $R_D \ge R_S$. But since S was an optimal policy, D is also an optimal policy. This proves that the algorithm does indeed return an optimal set of decoding nodes.

The only case in which this proof does not go through is when the algorithm returns $D = \emptyset$ and $S \neq \emptyset$. In this case, consider node $v \in L \subseteq S$, where L is as defined earlier. Since the algorithm does not pick up v, we have $R_{\emptyset} > R_{\emptyset}(v)$. But, $R_S \leq R_S(v) = R_{\emptyset}(v)$ from (4.5). Thus, $R_S < R_{\emptyset}$; however, this contradicts the optimality of S. Therefore, if there exists an optimal, non-empty S, the algorithm cannot return an empty D.

Corollary 4.2. The algorithm of Section 4.7 returns the largest optimal policy D.

Proof. In the proof above, we have shown that for any optimal policy S, we have $S \subseteq D$. This implies that D is the largest optimal policy.

Remark 4.3. In this chapter we consider a unicast problem. However, it is possible to easily generalize the algorithm of Section 4.7 for single source multicast problems where there is more than one destination. The only difference in the algorithm would be in the definition of R_D defined in (4.3). For a multicast problem with destination nodes d_1, \ldots, d_k , R_D should be defined as $R_D = \min_{i \in D \cup \{d_1, \ldots, d_k\}} R_D(i)$.

4.9 Examples

In this section we present some examples of networks and show how the algorithm runs on them.

4.9.1 Multistage Erasure Relay Networks

In Figure 4.4(a) we have depicted a multistage relay network. In this we have a single source and destination and k layers of relay nodes. The *i*-th layer consists of l_i nodes.

Between the *i*-th and the (i + 1)-th layer we have a complete bipartite graph where all the edges are directed from the *i*-th layer to the (i + 1)-th. We assume that each of these edges has erasure probability ϵ_i . The source is connected to all the nodes in the first layer by erasure channels with erasure probability ϵ_0 , and all the nodes in the *k*-th layer are connected to the destination by erasure channels with erasure probability ϵ_k . We will also call *d* the (k + 1)-th layer and $l_{k+1} = 1$.

Because of the structure of this network, finding the rate under a particular policy is easier than indicated in Section 4.6.2. Denote by $Q_{i,j}$ the probability that in layer *i* there are *j* nodes that do not see an erasure. This defines $Q_{i,j}$ for i = 1, 2, ..., (k+1), and $j = 0, 1, ..., l_i$. With this, for i = 1 we obtain

$$Q_{1,k} = {\binom{l_1}{k}} \epsilon_0^{l_1-k} (1-\epsilon_0)^k.$$
(4.7)

For i > 1, we can show the recursion below:

$$Q_{i,k} = \binom{l_i}{k} \sum_{t=0}^{l_{i-1}} \epsilon_{i-1}^{t(l_i-k)} (1 - \epsilon_{i-1}^t)^k Q_{i-1,t}.$$
(4.8)

Denote by e_i the probability that the at least one node in the *i*-th layer does not see an erasure. We can show that

$$e_i = \sum_{k=0}^{l_i} Q_{i,k} \left(1 - \frac{k}{l_i} \right).$$

Note that, by symmetry, whenever a node decides to decode, all the nodes in that layer decode. When layer *i* decides to decode, we set $Q_{i,l_i} = 1$ and $Q_{i,j} = 0$ for $j \neq l_i$ and continue with the recursion of (4.8) for the other layers. This also extends to the case when more than one layer decodes.

Now, our algorithm proceeds as before, but operates on layers rather than nodes, and the effective erasure probability at layer i is e_i . As an explicit example, consider



Figure 4.4: Multistage relay network.

a multistage relay network with four layers between the source and destination. Let $l_1 = 3, l_2 = 6, l_3 = 4, l_4 = 5$ and $\epsilon_0 = p, \epsilon_1 = p^2, \epsilon_2 = p, \epsilon_3 = p^3, \epsilon_4 = p$, where p is any number in the interval [0,1]. For a fixed value of p, we can find the optimum policy for the network, and this will give us the optimal rate. Figure 4.4(b) shows this optimal rate for the parameter p going from 0 to 1 (solid curve). This is not a smooth curve. The point where the right and left derivatives do not match is where either the optimum policy or the rate-determining layer changes. The rate with all nodes decoding has also been plotted (dashed curve). This rate is 1 - p, and we see that the algorithm gives us dramatically higher rates.

4.9.2Multistage Gaussian Relay Networks

We consider a multistage network similar to the one of the previous section, but in which the links represent Gaussian channels with fading coefficients h_i and with additive noise σ_i^2 at layer *i*. The indexing is identical to that in the erasure network.

Because of the structure of the network, it is easy to compute SNRs. Let $\rho(i)$ denote the SNR at layer *i*. Then, in the situation where all the nodes are forwarding,

117

the following recursion gives us the SNR. We initialize the recursion as follows:

$$a(1) = h_0^2 P \quad b(1) = \sigma_1^2 \quad \rho(1) = \frac{a(1)}{b(1)}.$$

For the rest of the layers, i.e., $i \ge 2$ we have

$$\begin{aligned} a(i) &= a(i-1)\frac{h_i^2 l_i^2}{1+\frac{1}{\rho(i-1)}} \\ b(i) &= b(i-1)\frac{h_i^2 l_i}{1+\frac{1}{\rho(i-1)}} + \sigma_i^2 \\ \rho(i) &= \frac{a(i)}{b(i)} \end{aligned}$$

As with the erasure relay network, whenever a node decides to decode, all the nodes in that layer decode. If some layers decide to decode, a simple modification of the above recursion gives us the new rates. If *i* is the smallest number such that the *i*-th layer decodes, then, clearly, the above recursion gives us rates for layers l_1 to l_i . For l_{i+1} , we set $a(i+1) = h_i^2 l_i^2 P$ and $b(i+1) = \sigma_{i+1}^2$. We have $\rho(i+1) = a(i+1)/b(i+1)$ as before, and we can continue with the recursion above for layers (i+2), etc. We repeat this modification for each layer that decodes.

Once the SNR at a layer is known, the rate is given by $\log(1 + \rho)$ as usual. With this procedure for calculating rates, we use the algorithm of Section 4.7. It now operates on layers rather than nodes.

As an explicit example, consider a multistage relay network with three layers between the source and destination. Each node is restricted to using power P = 1. Let $l_1 = 2, l_2 = 5$, and $l_3 = 3$ and $h_0 = 0.7, h_1 = 10, h_2 = 0.1$, and $h_3 = 1$. We will have $\sigma_1^2 = m^2, \sigma_2^2 = m, \sigma_3^2 = m^3$, and $\sigma_4^2 = m^2$, where *m* can be any positive real number. For a fixed value of *m*, we can find the optimum policy for the network, and this will give us the optimal rate. Figure 4.4(b) shows this optimal rate for the parameter *m* going from 0.5 to 1.5 (solid curve). As with the multistage erasure network, the curve is not smooth at points where the optimum policy or the ratedetermining layer changes. We also see the advantage compared to the case when all nodes decode (dashed curve).

4.9.3 Erasure Network with Four Relay Nodes

Consider the relay network of Figure 4.5(a). All the links have the same erasure probability p, where p is any number between 0 and 1. For this range of p, the algorithm has been used to find the optimum rates and policies. The rate is plotted in Figure 4.5(b) (solid curve). Throughout, the optimal policy is $D = \{2, 3, 5\}$. The rate with all nodes decoding is 1 - p and is also plotted (dashed curve). As expected, the algorithm outperforms the all-decoding scheme.



Figure 4.5: Erasure network with four relay nodes.

4.9.4 Gaussian Network with Three Relay Nodes

In Figure 4.6(a) we see a Gaussian network with three relay nodes. We assume that each node is restricted to use power P = 1. Let the additive noise variances be $\sigma_2^2 = m, \sigma_3^2 = m^3, \sigma_4^2 = m^2$, and $\sigma_5^2 = m^1$ where m can be an arbitrarily chosen real number. In Figure 4.6(b) we see the rate returned by the algorithm for the optimal



Figure 4.6: Gaussian network with three relay nodes.

policy for $m \in [0.5, 1.5]$ (solid curve). The rate with all nodes decoding is also plotted (dashed curve). In the region $m \in [0.5, 0.58]$ we see that the optimal policy is in fact that of decoding at all nodes and the two curves match. After that, the optimal policy changes, and hence we see that the optimal rate curve is not smooth.

4.9.5 Gaussian Network with Four Relay Nodes

In Figure 4.7 we see a Gaussian network with four relay nodes. Each node, including the source, is restricted to using power P = 1. The attenuation factors associated with the edges are $h_{1,2} = 1, h_{1,4} = 2, h_{4,2} = 3, h_{2,3} = 4, h_{4,3} = 5, h_{4,5} = 1, h_{3,6} = 3, h_{5,3} = 2,$ and $h_{5,6} = 4$. The additive noise variances associated with the nodes are $\sigma_2^2 = m, \sigma_3^2 =$ $m^3, \sigma_4^2 = m^2, \sigma_5^2 = m$, and $\sigma_6^2 = m^3$, where m can be any positive real number. In Figure 4.10 we see the rate returned by the algorithm for the optimal policy for $m \in [1,3]$ (solid curve). The rate with all nodes decoding is also plotted (dashed curve). We see that the forward/decode scheme gives us significant improvements in the rate.



Figure 4.7: Gaussian network with four relay nodes.



Figure 4.8: Rates for the Gaussian network of Figure 4.7.

4.10 A Distributed Algorithm for the Optimal Pol-

icy

The algorithm as proposed in Section 4.7 requires that the network parameters (viz., noise variances or erasure probabilities) be known before the network operation begins so that the optimum policy is known beforehand. With the algorithm in its current form the nodes cannot determine for themselves if they should decode or forward. In this section we propose a scheme that can permit nodes to determine their own operation.

The algorithm works iteratively to converge to a rate. In each iteration, the rate of operation of the network is incremented or decremented depending on whether the previous transmission was successful or not. In every iteration, all the nodes get to decide their operation for themselves.

Let R^* be the maximum rate of the network. This is not known beforehand. We assume that parameters R, δ , and N are known to all the nodes beforehand. The blocklength n is also predetermined and known to all the nodes. In addition, we require that the nodes have a common source of randomness so that they can generate the *same* random codebook individually. With this, consider the following algorithm:

- (a) All nodes generate the (same) codebook for rate R. They all set k = 0.
- (b) s transmits a randomly chosen codeword X_1 .
- (c) Every relay node i attempts to decode the received message Y_i . If it can decode without error, it transmits the decoded codeword. ³ Otherwise, it forwards the received message (with appropriate scaling, for the Gaussian network).
- (d) The destination attempts to decode the received message.If it decodes without error, it sends back bit 1 to all the other nodes to indicate successful decoding.Otherwise, it sends back bit 0 to all other nodes.
- (e) All nodes increment k. k = k + 1. If transmitted bit was zero, all nodes set $R = R - \delta/2^k$. If transmitted bit was one, all nodes set $R = R + \delta/2^k$.
- (f) While $k \leq N$, go to Step 1.

Theorem 4.3. If the maximum rate of the network, viz. R^* is in the range [R -

 $\delta, R+\delta$, the algorithm above converges to it with an accuracy of $\frac{\delta}{2^N}$.

³One method of error detection is for a node to perform typical set decoding and assume an error if it finds more than one codeword that is jointly typical with the received message. Other methods of error detection are the introduction of Cyclic Redundancy Checks (CRCs) or an ARQ protocol, e.g., [63].

Proof. The source starts by transmitting at rate R. Each relay node receives messages on all incoming links and decodes the message if it can. If it cannot, it simply forwards what it has received. With this procedure, nodes decide their own operation. (The order in which they decide this is a partial order in the sense defined in Section 2.2.2.) After the destination receives all its incoming messages, it tries to decode. If $R > R^*$, the destination will definitely not be able to decode. If $R \le R^*$, we claim that the destination will be able to decode. This is because when a node decodes, it only improves the rates for other nodes. Also, note that an arbitrary node v decides whether to decode or not only after all the nodes before it in the partial order have already determined if the rate they can support is greater or smaller than R. Since, by Lemma 4.1, these are the only nodes that affect the rate for v and they decode whenever they can, node v always gets to see the best situation it can as far as rate R is concerned. This is true for the destination also.

Therefore, depending on whether the destination can decode or not, we can say if R^* is greater or smaller than R. If this bit of information is transmitted back to the source and other nodes, they can accordingly decide whether to increase or decrease the rate for the next transmission. Thus, we have a decision tree of rates such that the ability or inability of the decoder tells us which path to traverse in that tree we can finally converge on a rate sufficiently close to the actual rate R.

This algorithm provides a very natural mode of network operation that obviates the need for a central agent to know the entire network and decide the optimum policy. Although some communication from the destination to the source and other nodes is required, this is minimal and should be easily possible in a practical network setting.

We mention that the algorithm we present can be made more sophisticated such that it works for all values of R^* , rather than just those in the interval $[R - \delta, R + \delta]$. We omit the details in the interests of brevity.

124

4.11 Upperbounds On the Maximum Rate

The algorithms of Section 4.7 as well as Section 4.10 converge to the maximum rate possible with the decode/forward scheme, but we have no way of simply looking at the network and saying what this maximum rate will be. In this section we present upperbounds on the rate achievable with the limited operations that we use in this chapter.

Before we derive upperbounds, let us remind ourselves of the notion of a cut-set defined in Section 2.2.2. An s - d cut is defined as a partition of the vertex set \mathcal{V} into two subsets, \mathcal{V}_s and $\mathcal{V}_d = \mathcal{V} - \mathcal{V}_s$, such that $s \in \mathcal{V}_s$ and $d \in \mathcal{V}_d$. The cutset $[\mathcal{V}_s, \mathcal{V}_d]$ is the set of edges passing across the cut. Also, \mathcal{V}_s^* is the subset of nodes in the source set that have an edge in the cutset. Similarly, \mathcal{V}_d^* is the subset of nodes in the destination set that have an edge in the cutset.

4.11.1 Upperbound for Gaussian Networks

For Gaussian networks, it is evident that making the additive noise zero at certain nodes can only increase the maximum rate available at d. In particular, let us make the additive noise zero at all nodes except \mathcal{V}_d^* . Therefore, the received messages (and the transmitted messages) at all nodes in \mathcal{V}_s are exactly the same as that transmitted by the source. Now, if we permit the nodes in the destination set to decode cooperatively, the rate at which they can decode will give us an upperbound on the rate that the destination can decode.

Note that the SNR at node $v_j \in \mathcal{V}_d^*$ is

$$\frac{P}{\sigma_j^2} \left(\sum_{i:(i,j)\in [\mathcal{V}_s,\mathcal{V}_d]} h_{i,j} \right)^2.$$

Since our codebook and noise are Gaussian distributed, the optimum scheme for

decoding co-operatively is taking a suitable linear combination of received messages and then decoding that. For optimal decoding, we find the linear combination that gives us the best SNR. It is easy to show that the best SNR possible is the sum of the SNRs seen by each node in \mathcal{V}_d^* .

Therefore, an upperbound on the rate is

$$R \le \log\left(1 + \sum_{j \in Y(\mathcal{V}_s)} \frac{P}{\sigma_j^2} \left(\sum_{i:(i,j) \in [\mathcal{V}_s, \mathcal{V}_d]} h_{i,j}\right)^2\right)$$

for every cut \mathcal{V}_s .

4.11.2 Upperbound for Erasure Networks

As in the above section, we can obtain an upperbound on the rate for erasure networks by making certain links perfect, or free of erasures. Therefore, we can obtain an upperbound on the rate by making all edges other than those in $[\mathcal{V}_s, \mathcal{V}_d]$ perfect. With this, all the received (and transmitted) messages in \mathcal{V}_s are exactly the same as the codeword transmitted by the source. Now, it is clear that the rate at which the nodes in \mathcal{V}_d^* can decode co-operatively is an upperbound on the rate available at the destination.

Clearly, the effective erasure probability seen by the set of nodes \mathcal{V}_s^* is $\prod_{(i,j)\in[\mathcal{V}_s,\mathcal{V}_d]} \epsilon_{ij}$. This gives us an upperbound on the rate. We have

$$R \le 1 - \prod_{(i,j) \in [\mathcal{V}_s, \mathcal{V}_d]} \epsilon_{ij}$$

for every cut \mathcal{V}_s .

Note that in Chapter 2 a different min-cut upperbound was proposed and was shown to be achievable. This gives the capacity of the network under the assumption that the destination has perfect side-information regarding erasure locations from across the network. This is very different from the setup of this chapter.

4.12 Conclusions

To summarize, we have shown that making each link error-free in a wireless network is sub-optimal. Thus, a multihop approach, in which every relay node decodes the received message, is not necessarily the correct approach for all wireless networks. We have proposed a scheme for network operation that is of use in practical networks and in which operations performed by a node are restricted to decoding and forwarding – both of which are common operations performed in a network setting. We have suggested an algorithm that finds the optimum policy without exhaustive search over an exponential number of policies and also proposed a method to converge to the correct policy without having a central decision-making agent.

The algorithm of Section 4.7 can find the maximum rate and optimum policy for any Gaussian or wireless erasure network. In addition, the bounds presented in Section 4.11 give us some idea of what sort of optimal rates to expect. However, we still do not know what sort of policies are optimal in what ranges of erasure probabilities or SNR. The examples of Section 4.3 suggest that when the links are poor (high erasure probabilities or low SNR), it is better to decode. It would be interesting to know if this is true for general networks and what thresholds exist below which a certain operation is always preferred.

Also, Corollary 4.2 tells us that the algorithm returns the largest decoding set. Since decoding is the more costly of the two operations considered here, an algorithm that finds the smallest decoding set such that the maximum rate is obtained is of interest.

Finally, we note that in this chapter we considered only two types of operations. However, it is possible to imagine a larger set of operations and the optimal choice of operation from among these. Finding practical schemes that improve upon the present algorithm is an interesting avenue for future work.
Chapter 5

Estimation over Wireless Erasure Networks

5.1 Introduction

Recent advances in Micro-Electro-Mechanical Systems (MEMS) technology have provided us with cheap, low power, customizable sensors capable of sensing, signal processing, and communication in wireless media (for university and industrial prototype of these sensors, see [64, 65]). These advances have given rise to an increasing number of applications for networks of sensors in different aspects of our life. As mentioned earlier in Chapter 1, examples of these applications appear in environmental monitoring, industrial, transportation, and home systems automation, control of distributed embedded systems (such as robots or UAVs), and even medical services [66, 67].

One important feature of these applications is that not in all of them the main objective is high-data *rate* communication between components of the network. Different tasks such as distributed computation, detection, and control can be the main purpose for deploying these networks.

Given the increasing use of wireless sensor networks for different tasks other than data communication, a theoretical framework for analysis of the ultimate performance and the optimal schemes of operation for each of these tasks is required.

Towards this end, recently a great deal of attention has been directed towards

networked control systems in which components communicate over wireless links or communication networks that may also be used for transmitting other unrelated data (see, e.g., [68, 69] and the references therein). The estimation and control performance in such systems is severely affected by the properties of the communication channels. Communication links introduce many potentially detrimental phenomena, such as quantization error, random delays, data loss, and data corruption to name a few, that lead to performance degradation or even stability loss. As emergent applications in distributed control mature, these issues have gained a lot of focus from the community.

In the previous chapters of this thesis, we looked at the performance of different classes of wireless networks for different network problems. In these problems the main objective is maximizing the reliable rate of *communication* between the nodes of the network. In this chapter, we look at another task, namely *control and estimation*, over these networks. We are interested in the problem of estimation and control of a dynamical process across the wireless erasure network model introduced in Section 2.3. We consider dynamical process evolving in time that is being observed by a sensor. The sensor needs to transmit the data over a network to a sink (destination) node, which can either be an estimator or a controller. However, the links in the network stochastically erase packets.

Prior work in this area has focused on studying the effect of packet erasures by a single link in an estimation or control problem. Assuming certain statistical models for the packet erasure process, stability and control performance of such systems were analyzed in [70]-[73]. To counteract the degradation in performance, some approaches have been proposed in the literature [74]-[79]. In particular, in [76], a sub-optimal estimator and regulator is proposed that minimizes a quadratic cost. This approach was later extended by [77, 78]. [79] also considered the related problem of optimal estimation across a packet erasure link that erases packets in an independent and identically distributed (i.i.d.) fashion, and obtained bounds on the expected error

covariance.

Most of the above designs aimed at designing a packet-loss compensator. The compensator accepts those packets that the link successfully transmits and comes up with an estimate for the time steps when data is lost. If the estimator is used inside a control loop, the estimate is then used by the controller. A more general approach is to design both an encoder and a decoder for the communication link to counteract the effect of stochastic packet erasures. This was considered for the case of a single communication link in [80] and [81]. It was demonstrated that using encoders and decoders can improve both the stability margin and the performance of the system.

In this chapter, we consider the design of encoders and decoders for wireless erasure network model introduced in Chapter 2. The optimal transmission strategy over general networks for the purpose of estimation and control is largely an open problem. In [82], Tatikonda studied some issues related to quantization rates required for stability when data was being transmitted over a network of digital memoryless channels. Also relevant is the work of Robinson and Kumar [83], who considered the problem of optimal placement of the controller when the sensor and the actuator are connected via a series of communication links. They ignore the issue of delays over paths of different lengths (consisting of different number of links), and under a *Long Packet Assumption* come up with the optimal controller structure. There are two main reasons why the problem of encoding data for transmission is much more complicated in the case of transmission over a network:

- If the intermediate nodes are allowed to process data, it introduces an element of memory. The network is not equivalent to an erasure channel with probability of successful transmission as the reliability of the network.
- There are potentially many paths from the source to any node that offer data with varying amounts of delay.

We begin by proving a separation principle that allows us to separate the control problem into one of designing a state-feedback optimal controller, and another of transmitting information across unreliable links. This also allows us to identify the information that needs to be made available to the controller for achieving optimal performance. We then propose a simple recursive algorithm that ensures that this information is available to the controller. Even though the algorithm requires a constant amount of memory, transmission, and processing at any node, it is optimal for *any* packet erasure pattern and has many additional desirable properties that we illustrate. The analysis of the algorithm identifies a property of the network called the max-cut probability that is relevant for the purpose of stability of the control loop. We also provide a framework to analyze the performance of our algorithm. The main contributions of this chapter are as follows:

- (a) We identify the optimal information processing strategy that should be followed by the nodes of the network to allow the sink to calculate the optimal estimate at every time step. This algorithm is optimal for *any* packet erasure process, yet requires a constant amount of memory, processing, and transmission by any node per time step. Due to a separation principle, the algorithm also solves the optimal control problem.
- (b) We analyze the stability of the expected error covariance for this strategy when the packet erasure events are independent from one time step to the next and across channels. For any other scheme (e.g., transmitting measurements without any processing), our conditions are necessary for stability. For channels with correlated erasures, we show how to extend this analysis.
- (c) We calculate the performance for our algorithm for channels that erase packets independently. We provide a mathematical framework for evaluating the performance for a general network and provide expressions for networks containing

links in series and parallel. We also provide lower and upper bounds for the performance over general networks. For any other strategy, these provide lower bounds for achievable performance.

Our results can also be used for *synthesis* of networks to improve estimation performance. We consider a simple example in which the optimal number of relay nodes to be placed is identified for estimation performance. We also consider optimal routing of data in unicast networks.¹ Simulation results are provided to illustrate the results.

The chapter is organized as follows. In the next section, we set up the problem and state the various assumptions. Then, we state a separation principle that allows us to focus on the optimal estimation problem. In Section 5.4 we identify a recursive yet optimal processing and transmission algorithm. We then specialize to the case of packet erasure events occurring in a memoryless fashion and independently across different links. We first do a stability analysis of the algorithm in Section 5.5 to obtain conditions on the packet erasure probabilities under which the estimate error at the sink retains a bounded covariance. Following that, in Section 5.6 we analyze the performance of the algorithm. We derive an expression for general networks and evaluate it explicitly for specific classes of networks. We also provide bounds for general networks. We then illustrate the results using some examples. Finally, we consider some extensions of the analysis by considering correlated erasures and using the results already derived for optimal routing in unicast networks and for network synthesis.



Figure 5.1: The set-up of the control across communication networks problem. For most of the discussion in the chapter, we will ignore the network between the controller and the actuator. See, however, Section 5.4.2.

5.2 Problem Setup

Consider the arrangement in Figure 5.1. Let a discrete-time linear process evolve according to the equation

$$x(k+1) = Ax(k) + Bu(k) + w(k),$$
(5.1)

where $x(k) \in \mathbf{R}^n$ is the process state, $u(k) \in \mathbf{R}^m$ is the control input, and w(k) is the process noise assumed to be white, Gaussian, and zero mean with covariance matrix R_w .² The initial condition x(0) is assumed to be independent of w(k) and to have mean zero and covariance matrix R(0). The state of the plant is measured by a sensor that generates measurements according to the equation

$$y(k) = Cx(k) + v(k).$$
 (5.2)

The measurement noise v(k) is white, zero-mean, Gaussian (with covariance matrix R_v), and independent of the plant noise w(k). We assume that the pairs (A, B) and

 $^{^1 \}mathrm{Unicast}$ network in this chapter refers to routing the data over only one particular path in the network.

 $^{^{2}}$ The results we present continue to hold for time-varying systems, but we consider the time-invariant case to simplify the presentation.

 $\{A, R_w^{\frac{1}{2}}\}$ are stabilizable and the pair (A, C) is observable.

The sensor communicates with a controller across a wireless erasure network that was introduced in Chapter 2. The sensor constitutes the source node and is denoted by s. The controller is designated as the sink, or the destination node d. As in Chapter 2, the network is represented by a directed graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$. The edges of the graph represents packet erasure channels. We do not make any assumptions a priori regarding the erasure process across the network. The following remarks regarding the model are in order:

- We assume that the links take in as input a finite vector of real numbers. Therefore, the assumption is that sufficient bits per data packet and a high enough data rate are present so that quantization error is negligible.
- We will nominally consider the delays introduced by the channel to be less than one time step according to which the discrete-time dynamical process evolves. Most of the results in the chapter can, however, be extended to the case when delays are present. In particular, the algorithm in the case of delays is provided in Section 5.4.1.
- In this chapter we also assume a global clock so that each node is synchronized.

If the packet erasure process is independent for different links and also from one time step to the next (or, in other words, memoryless), the probability of dropping a packet on link $(i, j) \in \mathcal{E}$ is given by ϵ_{ij} independent of time. More sophisticated models of erasure process such as erasures occurring according to a Markov chain to capture the *bursty* nature of them can readily be thought of and will be considered towards the end of this chapter.

We refer to individual realizations of the erasure process as a packet drop (erasure) sequence. The operation of the different nodes in the network at every time-step k can be described as follows:

- (a) Every node computes a function of all the information it has access to at that time.
- (b) It transmits the function on all the out-going edges. We allow some additional information in the message that tells us the time step j such that the function that the node transmits corresponds to the state x(j). The sink node calculates the control input u(k) based on the information it possesses.
- (c) Every node observes the messages from all the incoming links and updates its information set for the next time step. For the source node, the message it receives at time step k corresponds to the observation y(k).

Note that the time line we have proposed ensures a strictly causal operation. Thus, at time step k, the function that the source node transmits depends on measurements $y(0), y(1), \dots, y(k-1)$. Similarly, even if there was no packet erasure, if the sink node is d hops away from the source node (i.e., the shortest path from the source node to the sink node involves d edges), its control input u(k) at time k can only depend on measurements $y(0), y(1), \dots, y(k-d-1)$. Thus, unlike the model in [83], every communication edge consumes one hop, or in other words, one time step, as data is transmitted over it. We can easily adapt the discussion presented below to the causal case.

The controller at every time step calculates a control input u(k) and transmits it to the actuator. For the time being, we ignore any communication channel between the controller and the actuator. We will revisit the presence of a controller - actuator channel later in the chapter and show how simple modifications to our design can take care of them. The controller aims at minimizing a quadratic cost function

$$J_T = E\left[\sum_{k=0}^T \left(x^T(k)Qx(k) + u^T(k)Ru(k)\right) + x^T(T+1)P_{T+1}^cx(T+1)\right], \quad (5.3)$$

where the expectation is taken over the uncorrelated variables x(0), $\{w(k)\}$ and $\{v(k)\}$. Note that the cost functional J_T above also depends on the random packet drop sequences in each link. However, we do not average across packet-drop processes; the solution we will present is optimal for arbitrary realizations of the packet dropping processes. For considering the stability of the system, we will consider the infinite horizon cost

$$J_{\infty} = \lim_{T \to \infty} \frac{J_T}{T}.$$

Without the communication network, the problem is thus the same as the classical LQG control synthesis problem. The presence of the network, however, alters the problem drastically. It is unclear *a priori* what the structure of the optimal control algorithm should be and in what way the links in the network should be used to transmit information. We aim to solve the following problems:

- (a) Identify the optimal processing and transmission algorithm at the nodes that allow the controller to minimize the cost J_T . Clearly, sending measurements alone might not be the optimal thing to do, since in such a scheme, erasing a packet would mean loss of information that cannot be compensated for in the future. We are particularly interested in strategies that do not entail an increasing amount of transmission and memory at the nodes.
- (b) Identify the conditions on the network that would lead to a stable system.
- (c) Identify the best possible performance of the system in terms of the quadratic cost that can be achieved.

For future reference, we will denote this problem set-up as problem \mathcal{P}_1 .

137

5.3 Preliminary Results

We wish to construct the optimal control input and the optimal information processing algorithm at each node that minimizes the cost function J_T . If the packet dropping links were not present, the optimal performance would have been achieved if every node in the network transmitted the latest measurement it received and the controller calculated the LQ optimal control input based on the estimate obtained by using a Kalman filter. However, the presence of packet erasures disrupts this operation since the Kalman filter requires continuous access to measurements, which is denied by the packet erasures. To solve for the optimal controller design and the optimal information processing algorithm at each node, we begin by introducing some notation.

For the node i, denote by $\mathcal{I}^{i}(k)$ the information set that it can use to generate the message that it transmits at time step k. This set contains the aggregate of the information the node has received on the incoming edges at time steps $t = 0, 1, \dots, k-1$. As an example, for the source node s,

$$\mathcal{I}^{s}(k) = \{y(0), y(1), \cdots, y(k-1)\}.$$

Without loss of generality, we can restrict our attention to information-set feedback controllers, i.e., controllers of the form $u(k) = u(\mathcal{I}^d(k), k)$. For a given information set at the destination $\mathcal{I}^d(.)$, let us denote the minimal value of J_T by $J_T^*(\mathcal{I}^d)$. The packet drops occur according to a random process. Let $\lambda_{pq}(k)$ be the binary random variable describing the packet erasure event on link $(p,q) \in \mathcal{E}$ at time k. $\lambda_{pq}(k)$ assumes the value "dropped" if the packet is dropped on link (p,q) at time k, and "received" otherwise. For a network with independent and memoryless packet erasures, $\lambda_{pq}(k)$ is distributed according to Bernoulli with parameter p_{pq} . We define $\lambda_{pp}(k) =$ "received". Given the packet drop sequences in each link, at time step k we can define a time stamp $t^i(k)$ for node *i* such that the packet erasures did not allow any information transmitted by the source after $t^i(k)$ to reach the *i*-th node in time for it to be a part of $\mathcal{I}^i(k)$.

Now consider an algorithm \mathcal{A}_1 that proceeds as follows. At time step k, every node takes the following actions:

- (a) Calculate the estimate of state x(k) based on the information set at the node.
- (b) Transmit its entire information set on the outgoing edges.
- (c) Receive any data successfully transmitted along the incoming edges.
- (d) Update its information set and affix a time stamp corresponding to the time of the latest measurement in it.

When this algorithm is executed for a particular drop sequence, the information set at node i will be of the form

$$\mathcal{I}^{i}(k) = \{y(0), y(1), \cdots, y(t^{i}(k))\},\$$

where $t^i(k) < k$ is the time stamp as defined above. This is the maximal information set $\mathcal{I}^{i,\max}(k)$ that the node *i* can possibly have access to with any algorithm. For any other algorithm, the information set will be smaller than this since earlier packets, and hence measurements, might have been dropped.

Note that for two information sets $\mathcal{I}^d(k,1)$ and $\mathcal{I}^d(k,2)$ related by $\mathcal{I}^d(k,1) \subseteq \mathcal{I}^d(k,2)$, we have $J_T^*(\mathcal{I}^d(k,1)) \leq J_T^*(\mathcal{I}^d(k,2))$. Thus, in particular, one way to achieve the optimal value of J_T is through the combination of an information processing algorithm that makes the information set $\mathcal{I}^{d,\max}(k)$ available to the controller and a controller that optimally utilizes the information set. Further, one such information processing algorithm is the algorithm \mathcal{A}_1 described above. However, this algorithm requires increasing data transmission as time evolves. Surprisingly, in a lot of cases, we can achieve performance equivalent to this naïve solution using a constant amount of transmission and memory.

To see this, we first state the following separation principle. For any random variable $\alpha(k)$, denote by $\hat{\alpha}(k|\beta(k))$ the minimum mean squared error (mmse) estimate of $\alpha(k)$ given the information $\beta(k)$.

Proposition 5.1. [Separation Principle] Consider the packet-based optimal control problem \mathcal{P}_1 defined in section 5.2. Suppose that each node transmits all the measurements it has access to at every time step, so that the decoder has access to the maximal information set $\mathcal{I}^{d,\max}(k)$ at every time step k. Then, for an optimizing choice of the control, the control and estimation costs decouple. Specifically, the optimal control input at time k is calculated by using the relation

$$u(k) = \hat{u}_{LQ} \left(k | \mathcal{I}^{d, \max}(k), \{ u(t) \}_{t=0}^{k-1} \right),$$

where $u_{LQ}(k)$ is the optimal LQ control law and $\hat{u}_{LQ}\left(k|\mathcal{I}^{d,\max}(k), \{u(t)\}_{t=0}^{k-1}\right)$ denotes the minimum mean squared error (mmse) estimate of $u_{LQ}(k)$ given the information set $\mathcal{I}^{d,\max}(k)$ and the previous control inputs $u(0), \dots, u(k-1)$.

Proof. The proof is along the lines of the standard separation principle (see, e.g., [84, Chapter 9]; see also [77, 80]) and is omitted for space constraints. ■

Informally, the separation principle states that if every node transmits all previous measurements at every time step, then the controller design consists of an estimator (e.g., a Kalman filter) that calculates the minimum mean squared error estimate of the current state value using all available measurements and the previous control inputs together with an LQ optimal controller. There are two reasons this principle is useful to us:

(a) We recognize that the optimal controller does not need to have access to the in-

formation set $\mathcal{I}^{d,\max}(k)$ at every time step k. The encoders and the decoder only need to ensure that the controller receives the quantity $\hat{u}_{LQ}\left(k|\mathcal{I}^{d,\max}(k), \{u(t)\}_{t=0}^{k-1}\right)$, or equivalently, $\hat{x}\left(k|\mathcal{I}^{d,\max}(k), \{u(t)\}_{t=0}^{k-1}\right)$.

(b) If we can ensure that the controller has access to this quantity, the controller design part of the problem is solved. The optimal controller is the solution to the LQ control problem.

Thus, the controller needs access to the estimate $\hat{x} \left(k | \mathcal{I}^{d,\max}(k), \{u(t)\}_{t=0}^{k-1} \right)$ based on the information set $\mathcal{I}^{d,\max}(k)$ (that is of the form $\{y(0), y(1), \dots, y(j)\}$ for some j < k) and the previous control inputs. We can make another simplification in the problem by separating the dependence of the estimate on measurements from the effect of the control inputs. In the context of our problem, this is useful since the nodes in the network do not then need access to the control inputs and can concentrate solely on the effect of measurements. The effect of the control inputs can be taken care of by the decoder or the controller that has access to all previous control inputs. To this end, we state the following Theorem.

Theorem 5.1. Consider the problem \mathcal{P}_1 defined in section 5.2. The quantity $\hat{x}\left(k|\mathcal{I}^{d,\max}(k), \{u(t)\}_{t=0}^{k-1}\right)$, where $\mathcal{I}^{d,\max}(k)$ is of the form $\{y(0), y(1), \dots, y(j)\}$ for some j < k, can be calculated as the sum of two quantities:

$$\hat{x}(k|\mathcal{I}^{d,\max}(k), \{u(t)\}_{t=0}^{k-1}) = \bar{x}(k) + \psi(k),$$

where $\bar{x}(k)$ depends only on $\mathcal{I}^{d,\max}(k)$ and should be provided to the controller by the network and $\psi(k)$ depends only on the control inputs and can be computed by a recursive linear filter at the controller. Further, term $\bar{x}(k)$ that the network needs to deliver is, in fact, the mmse estimate of the state x(k) of a process evolving as

$$x(k+1) = Ax(k) + w(k), (5.4)$$

given the measurements $y(0), y(1), \dots, y(j)$ that are assumed to originate from a sensor of the form (5.2).

Proof. The proof of the above theorem is provided in [90].

As mentioned above, the advantage of separating the effects of measurements and the control inputs is that the nodes in the network can concentrate on delivering $\bar{x}(k)$ to the controller, the controller (which has access to all the control inputs) can then calculate $\psi(k)$ and, in turn, $\hat{x}\left(k|\{y(t)\}_{t=0}^{j}, \{u(t)\}_{t=0}^{k-1}\right)$. The nodes in the network do not need access to the control inputs.

Thus, consider an alternative estimation problem \mathcal{P}_2 . A process of the form (5.4) is observed by a sensor of the form (5.2). There is an estimator across the network that needs to estimate the state x(k) of the process in the mmse sense at every time step k. The network is modeled in the same manner as in the original problem \mathcal{P}_1 . We can once again define the information set $\mathcal{I}^i(k)$ that the node i has access to at time k and the corresponding maximal information set $\mathcal{I}^{i,\max}(k)$. What is the optimal information processing algorithm to be followed by each node that allows the estimator to calculate the estimate of x(k) based on the information set $\mathcal{I}^{d,\max}(k)$? By the arguments above, the optimal information processing algorithm for the nodes in the network in the problems \mathcal{P}_1 and \mathcal{P}_2 is identical. For the presentation of the algorithm and analysis of its properties, we will consider this equivalent problem \mathcal{P}_2 while keeping in mind that, to solve problem \mathcal{P}_1 , the controller can then calculate $\psi(k)$ to include the effect of the previous control inputs and, finally, the new control input u(k) by utilizing the separation principle. We now move on to present the algorithm.

142

5.4 Optimal Encoding at Each Node

We saw that one algorithm that ensures that the estimator (or the sink node) has access to $\mathcal{I}^{d,\max}(k)$ and hence can calculate the mmse estimate of x(k) is \mathcal{A}_1 . However, that algorithm involves an increasing amount of memory and transmission at each node. We will now describe an algorithm \mathcal{A}_2 that achieves the same performance at the expense of constant memory, processing, and transmission (modulo the transmission of the time stamp). The algorithm proceeds as follows. At each time step k, every node i takes the following actions:

- (a) Calculate its estimate $\hat{x}^i(k)$ of the state x(k) based on any data received at the previous time step k - 1 and its previous estimate. The estimate can be computed using a switched linear filter, as shown later.
- (b) Affix a time stamp corresponding to the latest measurement used in the calculation of the estimate in step 1, and transmit the estimate on the outgoing edges.
- (c) Receive data on the incoming edges, if any, and store it for the next time step.

To prove that algorithm \mathcal{A}_2 is indeed optimal, we need the following intermediate result.

Lemma 5.2. Consider any edge (i, j) and any packet drop pattern. At time step k, let the node i transmit the measurement set

$$S^{ij} = \{y(0), y(1), \cdots, y(l)\}$$

on the edge (i, j) if algorithm \mathcal{A}_1 is executed. If, instead, algorithm \mathcal{A}_2 is executed, the node *i* transmits the estimate

$$\hat{x}(k|S^{ij}) = \hat{x}(k|\{y(0), y(1), \cdots, y(l)\})$$

along the edge (i, j) at time step k.

Proof. The proof readily follows by induction on the time step k. For time k = 1, the source node s transmits $\{y(0)\}$ along all edges of the form (s, .) while following algorithm \mathcal{A}_1 and the estimate $\hat{x}(1|y(0))$ while executing algorithm \mathcal{A}_2 . If any edge is not of the form (s, .), there is no information transmitted along that edge in either algorithm. Thus, the statement is true for k = 1. Now assume that the statement is true for k = n. Consider the node i at time k = n + 1. If the node i is the source node, the statement is true by an argument similar to that at k = 1. Let us assume that node i is not the source node. Consider all edges that transmitted data at time step k = n to node i. Let each of the edges $(j, i) \in \mathcal{N}_I(i)$ transmits the measurement set

$$S^{ji} = \{y(0), y(1), \cdots, y(t(j))\}$$

if algorithm \mathcal{A}_1 is being executed. Also, denote the measurement set that the node *i* has access to from time step k = n - 1 as

$$S^{ii} = \{y(0), y(1), \cdots, y(t(i))\}.$$

Note that at time step k = n, the node *i* transmitted the set S^{ii} along all outgoing edges in $\mathcal{N}_O(i)$. Let *v* be the node for which

$$t(v) = \max\{t(i) \cup \{t(j) | (j, i) \in \mathcal{N}_I(i)\}\}.$$

Then at time k = n+1, the node *i* transmits along all outgoing edges the measurement set

$$\mathcal{S}^1 = \{y(0), y(1), \cdots, y(t(v))\}.$$

Now consider the case when algorithm \mathcal{A}_2 is being executed. By the assumption of

the statement being true at time step k = n, the edges (j, i) transmit the estimate

$$\hat{x}(n|S^{ji}) = \hat{x}(n|\{y(0), y(1), \cdots, y(t(j))\})$$

for all node js such that $(j, i) \in \mathcal{N}_I(i)$. Also, since at time k = n the node transmitted S^{ii} on any edge (i, .) in algorithm \mathcal{A}_1 , it has access to the estimate $\hat{x}(n|S^{ii})$ when algorithm \mathcal{A}_2 is executed. Clearly, the set S^{vi} is the superset of all sets S^{ii} and S^{ji} where $(j, i) \in \mathcal{N}_I(i)$ and v have been defined above. Thus, the estimate that the node i calculates at time k = n + 1 is $\hat{x}(n + 1|S^{vi})$. But the measurement set S^{vi} is simply the set \mathcal{S}^1 . Hence, at time step k = n + 1, the node i transmits along all outgoing edges the estimate $\hat{x}(n + 1|\mathcal{S}^1)$. Thus, the statement is true at time step k = n + 1 along all edges of the form (i, .). Since the node i was arbitrary, the statement is true at time k = n, it is true at time k = n + 1. But it is also true at time k = 1. Thus, by the principle of mathematical induction, it is true at all time steps.

Note that we have also shown that if at time step k, the node has access to the measurement set S^{ii} from time step k-1 when algorithm \mathcal{A}_1 is executed, it has access to the estimate $\hat{x}(k-1|S^{ii})$ from time step k-1 when algorithm \mathcal{A}_2 is executed. We can now state the following result.

Proposition 5.2. The algorithm A_2 is optimal in the sense that it leads to the minimum possible error covariance at any node at any time step.

Proof. Consider a node *i*. At time *k*, let $j \in \{i\} \cup \{q | (q, i) \in \mathcal{N}_I(i)\}$ such that $\lambda_{ji}(k-1) =$ "received". Denote the measurement set that is transmitted from node *j* to node *i* at time step *k* under algorithm \mathcal{A}_1 by S^{ji} . As in the proof of Lemma 5.2, there is a node *v* such that S^{vi} is the superset of all the sets S^{ji} . Thus, the estimate

of node *i* at time *k* under algorithm \mathcal{A}_1 is

$$\hat{x}^{\mathcal{A}_1}(k) = \hat{x}(k|S^{vi}).$$

From Lemma 5.2, when algorithm \mathcal{A}_2 is executed at time step k, the node i has access to the estimates $\hat{x}(k-1|S^{ji})$. Once again, since S^{vi} is the superset of all the sets S^{ji} , the estimate of node i at time step k is simply

$$\hat{x}^{\mathcal{A}_2}(k) = A\hat{x}(k-1|S^{vi}) = \hat{x}(k|S^{vi}).$$

Thus we see that for any node *i*, the estimates $\hat{x}^{\mathcal{A}_1}(k)$ and $\hat{x}^{\mathcal{A}_2}(k)$ are identical for any time step *k* for any packet drop pattern. But algorithm \mathcal{A}_1 leads to the minimum possible error covariance at each node. Thus, algorithm \mathcal{A}_2 is optimal.

The following remarks regarding the above algorithm are in order:

(a) The step of calculating the estimate at each node in the algorithm \mathcal{A}_2 can be implemented as follows. The source node implements a Kalman filter and updates its estimate at every time step with the new measurement received. Every other node *i* checks the time-stamps on the data coming on the incoming edges. The time-stamps correspond to the latest measurement used in the calculation of the estimate being transmitted. Then, node *i* updates its timestamp using the relation

$$t^{i}(k) = \max_{j \in \{q \mid (q,i) \in \mathcal{N}_{I}(i)\} \cup \{i\}} \lambda_{ji}(k-1)t^{j}(k-1).$$
(5.5)

Suppose the maximum of (5.5) is achieved by node $n \in \{q | (q, i) \in \mathcal{N}_I(i)\} \cup \{i\}$. Then the node *i* updates its estimate as

$$\hat{x}^i(k) = A\hat{x}^n(k-1),$$

where $\hat{x}^t(k)$ denotes the estimate of the state x(k) maintained by the node t. Thus, the processing can be done as a switched linear filter.

- (b) We have made no assumptions on the packet drop pattern. The algorithm provides the optimal estimate based on $\mathcal{I}^{d,\max}(k)$ for an arbitrary packet drop sequence, irrespective of whether the packet drop can be modeled as an i.i.d. process or a more sophisticated model like a Markov chain or even adversarial. The algorithm results in the optimal estimate at every time step for any instantiation of the packet drop sequence, not merely in the optimal average performance. We also do not assume any knowledge of the statistics of the packet drops at any of the nodes.
- (c) We have proved that the algorithm is optimal for any node. Thus we do not need to assume only one sink. The algorithm is also optimal for multiple sources if all sources have access to measurements from the same sensor. For multiple sources with each source obtaining measurements from a different sensor, the problem remains open.
- (d) Any received data vector x̂(k|j) "washes away" the effect of all previous packet drops. It ensures that the estimate at the receiving node is identical to the case when all measurements y(0), y(1), ..., y(j) were available, irrespective of which previous data packets had been dropped.
- (e) A priori we had not made any assumption about a node transmitting the same message along all the out-going edges. It turned out that in this optimal algorithm, the messages are the same along all the edges. This property is especially useful in the context of wireless communication, which is inherently broadcast in nature.
- (f) The communication requirements can be reduced by adopting an event-based

protocol in which a node transmits only if it updated its estimate based on data arriving on an incoming edge. This will not degrade the performance but reduce the number of transmissions, especially if packet drop probabilities are high.

In a sense our algorithm corresponds to communication of information over a digital channel, while the strategy of using no encoding is an analog communication scheme. Our algorithm allows the intermediate nodes to play the role of repeaters that help to limit the effect of the channel by decoding and re-encoding the data along the way. In analog channels, repeaters make no difference in the received SNR and hence the signal quality. Similarly, in our setting, if raw measurements are being transmitted, presence of intermediate nodes does not help in improving the estimation performance.

5.4.1 Presence of Delays

If the links introduce random delays, the algorithm remains optimal irrespective of the possibility of packet rearrangements. Each node, at every time step, still calculates the estimate of the state x(k) based on any information received at that time step and the previous estimate from its memory, affixes the correct time stamp, and transmits it along out-going edges. Further, if the graph is finite, the stability conditions of the algorithm we present in the next section also do not change.

5.4.2 Channel Between the Controller and the Actuator

If we look at the proof of the separation principle given above, the crucial assumption was that the controller knows what control input is applied at the plant. Thus, if we have a channel between the controller and the plant, the separation principle would still hold, provided there is a provision for acknowledgment from the receiver to the transmitter for any packet successfully received over that channel.³ The optimal information processing algorithm presented above carries over to this case as well. We can also ask the question of the optimal encoder-decoder design for the controller-actuator channel. The optimal decoding at the actuator end will depend on the information that is assumed to be known to the actuator (e.g., the cost matrices Q and R and the measurements from the sensor). Design of the decoder for various information sets is an interesting open problem.

For Sections 5.5 and 5.6, we will analyze the stability and performance of the above algorithm by assuming that packets are erased independently from one time step to the next and are uncorrelated in space. We will return to more general packet dropping processes in Section 5.8.

5.5 Stability Analysis

We are interested in stability in the bounded second moment, or the mean squared sense. Thus, for the problem \mathcal{P}_1 , we say that the system is stable if $E[J_{\infty}]$ is bounded, where the expectation is taken over the packet dropping processes in the network. For the problem \mathcal{P}_2 , denote the error at time step k as

$$e^d(k) = x(k) - \hat{x}^d(k),$$

where $\hat{x}^d(k)$ is the estimate of the destination node. We can compute the covariance of the error e(k) at time k as

$$R^{d}(k) = E\left[e^{d}(k)(e^{d}(k))^{T}\right],$$

³Note that we do not require acknowledgements for the sensor-controller channel.

where the expectation is taken over the initial condition x(0), the process noise w(j), and the measurement noise v(j). We can further take the expectation with respect to the packet dropping process in the network and denote

$$P^d(k) = E\left[R^d(k)\right].$$

We consider the steady-state error covariance in the limit as k goes to infinity, i.e.,

$$P^{d}(\infty) = \lim_{k \to \infty} P^{d}(k).$$
(5.6)

If the limit exists and is bounded, we will say that the estimate error is stable; otherwise it is unstable.⁴ Note that because of the separation principle, the cost J_T and the estimation error covariance $R^d(k)$ are related through the equation

$$J_{T} = E\left[x^{T}(0)S(0)x(0)\right] + \operatorname{tr}\left(S(0)R^{d}(0)\right) + \left(\sum_{k=0}^{T-1}\operatorname{tr}\left(S(k+1)R_{w} + \left(A^{T}S(k+1)A + Q - S(k)\right)R^{d}(k)\right),\right)\right]$$
(5.7)

where S(k) is the Riccati variable that arises because of the LQ optimal control being calculated and evolves as

$$S(k) = A^{T}S(k+1)A + Q - A^{T}S(k+1)B\left(B^{T}S(k)B + R\right)^{-1}B^{T}S(k+1)A.$$

Because of the stabilizability assumptions, S(k) would tend to a constant value S as the horizon T becomes longer. Thus, in the limit as $T \to \infty$, we obtain

$$J_{\infty} = \operatorname{tr} (SR_w) + \operatorname{tr} \left(\left(A^T S A + Q - S \right) R^d(\infty) \right).$$

⁴Our definition of stability requires the average estimation error or quadratic control cost to be bounded. Other metrics that require the probability density function of the cost to decay at a fast enough rate are also possible.

If we now take the expectation with respect to the packet dropping processes in the network, we obtain

$$E[J_{\infty}] = \operatorname{tr} (SR_w) + \operatorname{tr} \left(\left(A^T S A + Q - S \right) P^d(\infty) \right).$$
(5.8)

Thus, the stability conditions for problems \mathcal{P}_1 and \mathcal{P}_2 are identical. We now proceed to evaluate these conditions.

For node d and time k, let $t^{d}(k)$ denote the time-stamp of the most recent observation used in estimating x(k) at the destination node d. This time-stamp evolves according to (5.5). The expected estimation error covariance at time k at node d can thus be written as

$$P^{d}(k) = E \left[e^{d}(k)(e^{d}(k))^{T} \right]$$

= $E \left[\left(x(k) - \hat{x}^{d}(k) \right) \left(x(k) - \hat{x}^{d}(k) \right)^{T} \right]$
= $\sum_{l=0}^{k} E \left[\left(x(k) - \hat{x}^{d}(k|t^{d}(k) = l) \right) \left(x(k) - \hat{x}^{d}(k|t^{d}(k) = l) \right)^{T} \right] \Pr \left(t^{d}(k) = l \right),$

where in the final equation we have evaluated the expectation with respect to the packet dropping process, and $\hat{x}^d(k|t^d(k) = l)$ denotes the estimate of x(k) at the destination node given all the measurements $\{y(0), y(1), \dots, y(l)\}$. We see that the effect of the packet dropping process shows up in the distribution of the time-stamp of the most recent observation used in estimating x(k). For future use, we denote the *latency* for the node d at time k as

$$l^{d}(k) = k - 1 - t^{d}(k).$$

Also, denote the mmse estimate of x(k) given all the measurements $\{y(0), y(1), \cdots, y(k-1)\}$ by P(k). It is well-known that P(k) evolves according to the Riccati

recursion

$$P(k+1) = AP(k)A^{T} + R_{w} - AP(k)C^{T} (CP(k)C^{T} + R_{v})^{-1} CP(k)A^{T}$$

We can now rewrite the error covariance $P^d(k)$ as

$$P^{d}(k) = \sum_{l=0}^{k-1} E\left[\left(x(k) - \hat{x}^{d}(k|l^{d}(k) = l)\right) \left(x(k) - \hat{x}^{d}(k|l^{d}(k) = l)\right)^{T}\right] \times \Pr\left(l^{d}(k) = l\right)$$
$$= \sum_{l=0}^{k-1} \left[A^{l}P(k-l) \left(A^{l}\right)^{T} + \sum_{j=0}^{l-1} A^{j}Q \left(A^{j}\right)^{T}\right] \times \Pr\left(l^{d}(k) = l\right).$$
(5.9)

The above equation gives the expected estimation error covariance for a general network with any packet dropping process. The effect of the packet dropping process appears in the distribution of the latency $l^d(k)$. As we can see from (5.9), the stability of the system depends on how fast the probability distribution of the latency decreases.

To analyze the stability, we use the following result from [80] restated here for independent packet drops.

Proposition 5.3. Consider a process of the form (5.4) being estimated using measurements from a sensor of the form (5.2) over a packet-dropping link that drops packets in an i.i.d. fashion with probability ϵ . Suppose that the sensor calculates the mmse estimate of the measurements at every time step and transmits it over the channel. Then, the estimate error at the receiver is stable in the bounded second moment sense if and only if

$$\epsilon |\rho(A)|^2 < 1,$$

where $\rho(A)$ is the spectral radius of the matrix A appearing in (5.4).

5.5.1 Network with Links in Parallel

We begin by considering a network consisting only of links in parallel. Consider the source and the sink node being connected by a network with m links in parallel with the probability of packet drop in the *i*-th link being ϵ_i . Since the same data is being transmitted over all the links, the distribution of the latency in (5.9) remains the same if the network is replaced by a single link that drops packets when all the links in the original network drop packets and transmits the information if even one link in the original network allows transmission. Thus, the packet drop probability of this equivalent link is $\epsilon_1 \epsilon_2 \cdots \epsilon_m$. The necessary and sufficient condition for the error covariance to diverge thus becomes

$$\epsilon^* |\rho(A)|^2 < 1$$

where

$$\epsilon^{\star} = \epsilon_1 \epsilon_2 \cdots \epsilon_m$$

5.5.2 Necessary Condition for Stability in Arbitrary Networks

Using the result for parallel networks, we can obtain a necessary condition for stability for general networks as follows.

Proposition 5.4. Consider a process of the form (5.4) being estimated using measurements from a sensor of the form (5.2) through a wireless erasure network. Now, for each s-d cut-set $[\mathcal{V}_s, \mathcal{V}_s^c]$ define the equivalent erasure probability as

$$\epsilon(\mathcal{V}_s) = \prod_{(i,j)\in[\mathcal{V}_s,\mathcal{V}_s^c]} \epsilon_{ij}.$$
(5.10)

Then, a necessary condition for the error covariance to converge is

$$\epsilon_{\rm net} |\rho(A)|^2 < 1,$$

where ϵ_{net} is the network erasure probability defined as

$$\epsilon_{\text{net}} = \max_{\mathcal{V}_s: s - d \operatorname{cut}} \epsilon(\mathcal{V}_s).$$
(5.11)

Proof. Denote the given network by \mathcal{G}_1 . Consider a cut-set of this network with source set \mathcal{V}_s . Form another network \mathcal{G}_2 by replacing all links in $\mathcal{E} - [\mathcal{V}_s, \mathcal{V}_d]$ by perfect links, i.e., links that do not drop packets and additionally do not consume one time step to transmit data across. Now, for any packet drop pattern, denote the information sets that the destination node has access to at any time step k over networks \mathcal{G}_2 and \mathcal{G}_1 by $\mathcal{I}^{d,\mathcal{G}_2}(k)$ and $\mathcal{I}^{d,\mathcal{G}_1}(k)$, respectively. It is obvious that

$$\mathcal{I}^{d,\mathcal{G}_1}(k) \subseteq \mathcal{I}^{d,\mathcal{G}_2}(k).$$

Thus, the estimate error covariances at the destination node for the two networks are related by

$$P^{d}(k|\mathcal{I}^{d,\mathcal{G}_{1}}(k)) \succeq P^{d}(k|\mathcal{I}^{d,\mathcal{G}_{2}}(k)).$$

Hence, by considering the stability of error covariance over network \mathcal{G}_2 , we can obtain a necessary condition for the stability of error covariance over network \mathcal{G}_1 . Since the edges within each source and the destination sets do not introduce any delay or error, \mathcal{G}_2 consists of the source and the sink joined by parallel edges in the cut-set only, i.e., $[\mathcal{V}_s, \mathcal{V}_d]$. The condition for the error covariance across \mathcal{G}_2 to converge is thus

$$\epsilon(\mathcal{V}_s)|\rho(A)|^2 < 1,$$

where $\epsilon(\mathcal{V}_s)$ is defined in (5.10). This is thus a necessary condition for error covariance across \mathcal{G}_1 to be stable. One such condition is obtained by considering each cut-set. Thus, a necessary condition for the error covariance to converge is

$$\epsilon_{\rm net} |\rho(A)|^2 < 1$$

where ϵ_{net} is defined in (5.11).

5.5.3 Network with Links in Series

Consider a network consisting of two links in series with probability of packet drops ϵ_{sr} and ϵ_{rd} , where the relay node is denoted by r, and s and d denote the source and destination nodes. Denote the estimate at node i at time k by $\hat{x}^i(k)$. Also, let $e^s(k)$ be the error between x(k) and $\hat{x}^r(k)$. Similarly, let $e^r(k)$ be the error between $\hat{x}^r(k)$ and $\hat{x}^d(k)$. We are interested in the second moment stability of $e^s(k) + e^r(k)$. Clearly, a sufficient condition is that both $e^s(k)$ and $e^r(k)$ individually be second moment stable. Applying Theorem 5.3, this translates to the condition

$$\epsilon_{sr} |\rho(A)|^2 < 1$$

 $\epsilon_{rd} |\rho(A)|^2 < 1.$

If ϵ^* is the greater of the probabilities ϵ_{sr} and ϵ_{rd} , the sufficient condition thus is

$$\epsilon^* |\rho(A)|^2 < 1.$$

But this is identical to the necessary condition stated in Theorem 5.4. Thus, the condition above is both necessary and sufficient. Clearly this argument can be extended to any finite number of links in series. If there are m links in series with the probability of drop of the *i*-th link being ϵ_i , then a necessary and sufficient condition

for the estimate error to converge at the sink node is

$$\epsilon^* |\rho(A)|^2 < 1,$$

where

$$\epsilon^{\star} = \max(\epsilon_1, \epsilon_2, \cdots, \epsilon_m).$$

5.5.4 Sufficient Condition for Arbitrary Networks

We now proceed to prove that the condition stated in Theorem 5.4 is sufficient as well for stability.

Theorem 5.3. Consider the assumptions of Theorem 5.4 on the erasure process and the network. Then the estimation error covariance under algorithm \mathcal{A}_2 will be stable if

$$\epsilon_{\rm net} |\rho(A)|^2 < 1,$$

where ϵ_{net} is defined in (5.11).

Proof. First note that if a packet erasure link between two nodes u and v with probability of erasure ϵ_{uv} is replaced by two parallel links with erasure probabilities $p_{uv}^{(1)}$ and $p_{uv}^{(2)}$ such that $p_{uv} = p_{uv}^{(1)} p_{uv}^{(2)}$, then the average error covariance of the estimation under algorithm \mathcal{A}_2 will not change at any node. This is true simply because the probability distribution of the latency in (5.9) will not change with this replacement.

Next consider the set $\Gamma = \{\gamma_1, \gamma_2, \cdots, \gamma_m\}$ of all simple directed paths from the source to the destination in the network graph. Note that these paths may share edges. Consider the following optimization problem

$$\min_{\beta_j} \prod_{j=1}^m \beta_j, \tag{5.12}$$

subject to the following constraints

$$\prod_{\substack{j:(u,v)\in\gamma_j\\1\geq\beta_j\geq0}}\beta_j\geq\epsilon_{uv}\quad\forall(u,v)\in\mathcal{E}$$
(5.13)

A simple change of variables,

$$\psi_j = -\log\beta_j,\tag{5.14}$$

transforms the above optimization problem into the following linear program in the variables ψ_j 's:

$$\max_{\psi_j} \sum_{j=1}^m \psi_j \tag{5.15}$$

subject to

$$\sum_{\substack{j:(u,v)\in\gamma_j\\\psi_j\geq 0}}\psi_j\leq -\log\epsilon_{uv}\quad\forall (u,v)\in\mathcal{E}$$

The solutions of the optimization problems (5.12) and (5.15), denoted by $\{\beta_j^*\}$ and $\{\psi_j^*\}$, are related through the relation

$$\psi_j^\star = -\log\beta_j^\star.$$

The structure of the linear program (5.15) is the same as the one used for finding the maximum flow possible in a fluid network [85, Page 59], which has the same topology as our packet dropping network with the capacity of the link (u, v) equal to $-\log \epsilon_{uv}$. The solution to the problem of finding the maximum flow through a fluid network is well-known to be given by the max-flow min-cut theorem. Using this fact, we see

that the solution to the optimization problem (5.15) is given by

$$\psi_j^{\star} = \min_{\mathcal{V}_s: s - d \operatorname{cut}} \sum_{(u,v) \in [\mathcal{V}_s, \mathcal{V}_s^c]} - \log \epsilon_{uv}.$$

Thus, for the optimization problem (5.12), the solution is given by

$$\beta_{j}^{\star} = \max_{\substack{\mathcal{V}_{s}:s-d \operatorname{cut}\\ (u,v) \in [\mathcal{V}_{s}, \mathcal{V}_{s}^{c}]}} \prod_{(u,v) \in [\mathcal{V}_{s}, \mathcal{V}_{s}^{c}]} -\log \epsilon_{uv}$$
(5.16)
$$= \max_{\substack{\mathcal{V}_{s}:s-d \operatorname{cut}\\ (\mathcal{V}_{s})}} \epsilon(\mathcal{V}_{s})$$

$$= \epsilon_{\operatorname{net}},$$

where $\epsilon(\mathcal{V}_s)$ and ϵ_{net} have been defined in (5.10) and (5.11), respectively.

Consider the paths in the set Γ . Form a new set \mathcal{B} of all those paths γ_j s for which the associated optimal variable β_j^* is strictly less than one. The remaining paths in Γ have equivalent erasure probability as unity and can thus be ignored. Now form a new network \mathcal{G}' as follows. The node set of \mathcal{G}' is the union of those nodes of the original network \mathcal{G} that are present on any path in \mathcal{B} . Each pair of nodes (u, v) in the node set of \mathcal{G}' is connected by (possibly) multiple links. If an edge (u, v) between two nodes u and v is present in a path $\gamma_j \in \mathcal{B}$, we add an edge between nodes u and v in \mathcal{G}' and associate with it an erasure probability β_j^* . By considering all the edges in \mathcal{G} and following this procedure, we construct the edge set of \mathcal{G}' . The following properties of \mathcal{G}' are easily verified.

- By construction, G' can be presented as a union of edge-disjoint paths. Each path in G' corresponds to one path in B. Furthermore, for each path, the probabilities of packet erasure on all the links of that path are equal.
- By virtue of (5.16) and the procedure followed to construct \mathcal{G}' , the product of the probabilities of packet erasure of the different paths is equal to the equivalent probability of the network, ϵ_{net} , for the network \mathcal{G} .

• For any pair of nodes that were connected by a link in \mathcal{G} , the product of the probabilities of packet erasure of the links in \mathcal{G}' connecting these two nodes is greater than or equal to the erasure probability of the link between the same pair of nodes in \mathcal{G} . This can be seen from the first inequality constraint of (5.13).

Therefore, the estimate error covariance at the sink by following algorithm \mathcal{A}_2 in the original network \mathcal{G} is less than or equal to the error covariance by following \mathcal{A}_2 in the new network \mathcal{G}' . Thus, to obtain a sufficient condition on stability, we can analyze the performance of \mathcal{A}_2 in the network \mathcal{G}' . For this we consider another algorithm, which we denote as \mathcal{A}_3 . In this algorithm we consider the disjoint paths given in \mathcal{G}' and assume that estimates on different paths are routed separately. Thus, if a node lies on many paths, on each path it forwards the packets it received on that path only. Clearly, the performance \mathcal{A}_3 cannot be better than \mathcal{A}_2 since in \mathcal{A}_2 we send the most recent estimate received from different paths at any node compared to forwarding the estimates on different paths separately from each other.

Therefore, to prove the theorem we only need to show the stability of estimation using protocol \mathcal{A}_3 , assuming that the condition of Theorem 5.4 holds. Since we do not mix the estimates obtained from different paths in \mathcal{A}_3 , the network can be considered as a collection of parallel paths, with each path consisting of links with equal drop probability. Therefore, using the stability analysis of serial networks presented earlier, each path (from a stability point of view) can be viewed as an erasure channel with drop probability equal to the drop probability of one link in that path. Using the stability analysis of parallel networks, we see that the stability of the new network under algorithm \mathcal{A}_3 is equivalent to the stability of a packet erasure link with probability of erasure equal to the product of the drop probabilities of different paths, which as mentioned earlier is ϵ_{net} . Thus, if

$$\epsilon_{\rm net}|\rho(A)|^2 < 1, \tag{5.17}$$

the network \mathcal{G}' is stable under protocol \mathcal{A}_3 . But the performance of \mathcal{A}_3 cannot be better than of \mathcal{A}_2 . Thus \mathcal{G}' is stable under \mathcal{A}_2 . Therefore, the original network \mathcal{G} is stable under protocol \mathcal{A}_2 assuming (5.17) is satisfied.

The following remarks are in order:

- (a) We have provided a necessary and sufficient condition for the expected error covariance to remain bounded for a network of arbitrary topology. For any other causal data processing algorithm, it provides a necessary condition for stability.
- (b) Let us, in particular, compare the stability conditions for the algorithm A_2 to those for a simpler algorithm \bar{A} in which the intermediate nodes do not have any memory. At each time step k, the source node forwards the measurement y(k-1). The intermediate nodes compare the time stamps of the measurements they received at the previous time step along different incoming edges and forward the most recent one. If they did not receive any measurement on the last time step, they do not transmit anything. It is clear that the probability that the destination node receives any particular measurement y(k) from the source over the network is upper-bounded by the reliability of the network (see, e.g., [86]). Let us consider a simple example of a line network in which nedges each with drop probability ϵ are combined in series. With our optimal algorithm, the necessary and sufficient condition for expected estimate error covariance to be stable is $\epsilon |\rho(A)|^2 < 1$. On the other hand, in algorithm \bar{A} , the probability that any measurement is received by the destination node is

 $q = 1 - (1 - \epsilon)^n$. By a method similar to the one used in [79], it can be proven that a *necessary* condition for stability is $q|\rho(A)|^2 < 1$. As an example, for n = 5 links and drop probability p = 0.2, q = 0.67. Thus, our algorithm yields a huge improvement for the stability margin, from $\rho(A) \leq 1.22$ for the simple algorithm to $\rho(A) \leq \sqrt{5}$ for our algorithm. This is an instance of the effect of having "repeaters" in the path as mentioned above.

5.6 Performance Analysis

In this section we calculate the performance of the algorithm \mathcal{A}_2 for drops independent in time and uncorrelated in space. Once again, from (5.8) we realize that the cost function for the problem \mathcal{P}_1 can be calculated easily as long as we are able to calculate the steady-state expected estimate error covariance defined in (5.6). We now provide a framework to calculate the expected error covariance at *any* node.

Let $L_{uv}(k)$ be the difference between k and the time at which the last successful transmission before time k occurred on link (u, v):

$$L_{uv}(k) = \min\{j \ge 1 | \lambda_{uv}(k+1-j) = "received"\}.$$

By convention, we adopt $L_{uu}(k) = 1$. Thus, the last time that any message is received at node v from link (u, v) is $k - L_{uv}(k) + 1$, and that message has time-stamp $t^u(k - L_{uv}(k))$. Then, (5.5) can be rewritten as

$$t^{v}(k) = \max_{u \in \{j \mid (j,v) \in \mathcal{N}_{I}(v)\} \cup \{v\}} t^{v}(k - L_{uv}(k)).$$

Now $L_{uv}(k)$ is distributed as a truncated geometric random variable with the density

function

$$\Pr(L_{uv}(k) = i) = \begin{cases} (1 - \epsilon_{uv})\epsilon_{uv}^{i-1} & \forall k > i \ge 1\\ 1 - \sum_{i=1}^{k-1} \Pr(L_{uv}(k) = i), \quad k = i. \end{cases}$$

As an example, for the source node s, without extending the definition we have $t^{s}(k) = k - 1$ for $k \ge 1$. We can get rid of the truncation by extending the definition of $t^{u}(k)$. For all k < 0, we define $t^{u}(k) = 0$. Thus, using the extended definition, $t^{s}(k) = (k - 1)^{+}$ for all k, where $x^{+} = \max\{0, x\}$. In general, using the extended definition of $t^{u}(k)$ for all k and for any node u, we can easily verify that

$$t^{v}(k) = \max_{u \in \{j \mid (j,v) \in \mathcal{N}_{I}(v)\} \cup \{v\}} t^{u}(k - L_{uv}(k)),$$
(5.18)

where $L_{uv}(k)$'s are now independent random variables distributed according to a geometric distribution with parameters ϵ_{uv} s. Thus

$$\Pr\left(L_{uv}(k)=i\right) = (1-\epsilon_{uv})\epsilon_{uv}^{i-1} \quad \forall i \ge 1, \forall k.$$
(5.19)

Since $L_{uv}(k)$ s do not depend anymore on k from now on, we will omit the argument k. From (5.18) we can write $t^{v}(k)$ in terms of the time-stamp at the source node $(k-1)^{+}$ as

$$t^{v}(k) = \max_{P:\text{an s-v path}} (k - 1 - \sum_{(u,v)\in P} L_{uv})^{+}, \qquad (5.20)$$

where the maximum is taken over all paths P in the graph \mathcal{G} from source s to the node v. Therefore, the latency at node v can be written as

$$l^{v}(k) = k - 1 - t^{v}(k) = \min\{k - 1, \min_{P:\text{an s-v path}} (\sum_{(u,v)\in P} L_{uv})\}$$

Since we are interested in the steady-state expected error covariance, we consider

the steady-state behavior of the latency $l^{v}(k)$. As $k \to \infty$, the distribution of $l^{v}(k)$ approaches that of the variable l^{v} defined as

$$l^{v} = \min_{P:\text{an s-v path}} (\sum_{(u,v)\in P} L_{uv}).$$
(5.21)

Let us now concentrate on the destination node.⁵ For the destination node d, we refer to l^d as the *steady-state latency* of the network. From (5.9), the steady-state error covariance can now be rewritten as

$$P(\infty) = \sum_{l=0}^{\infty} \Pr\left(l^{d} = l\right) \left[A^{l} P^{\star} A^{l} + \sum_{j=0}^{l-1} A^{j} Q A^{j} \right], \qquad (5.22)$$

where P^* is the steady-state estimation error covariance of x(k) based on $\{y(0), y(1), \dots, y(k-1)\}$ and is the solution to the Discrete Algebraic Riccati Equation (DARE)

$$P^{\star} = AP^{\star}A^T + R_w - AP^{\star}C^T(CP^{\star}C^T + R_v)^{-1}CP^{\star}A^T.$$

Since the pair $\{A, R_w^{\frac{1}{2}}\}$ is stabilizable, the rate of convergence of P(0) to P^* is exponential [87], and the substitution of P^* for P(k-1) in (5.9) does not change the steady-state error covariance.

Let us define the generating function of the complementary density function G(X)and the moment generating function F(X) of the steady state latency l_d

$$G(X) = \sum_{l=0}^{\infty} \Pr\left(l^{d} \ge l+1\right) X^{l}, \text{ and } F(X) = \sum_{l=0}^{\infty} \Pr\left(l^{d} = l\right) X^{l},$$
 (5.23)

where X is an arbitrary matrix. Thus

$$F(X) = (X - I)G(X) + I.$$
(5.24)

⁵If we are interested in the error covariance at some other node v, simply denote v as the destination node.

On vectorizing (5.22), we obtain

$$\operatorname{vec}(P(\infty)) = F(A \otimes A)\operatorname{vec}(P^{\star}) + G(A \otimes A)\operatorname{vec}(Q)$$
$$= ((A \otimes A - I)G(A \otimes A) + I)\operatorname{vec}(P^{\star}) + G(A \otimes A)\operatorname{vec}(Q) (5.25)$$

where $A \otimes B$ is the Kronecker product of matrices A and B. Thus, the performance of the system depends on the value of G(X) evaluated at $X = A \otimes A$. In particular, the system is stable if and only if G(X) is bounded at $A \otimes A$. Since G(X) is a power series, boundedness of G(x) at $A \otimes A$ is equivalent to the boundedness of G(x)(evaluated for a scalar x) at the square of the spectral radius of A. We thus have the following result.

Theorem 5.4. Consider a process of the form (5.4) being observed using a sensor of the form (5.2) through a wireless erasure network. Let the packet erasures be independent from one time step to the next and across links. Then, the minimum expected steady-state estimation error covariance at the receiver is given by (5.25). Furthermore, the error covariance is stable in the sense of bounded expected steadystate error iff $|\rho(A)|^2$ lies in the region of convergence of G(x), where $\rho(A)$ is the spectral radius of A.

The above theorem allows us to calculate the steady state expected error covariance for any network as long as we can evaluate the function G(X) for that network. We now consider some special networks and evaluate the performance explicitly. We start with a network consisting of links in series, or a line network.

5.6.1 Networks with Links in Series

In this case, the network consists of only one path from the source to the destination. Thus, we have

$$F(X) = E\left[X^{l^d}\right] = E\left[X^{\sum_{(u,v)}L_{uv}}\right],$$
where the summation is taken over all the edges in the path. Since the drops across different links are uncorrelated, the variables L_{uv} s are independent. Thus, we have

$$F(X) = E\left[X^{\sum_{(u,v)} L_{uv}}\right] = \prod_{(u,v)} E\left[X^{L_{uv}}\right],$$

where we have used the independence of L_{uv} s. Since L_{uv} is a geometric random variable (5.19),

$$E\left[X^{L_{uv}}\right] = (1 - \epsilon_{uv})X\left(I - \epsilon_{uv}X\right)^{-1}$$

provided that $\rho(X)\epsilon_{uv} < 1$, where $\rho(X)$ is the spectral radius of matrix X. Therefore,

$$F(X) = E\left[X^{l_d}\right] = \prod_{(u,v)} \left[(1 - \epsilon_{uv}) X \left(I - \epsilon_{uv} X\right)^{-1} \right].$$

Using partial fractions and the relation in (5.24), we then obtain

$$G(X) = \sum_{i=0}^{n-1} X^{i} + X^{n} \sum_{(u,v)} c_{uv} \frac{\epsilon_{uv}}{1 - \epsilon_{uv}} (I - \epsilon_{uv} X)^{-1},$$

where

$$c_{uv} = \left(\prod_{(r,s)\neq(u,v)} \left(1 - \frac{\epsilon_{uv}}{\epsilon_{rs}}\right)\right)^{-1}.$$

Therefore, the cost can be written as

$$\operatorname{vec}\left(P(\infty)\right) = \prod_{(u,v)} \left[\left(A \otimes A\right) \left(\frac{I - \epsilon_{uv} A \otimes A}{1 - \epsilon_{uv}}\right)^{-1} \right] \operatorname{vec}\left(P^{\star}\right) + G\left(A \otimes A\right) \operatorname{vec}\left(Q\right).$$
(5.26)

We can also see from the above argument that the system is stable if for every link (u, v) we have $\epsilon_{uv} |\rho(A)|^2 < 1$ or, equivalently, $\max_{(u,v)} \epsilon_{uv} |\rho(A)|^2 < 1$. This matches with the condition in section 5.5 Also note that for the case that some of ϵ_{uv} s are equal, a different partial fraction expansion applies. In particular, for the case when

there are n links all with the erasure probability ϵ , we obtain

$$\operatorname{vec}\left(P(\infty)\right) = (A \otimes A)^{n} \left(\frac{I - \epsilon A \otimes A}{1 - \epsilon}\right)^{-n} \operatorname{vec}\left(P^{\star}\right) + \sum_{i=0}^{n-1} \left[\frac{\epsilon}{1 - \epsilon} (A \otimes A)^{n} \left(\frac{I - \epsilon A \otimes A}{1 - \epsilon}\right)^{-i-1}\right] \operatorname{vec}(Q) + \sum_{i=0}^{n-1} (A \otimes A)^{i} \operatorname{vec}(Q). \quad (5.27)$$

Finally, when there is only one link between the source and the destination, the cost reduces to a particularly simple form. It is easily seen that, in that case, the steady state error covariance will be the solution to the Lyapunov equation

$$P(\infty) = \sqrt{\epsilon}AP(\infty)\sqrt{\epsilon}A + (Q + (1 - \epsilon)AP^*A).$$

This expression can alternately be derived using Markov jump linear system theory as in [80].

5.6.2 Network of Parallel Links

Consider a network with one sensor connected to a destination node through n links with probabilities of erasure $\epsilon_1, \ldots, \epsilon_n$. Since the same data is being transmitted over all the links, using (5.21) the steady state latency can be written as

$$l_d = \min_{1 \le i \le n} (L_i).$$

Since L_i s are all independent geometrically distributed variables with parameters ϵ_i s respectively, their minimum is itself geometrically distributed with parameter $\epsilon_{eq} = \prod_i \epsilon_i$. Thus, F(X) can be evaluated as

$$F(X) = (1 - \epsilon_{eq})X(I - \epsilon_{eq}X)^{-1},$$

and G(X) can, in turn, be written as

$$G(X) = (I - \prod_{i} \epsilon_i X)^{-1}.$$

Thus, the steady-state error can be evaluated using (5.25). Note that the region of convergence of G(X) enforces for stability $\prod_i \epsilon_i |\rho(A)|^2 < 1$, which again matches with the condition in Section 5.5.

5.6.3 Arbitrary Network of Parallel and Serial Links

We can similarly find the steady-state error covariance of any network derived from the parallel and serial concatenations of sub-networks using the following two simple rules. Let $l_d(\mathcal{G})$ denote the steady-state latency function of network \mathcal{G} . Also, given two subnetworks \mathcal{G}_1 and \mathcal{G}_2 , denote their series combination by $\mathcal{G}_1 \oplus \mathcal{G}_2$ and their parallel combination by $\mathcal{G}_1 || \mathcal{G}_2$.

(a) Suppose the network \mathcal{G} can be decomposed as a series of two subnetworks \mathcal{G}_1 and \mathcal{G}_2 . Since packet erasures in the two subnetworks are independent of each other, we have

$$l^d(\mathcal{G}_1 \oplus \mathcal{G}_2) = l^d(\mathcal{G}_1) + l^d(\mathcal{G}_2).$$

Thus, we obtain

$$F_{\mathcal{G}_1 \oplus \mathcal{G}_2} (X) = E \left[X^{l^d(\mathcal{G}_1 \oplus \mathcal{G}_2)} \right]$$

$$= E \left[X^{l^d(\mathcal{G}_1) + l^d(\mathcal{G}_2)} \right]$$

$$= E \left[X^{l^d(\mathcal{G}_1)} \right] E \left[X^{l^d(\mathcal{G}_2)} \right]$$

$$= F_{\mathcal{G}_1} (X) F_{\mathcal{G}_2} (X).$$

Finally, using (5.24), the complementary density function of the network \mathcal{G} is

given by

$$G_{\mathcal{G}_{1}\oplus\mathcal{G}_{2}}(X) = (X-I)^{-1} (F_{\mathcal{G}_{1}\oplus\mathcal{G}_{2}}(X) - I)$$

= $G_{\mathcal{G}_{1}}(X) (X-I) G_{\mathcal{G}_{2}}(X) + G_{\mathcal{G}_{1}}(X) + G_{\mathcal{G}_{2}}(X)$
= $(X-I) G_{\mathcal{G}_{1}}(X) G_{\mathcal{G}_{2}}(X) + G_{\mathcal{G}_{1}}(X) + G_{\mathcal{G}_{2}}(X),$

where in the last line we have used the fact that

$$G(X)(X - I) = (X - I)G(X).$$

(b) If the network \mathcal{G} can be decomposed as parallel combination of two sub-networks \mathcal{G}_1 and \mathcal{G}_2 , we have

$$l^{d}(\mathcal{G}_{1} \| \mathcal{G}_{2}) = \min\{l^{d}(\mathcal{G}_{1}), l^{d}(\mathcal{G}_{2})\}.$$

Once again, the erasures in the two subnetworks are independent of each other. Thus

$$\Pr\left(l^d(\mathcal{G}_1 || \mathcal{G}_2) \ge l\right) = \Pr\left(l^d(\mathcal{G}_1) \ge l\right) \Pr\left(l^d(\mathcal{G}_2) \ge l\right).$$

Thus, we see that if

$$G_{\mathcal{G}_1}(X) = \sum_{i=0}^{\infty} a_i X^i$$
 and $G_{\mathcal{G}_2}(X) = \sum_{i=0}^{\infty} b_i X^i$,

then

$$G_{\mathcal{G}_1 \parallel \mathcal{G}_2}(X) = \sum_{i=0}^{\infty} a_i b_i X^i.$$

Thus, we can use (5.25) in conjunction with the above two rules to derive the steady state error of any network consisting of links in series and in parallel with each other. As an example, consider the network depicted in Figure 5.2. In this case



Figure 5.2: Example of a network of combination of parallel and serial links.

$$\mathcal{G} = \left(\left(\left(\mathcal{G}_0 \oplus \mathcal{G}_1 \right) \| \mathcal{G}_2 \right) \oplus \mathcal{G}_3 \right) \| \mathcal{G}_4 \right),$$

where each of the sub-networks \mathcal{G}_i is just a link with probability of packet drop ϵ . The generating function of a link with erasure probability ϵ is given by $G(X) = (I - \epsilon X)^{-1}$. Moreover, for a subnetwork with generating function G(X) in parallel with a link with erasure probability ϵ , the generating function of the entire network is given by $\mathcal{L}_{\epsilon}(G)(X)$, where \mathcal{L}_{ϵ} is an operator such that $\mathcal{L}_{\epsilon}(G)(X) = G(\epsilon X)$. Thus, the generating function of the network can be written as

$$G(X) = \mathcal{L}_{\epsilon}(\mathcal{L}_{\epsilon}(G_0 * G_1) * G_3)(X),$$

where

$$G_i(X) = (I - \epsilon X)^{-1}, \ i = 0, 1, 3$$

is the generating function for the *i*-th link and $G_i * G_j$ denotes the generating function of the series combination of link *i* and *j*. The steady state error covariance can thus be evaluated.

5.6.4 Networks with Arbitrary Topology

In this section, we consider the performance of general networks. Finding the distribution of the steady-state latency l_d of a general network is not an easy task because different paths may overlap. This can introduce dependency in the delays incurred along different paths and the calculation of the minimum delay and, hence, the steadystate latency becomes involved. However, using a method similar to the one used in Section 5.5, we can provide upper and lower bounds on the performance. We first mention the following intuitive lemma without proof.

Lemma 5.5. Let $P^{\infty}(\mathcal{G}, \{\epsilon_{uv}, (u, v) \in \mathcal{E}\})$ denote the expected steady-state error of a system with a communication network represented by graph $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ and probabilities of packet drop $\epsilon_{uv}, (u, v) \in \mathcal{E}$. Then, the expected steady-state error is non-increasing in $\epsilon_{uv}s$, i.e., if $\epsilon_{uv} \leq q_{uv} \quad \forall (u, v) \in \mathcal{E}$:

$$P^{\infty}(\mathcal{G}, \{\epsilon_{uv}, (u, v) \in \mathcal{E}\}) \preceq P^{\infty}(\mathcal{G}, \{q_{uv}, (u, v) \in \mathcal{E}\}),$$

where $A \preceq B$ means that B - A is positive semi-definite.

5.6.4.1 Lower Bound

Using the above lemma we can lower-bound the steady-state error by making a subset of links erasure free. This is similar to the method we used to obtain a necessary condition for stability in Section 5.5. Thus, once again consider any s-d cut \mathcal{V}_s of the network. Setting the probability of erasure equal to zero for every link except those crossing the cut (i.e., of the form $(u, v) \in [\mathcal{V}_s, \mathcal{V}_s^c]$) gives a lower bound on the error. Therefore,

$$P^{\infty}(\mathcal{G}, \{\epsilon_{uv}, (u, v) \in \mathcal{E}\}) \succeq P^{\infty}(\mathcal{G}, \{q_{uv}, (u, v) \in \mathcal{E}\}),$$

where

$$q_{uv} = \begin{cases} \epsilon_{uv} & (u, v) \in [\mathcal{V}_s, \mathcal{V}_s^c] \\ 0 & \text{otherwise.} \end{cases}$$

Now $P^{\infty}(\mathcal{G}, \{q_{uv}, (u, v) \in \mathcal{E}\})$ can be evaluated using the results given above for a network of parallel links. By considering the maximum along all possible cut-sets, we obtain the closest lower bound.

5.6.4.2 Upper Bound

We use a method similar to the one used to obtain the sufficient condition for stability in Section 5.5. In the proof of Theorem 5.3, it is shown that the performance of the network \mathcal{G} is lower-bounded by the performance of another network \mathcal{G}' that has series and parallel links only and has the following properties:

- \mathcal{G} and \mathcal{G}' have the same node set.
- \mathcal{G}' is the combination of edge-disjoint paths from the source to destination.
- The value of the max-cut in \mathcal{G}' is the same as in the original network \mathcal{G} .

The performance of \mathcal{G}' can be computed based on the results given above for arbitrary networks composed of subnetworks in series and parallel. This provides an upper bound on the performance of the original network.

5.7 Examples

We now illustrate the above results using some simple examples. Consider a scalar process evolving as

$$x(k+1) = 0.8x(k) + w(k)$$

that is being observed through a sensor of the form

$$y(k) = x(k) + v(k).$$

The noises w(k) and v(k) are assumed zero-mean, white, independent, and Gaussian with unit variances.

Suppose that the source and the destination nodes are connected using two links in series, each with a probability of packet erasure p. Figure 5.3 shows the performance of our strategy for the estimation problem \mathcal{P}_2 as the probability p is varied. The simulation results refer to data generated by a random run averaged over 100000 time steps, while the theoretical values refer to the value predicted by (5.25). We can see that the two sets of values match quite closely.



Figure 5.3: Simulated and theoretical results for a line network.

We also carried out a similar exercise for the source and destination nodes connected by two links in parallel, each with packet erasure probability p. The results are plotted in Figure 5.4. We can once again see that the simulated values match quite closely with the theoretical values.

Finally, we consider the source and destination nodes connected by a bridge network shown in Figure 5.5. We assume all the links in the network to have probability of erasure p. This network cannot be reduced to a series of series and parallel subnetworks. We can, however, calculate the performance analytically in this particular case and compare it to the upper and lower bounds presented earlier. The networks



Figure 5.4: Simulated and theoretical results for a parallel network.



Network used for upper bound

Network used for lower bound

Figure 5.5: Bridge network and the networks used for calculating lower and upper bounds.

used for calculating the bounds are also shown in Figure 5.5. The bounds can be computed analytically from series and parallel network results. Figure 5.6 shows a comparison of the analytical and simulated values with the lower and upper bounds. We can see that the bounds are quite tight.



Figure 5.6: Simulated values and theoretical bounds for the bridge network.

We can also compare the performance of our algorithm with that obtained if no encoding were done and only measurements were transmitted. Consider the process

$$x(k+1) = \begin{bmatrix} 1.25 & 0\\ 1 & 1.1 \end{bmatrix} x(k) + w(k)$$

being observed by a sensor of the form

$$y(k) = x(k) + v(k),$$

where w(k) and v(k) are white independent Gaussian noises with means zero and covariance identity. We consider transmission of data across a series of n channels. Figure 5.7 shows the difference in simulated performance of the algorithm in which no encoding is done and for our algorithm for various values of n. It can be seen that the error covariance is much higher if no encoding is being done, even for a few number of links and moderate values of drop probability. For each point we did 50000 simulations, with each simulation being 1000 time steps long. As a final example, we



Figure 5.7: Simulated difference in performance between algorithm in which no encoding is done and our optimal algorithm for series connection of n links.

consider the process

$$x(k+1) = 1.2x(k) + u(k) + w(k)$$

being observed through a sensor of the form

$$y(k) = x(k) + v(k),$$

which communicates with the controller over a series network of n links each with packet drop probability p. The controller is interested in minimizing the quadratic cost

$$J = \lim_{K \to \infty} \frac{1}{K} E\left[\sum_{k=0}^{K} x^{T}(k)Qx(k) + u^{T}(k)Ru(k)\right].$$

The cost matrices Q and R as well as the noise variances R_w and R_v are assumed to be unity. Figure 5.8 shows the variation of the cost with the probability for different number of links n. The loss in performance is very rapid with the increase in number of links.



Figure 5.8: Loss in performance as a function of packet drop probability for n links in series.

5.8 Generalizations

In this section we consider problems that are related to the problem set-up we considered above.

5.8.1 Correlated Erasure Events

Even though the algorithm \mathcal{A}_2 is optimal for any packet dropping pattern, the stability and performance analysis so far assumed that the erasure events are memoryless and independent across different links in the network. We could thus formulate the performance in terms of a generating function of the steady-state latency distribution as defined in (5.21). We now look at the effect of dropping these assumptions.

5.8.1.1 Markov Events

If we assume that the erasure events on each link are governed by a Markov chain (but are still independent of other links), we can obtain the performance as follows. Suppose that the packet drop event on link (u, v), denoted by $\lambda_{uv}(k) = "dropped"$ evolves according to a Markov chain with transition matrix M_{uv} . We further assume that M_{uv} is irreducible and reversible. Let us first consider the case where the initial distribution of packet erasure on each link is the stationary distribution of the Markov chain on that link. Then we can rewrite (5.5) in a similar fashion as (5.20), with L_l being a geometric random variable with distribution

$$\Pr(L_{uv} = l) = \begin{cases} \alpha_{uv} M_{uv}(1,2) M_{uv}(1,1)^{l-2} & \forall l \ge 2\\ 1 - \alpha_{uv} & l = 1 \end{cases}$$

where α_{uv} is the probability of packet drop based on the stationary distribution of link e = (u, v), and $M_{uv}(i, j)$ as the (i, j)-th element of M_{uv} . Thus, all the previous analysis goes through. In particular, the stability condition is

$$\max_{\mathcal{V}_s: s - d \text{cut}} \prod_{(u,v) \in \mathbf{c}} M_{uv}(1,1) |\rho(A)|^2 < 1.$$

Now, if the initial distribution is not the stationary distribution, the variables $L_{uv}(k)$ will not be time-independent, and the analysis does not goes through. However, since for large k the Markov chains will approach their stationary distribution, the stability condition remains unchanged.

5.8.1.2 Spatially Correlated Events

Suppose that the erasure events are correlated across the network but memoryless over time. In other words, at each time step k, the packet erasure occurs accord-

ing to distribution $\Pr(\lambda_{uv}, (u, v) \in \mathcal{E})$. Now, $L_{uv}(k)$ s are not independent across the network, and hence finding the steady-state error covariance does not seem to be tractable. However, we can find the condition for stability. For this we define a generalized notion of equivalent probability of packet drop for correlated events. Consider a cut-set with source set \mathcal{V}_s . Then, the equivalent probability of packet drop for this cut is

$$\epsilon_{eq}(\mathcal{V}_s) = \Pr(\lambda_{uv} = "dropped", \quad \forall (u, v) \in [\mathcal{V}_s, \mathcal{V}_s^c]).$$

The value of the max-cut for the network is the maximum of $\epsilon_{eq}(\mathcal{V}_s)$ over all the cuts, $\epsilon_{\text{net}}(\mathcal{G}) = \max_{\mathcal{V}_s:s-dcut} \epsilon_{eq}(\mathcal{V}_s)$. We can then show that the condition for stability of the system is $\epsilon_{\text{net}}(\mathcal{G})|\rho(A)|^2 < 1$. To see this, consider the scenario when only one packet is to be routed from the source to destination starting at time t_0 . For each time-step $t \geq t_0$, let $\mathcal{V}_r(t)$ denote the set of nodes that have received the packet at time t. Clearly $\mathcal{V}_r(t_0) = \{s\}$. We want to bound the probability that at time $t_0 + T$, the destination node has not yet received the packet. Note that for every time-step between t_0 and $t_0 + T$, $\mathcal{V}_r(t)$ defines a cut-set since it contains s and not d. Now, the size of $\mathcal{V}_r(t+1)$ does not increase with respect to time-step t iff all the links that cross the cut generated by $\mathcal{V}_r(t)$ drop packets. However, by the definition of $\epsilon_{\text{net}}(\mathcal{G})$, the probability of this event is at most $\epsilon_{\text{net}}(\mathcal{G})$. Therefore, we have

$$\left|\mathcal{V}_{r}(t+1)\right| \begin{cases} \geq \left|\mathcal{V}_{r}(t)\right| + 1 & \text{with prob. at most } \epsilon_{\text{net}}(\mathcal{G}) \\ = \left|\mathcal{V}_{r}(t)\right| & \text{with prob. at least } 1 - \epsilon_{\text{net}}(\mathcal{G}) \end{cases}$$

Thus, for large T the probability that at time $t_0 + T$ the destination node has not received the packet is upper-bounded by $n(1 - \epsilon_{\text{net}}(\mathcal{G}))^n T^n \epsilon_{\text{net}}(\mathcal{G})^{T-n}$, where n is the number of nodes in the network. Although a new packet is generated at the source at each time step, the importance of the packets is increasing with time; hence the error can be upper-bounded by considering that the network is only routing the packet generated at time k - l. The probability that the latency is larger than l grows like $f(l)\epsilon_{net}(\mathcal{G})^l$, where f(l) is polynomial in l with bounded degree and thus the sufficiency of the stability condition follows. The necessity part involves similar ideas and is omitted.

5.8.2 Synthesis of a Network

One can use the performance results above to design networks for the purpose of estimation and control. To consider a simple example, consider a scalar system observed by sensor s. Assume that the destination is located at distance d_0 from the sensor. The probability of dropping a packet on a link depends on its physical length. A reasonable model for probability of dropping packets is given by⁶ $\epsilon(d) = 1 - \exp(-\beta d^{\alpha})$, where β and α are positive constants. α denotes the exponent of power decay in the wireless environment. We are interested in the optimal number n of relay nodes that we should place between the sensor and the destination so as to minimize the expected steady-state error covariance (or in turn to minimize the cost J_{∞}). Clearly, there is a trade-off involved since more nodes will reduce the probability of erasure but at the same time lead to a higher delay before the destination receives a packet. Assuming that n sensors are uniformly placed, there are n + 1 links each with erasure probability q. Thus, from (5.27), $P(\infty)$ can be written as

$$P(\infty) = \left(\frac{a^2(1-q)}{1-qa^2}\right)^{n+1} \left(P^* + \frac{R_w}{a^2-1}\right) - \frac{R_w}{a^2-1}$$

Thus, the optimal number of relay nodes, (assuming that $a^2 > 1$) is the solution to the problem

$$\min_{n} \left(\frac{a^2 (1 - \epsilon(\frac{d_0}{n+1}))}{1 - \epsilon(\frac{d_0}{n+1})a^2} \right)^{n+1}$$

 $^{^{6}\}mathrm{This}$ expression can be derived by considering the probability of outage in a Rayleigh fading environment.

If $a^2 < 1$, then the minimization in the above problem is replaced with maximization.

5.8.3 Unicast Networks

So far, we have assumed that the topology of the network was fixed and that a node could transmit a message on all the out-going edges. We can consider networks that are unicast in the sense that each node should choose one out of a set of possible edges to transmit the message on. There are two parts of the problem:

- (a) Choose the optimal path for data to flow from the source node to the sink node.
- (b) Find the optimal information processing scheme to be followed by each node.

The two parts can clearly be solved separately in the sense that given any path, the optimal processing strategy is the algorithm described in Section 5.4.

To choose the optimal path, we need to define a metric for the cost of a path. We can consider two choices:

- (a) If the metric is the condition for stability, then the problem can be recast as choosing the shortest path in a graph with the length of a path being given by its equivalent probability of packet drop. Thus, we need to find the path that has the minimum ϵ_{path} . Since each path is just a line network, this reduces to the problem $\min_{P:s-d \text{ path}} \max_{(u,v) \in P} \epsilon_{uv}$. The above problem is well studied in the computer science community and can be solved as a shortest-path problem over a min-max semi-ring in a distributed fashion [88].
- (b) If the metric is performance, the problem is more complicated in general. We can consider the special case of a scalar system and no process noise. In this case, from (5.26), we have for path q

$$\log P_q(\infty) = \sum_{e \in q} \log(\frac{(1 - \epsilon_e)a^2}{1 - \epsilon_e a^2})$$

Now the problem is equivalent to

$$\min_{q: \text{s-d path}} \sum_{e \in q} \log(\frac{(1 - \epsilon_e)a^2}{1 - \epsilon_e a^2}).$$

This problem can also be solved in a distributed way [89].

5.9 Conclusions

In this chapter we considered the problem of estimation and control of a dynamical process across a wireless erasure network. We identified the optimal information processing strategy to be followed by each node in the network that allows the estimator to calculate the best possible estimate in the minimum mean square sense and the controller to minimize a quadratic cost. The recursive algorithm that we propose requires a constant amount of memory, processing, and transmission at every node in the network per time step, yet is optimal for any erasure process and at every time step. It has numerous other important properties as well, such as being able to take care of delays and packet reordering. For the case when the erasure probabilities are memoryless and independent across links, we also carried out the stability and performance analysis for this algorithm. The results can also be used for the problems of routing of data through a network and synthesis of networks for the purpose of estimation and control.

The work in this chapter can be extended in many ways. We have ignored issues of quantization so far. Finding stability condition and performance bounds for a limited bit rate erasure channels is a possible avenue. Another important extension is analysis of stability conditions for the case that more than one sensor is providing measurements in the network. This seems like a very challenging and interesting problem that merits further investigation. And finally, at a more general level, given the variety of tasks, e.g. communication, computation, estimation, and control that wireless networks are deployed for today, a mathematical framework capable of analyzing their ultimate performance is necessary. Such a framework enables us to find the important features of the network affecting the performance of each of these tasks and design networks that are efficient for different applications. The work presents in the present and previous chapters is a step toward this goal.

Chapter 6

Power Efficiency of Sensor and Ad-Hoc Networks

6.1 Introduction

In recent years, there has been great interest in the analyses of capabilities of wireless networks from different aspects. Most of the analysis have dealt with the analysis of the following two types of networks [91], [92]:

- (a) **Sensory networks:** A sensory network consists of n + 1 fixed nodes with a single receiver that collects data/information from the sensor nodes. At any given time, there can be at most one sensory transmitter. All other nodes in the network can be thought of as relay nodes. (See Figure 6.1.(a).)
- (b) Ad-hoc networks: At any time, an ad-hoc network consists of n fixed relay nodes and r fixed simultaneous transmitter/receiver pairs, where $r \leq n$. In this network relay nodes cooperate for transmission of information from one transmit node to the corresponding receiver node. (See Figure 6.1.(b).)

Unfortunately, finding the exact ultimate performance of a general wireless network for different applications, e.g., communications, estimation, and control, is very much open. For instance, finding the maximum reliable rate of communication for a network with only one relay is still open [53]. Therefore, most of the research



Figure 6.1: Sensory and Ad-hoc wireless networks.

for wireless sensory and ad-hoc networks involves asymptotically large networks and gives scaling laws for performance measures such as capacity or distortion rather than exact results for general networks. For instance, it is shown in [91] that for a sensory network, the capacity scales as $\Theta(\log n)$.¹ For ad-hoc networks, the problem is much more challenging. The groundbreaking work of [92] shows that the capacity grows at least as $\Theta(\sqrt{n})$. Using information theoretic tools, it is shown in [93, 94, 95] that under some mild assumptions on the channel model, $\Theta(\sqrt{n})$ is an upperbound on the sum-capacity in the extended wireless networks, i.e., networks where the density of the nodes per area does not increase with the number of nodes. In both sensory and ad-hoc wireless networks, these results are discouraging from a practical point of view because they suggest that for sensory and ad-hoc wireless networks, the peruser capacity scales as $\Theta(\frac{\log n}{n})$ and $\Theta(\frac{\sqrt{n}}{n})$, respectively. This represents rewards that rapidly diminish to zero as the number of nodes (users) in the network increases.

Therefore, one interesting problem is to see whether there exists any favorable scalings in ad-hoc and sensory wireless networks. In other words, are there any

¹The following notation will be used in this thesis. For two functions f, g defined on natural numbers, we have f(n) = O(g(n)) if $\lim_{n\to\infty} \inf f(n)/g(n) < \infty$; we have $f(n) = \Omega(g(n))$ if $\lim_{n\to\infty} \inf f(n)/g(n) > 0$; and we have f(n) = o(g(n)) if $\lim_{n\to\infty} \inf f(n)/g(n) = 0$. Finally, we have $f(n) = \Theta(g(n))$ if $f(n) = \Omega(g(n))$ and f(n) = O(g(n)).

scenarios in which it is actually beneficial to form a sensory or ad-hoc network and obtain increasing gains as the network size grows? Several researchers have looked at this problem from different points of view. In [96] the authors look at a wireless network in which users are mobile (not fixed), and they show that the total capacity of such a network scales like $\Theta(n)$. The work in [97],[98] and [99] also consider the feasibility of wireless networks from a distributed source coding point of view.

In previous chapters we were concerned with performance analysis of special classes of wireless networks (mainly wireless erasure networks) with *arbitrary* size. We found exact ultimate performance measures for applications like communications and estimation across the network. In this chapter we look at a more general class of wireless networks, but instead of finding the exact performance measure for arbitrary size networks, we analyze the performance for random and asymptotically large size networks. The performance measure considered in this chapter, unlike the references mentioned above, will be the power consumption in the network.

Power consumption is one of the main concerns in wireless networks, especially in sensory networks [100]. Since the source of energy for each user is limited (usually a battery)[65], users in these networks need to use power efficiently. Two major sources of power consumption at each node are the computation power and the transmit power. We only consider the power consumption due to transmission and not due to computation. However, we should mention that it is not clear whether at low SNR (where many wireless networks usually operate at) the computation power is negligible compared to transmit power. Incorporating computation power as well is very interesting and can be a subject of research in itself.

We will show that it is beneficial to form large networks of users in terms of power consumption. We consider sensory and ad-hoc wireless networks where the users are placed randomly in a domain of fixed area A. We show that users in these networks can support the same rate as a single user system, but by expending less power. Furthermore, the power that each user needs to expend decreases as we increase the size of the network where the rate of communication is kept fixed. To look at the power efficiency of these networks we will follow the same approach and concept as in [101]. The power efficiency η of a communications channel is defined as the ratio between the capacity (data rate) of the channel and the transmit power (energy rate). For AWGN channels, this is given by

$$\eta = \frac{\log(1 + \sigma_s^2/\sigma_n^2)}{\sigma_s^2},\tag{6.1}$$

where σ_s^2 represents the transmit power and σ_n^2 represents the noise power. Clearly, for fixed σ_n^2 as $\sigma_s^2 \to \infty$, the value of η approaches zero, meaning that we are highly power inefficient at high SNR. On the other hand, we are power efficient at low SNR and, in fact,

$$\max_{\sigma_s^2} \eta = \lim_{\sigma_s^2 \to 0} \eta = \frac{\log e}{\sigma_n^2}.$$
(6.2)

This implies that, at low SNR, capacity is proportional to the transmit power. In [101], the power efficiency (or capacity per unit cost as the author defines it) of several other communication systems is computed.

In this chapter we will find a lower bound for the power efficiency of sensory and ad-hoc wireless networks formed in a domain of fixed area. For this, we will propose a protocol for communication among the nodes. The key idea used in the protocol is to exploit features of wireless networks and operate the network at low SNR (thereby avoiding the logarithmic scaling of the capacity). The main features that distinguish wireless networks from wireline networks are path-loss, fading, and interference. Path-loss has been exploited in cellular networks. Fading also is exploited in multi-user systems by scheduling transmissions when a user has favorable channel conditions [102, 103]. However, most current approaches avoid interference avoidance and the construction of a multi-hop network. In our protocol we will exploit the interference and fading inherent in any wireless network for achieving good power efficiency. Also, the protocol proposed in this chapter is a double-hop protocol. Although it is thought that the power efficiency of multi-hop networks is better than that of doublehop, networks it can be shown that if the nodes are placed in a domain of fixed area, this is not true. A similar observation is made in [104]. The authors have observed that the most energy-efficient protocol to use depends on the network topology and the radio parameters of the system.

We have shown in [105] that for sensory and ad-hoc wireless networks for which the channel coefficients between users can be modeled by independent zero-mean, unit variance, and bounded fourth order moment random variables, the power efficiency scales at least as $\Theta(\sqrt{n})$. However, the model used for channel coefficients in this chapter is more general. We will see that even with this general model we are still able to achieve a power efficiency that scales favorably as the size of the network grows. The net result is that under some mild assumptions on the channel coefficients that will be mentioned in Section 6.2.2, with high probability the power efficiency of a random network, i.e., the data rate per energy rate, scales as $\Theta(\sqrt{n})$ for each user.

This chapter is organized as follows. Section 6.2 describes the system model and assumptions and presents the statement of the problem considered in this chapter. In Section 6.3 we will compute the power efficiency of a multi-antenna communication system for comparison. In Section 6.4 we consider the power efficiency of sensory wireless networks. We describe the proposed "Listen and Transmit" protocol for achieving scalable power efficiency for sensory networks. In Section 6.5 power efficiency of adhoc wireless networks is considered and analyzed. We first present a generalization of the "Listen and Transmit" protocol for adhoc networks and then optimally allocate powers to achieve a scalable power efficiency for the network. At the end of this section we will compare the performance of our protocol with an interference suppression

scheme that requires complete knowledge of the channel. Conclusions and proposals of further work are provided in Section 6.6.

6.2 Notation and System Model

6.2.1 Notation and Definitions

Throughout this chapter matrices and vectors are denoted by boldface characters. tr (**A**), $\lambda_{max}(\mathbf{A})$, and $\lambda_{min}(\mathbf{A})$ denote the trace, the maximum eigenvalue, and the minimum eigenvalue of a square hermitian matrix **A**. The superscript * denotes conjugate transposition for matrices and complex conjugate for scalars. Complex conjugation for matrices is shown by using bar. Transposition is also denoted by superscript T. \mathbf{A}^* , \mathbf{A}^T , and $\mathbf{\bar{A}}$ are the conjugate transpose, transpose, and conjugate of the matrix **A**, and α^* is the complex conjugate of the scalar α . \mathbf{I}_r is the $r \times r$ identity matrix. For a matrix **A**, vec (**A**) denotes the vector obtained from stacking all the columns of **A**, one on top of another. For a vector $\mathbf{g} = (g_1, \ldots, g_n)$, diag (**g**) denotes the $n \times n$ diagonal matrix with *i*-th diagonal element equal to g_i . We may also write diag (**g**) as diag (g_1, \ldots, g_n) . Finally, $\|\mathbf{g}\|$ denotes the Euclidean norm of vector **g**.

We consider random wireless networks over a fixed area A. We randomly select points in A to form the nodes of the network (either as transmitters, receivers, or relay nodes). Since the network is wireless, the connections between any two nodes will be subject to fading. Thus, the randomness in the network will be due to two sources: the random choice of points in A, and the random fading between the connections. When we fix the position of the nodes, we denote the expectation over channel fading by E_f [.]. The expectation over the location of some set of the nodes, say P, in a random network is denoted by E_P [.], which from now on we shall call the spatial average . For instance, the spatial average of the mean value of the channel coefficient between node *i* and node *j*, h_{ij} , over the position of node *i* while node *j* is fixed, can be written as $E_{\{i|j\}}[E_f[h_{ij}]]$. The expectation over the location of all the nodes is denoted by E_{loc} , and whenever E[.] is used without any subscript, expectation over both fading and the location of the nodes is implied. Channel coefficients are denoted by h, g, or *c* depending on the context. Usually *c* is used as a generic channel between two arbitrary points in the domain.

6.2.2 System Model and Problem Statement

Sensory Networks

As mentioned earlier, by a sensory wireless network we mean one with n relay nodes and a single transmitter/receiver pair (see Figure 6.1). We assume that the nodes are placed randomly and independently according to some distribution function (not necessarily uniform) in a domain of fixed area, say A. We denote the channel coefficient from the transmitter to the relay node i by g_i , and the channel coefficient from the relay node i to the receiver by h_i . We assume that, averaged over the fading, different channels are independent. Furthermore, we assume that each node i knows only its local connections h_i , g_i , but not the other connections in the network.

Ad-hoc Networks

As mentioned earlier, for ad-hoc networks we assume that at any time there are n relay nodes and at most r simultaneous transmit/receive pairs in the network. The nodes are placed randomly and independently according to some distribution function (not necessarily uniform) in a domain of fixed area, say A. The channel coefficient from transmitter i (i = 1, ..., r) to relay node j (j = 1, ..., n) is denoted by g_{ij} , and from relay node j to receive node k (k = 1, ..., r) is denoted by h_{jk} . Similar to the sensory case, we assume that, averaged over the fading, distinct channels are

independent. Furthermore, if we fix the location of the transmitters and the receivers and randomly choose relay nodes j and j', the channel coefficients $\{g_{ij}, h_{jk}\}$ and $\{g_{ij'}, h_{j'k}\}$ are independent for all i and k. As with the sensory case, we assume that all the relay nodes know their local connections, but not the remaining connections in the network. In other words, node j knows all the connections $\{g_{ij}, i = 1, ..., r\}$ and $\{h_{jk}, k = 1, ..., r\}$.

Additional Assumptions for Ad-hoc Networks

For ad-hoc networks, we have a few more assumptions. Thus, denote the channel coefficient between two points x_i and x_j by c_{ij} . With this notation, we have the following additional assumptions:

- A. $E_{\{x_5|x_2\}}[E_f[c_{52}]] = 0, \quad \forall x_2 \in A$
- B. $E_{\{x_5|x_1,x_2\}}[E_f[c_{51}c_{52}^*]] = 0 \quad \forall x_1 \neq x_2 \in A$
- C. $E_{\{x_5|x_1,x_2,x_3,x_4\}}[E_f[c_{51}^*c_{52}^*c_{53}c_{54}]] = 0 \quad \forall x_1, x_2, x_3, x_4 \in A \text{ at least 3 of } x_i \text{ are distinct.}$

Note that the above conditions are clearly met if the fading is zero mean. In general, however, there may be line-of-sight between different nodes in the network, and the fading may be nonzero mean. The above conditions are more general and do not require zero mean fading. The first assumption says that the spatial average of the mean of a channel coefficient between a random point and a fixed point is zero. The second assumption is that the channel coefficients between one random point, x_5 , and two different points, x_1, x_2 , are uncorrelated when averaged over both the fading and the point placement of x_5 . In other words, although the channels c_{51} and c_{52} , given that x_1 and x_2 are fixed, are not independent and may be correlated, the spatial average of the correlation between these two channels is zero. The last condition also says that the expectation of the product of the channel coefficients between one random point and four fixed points averaged over the location of the random point



Figure 6.2: Condition on the channel coefficients.

is zero (See Figure 6.2). These assumptions appear to be reasonable, especially if we assume that the environment is rich in scattering. We obtain two achievable bounds for the power efficiency of ad-hoc networks. The first bound (Theorem 6.9) relies only on the first and second assumptions, and the second bound (Theorem 6.12) requires the last assumption as well.

Power Assumptions

In the sensory network we assume that the transmit power is p. For ad-hoc networks we assume that *all* the transmitters transmit with the same power p. In both cases, we will assume that the relay nodes transmit with identical power σ_r^2 . The noise introduced in every reception is an additive white circularly-symmetric Gaussian noise with zero mean and variance σ_n^2 , which is denoted by $\mathcal{CN}(0, \sigma_n^2)$.

Path Loss

In this work we will not be concerned with explicit path loss models. The main reason is that since we consider a fixed domain A, the only characteristics of the path-loss that enter our analysis are the second and fourth order moments of the channel. In fact, a strength of our results is that the asymptotics are not sensitive to the path-loss



Figure 6.3: Single-hop versus Multi-hop.

model (the model and the geometry of the domain A affect the constants but not the scaling behavior). We further discuss path-loss models when comparing single-hop and multi-hop systems below.

Single-hop vs. Multi-hop Communication

The proposed protocol in this work achieves a power efficiency that scales with the number of nodes, n. The communication model that we are using is a double hop (transmit and relay) communication protocol. In this protocol, which will be explained in detail later on, the communication is done in two intervals. In the first interval, the transmit node(s) sends their data signal. In the second interval, relay nodes send a signal based on what they have received in the first interval.

Typically, in order to increase power efficiency in wireless networks, one must move toward a multi-hop system so as to avoid long hops (which are subject to severe path loss) [106], [107]. While this is certainly true for networks that grow in physical size as the number of nodes increases (thereby increasing the size of the hops), it is not true for networks in which the physical domain is *fixed* while the number of nodes increases.² In this case, there is nothing to be gained by using multi-hop schemes in which the number of hops scales with the number of nodes in the network and the

²If the area of the network increases with the number of nodes, a combination of multi-hop routing and the "Listen and Transmit" protocol described here is necessary to achieve a power efficiency that scales with the number of nodes.

length of the hops becomes shorter and shorter as the number of nodes increases. To make this more explicit, we use the following qualitative argument. Suppose that nnodes are located in a domain of fixed area A. Consider two nodes (users) of distance d which want to communicate with each other (Figure 6.3.(a)). Assume that the channel is AWGN and that the power loss between any two points is a decreasing function of their distance d and is denoted by f(d). In this case, the relation between the transmitted signal s from A and the received signal y at B is

$$y = \sqrt{f(d)}s + v.$$

The capacity is clearly $\log(1 + f(d)\frac{p}{\sigma_n^2})$, and the power efficiency achieved at low SNR is

$$\eta = \frac{f(d)}{\sigma_n^2}$$

Assume now that we employ a multi-hop scheme to communicate between A and B where each node relays to its nearest neighbor (Figure 6.3.(b)). Since we have nnodes, the distance to a nearest neighbor will be $d' = O(\frac{1}{\sqrt{n}})d$, and the number of hops will be of $\Omega(\sqrt{n})$. Here each relay will communicate at rate $\log(1 + f(d')\frac{\sigma_r^2}{\sigma_n^2})$, and since the total transmit power is $\Omega(\sqrt{n})\sigma_r^2$, the power efficiency achieved at low SNR will be

$$\eta^{'} = \frac{f(d^{\prime})}{\sigma_n^2 \Omega(\sqrt{n})} = \frac{f(dO(\frac{1}{\sqrt{n}}))}{\sigma_n^2 \Omega(\sqrt{n})}.$$

For any reasonable path-loss model $\lim_{d\to 0} f(d) = constant.^3$ Therefore, the power efficiency of the multi-hop system scales like $O(\frac{1}{\sqrt{n}})$ as *n* increases. This says that for a fixed-size network, increasing the number of hops in fact reduces the power efficiency.

Remark 6.1. Note that if the size of the domain also increases with n, then d will

³Note that the common power law function used in literature, $f(d) = \frac{1}{d^n}$ $n \ge 2$, does not satisfy this property since this model is only valid for far field approximation.

also increase. In this case, since

$$\frac{\eta'}{\eta} = \frac{f(dO(\frac{1}{\sqrt{n}}))}{f(d)\Omega(\sqrt{n})}$$

depending on the path-loss model f(.) and how d scales with n, it may be more power efficient to use multi-hop.

Channel Knowledge and Synchronicity

As mentioned earlier, we have assumed that the nodes have knowledge of their local connections. This is a much more reasonable assumption than the nodes knowing the entire network. However, it does require that the network remain relatively stationary in time so that the local connections can be learned via the transmission of pilot symbols, etc. Furthermore, we assume a synchronous system. In other words, all the transmissions and receptions are synchronized. Later on we will argue that the system performance is not very sensitive to timing errors and lack of perfect synchronicity.

6.3 An Example: Multi-antenna Systems

In order to obtain some insight into how the power efficiency of a sensory or ad-hoc wireless network might scale, it is useful to look at the example of a multi-antenna system. For more details see [108].

Consider an n transmit single receive multi-antenna channel, described by the channel vector

$$H = \left[\begin{array}{ccc} h_1 & h_2 & \dots & h_n \end{array} \right],$$

where h_i denotes the channel coefficient from the *i*-th transmitter to the receiver. (Assume that the channel coefficients are zero-mean and unit variance and have fourth order moment κ .) Two cases can be envisioned.

• The channel matrix is *known* to the transmitter: In this case the optimal scheme is for beam-forming. Thus, if each antenna transmits with power *p*, the power efficiency becomes

$$\eta = \frac{\mathrm{E}\,\log\left(1 + \frac{p}{\sigma_n^2} (\sum_{i=1}^n |h_i|^2)^2\right)}{np}.$$

This is maximized when $p \to 0$, which yields

$$\eta = \frac{\log e}{n\sigma_n^2} \operatorname{E}\left(\sum_{i=1}^n |h_i|^2\right)^2$$
$$= \frac{\log e}{n\sigma_n^2} \left(n\kappa + n(n-1)\right) = \Theta(n).$$
(6.3)

• The channel matrix is *unknown* to the transmitter. In this case, beam-forming cannot be done. However, the capacity is known from [109], and so the power efficiency becomes

$$\eta = \frac{\mathrm{E}\,\log\left(1 + \frac{p}{\sigma_n^2}\sum_{i=1}^n |h_i|^2\right)}{np}.$$

Looking at the low SNR we have

$$\eta = \frac{\log e}{n\sigma_n^2} \mathbb{E}\left(\sum_{i=1}^n |h_i|^2\right) = \frac{\log e}{n\sigma_n^2} n = \Theta(1).$$
(6.4)

What distinguishes an $n \times 1$ multi-antenna system from an n node sensory network is that the n antenna elements are allowed to cooperate, but the n nodes in a sensory network are not. What the above result says is that when the nodes are allowed to cooperate and the nodes know the channel coefficients, the power efficiency scales as $\Theta(n)$. However, even if the nodes are allowed to cooperate, as long as they do not know the channel coefficients, the power efficiency over $\Theta(1)$. But what about a sensory network where the nodes are not allowed to cooperate but know the local channel coefficients? Moreover, what about ad-hoc networks? These are the questions we shall address.

6.4 Sensory networks

We begin by describing a simple protocol that achieves a power efficiency of $\Theta(\sqrt{n})$ for random sensory and, as we shall see in the next section, with some modification for ad-hoc wireless networks. As mentioned earlier, the protocol assumes synchronous transmission and receptions, as well as local channel knowledge at the nodes.

6.4.1 "Listen and Transmit" Protocol

Consider a random sensory network with n relay nodes and one transmitter/receiver pair. We are interested in a probabilistic bound for the achievable power efficiency in this network, i.e., a bound that with high probability is achievable for a random network in the domain. We begin by explaining the protocol that achieves power efficiency of $\Theta(\sqrt{n})$ for sensory wireless networks. In this, so called "Listen and Transmit," protocol communication is done in two intervals:

(a) Listen interval: In this interval the transmitter sends the data and the relay nodes only listen. Relay node *i* receives:

$$r_i = g_i s + v_i \qquad i = 1, 2, \dots, n,$$
 (6.5)

where v_i is $\mathcal{CN}(0, \sigma_n^2)$.

(b) **Transmit interval:** Each node, using its knowledge of the local connections, transmits a scaled version of the signal it has received in the first interval:

$$t_i = \frac{\sigma_r g_i^* h_i^*}{|g_i| |h_i| \sqrt{|g_i|^2 p + \sigma_n^2}} r_i \qquad i = 1, 2, \dots, n.$$
(6.6)

The scalar is chosen so that the relay node power is σ_r^2 and so that the signal parts coherently add at the receiver.

This protocol is similar to the protocol proposed in [110]. In [110], the relay nodes transmit the exact signals they have received, scaled to meet the power constraint. In the "Listen and Transmit" protocol, the channel coefficients can be complex. There-fore, the relay nodes change the phase of their received signal appropriately so that the signal parts of the received signal (at the receiver) add up coherently. The received signal at the receiver is:

$$y = \sum_{i=1}^{n} h_i t_i + w$$

= $\sum_{i=1}^{n} \frac{\sigma_r |g_i| |h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}} s + \sum_{i=1}^{n} \frac{\sigma_r |h_i| g_i^*}{|g_i| \sqrt{|g_i|^2 p + \sigma_n^2}} v_i + w,$ (6.7)

where w is $\mathcal{CN}(0, \sigma_n^2)$. From (6.7) it is clear that the signal part from different relay nodes adds up coherently, but the noise part does not. In this sense, the "Listen and Transmit" protocol can be regarded as performing *distributed beamforming*.

6.4.2 Finding a Lower Bound

We break α , defined in (6.7), into $\bar{\alpha}$ and $\tilde{\alpha} = \alpha - \bar{\alpha}$, where $\bar{\alpha} = E_f[\alpha]$ and $E_f[\tilde{\alpha}] = 0$. Now, if we rewrite (6.7) as $y = \bar{\alpha}s + \tilde{\alpha}s + w'$, then as it is shown in [111], the capacity of this system can be lower-bounded by the capacity of the AWGN channel $y = \bar{\alpha}s + w''$, where w'' is a Gaussian noise with variance equal to the variance of $\tilde{\alpha}s + w'$ (in this analysis we assume that the receiver is provided with the mean of α). Therefore, the capacity of the system in (6.7) may be lower-bounded by

$$C \ge \frac{1}{2} \log \left(1 + \frac{|\bar{\alpha}|^2 p}{\operatorname{Var}_{\mathrm{f}}[w'] + \operatorname{Var}_{\mathrm{f}}[\tilde{\alpha}]p} \right).$$
(6.8)

Note that the $\frac{1}{2}$ in front of the logarithmic term comes from the fact that the transmitter transmits half of the time. By substituting $\bar{\alpha}$, $\operatorname{Var}_{\mathrm{f}}[w']$, and $\operatorname{Var}_{\mathrm{f}}[\tilde{\alpha}]$ in (6.8) with

$$\bar{\alpha} = \mathcal{E}_{f} \left[\alpha \right] = \sigma_{r} \sum_{i=1}^{n} \mathcal{E}_{f} \frac{|g_{i}||h_{i}|}{\sqrt{|g_{i}|^{2}p + \sigma_{n}^{2}}},$$

$$\operatorname{Var}_{f} \left[\tilde{\alpha} \right] = \operatorname{Var}_{f} \left[\alpha \right] = \sigma_{r}^{2} \sum_{i=1}^{n} \mathcal{E}_{f} \frac{|g_{i}|^{2}|h_{i}|^{2}}{|g_{i}|^{2}p + \sigma_{n}^{2}} - \sigma_{r}^{2} \sum_{i=1}^{n} \left(\mathcal{E}_{f} \frac{|g_{i}||h_{i}|}{\sqrt{|g_{i}|^{2}p + \sigma_{n}^{2}}} \right)^{2}, \quad \text{and}$$

$$\operatorname{Var}_{f} \left[w' \right] = \sigma_{n}^{2} \left(1 + \sigma_{r}^{2} \sum_{i=1}^{n} \mathcal{E}_{f} \frac{|h_{i}|^{2}}{|g_{i}|^{2}p + \sigma_{n}^{2}} \right),$$

and rearranging the terms, we get

$$C \ge \frac{1}{2} \log \left(1 + \frac{\sigma_r^2 p \left(\sum_{i=1}^n \mathrm{E}_{\mathrm{f}} \frac{|g_i||h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}} \right)^2}{\sigma_n^2 + \sigma_r^2 \left(\sum_{i=1}^n \mathrm{E}_{\mathrm{f}} |h_i|^2 - \sum_{i=1}^n p \left(\mathrm{E}_{\mathrm{f}} \frac{|g_i||h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}} \right)^2 \right)} \right).$$
(6.9)

Define $c_i = E_f \frac{|g_i||h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}}$ and $b_i = E_f |h_i|^2$. By ignoring the negative term in the denominator we can rewrite (6.9) as

$$C \ge \frac{1}{2} \log \left(1 + \frac{n^2 \sigma_r^2 p \left(\frac{\sum_{i=1}^n c_i}{n}\right)^2}{\sigma_n^2 + n \sigma_r^2 \left(\frac{\sum_{i=1}^n b_i}{n}\right)} \right).$$
(6.10)

The above lower bound holds for every fixed network. For a random network, the capacity, the power efficiency, and the above lower bound are random variables depending on the placement of the nodes in the network. Since the nodes are placed independently and according to the same distribution on the available area, the b_i s and c_i s are i.i.d random variables for different relay nodes (i.e., different *is*). Therefore,

denoting the transmitter and receiver location by $\{t, r\}$, we have

$$E_{loc}(\sum_{i=1}^{n} b_{i}) = E_{\{t,r\}} \sum_{i=1}^{n} E_{\{i|t,r\}}[b_{i}] = nE_{loc}[b]$$

$$Var_{loc}(\sum_{i=1}^{n} b_{i}) = E_{\{t,r\}} \sum_{i=1}^{n} E_{\{i|t,r\}}([b_{i}^{2}] - [b_{i}]^{2}) = nVar_{loc}[b]$$

$$E_{loc}(\sum_{i=1}^{n} c_{i}) = E_{\{t,r\}} \sum_{i=1}^{n} E_{\{i|t,r\}}[c_{i}] = nE_{loc}[c]$$

$$Var_{loc}(\sum_{i=1}^{n} c_{i}) = E_{\{t,r\}} \sum_{i=1}^{n} E_{\{i|t,r\}}([c_{i}^{2}] - [c_{i}]^{2}) = nVar_{loc}[c],$$

where $c = E_f \frac{|g||h|}{\sqrt{|g|^2 p + \sigma_n^2}}$ and $b = E_f |h|^2$ are the random variables depending on the channel coefficients between one random point and two other random points. Now, for any $\epsilon > 0$, using Chebyshev's inequality and the union bound on the probability of the events, we have

$$\Pr\left\{\left|\frac{\sum_{i=1}^{n} c_{i}}{n} - \mathcal{E}_{loc}c\right| \le \epsilon, \left|\frac{\sum_{i=1}^{n} b_{i}}{n} - \mathcal{E}_{loc}b\right| \le \epsilon\right\} \ge 1 - \frac{\operatorname{Var}_{loc}[b] + \mathcal{E}_{loc}[c^{2}]}{n\epsilon^{2}}.$$
 (6.11)

The inequality of (6.11) shows that as $n \to \infty$, the quantities $\frac{\sum_{i=1}^{n} c_i}{n}$ and $\frac{\sum_{i=1}^{n} b_i}{n}$ behave like their spatial averages. This implies that with high probability, C in (6.10) is bounded by

$$\frac{1}{2}\log\left(1+\frac{n^2\sigma_r^2 p \mathcal{E}_{loc}^2[c]}{\sigma_n^2+n\sigma_r^2 \mathcal{E}_{loc}[b]}\right).$$

Remark 6.2. Note that $E_{loc}[b]$ and $E_{loc}[c]$ depend only on the domain A on the fading characteristics and on the distribution of the points, and they do not depend on n. Thus, for fixed p and σ_r^2 , as $n \to \infty$ the lower bound on capacity with high probability behaves like

$$\frac{1}{2}\log n + O(1).$$

Note that this is the same asymptotic growth obtained for Gaussian relay channels in [91]. Thus, we conclude that the "Listen and Transmit" protocol (i.e., distributed beamforming) achieves the optimal asymptotic capacity growth. We, of course, are not primarily interested in capacity, but rather in power efficiency.

Now we will focus on how to optimally allocate the powers $(p \text{ and } \sigma_r^2 \text{ as a function}$ of n) to maximize the power efficiency. As mentioned earlier, $E_{loc}[b]$ and $Var_{loc}[b]$ do not depend n. Using the Taylor series expansion of $E_{loc}[c]$ and $E_{loc}[c^2]$ in p, we have

$$E_{loc}[c] = \frac{E_{loc}E|h||g|}{\sigma_n} - \frac{E_{loc}E|h||g|^3}{\sigma_n^3}p + o(p)$$

$$E_{loc}[c^2] = \frac{E_{loc}E|h|^2|g|^2}{\sigma_n^2} - \frac{E_{loc}E|h|^2|g|^4}{\sigma_n^4}p + o(p).$$
(6.12)

Note that |h| and |g| do not depend on n, so the only dependence of $E_{loc}[c]$ and $E_{loc}[c^2]$ on n can be through p. Since the total power consumed in the network is $\frac{1}{2}(p + n\sigma_r^2)$ ($\frac{1}{2}$ comes from the fact that each node is sending only half of the time), the power efficiency is

$$\eta = \frac{2C}{p + n\sigma_r^2}.$$

From (6.11) and (6.10) we can find a probabilistic lower bound for the power efficiency of the network. In other words, for a random placement of the nodes in the domain we have

$$\Pr\left\{\eta = \frac{2C}{n\sigma_r^2 + p} \ge \frac{1}{p + n\sigma_r^2} \log\left(1 + \frac{n^2 \sigma_r^2 p \left(\mathbb{E}_{loc}[c] - \epsilon\right)^2}{\sigma_n^2 + n\sigma_r^2 \left(\mathbb{E}_{loc}[b] + \epsilon\right)}\right)\right\} \ge 1 - \frac{\operatorname{Var}_{loc}[b] + \operatorname{E}_{loc}[c^2]}{n\epsilon^2}.$$

By choosing the transmit power, $p = \frac{1}{\sqrt{n}}$, and the relay node power, $\sigma_r^2 = \frac{1}{n\sqrt{n}}$, we have from (6.12) and the above equation that

$$\Pr\left\{\eta \ge \frac{\sqrt{n}}{2}\log\left(1 + \frac{(\beta_1 + o(\frac{1}{\sqrt{n}}) - \epsilon)^2}{\sigma_n^2 + \frac{\mathbb{E}\log[b] + \epsilon}{\sqrt{n}}}\right)\right\} \ge 1 - \frac{\operatorname{Var}_{loc}[b] + \beta_2 + o(\frac{1}{\sqrt{n}})}{n\epsilon^2},$$

where $\beta_1 = \frac{E_{loc}E|h||g|}{\sigma_n}$ and $\beta_2 = \frac{E_{loc}E|h|^2|g|^2}{\sigma_n^2}$ are some constants independent of n. Close inspection of the above inequality reveals that the term in the logarithm is of order
one. This says that the capacity achieved with the "Listen and Transmit" protocol is of $\Theta(1)$. Moreover, there exists a constant $K_1 \ge 0$ such that

$$\Pr\{\eta \ge K_1 \sqrt{n}\} \ge 1 - \frac{\operatorname{Var}_{loc}[b] + \beta_2 + o(\frac{1}{\sqrt{n}})}{n\epsilon^2}.$$
(6.13)

From the above inequality we can see that for a random placement of the nodes in A, with a high probability that approaches one as the number of nodes increases, we can achieve a power efficiency that grows like \sqrt{n} . Also, the rate achieved is of $\Theta(1)$. The choice of transmit and relay node power in this case is $p = n\sigma_r^2 = \frac{1}{\sqrt{n}}$.

6.4.3 Finding an Upper Bound

We can also find an upper bound on the achievable rates using the "Listen and Transmit" protocol. For this, we consider the case where the receiver in (6.7) knows $\{h_i, g_i, v_i\}$ for i = 1, 2, ..., n. In this case we have

$$C = \sup_{p(s)} I(y; s) \le \sup_{p(s)} I(s; y | \{h_i, g_i, v_i\}),$$
(6.14)

where I(y; s) is the mutual information between y and s. Now, if the receiver knows the channel coefficients and the v_i s, then the system in (6.7) becomes an AWGN channel and therefore

$$\sup_{p(s)} I(s; y | \{h_i, g_i\}) = \frac{1}{2} E_{f\{h_i, g_i\}} \left[\log \left(1 + \frac{|\alpha|^2 p}{\sigma_n^2} \right) \right].$$
(6.15)

The v_i s do not contribute to the noise power in the denominator of (6.15) since the receiver has complete knowledge of them and can cancel out their effect. Combining (6.14) and (6.15) and using the convexity of the log function, we may write

$$C \le \frac{1}{2} \log \left(1 + \frac{\mathrm{E}_{\mathrm{f}} |\alpha|^2 p}{\sigma_n^2} \right). \tag{6.16}$$

Substituting the value of α from (6.7) in the above equation gives

$$C \le \frac{1}{2} \log \left(1 + \frac{\sigma_r^2 p}{\sigma_n^2} \mathbf{E}_{\mathbf{f}} \left(\sum_{i=1}^n \frac{|g_i| |h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}} \right)^2 \right),$$

where again the expectation is taken over the fading of the channels for a fixed placement of the nodes. Without loss of generality we can ignore the $|g_i|^2 p$ term in denominator and rewrite the above equation as

$$C \le \frac{1}{2} \log \left(1 + \frac{\sigma_r^2 p}{\sigma_n^4} \mathbb{E}_f \left(\sum_{i=1}^n |g_i| |h_i| \right)^2 \right).$$
(6.17)

Since, averaged over the fading, the $|h_i||g_i|$ s are independent for different *i*s, we have $E_f \left(\sum_{i=1}^n |g_i||h_i|\right)^2 \leq \left(\sum_{i=1}^n E_f |g_i||h_i|\right)^2 + \sum_{i=1}^n E_f |h_i|^2 |g_i|^2$, and therefore

$$C \le \frac{1}{2} \log \left(1 + \frac{n^2 \sigma_r^2 p}{\sigma_n^4} \left(\frac{\sum_{i=1}^n \mathcal{E}_f |g_i| |h_i|}{n} \right)^2 + \frac{n \sigma_r^2 p}{\sigma_n^4} \frac{\sum_{i=1}^n \mathcal{E}_f |h_i|^2 |g_i|^2}{n} \right).$$
(6.18)

Given that the location of the transmitter and the receiver is fixed, $E_f |h_i| |g_i|$ and $E_f |h_i|^2 |g_i|^2$ for all *i* depend only on the placement of the relay nodes and are i.i.d random variables. Thus, according to the law of large numbers, their average converges to their statistical mean. More specifically, for any $\epsilon > 0$, using Chebyshev's inequality and the union bound on the probability of the events we have,

$$\Pr\left\{\left|\frac{\sum_{i} \mathcal{E}_{f} |h_{i}| |g_{i}|}{n} - \kappa\right| \geq \epsilon, \left|\frac{\sum_{i} \mathcal{E}_{f} |h_{i}|^{2} |g_{i}|^{2}}{n} - \mu\right| \geq \epsilon\right\}$$
$$\geq 1 - \frac{\operatorname{Var}_{loc} \mathcal{E}_{f} |h| |g| + \operatorname{Var}_{loc} \mathcal{E}_{f} |h|^{2} |g|^{2}}{n\epsilon^{2}}, \tag{6.19}$$

where $\kappa = E |h| |g|$ and $\mu = E |h|^2 |g|^2$. Combining (6.19) and (6.18) gives

$$\Pr\left\{\eta \leq \frac{1}{p + n\sigma_r^2} \log\left(1 + \frac{n\sigma_r^2 p}{\sigma_n^4} \left(n(\kappa + \epsilon)^2 + \mu + \epsilon\right)\right)\right\}$$

$$\geq 1 - \frac{\operatorname{Var}_{loc} \operatorname{E}_{\mathrm{f}}|h||g| + \operatorname{Var}_{loc} \operatorname{E}_{\mathrm{f}}|h|^2|g|^2}{n\epsilon^2}.$$
(6.20)

It can be easily verified that for the extreme point of the above upper bound (with respect to p and $n\sigma_r^2$), we have $p = n\sigma_r^2$. Therefore,

$$\eta \le \max_{p} \left\{ \frac{1}{4p} \log \left(1 + \frac{(n(\kappa + \epsilon)^2 + \mu + \epsilon)p^2}{\sigma_n^4} \right) \right\}.$$
(6.21)

By defining $x = \frac{\sqrt{n(\kappa+\epsilon)^2 + \mu + \epsilon}}{\sigma_n^2} p$, (6.21) may be written as

$$\eta \le \frac{\sqrt{n(\kappa+\epsilon)^2 + \mu + \epsilon}}{4\sigma_n^2} \underbrace{\max_x \frac{\log(1+x^2)}{x}}_{<2}.$$
(6.22)

Since κ , μ , ϵ , and σ_n^2 do not depend on n, it is clear from the above equation that $\eta \leq O(\sqrt{n})$. Also the maximization over x is uniquely achieved by some constant x in the interval [0, 2]. Therefore, from the definition of x, the optimal value of p, and hence $n\sigma_r^2$, is $\Theta(\frac{1}{\sqrt{n}})$. Thus, we have shown that for a random placement of nodes in the domain A, the "Listen and Transmit" protocol with high probability achieves a power efficiency of at most of order \sqrt{n} .

6.4.4 Main Result: Sensory Case

In previous sections we found a lower bound on the power efficiency of sensory networks. Combining these bounds together, we have the following theorem.

Theorem 6.3. Consider a random sensory network with a transmitter/receiver pair and n relay nodes where all the nodes are placed randomly and independently on a domain of fixed area A. Assume that averaged over the fading, the various channels are independent, i.e., for every two different channels c_1, c_2 we have $E_f[c_1c_2] =$ $E_f[c_1]E_f[c_2]$, and the measurement noises are all i.i.d $C\mathcal{N}(0, \sigma_n^2)$. Furthermore, assume that the relay nodes have knowledge of their channels to and from the receiver and the transmitter and that the receiver knows the mean of α in (6.7). Then, with high probability the power efficiency of the network is at least $\Theta(\sqrt{n})$, i.e., there exists a scheme such that

$$\Pr\left\{K_2\sqrt{n} \ge \eta \ge K_1\sqrt{n}\right\} \ge 1 - \frac{K}{n},\tag{6.23}$$

where K_1, K_2 , and K are constants depending on the domain and the fading characteristics, but not on n. Moreover, the listen transmit protocol achieves $\eta = \Theta(\sqrt{n})$ with the power allocation $p = n\sigma_r^2 = \Theta(\frac{1}{\sqrt{n}})$.

Remark 6.4. It was shown that in the "Listen and Transmit" protocol the rate of communication is of order constant. Therefore we are getting the same rate of communication as the case when the transmitter and the receiver communicate in isolation. The difference is that in the former protocol, the total power consumption is of order $\Theta(\frac{1}{\sqrt{n}})$ which is \sqrt{n} time less than the power consumption in the later case. Thus, we are getting a fixed rate with less power consumption.

Remark 6.5. Implicit in the "Listen and Transmit" protocol there is a notion of fairness: nodes in relay mode consume, n times less power that the node in transmit mode.

Remark 6.6. Comparing the power efficiency achieved in the sensory networks with the power efficiency of multi-antenna systems, we observe that it is better than the power efficiency of a $n \times 1$ multi-antenna system with no channel knowledge at the transmit antennas where unlike the sensory case, cooperation between different antennas is allowed. However, as we expected it is worse than the power efficiency of a $n \times 1$ multi-antenna system with perfect knowledge at the transmit antennas.

6.4.5 Discussion on Synchronicity

The key idea in the "Listen and Transmit" protocol is to scale the received signals at the relay stage in such a way that the information-bearing signal parts add up coherently at the receiver. Therefore, the protocol is sensitive to any error in the phase, and hence to synchronicity. In this section we try to make a qualitative analysis of the effect of asynchronicity on the "Listen and Transmit" protocol.

Instead of considering an asynchronous system, we consider the lack of synchronicity by introducing a phase error in the channel knowledge used by the relay nodes. More precisely, we assume that instead of knowing the channel h_i perfectly, the *i*-th relay node uses $h_i e^{j\delta_i}$ for processing its received signa, where δ_i is the phase error that models the time lag corresponding to the transmission from *i*-th relay node to the receiver. We assume that the phase errors are i.i.d random variables and independent from the channel coefficients. Furthermore, we assume that $E[e^{j\delta_i}]$ is not zero and is equal to some constant $|\lambda| \neq 0$. In other words, we assume that by the aid of the receiver and by using a training sequence, the relay nodes have some estimate of their time lag and therefore the phase error is not distributed uniformly over the unit circle.

In this case, the received signal at the receiver is

$$y = \sum_{i=1}^{n} h_i t_i e^{j\delta_i} + w,$$

where t_i , defined in (6.6), is the transmitted signal in the case of perfect synchronicity. By plugging in t_i from (6.6) and using the same approach as before, we have

$$C \ge \frac{1}{2} \log \left(1 + \frac{n^2 \sigma_r^2 p \left| \frac{1}{n} \sum_{i=1}^n \mathcal{E}_{f} \frac{|g_i|h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}} e^{j\delta_i} \right|^2}{\sigma_n^2 + \sigma_r^2 \sum_{i=1}^n \mathcal{E}_{f} |h_i|^2} \right)$$
(6.24)

Note that the lack of synchronicity appears as the phase errors $e^{j\delta_i}$ in the lower bound. Looking at the numerator of the lower bound, since the phase errors are independent of the channels, we can see that as we increase n

$$\frac{1}{n}\sum_{i=1}^{n} \mathcal{E}_{\mathbf{f}} \frac{|g_i|h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}} e^{j\delta_i} \to \lambda \mathcal{E}_{loc} \left[\mathcal{E}_{\mathbf{f}} \frac{|g_i|h_i|}{\sqrt{|g_i|^2 p + \sigma_n^2}}\right] \quad a.s.$$

In other words, as the number of nodes increases, for any random network, the term in the numerator of (6.24) with high probability is close to its average over the phase error and the location of the points. Therefore, using the same approach as in the previous section, with high probability, the power efficiency is lower-bounded by

$$\sqrt{n}\log(1+K'|\lambda|^2),$$

where K' depends only on the geometry of the domain and the fading characteristics. From the above discussion, we have the following observations:

- As |λ| decreases, i.e., as we become more and more uncertain about the phase of the channel, the power efficiency also decreases. For the case where |λ| = 0, i.e., the case where we have no estimate of the phase, the lower bound on the achieved power efficiency become zero.
- In terms of n, we see that, as long as $|\lambda| \neq 0$, the asymptotics of the lower bound do not change, and we can still achieve a power efficiency of $\Theta(\sqrt{n})$ with the "Listen and Transmit" protocol.

6.5 Ad-hoc Networks

We now turn our attention to ad-hoc networks. The key difference, compared to the sensory networks, is that we now have r simultaneous transmitter/receiver pairs. Therefore, we are interested in the following question. Assume that in isolation, to maintain some fixed communication rate, each transmitter/receiver pair needs to operate at some power. Now, if the r transmitter/receiver pairs are required to communicate simultaneously and are members of a random wireless network with nnodes, how much can the total power consumption in the network be reduced (from the power consumption required in isolation) to maintain the same communication rate between the transmitters and receivers? We remark that since the capacity of an ad-hoc wireless network scales as $\Theta(\sqrt{n})$ [92, 93, 94, 95], to maintain a fixed rate for each transmitter/receiver pair we need to assume that $r \leq \sqrt{n}$. This will be our standing assumption throughout. To answer the question above, we will construct an extension of "Listen and Transmit" protocol developed for sensory networks. As in the sensory case, the main idea is to exploit interference in the network. For ad-hoc networks the power efficiency is defined as the ratio between the sum of the mutual information of different transmitter/receiver pairs and the total power consumption of the network:

$$\eta = \frac{C_{total}}{P_0}$$

Note that both C_{total} and P_0 are random variables that depend on the placement of the points. We first consider an alternative form of power efficiency, namely $\eta' = \frac{C_{total}}{E_{loc}[P_0]}$, where the denominator is averaged over all point placements in the network. The reason is that it is easier to establish scaling laws for η' . We then show that similar scaling laws apply to η .

6.5.1 "Listen and Transmit" Protocol

As in the sensory case, the communication in the "Listen and Transmit" protocol is divided into two intervals.

(a) **Listen interval:** Each of the *r* transmit users transmit the signal s_i , where $\{s_i\}$ are independent random variables. All other nodes listen. Relay node *j*

receives:

$$r_j = \sum_{i=1}^r g_{ij} s_i + v_j \qquad i = 1, 2, \dots, r \quad j = 1, 2, \dots, n.$$
 (6.25)

(b) **Transmit interval:** Each relay node j transmits t_j , a scaled version of what it has previously received:

$$t_j = \sigma_r d_j r_j \qquad j = 1, 2, \dots, n,$$

where the scalar d_j can depend only on the local knowledge of the channel coefficients at relay node j. Before describing the particular choice of $\{d_j\}$, it is instructive to consider what can be accomplished by having the relay nodes just scale their received signals. To this end, if, for a particular choice of the $\{d_j\}$, we focus on the $r \times r$ channel matrix relating the r transmit signals to the rreceive signals, it is clear that the entries of this matrix are linear combinations of the $\{d_j\}$ (see Section 6.5.7 and (6.64) below). Since the channel matrix has r^2 entries, if $n \ge r^2$, then we have enough free parameters in the $\{d_j\}$ to "generically" make the channel matrix diagonal. This totally suppresses the interference and yields r independent channels. Therefore, in principle, a sumrate of order r is achievable.⁴

The problem with this approach is that it requires complete knowledge of all the channel coefficients at every node of the network (so that each node can solve the system of linear equations required to diagonalize the channel). Since this is not acceptable, we need to introduce a method that only uses local channel

⁴Of course, one should worry about satisfying the power constraints. As we will see later, this scheme is not power efficient.

knowledge, and so we propose the following choice for d_j

$$d_{j} = \frac{\left(\sum_{i=1}^{r} g_{ij}^{*} h_{ji}^{*}\right)}{\sqrt{r\kappa_{1}\sigma_{n}^{2} + r((r-1)\kappa_{2} + \kappa_{3})p}},$$
(6.26)

where κ_1 , κ_2 , and κ_3 are defined as

$$\kappa_{1} = E_{\{j\}} (E_{\{\gamma|j\}} E_{f} |g_{\gamma j}|^{2})^{2}$$

$$\kappa_{2} = E_{\{j\}} (E_{\{\gamma|j\}} E_{f} |g_{\gamma j}|^{2})^{3}$$

$$\kappa_{3} = E_{\{j\}} (E_{\{\gamma|j\}} E_{f} |g_{\gamma j}|^{2} E_{\{\gamma|j\}} E_{f} |g_{\gamma j}|^{4}),$$
(6.27)

where γ is a location of a random point in the domain. Note that these parameters do not depend on n and r and depend only on the geometry of the domain and the fading characteristics.

With the above choice of d_j s, the operation of the relay nodes can be regarded as performing *distributed performing*. It can be further shown that with this choice of d_j , the average transmit power for the relay nodes of the random network is σ_r^2 , where the averaging is over both placement of the network as well as channel fading [113]. Since d_j depends only on the local knowledge of the channel coefficients at relay node j, the d_j s are identical and independent random variables when the location of the relay nodes is random and the transmitters and the receivers are fixed. We will use this fact later on in our results. Finally, we remark that the above mentioned scheme may be interpreted as follows:

• Each relay node estimates each of the r transmitted signals as

$$\hat{s}_{i}^{j} = g_{ii}^{*}r_{j}$$
 $i = 1, 2, \dots, r$ $j = 1, 2, \dots, n$.

Of course, these are very inaccurate estimates.

• Each node attempts to coherently add its estimate of signal s_i for *i*-th receiver via multiplication by h_{ji}^* and normalize the sum to power σ_r^2 :

$$t_j = \frac{1}{\sqrt{r\kappa_1 \sigma_n^2 + r((r-1)\kappa_2 + \kappa_3)p}} \sum_{i=1}^r \hat{s}_i^j h_{ij}^*.$$

Note that in both steps of the protocol, we are exploiting interference. Since the wireless medium is a shared medium, each relay node can estimate each of the r transmitted signals. Also, because of the interference, each receiver will receive a summation of the scaled versions of the signals that the relay nodes have sent. So there are n indirect paths for the signal transmitted from transmitter i to receiver i, each passing through one relay node. Each of the relay nodes has transmitted a signal that has a part that adds coherently for receiver i. Therefore, there are n signal parts that add up coherently at receiver i.

The received signal at receiver k is

$$y_{k} = \sum_{j=1}^{n} h_{jk} t_{j} + w_{k} = \sum_{j=1}^{n} \sigma_{r} h_{jk} d_{j} r_{j} + w_{k} = \sum_{j=1}^{n} \sigma_{r} h_{jk} d_{j} (\sum_{i=1}^{r} g_{ij} s_{i} + v_{j}) + w_{k}$$
$$= \sigma_{r} \sum_{i=1}^{r} \underbrace{(\sum_{j=1}^{n} g_{ij} h_{jk} d_{j})}_{\alpha^{k}} s_{i} + \sigma_{r} \sum_{j=1}^{n} \underbrace{h_{jk} d_{j}}_{\beta^{k}_{j}} v_{j} + w_{k}.$$
(6.28)

We should remark that α_i^k for all *i* is a sum of *n* independent random variables. Also notice that since the relay nodes are placed independently and the β_j^k s depend only on the channel coefficients between relay node *j* and the transmitters and the receivers, they are independent for different *j*'s.

6.5.2 Finding Upper and Lower Bounds

By using the same technique as Section 6.4.1, we can find a lower and upper bound for the mutual information between y_k and s_k . Using the results of [111] again, the maximum value of the mutual information $C_k = \sup_{p(s_k)} I(y_k; s_k)$ can be lowerbounded by the capacity of the AWGN channel with input/output equation:

$$y = \sigma_r \mathcal{E}_{\mathbf{f}} \left[\alpha_k^k \right] s_k + w'_k, \tag{6.29}$$

where w' is a zero mean complex Gaussian noise with variance

$$\operatorname{Var}_{f}[w_{k}'] = \operatorname{Var}_{f}[\sigma_{r}(\alpha_{k}^{k} - \operatorname{E}_{f}[\alpha_{k}^{k}])s_{k} + \sigma_{r}\sum_{i=1,i\neq k}^{r}\alpha_{i}^{k}s_{i} + \sigma_{r}\sum_{j=1}^{n}\beta_{j}^{k}v_{j} + w_{k}]$$

$$= \operatorname{Var}_{f}[\alpha_{k}^{k}]\sigma_{r}^{2}p + (\sum_{i=1,i\neq k}^{r}\operatorname{E}_{f}|\alpha_{i}^{k}|^{2})\sigma_{r}^{2}p + (\sigma_{r}^{2}\sum_{j=1}^{n}\operatorname{E}_{f}|\beta_{j}^{k}|^{2} + 1)\sigma_{n}^{2}.$$
(6.30)

Therefore, we have

$$C_{k} = \sup_{p(s_{k})} \mathrm{I}(y_{k}; s_{k}) \ge \sup_{p(s_{k})} \mathrm{I}(y; s_{k}) = \frac{1}{2} \log \left(1 + \frac{\sigma_{r}^{2} p |\mathrm{E}_{\mathrm{f}}[\alpha_{k}^{k}]|^{2}}{\mathrm{Var}_{\mathrm{f}}[w_{k}']} \right).$$
(6.31)

Note that the $\frac{1}{2}$ in front of the logarithmic term comes from the fact that the transmitters transmit half of the time. We can obtain an upper bound on C_k , considering the case that the receiver k knows $\{v_j, s_i, i = 1, ..., r, i \neq k, j = 1, ..., n\}$ and all the channel coefficients. For this case we have

$$C_{k} = \sup_{p(s_{k})} I(y_{k}; s_{k}) \leq \sup_{p(s_{k})} I(y_{k}; s_{k} | \{v_{j}, s_{i}, i \neq k, g_{ij}, h_{jk}\}) = \frac{1}{2} \mathbb{E}_{f\{g_{ij}, h_{jk}\}} \left[\log \left(1 + \frac{|\alpha_{k}^{k}|^{2} \sigma_{r}^{2} p}{\sigma_{n}^{2}} \right) \right]$$
(6.32)

Using the convexity of the log function, we may rewrite the above equation as

$$C_{k} = \sup_{p(s_{k})} I(y_{k}; s_{k}) \leq \frac{1}{2} \log \left(1 + \frac{\mathrm{E}_{\mathrm{f}} |\alpha_{k}^{k}|^{2} \sigma_{r}^{2} p}{\sigma_{n}^{2}} \right).$$
(6.33)

In order to compute the lower and the upper bound in (6.31) and the above equation, first and second order moments of $\{\alpha_i^k\}$ and $\{\beta_j^k\}$ are required. In the following lemma we give probabilistic bounds on Var_f $[w'_k]$ and $E_f [\alpha_k^k]$. The proof of this lemma can be found in [113].

Lemma 6.7. For every domain A of fixed area and every placement of the nodes of the network, there exist constants B_i , i = 1, ..., 8, κ , κ_1 , and κ' ($\kappa' \leq \kappa$) that depend only on the domain A and the fading characteristics such that for every positive ϵ_i , i = 1, ..., 5 that $\epsilon_4 < \frac{B_7}{\sqrt{r(\kappa_1 \sigma_n^2 + rp\kappa)}}$ and every positive ρ and ζ , we have the following relations:

$$\Pr\left\{ |\mathbf{E}_{\mathbf{f}} \left[\alpha_{k}^{k} \right] | > n\left(\frac{B_{7}}{\sqrt{r(\kappa_{1}\sigma_{n}^{2} + rp\kappa)}} - \epsilon_{4} \right) \right\} \ge 1 - \frac{B_{8}}{n\epsilon_{4}^{2}(\kappa_{1}\sigma_{n}^{2} + rp\kappa')}$$

$$\Pr\left\{ \mathbf{E}_{\mathbf{f}} \left| \alpha_{k}^{k} \right|^{2} < n\left(\frac{B_{2}}{r\kappa'p + \kappa_{1}\sigma_{n}^{2}} + \epsilon_{2} \right) + n^{2}\left(\frac{B_{7}}{\sqrt{r(\kappa_{1}\sigma_{n}^{2} + r\kappap)}} + \epsilon_{4} \right)^{2} \right\}$$

$$\ge 1 - \left(\frac{B_{8}}{n\epsilon_{4}^{2}(\kappa_{1}\sigma_{n}^{2} + rp\kappa')} + \frac{3B_{3}}{n\epsilon_{2}^{2}(\kappa'rp + \kappa_{1}\sigma_{n}^{2})^{2}} \right)$$

$$\Pr\left\{ \operatorname{Var} \left[w_{k}^{'} \right] < \sigma_{r}^{2}p\left(n\epsilon_{2} + nr\epsilon_{3} + \frac{nr(B_{2} + B_{5})}{\kappa'rp + \kappa_{1}\sigma_{n}^{2}} + n^{\rho}r^{\zeta}\epsilon_{5} \right) + \sigma_{n}^{2}\left(1 + n\sigma_{r}^{2}\left(\epsilon_{1} + \frac{\kappa}{\kappa rp + \kappa_{1}\sigma_{n}^{2}}\right) \right) \right\}$$

$$\ge 1 - \frac{1}{(\kappa'rp + \kappa_{1}\sigma_{n}^{2})^{2}} \left(\frac{3B_{1}}{n\epsilon_{1}^{2}} + \frac{3B_{3}}{n\epsilon_{2}^{2}} + \frac{3(r - 1)B_{4}}{n\epsilon_{3}^{2}} + h(n, r, \epsilon_{5}) \right),$$

$$(6.34)$$

where

$$h(n, r, \epsilon_5) = \begin{cases} \frac{rn(\kappa' rp + \kappa_1 \sigma_n^2)}{n^{\rho_r \zeta} \epsilon_5} & Using \ Assumptions \ A, B\\ \frac{rn(r+n)B_6}{n^{2\rho_r 2\zeta} \epsilon_5^2} & Using \ Assumptions \ A, B, C \end{cases}$$

Using Lemma 1, we can combine (6.31) and (6.33) to get upper and lower bounds for C_k . For this, define

$$C_{lower} = \frac{1}{2} \log \left(1 + \frac{n^2 \sigma_r^2 p (\frac{B_7}{\sqrt{r(\kappa_1 \sigma_n^2 + rp\kappa)}} - \epsilon_4)^2}{p \sigma_r^2 (n \epsilon_2 + nr \epsilon_3 + \frac{nr(B_2 + B_5)}{\kappa' rp + \kappa_1 \sigma_n^2} + n^\rho r^\zeta \epsilon_5) + \sigma_n^2 (1 + n \sigma_r^2 (\epsilon_1 + \frac{\kappa}{\kappa rp + \kappa_1 \sigma_n^2}))} \right)$$

$$C_{upper} = \frac{1}{2} \log \left(1 + n \frac{\sigma_r^2 p}{\sigma_n^2} (\frac{B_2}{r \kappa' p + \kappa_1 \sigma_n^2} + \epsilon_2) + n^2 \frac{\sigma_r^2 p}{\sigma_n^2} (\frac{B_7}{\sqrt{r(\kappa_1 \sigma_n^2 + r\kappa p)}} + \epsilon_4)^2 \right).$$
(6.35)

then we have the following relations as probabilistic lower and upper bounds for C_k :

$$\Pr\left\{C_{k} \ge C_{lower}\right\} \ge 1 - \frac{\left(\frac{3B_{1}}{n\epsilon_{1}^{2}} + \frac{3B_{3}}{n\epsilon_{2}^{2}} + \frac{3(r-1)B_{4}}{n\epsilon_{3}^{2}} + \frac{B_{8}(\kappa'rp + \kappa_{1}\sigma_{n}^{2})}{n\epsilon_{4}^{2}} + h(n,r,\epsilon_{5})\right)}{(\kappa'rp + \kappa_{1}\sigma_{n}^{2})^{2}} \quad (6.36)$$

$$\Pr\left\{C_{k} \le C_{upper}\right\} \ge 1 - \left(\frac{B_{8}}{n\epsilon_{4}^{2}(\kappa_{1}\sigma_{n}^{2} + rp\kappa')} + \frac{3B_{3}}{n\epsilon_{2}^{2}(\kappa'rp + \kappa_{1}\sigma_{n}^{2})^{2}}\right).$$

In the "Listen and Transmit" protocol, since there are r transmitters and n relay nodes and all the nodes are transmitting half of the time, the average total power consumption is $\frac{1}{2}(rp + n\sigma_r^2)$. The total capacity of the network is $C_{total} = \sum_{k=1}^r C_k$. Therefore, the power efficiency of the network is

$$\eta' = \frac{2C_{total}}{rp + n\sigma_r^2} = \frac{2\sum_{k=1}^n C_k}{rp + n\sigma_r^2}.$$
(6.37)

Remark 6.8. As mentioned earlier in Lemma 1, the constants B_i and κ , κ_1 , and κ' do not depend on n and r. Now, if we fix σ_r^2 , p, and ϵ_i i = 1, 2, 5, and set $\epsilon_3 = \frac{1}{r}$, $\epsilon_4 = \frac{B_7}{2\sqrt{r(\kappa_1\sigma_n^2 + rp\kappa)}}$ and $\zeta = 0$ in (6.35) and (6.36), then the total capacity achieved by the "Listen and Transmit" protocol is bounded probabilistically as

$$\Pr\left\{C_{total} \ge K_1' r \log(1 + \frac{n^2}{r^2(n+n^\rho)})\right\} > 1 - \frac{rK_2'}{r^2} \left(\frac{r^3}{n} + \frac{r}{n^{2(\rho-1)}}\right),$$

where K'_1 and K'_2 are some constants and we have considered assumptions A, B, and C in Section 6.2.2. Therefore, by setting $\rho = 1 + \theta$, $\theta > 0$, we have

$$\Pr\left\{C_{total} \ge K_1' r \log(1 + \frac{n^{1-\theta}}{r^2})\right\} \ge 1 - K_2' (\frac{r^2}{n} + \frac{1}{n^{2\theta}})$$

Now note that the maximum of the bound is achieved for $r = \Theta(n^{\frac{1-\theta}{2}})$, and in that case we have

$$\Pr\left\{C_{total} \ge K_1' n^{\frac{1-\theta}{2}}\right\} \ge 1 - \frac{K_2'}{n^{\theta}} \quad \forall \theta > 0.$$
(6.38)

From (6.38) we see that, with high probability, by using the "Listen and Transmit" protocol we can get arbitrarily close to the $\Theta(\sqrt{n})$ result of [92]. This result is interesting since we are only using the local knowledge of the channel coefficients at the relay nodes, and the protocol is very simple (it is double-hop and requires no routing).

6.5.3 Power Allocation

We will now focus on how to optimally allocate the transmit and relay node powers (i.e., p and σ_r^2 as functions of r and n) to maximize the power efficiency. Define $\eta'_{lower} = \frac{2rC_{lower}}{rp+n\sigma_r^2}$ and $\eta'_{upper} = \frac{2rC_{upper}}{rp+n\sigma_r^2}$. By using union bound on the probability of events, we get the following probabilistic lower and upper bounds for the power efficiency of the network using the "Listen and Transmit" protocol:

$$\Pr\left\{\eta_{lower}' \leq \eta\right\} \geq \Pr\left\{C_{lower} \leq C_k \quad k = 1, \dots, r\right\} \geq 1 - r\Pr\left\{C_k < C_{lower}\right\}$$

$$\Pr\left\{\eta_{upper}' \geq \eta\right\} \geq \Pr\left\{C_{upper} \geq C_k \quad k = 1, \dots, r\right\} \geq 1 - r\Pr\left\{C_k > C_{upper}\right\}.$$
(6.39)

We will consider the lower bound first. We try to choose the values for p, σ_r^2 , ζ , and ρ so that with high probability we can achieve a power efficiency that scales with the number of nodes in the network. For this goal we take $\epsilon_1, \epsilon_2, \epsilon_3$, and ϵ_5 all to be equal to a positive constant denoted by ϵ . We also choose $\epsilon_4 = \frac{B_7}{2\sqrt{r(\kappa_1\sigma_n^2 + r\kappa_p)}}$. We further consider the network operating in the low SNR regime so that rp is at most constant (in terms of how it scales with n). Later on, when we are looking at the upper bound, we will show that the optimal operating point for this protocol is indeed when rp is of O(1). Using these assumptions in (6.35) and adding K_3rp to the denominator, we have the following lower bound for the power efficiency η'

$$\eta_{lower}' = \frac{r}{rp + n\sigma_r^2} \log \left(1 + \frac{K_1 \frac{n^2}{r} \sigma_r^2 p}{p\sigma_r^2 (n + nrK_2 + n^\rho r^\zeta)\epsilon + \sigma_n^2 + K_3 (n\sigma_r^2 + rp)} \right).$$

$$\Pr\left\{\eta_{lower}' \le \eta\right\} \ge 1 - r \left(\frac{rK_4}{n\epsilon^2} + g(n, r, \epsilon)\right).$$
(6.40)

 K_i , i = 1, ..., 5 are constants and do not depend on r or n. $g(n, r, \epsilon)$ is also derived from $h(n, r, \epsilon)$ after applying the simplifications:

$$g(n, r, \epsilon_5) = \begin{cases} \frac{K_5 r n}{n^{\rho} r^{\zeta} \epsilon} & \text{Using Assumptions A,B} \\ \frac{r n (r+n) B_6}{n^{2\rho} r^{2\zeta} \epsilon^2} & \text{Using Assumptions A,B,C.} \end{cases}$$
(6.41)

Looking at (6.40), the following conditions are necessary in order to have $\Pr \{\eta'_{lower} \leq \eta'\} \rightarrow 1$ for large n and r

$$\frac{r^2}{n} \to 0 \qquad g(n, r, \epsilon)r \to 0. \tag{6.42}$$

Therefore, from the above equation it is clear that this analysis is valid for the case where $r = O(\sqrt{n})$. ζ and ρ should be chosen so that the second condition in (6.42) is satisfied. These two parameters determine the rate of convergence in probability. By looking at (6.41) we observe that the second condition in (6.42) implies that $\frac{1}{n^{\rho-1}r^{\zeta-1}} \to 0$ an n grows. In this stage we maximize the power efficiency with respect to the total transmit power, rp, and total relay power, $n\sigma_r^2$, and subject to the constraints in (6.42). It can be easily verified by taking partial derivatives with respect to rp and $n\sigma_r^2$ that the expression is maximized for $rp = n\sigma_r^2 = x$. Hence, we can write the maximization problem as

$$\eta_{lower}^{\star} = \max_{rp, n\sigma_r^2} \eta_{lower}' = \max_{x \ge 0} \frac{r}{2x} \log \left(1 + \frac{K_1 \frac{n}{r^2} x^2}{\sigma_n^2 + 2K_3 x + (n + nrK_2 + n^{\rho} r^{\zeta}) \epsilon x^2} \right).$$
(6.43)

Let $x = \Theta(n^{\alpha})$, $\alpha \leq 0$, and $r = \Theta(n^{\nu})$, $0 \leq \nu \leq 1/2$. Using the fact that $\alpha \leq 0$ and $\frac{1}{n^{\rho-1}r^{\zeta-1}} \to 0$ in (6.43), we can write

$$\eta_{lower}^{\star} = \Theta\left(\max_{\alpha} n^{\nu-\alpha} \log\left(1 + \frac{n^{1+2\alpha}}{n^{2\nu}(\sigma_n^2 + n^{\rho+2\alpha+\nu(\zeta-1)-1})}\right)\right).$$
(6.44)

With the following constraints:

$$\alpha \le 0, \nu \le \frac{1}{2},$$

$$\begin{cases}
\rho - 1 + \nu(\zeta - 2) > 0 & \text{If using assumptions A, B} \\
\rho - 1 + \nu(\zeta - 1) > 0 & \text{If using assumptions A, B, C,}
\end{cases} (6.45)$$

where the last two constraints are consequences of (6.42).

Consider that we use assumptions A, B, and C. Later on, we analyze the performance of the protocol when only assumptions A and B are used. Set $\rho - 1 + \nu(\zeta - 1)$ fixed and equal to μ . In this case, the rate of convergence in the probability expression of (6.40) is min $\{1 - 2\nu, \mu\}$. Now we are interested in the maximum achievable power efficiency for a fixed μ . We consider the following cases:

(a) $\mu + 2\alpha \leq 0$: In this case we can see that $\rho + 2\alpha + \nu(\zeta - 1) - 1 \leq 0$. Therefore, the noise power is dominant to the interference in (6.44), and we can simplify the expression as

$$\eta_{lower}^{\star} = \Theta\left(n^{\nu-\alpha}\log(1+n^{1-2(\nu-\alpha)})\right) = \Theta\left(n^{\beta}\log(1+n^{1-2\beta})\right), \qquad (6.46)$$

where β is defined as $\beta = \nu - \alpha$. Note that we have $\mu + 2\nu \leq 2\beta$. Now we consider the following cases:

β ≥ ½: In this case the power of n in the log function in (6.46) is negative, so it is of Θ(n^{1-2β}). Therefore we have

$$\eta_{lower}^{\star} = \Theta(n^{1-\beta}).$$

The total rate of transmission R_{sum} is of $\Theta(n^{\nu+1-2\beta})$ in this case. The maximum order of power efficiency is achieved when β takes its smallest possible value, i.e., $\beta = \max\{\frac{1}{2}, \frac{\mu+2\nu}{2}\}$. For this case, the maximum

achievable power efficiency and the total rate of transmission R_{sum} are respectively

$$\eta_{lower}^{\star} = \Theta(n^{1 - \max\{\frac{1}{2}, \frac{\mu + 2\nu}{2}\}}) \text{ and }$$
$$R_{sum} = \Theta(n^{\min\{\nu, 1 - (\nu + \mu)\}}).$$

The transmit power and the relay node power for achieving the maximum power efficiency are

$$rp = n\sigma_r^2 = \Theta(n^{\alpha}) = \Theta(n^{\nu-\beta}) = \Theta(n^{\min\{\frac{2\nu-1}{2},\frac{-\mu}{2}\}}) = \Theta(\min\{\frac{r}{\sqrt{n}},\frac{1}{\sqrt{n^{\mu}}}\}).$$
(6.47)

Therefore, with the choice of the transmit and relay node power as above, we have

$$\Pr\left\{\eta' = \Theta(n^{1-\max\left\{\frac{1}{2},\frac{\mu+2\nu}{2}\right\}})\right\} \ge 1 - \Theta(\frac{1}{n^{\min\left\{\mu,1-2\nu\right\}}}).$$
(6.48)

• $\beta < \frac{1}{2}$: In this case, (6.46) can be rewritten as

$$\eta_{lower}^{\star} = \Theta(n^{\beta} \log n^{1-2\beta}) = (1 - 2\beta)\Theta(n^{\beta} \log n).$$

Now, since $\log n$ grows slower than any polynomial function in n, and β is strictly less than $\frac{1}{2}$, the maximum achievable power efficiency in this case cannot be better than the previous case, and thus operating the network in this region is not favorable.

(b) $\mu + 2\alpha \ge 0$: For this regime the interference will be the dominant term in (6.44), and therefore we have

$$\eta_{lower}^{\star} = \Theta\left(n^{\nu-\alpha}\log(1+n^{1-(2\nu+\mu)})\right)$$

As we can see from the above equation, the power efficiency is maximized when α takes is greatest possible value $\alpha = \frac{-\mu}{2}$. In this case, the power efficiency of the network is

$$\eta_{lower}^{\star} = \Theta\left(n^{\frac{2\nu+\mu}{2}}\log(1+n^{1-(3\nu+\mu)})\right).$$
(6.49)

It can be easily checked that the maximum power efficiency achieved in this region is always less than or equal to the case where $\mu + 2\alpha \leq 0$.

The discussion above gives the probabilistic lower bound of (6.48) for power efficiency when assumptions A,B, and Cfrom Section 6.2.2 can be used. If only assumptions A and B can be used, then applying the same technique as above, we can easily check that the power efficiency is maximized when the network operates in the noise-dominant regime (i.e., $\mu + \nu + 2\alpha \leq 0$) rather than interference-dominant regime (i.e., $\mu + \nu + 2\alpha \geq 0$). Also, similar to the previous discussion, $\beta \triangleq \nu - \alpha$ should be greater than equal to $\frac{1}{2}$.

In this region $\rho + 2\alpha + \nu(\zeta - 1) - 1 \leq 0$, and we are in the noise dominant regime. Hence, we can simplify (6.44) to

$$\eta_{lower}^{\star} = \Theta\left(n^{\nu-\alpha}\log(1+n^{1-2(\nu-\alpha)})\right). \tag{6.50}$$

Therefore, we have $\mu + 3\nu \leq 2\beta$. Also, since $\beta \geq \frac{1}{2}$, the power of n in the log(.) function in (6.50) is negative, so it is of $\Theta(n^{1-2\beta})$. Therefore, we have

$$\eta_{lower}^{\star} = \Theta(n^{1-\beta})$$
 and
 $R_{sum} = \Theta(n^{\nu+1-2\beta})$

The best achievable power efficiency is for the case when β takes its smallest possible value. In this case, $\beta = \max\{\frac{1}{2}, \frac{\mu+3\nu}{2}\}$ and the maximum achievable power efficiency

and the total rate of transmission R_{sum} are respectively

$$\eta_{lower}^{\star} = \Theta(n^{1 - \max\{\frac{1}{2}, \frac{\mu + 3\nu}{2}\}})$$
$$R_{sum} = \Theta(n^{\min\{\nu, 1 - (2\nu + \mu)\}}).$$

The transmit power and the relay node power for achieving the maximum power efficiency are

$$rp = n\sigma_r^2 = \Theta(n^{\min\{\frac{2\nu-1}{2}, \frac{-\nu-\mu}{2}\}}) = \Theta(\min\{\frac{r}{\sqrt{n}}, \frac{1}{\sqrt{n^{\mu}r}}\}).$$
 (6.51)

With this choice of transmit and relay node power we have the following probabilistic lower bound on the power efficiency (using assumptions A and B only):

$$\Pr\left\{\eta = \Theta(n^{1-\max\left\{\frac{1}{2}, \frac{\mu+3\nu}{2}\right\}})\right\} \ge 1 - \Theta(\frac{1}{n^{\min\left\{\mu, 1-2\nu\right\}}}).$$
(6.52)

6.5.4 Main Result: Ad-hoc Case

The analysis in the previous section shows the following result:

Theorem 6.9. Consider an n node random ad-hoc network where the nodes are placed randomly and independently on a domain of fixed area where, averaged over the fading, the various channels are independent, i.e., for every two different channels c_1, c_2 we have $E_f[c_1c_2] = E_f[c_1].E_f[c_2]$. Furthermore, assume conditions A, B, and C given in Section 6.2.2 and that at any given time there are $r = O(n^{\nu})$ $\nu \leq \frac{1}{2}$ transmit/receive pairs. Also, the measurement noises are all i.i.d $C\mathcal{N}(0, \sigma_n^2)$. If we denote the power efficiency of the network by η' (i.e, $\eta' = \frac{C_{total}}{E_{loc}[P_0]}$), then for every $\mu > 0$

$$\Pr\left\{\eta' \ge K_1(n^{1-\max\left\{\frac{1}{2},\frac{\mu+2\nu}{2}\right\}})\right\} \ge 1 - \frac{K_2}{n^{\min\left\{1-2\nu,\mu\right\}}},\tag{6.53}$$

where K_1 and K_2 are independent of n and r but depend on the domain and the fading

characteristics. Moreover, the listen transmit protocol achieves this lower bound. The transmit and the relay node powers that achieve this power efficiency are given in (6.47).

The following corollary is an immediate consequence of Theorem 6.9 by setting $\mu = 1 - 2\nu$.

Corollary 6.10. Consider the network model described in Theorem 6.9. If the number of transmitter/receiver pairs in the network is of $O(n^{\frac{1-\epsilon}{2}})$, where $\epsilon > 0$, then we have

$$\Pr\left\{\eta \ge K_3\sqrt{n}\right\} \ge 1 - \frac{K_4}{n^{\epsilon}},\tag{6.54}$$

where K_3 and K_4 are independent of n and r but depend on the domain and the fading characteristics. Moreover, by choosing the transmit and relay node powers as $rp = n\sigma_r^2 = \Theta(\frac{r}{\sqrt{n}})$, the listen and transmit protocol achieves this lower bound.

Remark 6.11. Note that from (6.54), we can see that there is a trade-off between the number of transmitter/receiver pairs r and the rate of convergence. As we increase r from $\Theta(1)$ to $\Theta(\sqrt{n})$, the convergence slows down.

For the case, when only assumptions A and B are used, we have the following result from (6.51) and (6.52):

Theorem 6.12. Consider an n node random ad-hoc network in a domain of fixed area where, averaged over the fading, the various channels are independent. Furthermore, assume conditions A, B given in Section 6.2.2 and that at any given time there are $r = O(n^{\nu})$ $\nu \leq \frac{1}{2}$ transmit/receive pairs. Also, the measurement noises are all i.i.d $\mathcal{CN}(0, \sigma_n^2)$. If we denote the power efficiency of the network by η' (i.e., $\eta' = \frac{C_{total}}{E_{loc}[P_0]}$), then for every $\mu > 0$

$$\Pr\left\{\eta' \ge K_5 \, n^{1 - \max\left\{\frac{1}{2}, \frac{\mu + 3\nu}{2}\right\}}\right\} \ge 1 - \frac{K_6}{n^{\min\left\{1 - 2\nu, \mu\right\}}}$$

where K_5 and K_6 are independent of n and r and depend on the domain and fading characteristics. Moreover, the listen transmit protocol achieves this lower bound. The transmit and relay node powers achieving this power efficiency are given in (6.51).

By considering the case where $\mu + 3\nu = 1$, we have the following corollary:

Corollary 6.13. Consider the network described in Theorem 6.12. If the number of transmit/receive pairs in the network is of $O(n^{\frac{1-\epsilon}{3}})$, where $\epsilon > 0$, then we have

$$\Pr\left\{\eta' \ge K_7 \sqrt{n}\right\} \ge 1 - \frac{K_8}{n^\epsilon}$$

where K_7 and K_8 only depend on the domain and fading characteristics. Moreover, by choosing the transmit and relay node powers as $rp = n\sigma_r^2 = \Theta(\frac{r}{\sqrt{n}})$, the listen and transmit protocol achieves this lower bound.

Corollary 6.13 implies that the maximum number of transmit/receive pairs that the network can support with power efficiency of $\Theta(\sqrt{n})$ is $\Theta(n^{\frac{1}{3}})$. On the other hand, considering the maximum power efficiency of the network with the number of transmit/receive pairs up to $n^{\frac{1}{2}}$, one can write the following corollary by setting μ equal to $1 - 2\nu$ in Theorem 6.9. We should also remark that comparing Corollaries 6.10 and 6.13, we see that the effect of assumption **C** is on the number of simultaneous transmitter/receiver pairs that can be in the network.

Corollary 6.14. Consider again the network described in Theorem 6.12. If the number of transmit/receive pairs in the network is of $\Theta(n^{\frac{1-\epsilon}{2}})$, where $0 < \epsilon < 1$, then we have

$$\Pr\left\{\eta' \ge K n^{\left(\frac{1+\epsilon}{4}\right)}\right\} \ge 1 - \frac{K'}{n^{\epsilon}},$$

where K, K' are some constants. Therefore, in this case if the number of transmit/receive pairs is near to \sqrt{n} , we can achieve a power efficiency that scales like $n^{\frac{1}{4}}$. The previous discussion shows that:

- If the number of the transmitter/receiver pairs is less than \sqrt{n} , it was shown that a power efficiency that scales with the number of nodes, n, is achievable. The rate per transmitter/receiver pair in this case is of order constant. If we increase the number of simultaneous transmissions to more than \sqrt{n} , we can still achieve power efficiency of \sqrt{n} using time-sharing and "Listen and Transmit" protocol together. In this case, at each time instant \sqrt{n} of the transmitters transmit and all the others act as relay nodes. However, in this case, the rate per transmitter/receiver will not be of order constant, but of order $\frac{1}{\sqrt{n}}$. This is in agreement with the result of [92, 93, 94] in that achieving a constant rate per node in this case would require a total sum-capacity larger than $\Theta(\sqrt{n})$, which is not possible.
- There is a notion of fairness implicit in the protocol in the sense that nodes in the relay mode consume $\frac{r}{n}$ -th the power of the nodes in the transmit mode.
- For the case where $r = \Theta(n^{\frac{1-\epsilon}{2}})$ (or $r = \Theta(n^{\frac{1-\epsilon}{3}})$, if we do not have assumption C in Section 6.2.2, the optimal choice of the transmit power and relay power is $p = \Theta(\frac{1}{\sqrt{n}}), \ \sigma_r^2 = \Theta(\frac{1}{n})$. The total power consumption is $\Theta(\frac{r}{\sqrt{n}})$, and the total rate is $\Theta(r)$.
- In the case of ad-hoc networks, by using the "Listen and Transmit" protocol, we can see that the we are keeping the rate of transmission for each transmitter/receiver fixed and of order Θ(1), but the total power consumption decreases as the number of nodes grows larger, as long as r = O(n^{1-ε}/₂) for some positive ε. If we do not have assumption C on the channel coefficients, we still have this property for r = O(n^{1-ε}/₃) for some positive ε.

We should mention that with high probability we cannot get a better power efficiency for ad-hoc networks with this protocol. We can show this by using (6.35), (6.36), and (6.39) to find a probabilistic upper bound. With an argument like the one for the lower bound or the one in [105], we can show that with high probability the maximum achievable power efficiency with this protocol is $O(\sqrt{n})$.

6.5.5 A Further Result

As mentioned earlier, the power efficiency that was considered up to now was defined as the ratio between the sum rate capacity for a specific placement of the nodes of the network and the average of the power consumption over all possible point placements of the network. In other words, for a specific placement of the nodes with sum rate capacity of C_{total} and power consumption of P_0 we defined power efficiency as $\eta' = \frac{C_{total}}{E_{toc}[P_0]}$. An alternative to this definition is to consider the ratio between the rate and power consumption for a specific network, i.e., $\eta = \frac{C_{total}}{P_0}$ as the power efficiency. In this case, the power efficiency η is a random variable depending on the placement of the nodes. However, because of the law of large numbers, as the size of the network increases, η will be close to its average, and we observe the same behavior as η' for the power efficiency. In order to state this formally, we will need the following lemma. We should remark that in proving this lemma one only needs assumptions A and B of Section 6.2.2. This lemma gives a bound on the power consumption at the relay stage.

Lemma 6.15. Consider an n node ad-hoc network with assumptions provided in Theorem 6.9; then for any specific placement of the nodes in the network, the total power consumption at relay nodes P_r can be bounded as

$$\Pr\{|P_r - n\sigma_r^2| < \frac{n^{1+\gamma}\sigma_r^2}{\sqrt{r}}(\sigma_n^2 + rp)\} \ge 1 - \frac{K'}{n^{2\gamma}(\kappa_1\sigma_n^2 + r\kappa p)^2}, \tag{6.55}$$

where γ is any positive number, and K', κ, κ_1 are constant independent of n and r.

Using this lemma and the fact that the network operates at low SNR regime, i.e.,

 $rp \ll 1$, we can bound the total power consumption of the network $P_0 = rp + P_r$ as follows

$$\Pr\{P_0 \le (1 + \frac{n^{\gamma}}{\sqrt{r}})(rp + n\sigma_r^2)\} \ge 1 - \frac{K}{n^{2\gamma}}.$$

Therefore, for $\eta = \frac{R_{sum}}{P_0}$ we have

$$\Pr\left\{ (1 + \frac{n^{\gamma}}{\sqrt{r}})^{-1} \eta' \le \eta \right\} \ge 1 - \frac{K}{n^{2\gamma}}.$$
(6.56)

One can combine this relation with the results on the power efficiency η' to get new bounds on η . The following theorems are immediate consequences of (6.56), Theorem 6.9 and Theorem 6.12.

Theorem 6.16. Consider an n node random ad-hoc network where the nodes are placed randomly and independently on a domain of fixed area where, averaged over the fading, the various channels are independent, i.e., for every two different channels c_1, c_2 we have $E_f[c_1c_2] = E_f[c_1].E_f[c_2]$. Furthermore, assume conditions A, B and, C given in Section 6.2.2 and that at any given time there are $r = O(n^{\nu})$ $\nu \leq \frac{1}{2}$ transmit/receive pairs. Also, the measurement noises are all i.i.d $\mathcal{CN}(0, \sigma_n^2)$. For a specific placement of the nodes of the random network, let C_{total} be the total rate of communication and P_0 be the total power consumption in the network. Then, for every $\mu, \gamma > 0$, the power efficiency of the network defined as $\eta = \frac{C_{total}}{P_0}$ satisfies

$$\Pr\left\{\eta \ge K_1 \frac{\left(n^{1-\max\left\{\frac{1}{2}, \frac{\mu+2\nu}{2}\right\}}\right)}{1+\frac{n^{\gamma}}{\sqrt{r}}}\right\} \ge 1 - \frac{K_2}{n^{\min\left\{1-2\nu, \mu, 2\gamma\right\}}},\tag{6.57}$$

where K_1 and K_2 are independent of n and r but depend on the domain and the fading characteristics. Moreover, the listen transmit protocol achieves this lower bound.

By setting $\gamma = \frac{\nu}{2}$ and $\mu = 1 - 2\nu$, we have the following corollary.

Corollary 6.17. For the network described in Theorem 6.16, we have

$$\Pr\left\{\eta \ge K_1 \sqrt{n}\right\} \ge 1 - \frac{K_2}{n^{\min\left\{1 - 2\nu, \nu\right\}}}.$$
(6.58)

Therefore, it becomes clear that by considering the power efficiency as the ratio between the sum-rate of transmission and the total power consumption for specific placement of the nodes in the network, we still have similar scaling behavior.

Remark 6.18. We should remark that the rate of convergence obtained for the probability of the event in (6.58) is not tight (for small r). One expects that as r, the number of pairs requesting service from the network, decreases, the rate of convergence improves (for instance, we can see from Theorem 6.3 that for sensory networks in which r = 1, the rate of convergence is proportional to $\frac{1}{n}$). Looking at (6.58), we observe that as r decreases to constant, the convergence slows down. This is an artifact of our approach in bounding η .

If we use only assumptions A and B in Section 6.2.2, then we can write the following theorem using (6.56) and Theorem 6.12.

Theorem 6.19. Consider an n node random ad-hoc network where the nodes are placed randomly and independently on a domain of fixed area where, averaged over the fading, the various channels are independent, i.e., for every two different channels c_1, c_2 we have $E_f[c_1c_2] = E_f[c_1].E_f[c_2]$. Furthermore, assume conditions A, B given in Section 6.2.2 and that at any given time there are $r = O(n^{\nu})$ $\nu \leq \frac{1}{2}$ transmit/receive pairs. Also, the measurement noises are all i.i.d $\mathcal{CN}(0, \sigma_n^2)$. For a specific placement of the nodes of the random network, let C_{total} be the total rate of communication and P_0 be the total power consumption in the network. Then, for every $\mu, \gamma > 0$, the power efficiency of the network defined as $\eta = \frac{C_{total}}{P_0}$ satisfies

$$\Pr\left\{\eta \ge K_5 \frac{\left(n^{1-\max\left\{\frac{1}{2},\frac{\mu+3\nu}{2}\right\}}\right)}{1+\frac{n^{\gamma}}{\sqrt{r}}}\right\} \ge 1 - \frac{K_6}{n^{\min\left\{1-2\nu,\mu,2\gamma\right\}}},\tag{6.59}$$

where K_5 and K_6 are independent of n and r but depend on the domain and the fading characteristics. Moreover, the listen transmit protocol achieves this lower bound.

By setting $\gamma = \frac{\nu}{2}$ and $\mu = 1 - 3\nu$, we have the following corollary:

Corollary 6.20. Consider the network model described in Theorem 6.19. Then, for $r = O(n^{\nu})$ $\nu \leq \frac{1}{3}$ we have

$$\Pr\left\{\eta \ge K_1 \sqrt{n}\right\} \ge 1 - \frac{K_2}{n^{\min\{1-3\nu,\nu\}}}.$$
(6.60)

6.5.6 Discussion on Synchronicity

Similar to the sensory case, the key idea of the "Listen and Transmit" protocol used for ad-hoc networks is the coherent and synchronous reception of the signals. Therefore, the protocol is sensitive to synchronicity. In this section we discuss the effect of asynchronocity on our protocol.

Like sensory networks, instead of considering an asynchronous system, we consider the lack of synchronicity by introducing a phase error in the channel knowledge used by the relay nodes. More precisely, we assume that instead of knowing the channel h_{ji} perfectly, the *j*-th relay node uses $h_{ji}e^{j\delta_{ji}}$ for processing its received signal. δ_{ji} is the phase error that models the time lag, corresponding to the transmission from *j*-th relay node to the *i* receiver. We assume that the phase errors are i.i.d random variables and are independent from channel coefficients. Furthermore, we assume that $E[e^{j\delta_{li}}]$ is not zero and is equal to some constant λ . In other words, we assume that by the aid of the receivers and by a using training sequence, the relay nodes have some estimate of their time lag and therefore the phase error is not distributed uniformly over the unit circle. In this case, the scalar \hat{d}_k used by the k-th relay node is proportional to

$$\sum_{l=1}^r g_{lk}^* h_{kl}^* e^{-j\delta_{lk}}.$$

Using these \hat{d}_k s, we can find the new α_k^i s and β_k^i s (6.28) in terms of δ_{li} . Following the lines of Section 6.5.1 and Section 6.5.2, it can be verified that we will still have the same asymptotic behavior for power efficiency in terms of n (i.e., the asymptotic behavior of the achieved power efficiency is still like \sqrt{n}), but the constants appearing in the relations will now depend on λ as well. Therefore, similar to the sensory case as $|\lambda|$ decreases, the power efficiency will also decrease and, finally, for the limiting case of $|\lambda| = 0$, the lower bound on the achieved power efficiency also becomes zero.

6.5.7 Complete Knowledge of the Channel

In the "Listen and Transmit" protocol we assumed that relay nodes have only local knowledge of the channels, i.e., they only know their connections to the transmitter and receiver nodes. We addressed another scenario in the previous sections, where the nodes have complete knowledge of all the channel coefficients and try to diagonalize the channel matrix between the transmitters and their corresponding receivers. In this section we analyze the effect of perfect knowledge of the channel on the power efficiency achieved by diagonalizing the channel matrix.

In this section we make an additional assumption that the channel coefficients are independent complex random variables with zero mean and unit variance. Using (6.25) and (6.28), we can describe our protocol by the following matrix relations

p

$$\mathbf{r} = \mathbf{s}\mathbf{G} + \mathbf{v}$$

$$\mathbf{t} = \mathbf{r}\mathbf{D}$$

$$\mathbf{y} = \mathbf{s}\mathbf{G}\mathbf{D}\mathbf{H} + \mathbf{v}\mathbf{D}\mathbf{H} + \mathbf{w}$$

$$\operatorname{tr}\left(\mathbf{D}^{*}\mathbf{G}^{*}\mathbf{G}\mathbf{D}\right) + \sigma_{n}^{2}\operatorname{tr}\left(\mathbf{D}^{*}\mathbf{D}\right) = n\sigma_{r}^{2}$$
(6.61)

where $\mathbf{s} \in \mathbb{C}^{1 \times r}$ is the transmitted vector, $\mathbf{y} \in \mathbb{C}^{1 \times r}$ is the received vector, and $\mathbf{r}, \mathbf{t} \in \mathbb{C}^{1 \times n}$ are the respective received and transmitted vectors at the relay stage. $\mathbf{w} \sim \mathcal{CN}(\mathbf{0}, \sigma_n^2 \mathbf{I}_r)$ and $\mathbf{v} \sim \mathcal{CN}(\mathbf{0}, \sigma_n^2 \mathbf{I}_n)$ are the corresponding vectors of noise introduced at the receivers and at the relay stage, respectively. $\mathbf{G} \in \mathbb{C}^{r \times n}$ is the channel matrix between the transmitters and the relay nodes, and $\mathbf{H} \in \mathbb{C}^{n \times r}$ is the channel matrix between the relay nodes and the receivers. Finally, $\mathbf{D} = \text{diag}(d_1, \ldots, d_n) \in \mathbb{C}^{n \times n}$ is a diagonal matrix with diagonal entries corresponding to the scalars chosen by the relay nodes. Notice that the d_j s depend on the channel gains. The last equation in (6.61) is a consequence of the power constraint for the relay nodes. We remark that the power constraint considered here is more general than what was assumed in previous sections.

From (6.61), the equivalent channel matrix between the transmitters and the receivers is **GDH**. Therefore, diagonalizing the channel matrix amounts to finding diagonal matrix **D** such that **GDH** = $\alpha \mathbf{I}_r$ for some complex scalar α . The number of complex equations is r^2 , and the number of variables is n. Therefore, generically, this equation has a solution for $r^2 \leq n$. In this case, by looking at (6.61) we can write the received signal at receive node k as

$$y_k = \alpha s_k + w_k + \sum_{j=1}^n h_{jk} d_j v_j$$

We can find an upper bound on the achievable rates using the scheme described

above by considering that the receiver node has knowledge of the different noises introduced in the relay stage. Hence, we can bound the capacity of the channel between the transmit/receive pair k as

$$C_k \le \frac{1}{2} \operatorname{E}_{\mathrm{f}} \log(1 + \frac{|\alpha|^2 p}{\sigma_n^2}) \le \frac{1}{2} \log(1 + \frac{p \operatorname{E}_{\mathrm{f}} |\alpha|^2}{\sigma_n^2}).$$

The power efficiency can be bounded as follows:

$$\eta' = \frac{2\sum_{i=1}^{r} C_i}{rp + n\sigma^2} \le \frac{2r}{rp + n\sigma_r^2} \log(1 + \frac{E_f |\alpha|^2 p}{\sigma_n^2}).$$
(6.62)

Thus, we only need to find the mean of the maximum possible value of $|\alpha|^2$ subject to the following constraints

$$\mathbf{GDH} = \alpha \mathbf{I}_r \quad \text{and}$$

$$p \operatorname{tr} \left(\mathbf{D}^* \mathbf{G}^* \mathbf{GD} \right) + \sigma_n^2 \operatorname{tr} \left(\mathbf{D}^* \mathbf{D} \right) = n \sigma_r^2.$$

$$(6.63)$$

First, we try to solve the first equation in (6.63). Define $\mathbf{d} = [d_1, \ldots, d_n]^t$, $\mathbf{b} = \operatorname{vec}(\mathbf{I}_r)$. (6.63) can be written in terms of \mathbf{d} and \mathbf{b} as

$$\mathbf{Ad} = \alpha \mathbf{b} \quad \text{and}$$

$$\mathbf{d}^* \mathbf{Ad} = n\sigma_r^2,$$
(6.64)

where it can be easily verified that $\mathbf{A} \in \mathbb{C}^{r^2 \times n}$ and $\mathbf{\Lambda} \in \mathbb{C}^{n \times n}$ are

$$\mathbf{A}_{r^{2} \times n} = \begin{bmatrix} \mathbf{H}^{T} & \mathbf{0} & \dots & \mathbf{0} \\ \mathbf{0} & \mathbf{H}^{T} & \dots & \mathbf{0} \\ \vdots & \dots & \ddots & \vdots \\ \mathbf{0} & \mathbf{0} & \dots & \mathbf{H}^{T} \end{bmatrix}_{r^{2} \times rn} \cdot \begin{bmatrix} \operatorname{diag} (\mathbf{G}_{1}) \\ \operatorname{diag} (\mathbf{G}_{2}) \\ \vdots \\ \operatorname{diag} (\mathbf{G}_{r}) \end{bmatrix}_{rn \times n},$$
$$\mathbf{A}_{n \times n} = \operatorname{diag} \left(p \| \mathbf{g}_{1} \|^{2} + \sigma_{n}^{2}, \dots, p \| \mathbf{g}_{n} \|^{2} + \sigma_{n}^{2} \right).$$
(6.65)

$$\begin{split} \mathbf{G}_{i}, \quad i = 1, \dots, r, \text{ and } \mathbf{g}_{j}, \quad j = 1, \dots, n, \text{ denote the } i\text{-th row and } j\text{-th column} \\ \text{of } \mathbf{G}, \text{ respectively. If we define } \mathbf{B}_{r^{2} \times n} &= \mathbf{A} \mathbf{\Lambda}^{-\frac{1}{2}} \text{ by using QR-type decomposition} \\ [112] \text{ for } \mathbf{B}^{*}, \text{ we can write } \mathbf{B}^{*} &= \mathbf{Q} \begin{pmatrix} \mathbf{R} \\ \mathbf{0} \end{pmatrix}, \text{ where } \mathbf{Q}_{n \times n} \text{ is unitary matrix (i.e., } \mathbf{Q} \mathbf{Q}^{*} = \\ \mathbf{Q}^{*} \mathbf{Q} &= \mathbf{I}_{n}), \text{ and } \mathbf{R}_{r^{2} \times r^{2}} \text{ is a lower triangular matrix with diagonal elements equal to} \\ \text{ unity. By writing } \mathbf{Q}^{*} \mathbf{\Lambda}^{\frac{1}{2}} \mathbf{d} &= \begin{pmatrix} \mathbf{d}_{r^{2} \times 1}^{(1)} \\ \mathbf{d}_{(n-r^{2}) \times 1}^{(2)} \end{pmatrix}, \text{ we can rewrite } (6.64) \text{ as} \end{split}$$

$$\mathbf{R}^* \mathbf{d}^{(1)} = \alpha \mathbf{b} \quad \text{and}$$

$$\|\mathbf{d}^{(1)}\|^2 + \|\mathbf{d}^{(2)}\|^2 = n\sigma_r^2.$$
(6.66)

Now notice that **R** is invertible and therefore we can find $\mathbf{d}^{(1)}$ from (6.66), and by substituting its value in the second relation of (6.66) we get

$$|\alpha|^2 \mathbf{b}^* (\mathbf{R}^* \mathbf{R})^{-1} \mathbf{b} + \|\mathbf{d}^{(2)}\|^2 = n\sigma_r^2.$$

It can be easily verified that $\mathbf{R}^*\mathbf{R} = \mathbf{B}\mathbf{B}^* = \mathbf{A}\Lambda^{-1}\mathbf{A}^*$. Also, the maximum of $|\alpha|^2$ is when $\mathbf{d}^{(2)} = \mathbf{0}_{n-r^2 \times 1}$. Therefore,

$$|\alpha|_{max}^2 = \frac{n\sigma_r^2}{\mathbf{b}^* (\mathbf{A} \mathbf{\Lambda}^{-1} \mathbf{A}^*)^{-1} \mathbf{b}}.$$
(6.67)

Now, using the following inequality for positive definite matrix B and any vector x ([112], page 452)

$$(\mathbf{x}^*\mathbf{B}\mathbf{x})(\mathbf{x}^*\mathbf{B}^{-1}\mathbf{x}) \ge (\mathbf{x}^*\mathbf{x})^2,$$

we have

$$|\alpha|_{max}^2 \leq \frac{n\sigma_r^2(\mathbf{b}^*(\mathbf{A}\mathbf{\Lambda}^{-1}\mathbf{A}^*)\mathbf{b})}{(\mathbf{b}^*\mathbf{b})^2}.$$

 $\mathbf{b}^*\mathbf{b} = r$ and $\frac{1}{\sigma_n^2}\mathbf{I}_n - \mathbf{\Lambda}^{-1}$ is positive semi-definite. Therefore, we have

$$\mathbf{E}_{\mathbf{f}} \left| \alpha \right|_{max}^{2} \leq \frac{n \sigma_{r}^{2}}{r^{2} \sigma_{n}^{2}} \mathbf{E}_{\mathbf{f}} \left[\mathbf{b}^{*} (\mathbf{A} \mathbf{A}^{*}) \mathbf{b} \right].$$
(6.68)

Using the structure of matrix \mathbf{A} in (6.65), it can be verified that

$$E_{f}[\mathbf{b}^{*}(\mathbf{A}\mathbf{A}^{*})\mathbf{b}] = \sum_{i,j=1}^{r} E_{f}[\mathbf{b}_{i}^{*}\mathbf{H}^{T} \operatorname{diag}(\mathbf{G}_{i})\operatorname{diag}^{*}(\mathbf{G}_{j})\bar{\mathbf{H}}\mathbf{b}_{j}],$$

where b_i is the *i*-th column of \mathbf{I}_r . Now since the entries of \mathbf{H} and \mathbf{G} are independent from each other, the expectation inside the above summation is zero for $i \neq j$. Therefore we have

$$E_{f} [\mathbf{b}^{*} (\mathbf{A}\mathbf{A}^{*})\mathbf{b}] = \sum_{i}^{r} E_{f} [\mathbf{b}_{i}^{*} (\mathbf{H}^{*}\mathbf{H})^{T} \mathbf{b}_{i}]$$

$$= E_{f} \operatorname{tr} ((\mathbf{H}^{*}\mathbf{H})^{T} \cdot \sum_{i=1}^{r} \mathbf{b}_{i} \mathbf{b}_{i}^{*})$$

$$= E_{f} \operatorname{tr} ((\mathbf{H}^{*}\mathbf{H})^{T} \cdot \mathbf{I}_{r})$$

$$= nr,$$

(6.69)

where we have used the fact that $\operatorname{tr}(\mathbf{AB}) = \operatorname{tr}(\mathbf{BA})$. Using the above result in (6.68), we have

$$\mathcal{E}_{f} \left| \alpha \right|_{max}^{2} \le \frac{n^{2} \sigma_{r}^{2}}{r \sigma_{n}^{2}}.$$
(6.70)

Combining (6.70) and (6.62) we have

$$\eta' \le \frac{r}{rp + n\sigma_r^2} \log(1 + \frac{n^2 \sigma_r^2 p}{r\sigma_n^2}).$$
(6.71)

Using an argument to previous sections (e.g., Section 4.3), we know that the maximum of the right hand side expression is less than $\kappa \sqrt{n}$ for some constant κ dependent on



Figure 6.4: Power efficiency for interference suppression case.

 σ_n^2 . Therefore,

$$\eta' \le \kappa \sqrt{n},$$

which is the same as the case when we have only local knowledge of the channel and "Listen and Transmit" protocol is used. We have found the upper bound for the maximum power efficiency, η' , using the actual value for $E_f |\alpha|^2$ from (6.67) and Matlab simulation. We have plotted the ratio $\frac{\log(\eta')}{\log n}$ for different values of $r \leq \sqrt{n}$ for n = 100, 400, 900, and 1600. As we can see from the plots, the upper bound suggests that we cannot do better than \sqrt{n} (or equivalently $\frac{1}{2}$ in the Figure 6.4). Also, as the number of the simultaneous transmitter/receiver pairs, r, increases, the upper bound on the power efficiency of the interference cancelation method becomes smaller. This suggests that this method is not power efficient.

Based on the above argument we have the following theorem.

Theorem 6.21. Consider a wireless ad-hoc network with n relay nodes and r transmit/receive pair in which $r^2 \leq n$. Moreover, assume the channel coefficients can be modeled by independent zero mean unit variance complex random variables. Also assume that the relay nodes have complete knowledge of the channel coefficients; then, if the relay nodes cancel out the interference at the receiver nodes, the power efficiency scales as $O(\sqrt{n})$.

6.6 Conclusion

In this chapter we address the power efficiency of random sensory and ad-hoc wireless networks formed in a domain of fixed area. Under some assumptions on the moments of the channel coefficients, we show that asymptotically, as the number of nodes in the network, n, grows larger, with high probability we can achieve a power efficiency of $\Theta(\sqrt{n})$ for sensory networks. For ad-hoc networks, if the number of transmitter/receiver pairs is of $O(\sqrt{n})$, we can achieve the same result. We also described the protocol used to achieve this power efficiency. Although the best results for capacity per node in sensory and ad-hoc wireless networks decrease as the size of networks grows larger [91], [92], we can see that it pays off to consider these networks in terms of power efficiency.

We can think of the protocol used in this chapter as a simple yet powerful memoryless linear coding scheme for the relay nodes, i.e., the relay nodes simply relay a scaled version of what they have heard. One can generalize this protocol by using other coding schemes for relay nodes. Another interesting problem is to look at the spectral efficiency of sensory and ad-hoc networks and its trade-off with the power efficiency. Partial results about the optimal trade-off is reported in [114].

Chapter 7

Differentiated Rate Scheduling for Cellular Systems

7.1 Introduction

The downlink scheduling in cellular systems is known to be one major bottleneck for future broadband communication systems. From an information-theoretic perspective, broadcast channels [53], and in particular the Gaussian broadcast channels (GBC), can be used to model the downlink in a cellular system. There exist an abundance of information-theoretic results describing the limits of the achievable rates¹ to the users in single-input single-output (SISO) Gaussian broadcast channels (see e.g., [115, 116]). For example, in a homogeneous network, i.e., a network where the fading and noise distributions of all the users are identical, if the transmitter wants to maximize the throughput (or the sum of the rates to all the receivers²), it is well known that the optimal strategy is to transmit to the user with the best channel condition at each channel use. This is often referred to as the *opportunistic* transmission strategy [102].

More recently, there has been growing interest in the use of multiple antennas (at the transmitter, receivers, or both) for wireless communication systems. The initial

 $^{^{1}}$ In the remainder of this work, we assume that the channel is ergodic and rate refers to the average rate over all channel realizations.

²We use users and receivers interchangeably.

focus has been on point-to-point communications, where it has been shown that the use of multiple antennas can significantly increase the rate and reliability of a wireless communication link. Given this, both the research and industrial communities have begun to study the use of multiple antenna systems in wireless networks. A most obvious application is in cellular systems, where the use of multiple antennas at the base station can potentially increase the capacity of each cell. This has led to an interest in the multiple-input multiple-output (MIMO) Gaussian broadcast channel, where the transmitter and the various users may be equipped with multiple transmit and receive antennas, respectively. First, the sum-rate of the MIMO broadcast channel, i.e., the maximum possible sum of the rates to all users [117], and then the entire capacity region [118], were shown to be achieved by an interference cancelation scheme referred to as *dirty paper coding* (DPC) [119].

Thus, from a theoretical point of view, the limits of reliable communication in MIMO Gaussian broadcast channels is well understood. Fortunately, the same is true if one takes a *computational* point of view. In other words, it is well known how to computationally obtain any point on the boundary of the capacity region. In a nutshell, the methodology can be explained as follows. Each point on the boundary of the capacity region is characterized by a set of covariance matrices (corresponding to how DPC is used at that particular boundary point). To obtain the desired covariance matrices, one may construct a *dual* multiple-access (MAC) system [120] where, due to the polymatroid structure of the problem, the solution can be found via standard convex optimization techniques [103].

While the aforementioned results gives a good mathematical understanding of the scheduling for downlink of cellular systems, there are many design issues and practical limitations that are not taken into account. For an efficient design of the downlink of cellular systems, these issues and limitations should be addressed. Examples of these issues follow:

- Channel state information: A crucial assumption in all the aforementioned results is that the channel coefficients to all the users is known–an assumption referred to as full channel state information (CSI)–at the transmitter. In fact, it is easy to show that with no CSI at the transmitter there is no capacity gain to be had by employing multiple antennas at the transmitter (provided all the users have single antennas) [121]. However, in practice, obtaining full CSI at the transmitter may not be feasible, especially for systems where the number of users is large and/or the users are mobile so that the channel coefficients vary rapidly with time.
- Computational complexity: When the number of users is large, the computational complexity of DPC, and even the convex optimization steps required to determine the optimal covariance matrices from the dual MAC, may become prohibitively large. Therefore there is interest in developing *simple* schemes that require *little* CSI at the transmitter, yet deliver on most of the capacity offered by the MIMO broadcast channel. One such scheme that achieves most of the throughput (sum-capacity) in MIMO broadcast channels in certain regimes is described in [121].
- Differentiated Quality of Service: In homogenous networks, the sum-rate point is a symmetrical point on the boundary of the capacity region, and so treats all the users equally. In systems which are provisioned to provide differentiated services to different users, the transmitter has to give different services (or rates) to different subsets of receivers, and yet at the same time maximize the throughput (see e.g., [122] for a discussion of the SISO case). Giving differentiated rates to users clearly means operating at non-symmetrical boundary points of the capacity region.

In the present and the following chapters, we will look at two of these issues. In
this chapter we consider different quality of service provisioning in the downlink of cellular systems. More specifically, we look at a MIMO broadcast channel where the transmitter has to provide different rates to different subset of receivers, and yet at the same time maximizes the throughput.

As mentioned earlier, this problem can, in principle, be solved since the duality to the MAC allows one to attain any point on the capacity region. However, since this solution requires full CSI at the transmitter and potentially prohibitive computations when the number of users is large, the main goal of this chapter is to develop *simple* schemes that require very *little* CSI, give *differentiated rates* to the users, and that operate *close* to the boundary of the capacity region.

We should also mention that in this chapter we will only be dealing with homogenous networks, in the sense that the SNR of the different users are assumed to have the same probability distribution. Of course many networks are, in fact, heterogenous, with different users having different distributions for their SNRs. The methodology of this paper (and many of the results, we suspect) can be straightforwardly carried over to the heterogenous case, with the caveat that the development will be much more involved and cumbersome. For this reason, and for reasons of space, although quite important in practice, we deem the heterogenous case beyond the scope of the current work. We only remark that a common practice to make a heterogenous network appear homogenous is to use an appropriate power control (after which all our results will directly apply).

In the first part of the chapter we consider channels with a small number of users (i.e., n = 2, 3). It turns out that the problem of determining a schedule that satisfies the rate constraints becomes analytically intractable as the number of users grows beyond 3. Therefore, in the second part of the chapter, we assume that the number of users is large. This is, of course, of practical interest since many systems operate in such a regime. Furthermore, it allows us to obtain explicit results in the asymptote

of large number of users, i.e., n.

7.2 Problem Formulation

We consider a fading Gaussian broadcast channel with M antennas at the transmitter and n users, each with N = 1 receive antennas.³ The channels to each user are assumed to be block fading with a coherence interval of T; in other words, the channels remain constant for T channel uses, after which they change to different independent values.⁴ Furthermore, over different users the fading is assumed to be independent. Thus, during any coherence interval, the signal to the *i*-th user, i = 1, 2, ..., n, can be written as

$$x_i(t) = \sqrt{\rho} H_i s(t) + w_i(t), \quad t = 1, \dots, T,$$
(7.1)

where $H_i \in \mathcal{C}^{1 \times M}$ is constant during the coherence interval and has i.i.d $\mathcal{CN}(0, 1)$ entries. For the purposes of this chapter it will not matter *how* the channels change from block to block, other than the fact that they vary in some stationary and ergodic way.⁵ $w_i(t)$ is additive white noise with distribution $\mathcal{CN}(0, 1)$, and $s(t) \in \mathcal{C}^{M \times 1}$ is the transmit symbol satisfying $E||s(t)||^2 = M$. In other words, the transmit power is assumed to be M. Therefore, the received signal to noise ratio (SNR) of the *i*-th user will be $E \rho |H_i s(t)|^2 = P = M\rho$; however, to simplify the notation we refer to ρ as the SNR of the users.

In terms of channel knowledge, we assume that H_i is known perfectly at the receiver. Also, we assume that the transmitter is provided with error-free information about the channel states. We denote the (average) rate of the *i*-th user, i = 1, ..., n,

³It is possible to extend our results to $N \neq 1$ in a straightforward fashion. However, for simplicity, we shall not do so here. From a practical point of view N = 1 is also very reasonable.

⁴We should remark that, although the assumption of a constant channel for T channel uses is critical, the requirement that the channels vary independently from one coherence interval to the next is not.

⁵This is because our focus is on the rate. If we had focused on other performance measures, such as delay, then how the channels vary with time would have been important.



Figure 7.1: A MISO broadcast channels with differentiated rate users.

over the different channel realizations by R_i .

In this chapter we are interested in analyzing differentiated rate scheduling schemes for broadcast systems with n users. We consider a partitioning of the users into Kgroups $\mathcal{G}_1, \ldots, \mathcal{G}_K$, as shown in Figure 7.1, where different groups require different rates from the transmitter. We also assume that the sizes of the groups are all of the same order and, hence, the cardinality of \mathcal{G}_k is $\alpha_k n$ where K and α_k 's are fixed numbers such that $\sum_{i=1}^{K} \alpha_i = 1$.

Assuming that the average rate of a user in the k-th group is denoted by $R^{k,6}$ without loss of generality we may assume $R^{1} < \ldots < R^{k}$. We further impose the constraint that the average rate of a user in the k-th group is β_{k} times the average rate of a user in the K-th group. $\beta_{1} < \ldots < \beta_{K-1} < \beta_{K} = 1$ are fixed numbers independent of n. In general we are interested in the following problem.

Problem 7.1. Consider the fading MIMO Gaussian broadcast channel described above. Let R_i denote the rate to *i*-th user and R^k denote the rate to a user in group

⁶Throughout the chapter we use superscript k to refer to any user in \mathcal{G}_k .

k. Then construct a transmission scheme such that

$$\max_{(R_1,\dots,R_n)\in\mathcal{C}_{BC}}\sum_{i=1}^n R_i$$

subject to
$$\frac{R^k}{R^K} = \beta_k, \quad k = 1, \dots, K,$$
 (7.2)

where C_{BC} is the capacity region of the broadcast channel given by Dirty Paper coding (or superposition coding in the single antenna case) [118, 40].

Clearly, the solution to Problem 7.1 is given by the intersection of the line $R^k/R^K = \beta_k$, k = 1, ..., K - 1 with the boundary of the capacity region of the broadcast channel nel. Using the duality between the capacity region of Gaussian broadcast channel and multiple access channel shown in [120, 133, 137], Problem 7.1 can be solved using bisection method in the following way.

- (a) Choose a set of rates R'^k satisfying the rate constraints of (7.2).
- (b) By appealing to the dual MAC, solve the problem, $\min \sum p_i$, subject to the rates R'^k .
- (c) If the minimum sum of powers, min $\sum p_i$, is less than total transmit power, ρM , then the rate vector is achievable. Increase the rate proportionately (according to vector $\underline{\beta} = (\beta_1, \ldots, \beta_K)$) and go to (a).
- (d) Otherwise decrease the rates proportionately (according to vector $\underline{\beta} = (\beta_1, \dots, \beta_K)$) and go to (a).

While this is all fine, the algorithm is computationally-intensive (even though the problem in step 2 is convex, it is time-consuming if n is large), requires full CSI, and finally requires implementation of DPC. Furthermore, the solution to Problem 7.1 does not give us insight into how much throughput loss we would incur by imposing the rate constraints.

In this chapter we look into devising simpler scheduling schemes such as timesharing and opportunistic transmission. We also compare the performance of different scheduling schemes in terms of their sum-rate.

It is clear that there is a price to pay in terms of throughput (sum-rate) to maintain the rate constraints. This is due to the fact that we are not working on the sumcapacity point and, therefore, the throughput will be reduced compared to the case where we had no rate constraint. In other words, we are interested in

Throughput Loss =
$$\sum_{\substack{i=1\\(R_1,...,R_n)\in \mathcal{C}_{BC}}}^n R_i - \sum_{\substack{i=1\\(R_1,...,R_n)\in \mathcal{C}_{BC} \text{ and } (7.2)}}^n R_i$$
(7.3)

for different scheduling schemes.

In Section 7.4 we consider a channel with a small number of users, namely n = 2and n = 3 and single antenna at the transmitter, i.e., M = 1. We also focus on the rate region achieved by weighted-opportunistic (WO) scheduling in which we transmit to the user that has the maximum "weighted" signal to noise ratio (SNR). We obtain the relationship between the weights for WO scheduling and the ratio of the rates. It turns out that finding an explicit relationship between the weights as a function of the given ratios is analytically intractable for n > 3, even if we allow for simplifying assumptions such as considering the low SNR regime. We further look into the throughput loss due to the rate constraints in (7.2).

In order to obtain explicit solutions, in the second part of the chapter we consider a system with many users and, rather than attempting to solve Problem 7.1 and optimization problem (7.2) directly, we propose three scheduling schemes to provide the rational rate constraints, namely weighted opportunistic beamforming (WO), time division opportunistic beamforming (TO), and superposition coding (SC) for single antenna systems. In WO, a generalization of opportunistic random beamforming scheduling, a beam is assigned to the user that has the largest "weighted" signal to noise and interference ratio (SINR) corresponding to that beam. In TO, each group has its own time slot in which the transmitter chooses the user with the best SINR from the corresponding group. Superposition coding (SC) is the scheme that achieves the information-theoretic capacity region for a single antenna broadcast channel. For each scheduling we give an explicit scheme to guarantee the rate constraints. We also analyze the throughput loss due to the rate constraints for all three different schemes. At the end of the chapter, we will provide simulation results showing the performance of the developed scheduling schemes.

7.3 Preliminary Results for MIMO GBC

In this section we will review some of the results for MIMO Gaussian broadcast channels (GBC). In particular, we will look at the sum-capacity of these channels. We will review the concept of "opportunistic" beamforming and analyze the difference between the achievable sum-rate using "opportunistic" beamforming and the sum-capacity point for large systems, i.e., large n.

7.3.1 The Capacity Region of MIMO GBC

The capacity region of MIMO Gaussian broadcast channels was recently found in [118]. It is shown that the capacity region of these channels is given by a precoding scheme at the transmitter, called Dirty Paper Coding (DPC)[119]. Furthermore, it is shown in [120, 133, 137] that the capacity region of the MIMO GBC is equal to the capacity region of a *dual* multiple access channel (MAC) with sum-power constraint equal to the GBC power constraint. Using this duality result we can write the capacity

region of the broadcast channel of (7.1) as

$$\mathcal{C}_{BC} = \bigcup_{\{\underline{P}:\sum P_i \le M\rho\}} \left\{ \underline{R} | \sum_{j \in \mathcal{A}} R_j \le \log \det(I + \sum_{j \in \mathcal{A}} H_j^* P_j H_j) \quad \forall \mathcal{A} \subseteq \{1, \dots, n\} \right\},\tag{7.4}$$

where $\underline{R} = (R_1, \ldots, R_n)$ and $\underline{P} = (P_1, \ldots, P_n)$. In particular, the ergodic sumcapacity of the channel is

$$C_{sum} = E \max_{0 \le P_i: \sum_{i=1}^{n} P_i \le M\rho} \log \det(I + \sum_{i=1}^{n} H_i^* P_i H_i),$$
(7.5)

where the expectation is over all the channel realizations.

Sum-Capacity Scaling Laws

In point-to-point multi-antenna systems, the throughput scaling is often equivalent to the "multiplexing gain," defined as $\lim_{P\to\infty} \frac{C_{sum}}{\log P}$ where C_{sum} denotes the ergodic sum-rate capacity (or throughput) of the channel achieved by coding over several coherence intervals.

In broadcast channels, as the number of users can also be large, two different throughput scaling laws can be envisioned with respect to $P = M\rho$ (or equivalently SNR) and with respect to the number of users n.

Theorem 7.1. (Large Power Regime)[124] Consider the fading MIMO Gaussian broadcast channel of Section 7.2 with received SNR of $P = M\rho$. Then, for a fixed M and n, we have

$$\lim_{P \to \infty} \frac{C_{sum}}{\log P} = M. \tag{7.6}$$

Theorem 7.2. (Large Number of Users Regime)[125] Consider the fading MIMO

Gaussian broadcast channel of Section 7.2. Then, for fixed M and ρ , we have

$$\lim_{n \to \infty} \frac{C_{sum}}{\log \log n} = M,\tag{7.7}$$

where C_{sum} refers to the maximum possible sum of the rates to all n users.

These are clearly two very different regimes and both confirm that the sum-rate is linearly scaled with the number of transmit antennas M. We argue that, from a practical perspective, the latter regime may be more interesting. There are three reasons that come to mind.

- (a) Many practical systems operate with a large number of per-cell users (n could be in the hundreds, whereas M may be no more than two, three, or four).
- (b) Significant rates can be obtained even at low to moderate SNR, $P = M\rho$.
- (c) The first gain requires channel knowledge with very high fidelity at the transmitter (indeed a fidelity that grows with the transmit power) [126], whereas the latter requires very little CSI (see, e.g., [121] and Chapter 8).

In view of the above, in this paper we will focus on the large n regime.

Note that in order to perform DPC to achieve the sum-capacity (or any other point in the capacity region), the transmitter requires exact knowledge of the channels. In the following, we briefly describe "opportunistic" beamforming, a simple scheme that achieves most of the sum-capacity in some regimes and yet requires very little CSI at the transmitter. For a complete study of this scheme, see [121, 127].

7.3.2 Opportunistic Beamforming

The main idea behind opportunistic beamforming is to exploit the multi-user diversity available in the network and transmit to the users with the best quality of reception. In this beamforming, during any coherence interval the transmitter constructs M random beams and transmits each beam to the user with the highest signal to interference plus noise ratio (SINR). Let $\phi_m(M \times 1)$, for $m = 1, \ldots, M$, be M random orthonormal vectors generated according to an isotropic distribution [149]. The transmitted signal is

$$s(t) = \sum_{m=1}^{M} \phi_m s_m(t), \qquad t = 1, \dots, T,$$

where each $s_m(t)$ is a scalar signal (with average unit power, i.e., $E |s_m|^2 = 1$) intended for one of the users. Assuming the users know their own channel coefficients (a much more reasonable assumption than the transmitter knowing *all* the channel gains to the different users), each user can compute its signal-to-interference-plus-noise-ratio (SINR) for every beam as

SINR_{*i*,*m*} =
$$\frac{|H_i \phi_m|^2}{\frac{1}{\rho} + \sum_{l \neq m} |H_i \phi_l|^2}, \qquad m = 1, \dots, M.$$
 (7.8)

If each user (or, in fact, only those users who have favorable SINRs) feeds back its best SINR and corresponding beam index to the transmitter, the transmitter can assign each beam to the user that has the best SINR for that beam (see [121] for more details.).

For this scheme the following result can be shown.

Theorem 7.3. [121] Consider the fading MIMO Gaussian broadcast channel of Section 7.2 and let C_{ob} denote the sum-rate obtained by the opportunistic beamforming technique described above. Then, for fixed P and M

$$\lim_{n \to \infty} \frac{C_{ob}}{\log \log n} = M.$$
(7.9)

However, for fixed n and M

$$\lim_{P \to \infty} \frac{C_{ob}}{\log P} = 0. \tag{7.10}$$

In other words, opportunistic beamforming is order-optimal in the large n regime, but not in the large P regime. The reason is that opportunistic beamforming is interference-dominated, and so the sum-rate does not scale with the logarithm of the power. (In fact, to obtain the multiplexing gain of M at high power requires essentially eliminating the interference, such as is done by a zero-forcing solution.)

Before we proceed further, it is useful to mention that the probability distribution function (PDF) of $SINR_{i,m}$, denoted by $f_s(x)$, can be written as [121]

$$f_s(x) = \frac{e^{-x/\rho}}{(1+x)^M} \left(\frac{1}{\rho}(1+x) + M - 1\right).$$
(7.11)

We can also calculate the cumulative distribution function (CDF) of $SINR_{i,m}$, $F_s(x)$, as

$$F_s(x) = 1 - \frac{e^{-x/\rho}}{(1+x)^{M-1}}, \qquad x \ge 0.$$
(7.12)

7.3.3 Tighter Scaling Laws

In this section we give a tighter result regarding the convergence of the sum-rate of opportunistic beamforming scheme to the sum-capacity.

Theorem 7.4. Consider the fading MIMO Gaussian broadcast channel of Section 7.2. For fixed ρ and M

$$C_{sum} = M \log \log n + M \log \rho + o(1), \qquad (7.13)$$

where C_{sum} refers to the sum-capacity of the broadcast channel with n users and o(1)is with respect to growing n. *Proof.* Using (7.5), we can write the sum-capacity as capacity as

$$C_{sum} = E \max_{\sum_{i=1}^{n} P_i = M\rho, P_i \ge 0} \log \det \left(I_M + \sum_{i=1}^{n} H_i^* P_i H_i \right)$$

$$\leq E \max_{\sum_{i=1}^{n} P_i = M\rho, P_i \ge 0} M \log \left(1 + \frac{1}{M} \operatorname{tr} \left(\sum_{i=1}^{n} H_i^* P_i H_i \right) \right),$$

$$= E \max_{\sum_{i=1}^{n} P_i = M\rho, P_i \ge 0} M \log \left(1 + \frac{1}{M} \sum_{i=1}^{n} P_i \| H_i \|^2 \right),$$

$$= E M \log \left(1 + \rho \max_i \| H_i \|^2 \right),$$
(7.14)

where we have used the Hadamard's inequality for $A \succeq 0$ that states det $(A) \leq (\frac{\operatorname{tr}(A)}{M})^{M}$ [129]. It is worth mentioning that $||H_{i}||^{2}$ s have $\chi^{2}(2M)$ distribution and they are independent. Therefore, using order statistics results [128], we can show that the random variable $\max_{i} ||H_{i}||^{2}$ with high probability behaves as $\log n$. More precisely,

$$\Pr\left(\log n + 2(M-2)\log\log n \le \max_{i} \|H_{i}\|^{2} \le \log n + 2M\log\log n\right) = 1 - O\left(\frac{1}{(\log n)^{2}}\right)$$
(7.15)

We can split the expectation in the right hand side of (7.14) in two parts; one is the expectation conditioned on $\max_i ||H_i||^2$ being less than or equal to $\log n + 2M \log \log n$, and the other is the expectation conditioned on $\max_i ||H_i||^2$ being greater than $\log n + 2M \log \log n$. This way, we can upper bound the expectation as

$$E \log \left(1 + \rho \max_{i} \|H_{i}\|^{2}\right) \leq \log \left(1 + \rho(\log n + 2M \log \log n)\right)$$
$$+ \int_{\log n + 2M \log \log n}^{\infty} \log \left(1 + \rho x\right) f_{m}(x) dx, \quad (7.16)$$

where $f_m(x)$ is the distribution of $x = \max_i ||H_i||^2$. It is quite straightforward to show that the second term in the right hand side behaves like o(1) for large n, and the first term scales like $\log \log n + \log \rho + o(1)$. This shows that

$$C_{sum} \le M \log \log n + M \log \rho + o(1).$$

To complete the proof we need a lower bound that has the same behavior. It turns out that the desired lower bound can be obtained by employing opportunistic beamforming. The required result is the next theorem. \blacksquare

Theorem 7.5. Consider the setting of Theorem 7.4. Then, if we use opportunistic beamforming

$$C_{ob} = M \log \log n + M \log \rho + o(1).$$
 (7.17)

Proof. In order to write the sum-rate achieved by random beamforming, we have to take into account the probability that there can be a user that has the best SINR for two different beams. One can lower bound the sum-rate by sending to one of the M best SINRs for each beam such that the user has not been chosen before. Using this selection, we make sure there is no user with the best SINRs for two beams; obviously, there is a rate hit, as we may transmit to the user that has the M-th best SINR as opposed to the best SINR. This, however, has little effect on the sum-rate, as the asymptotic behavior of the M-th best and the best of the SINRs is quite similar (when M is fixed and n grows). In particular,

$$C_{ob} \geq ME \log \left(1 + \rho \max_{1 \leq i \leq n}^{M} \operatorname{SINR}_{1,i}\right)$$

$$\geq M \log \left(1 + \rho (\log n - 2M \log \log n)\right) \underbrace{\operatorname{Pr} \left(\max_{1 \leq i \leq n}^{M} \operatorname{SINR}_{1,i} > \log n - 2M \log \log n\right)}_{=1 - O\left(\frac{1}{\log n}\right)}$$

$$= M \log \log n + M \log \rho + o(1),$$

where $\max^{M} x_i$ denotes the *M*'th maximum of x_i 's. Using theorem 7.4, C_{ob} matches the upper bound, and that completes the proof.

Theorems 7.4 and 7.5 imply that the difference of the sum-rate achieved by beamforming and DPC tends to zero, i.e.,

$$\lim_{n \to \infty} \left(C_{sum} - C_{ob} \right) = 0, \tag{7.18}$$

which is a much stronger result than being simply order optimal. In fact, a careful analysis of the different in the sum-rates shows that

$$(C_{sum} - C_{ob}) = O(\frac{\log \log n}{\log n}).$$

$$(7.19)$$

For the rest of this chapter we look at the differentiated rate scheduling problem.

7.4 Channels with a Small Number of Users

In this section we start with characterizing the achievable rate region using weightedopportunistic (WO) scheduling for single antenna (i.e., M = 1) systems. In WO, at each channel use we send to only the user that has the maximum weighted signal to noise ratio, i.e., the user for which

$$\max_{1 \le i \le n} \mu_i |h_i|^2. \tag{7.20}$$

Let us look at the rate transmitted to one of the users, say the first user (denoted by R_1^w). Clearly,

$$R_1^w = \int_0^\infty \log(1+x) f_s(x) \Pr(\mu_1 x_1 \ge \mu_i x_i, \ i = 2, \dots, n | x_1 = x) \mathrm{d}x,$$

where $x_i = \rho |h_i|^2$, i = 1, ..., n and $f_s(\cdot)$ is the PDF of the SNR defined in (7.11). The probability of the event inside the above integral can be written as

$$\Pr(\mu_1 x_1 \ge \mu_i x_i, \ i = 2, \dots, n | x_1 = x) = \prod_{i=2}^n (1 - e^{-\frac{\mu_1 x}{\mu_i \rho}}).$$

Replacing this expression into the formula for R_1^w yields

$$R_1^w = \int_0^\infty \log(1+x) \frac{e^{-x/\rho}}{\rho} \prod_{i=2}^n \left(1 - e^{-\frac{\mu_1 x}{\mu_i \rho}}\right) \mathrm{d}x.$$
 (7.21)

The rates to the other users can be found in a similar fashion. Let us now focus on (7.21) for the case of two and three users.

7.4.1 Case 1: Two-User Channels

In this case (7.21) simply reduces to

$$R_1^w = \int_0^\infty \log(1+x) \left(1 - e^{-\frac{\mu_1 x}{\mu_2 \rho}}\right) \frac{e^{-x/\rho}}{\rho} dx.$$
 (7.22)

Similarly, the rate for the second user is as in (7.22), with the only difference that μ_1 should be exchanged by μ_2 . We would like to find μ_1 and μ_2^7 such that the rate constraint, i.e., $\frac{R_1^w}{R_2^w} = \beta$, is satisfied.

We can simplify (7.22) as

$$R_{1}^{w} = \int_{0}^{\infty} \log(1+x) \frac{e^{-x/\rho}}{\rho} dx - \int_{0}^{\infty} \frac{\mu_{2}}{\mu_{1}+\mu_{2}} \log\left(1+\frac{\mu_{2}}{\mu_{1}+\mu_{2}}x\right) \frac{e^{-x/\rho}}{\rho} dx$$
$$= -e^{\frac{1}{\rho}} \operatorname{Ei}\left(-\frac{1}{\rho}\right) + \frac{\mu_{2}}{\mu_{1}+\mu_{2}} e^{\frac{\mu_{1}+\mu_{2}}{\rho\mu_{2}}} \operatorname{Ei}\left(-\frac{\mu_{1}+\mu_{2}}{\rho\mu_{2}}\right),$$
(7.23)

where we used the definition of the exponential integral function defined as -Ei(-x) =

⁷It is worth mentioning that without loss of generality we can set the sum of μ_i s equal to 1.

 $\int_{1}^{\infty} \frac{e^{-tx}}{t} dt$. We can similarly write the rate for the second user as

$$R_2^w = -e^{\frac{1}{\rho}} \operatorname{Ei}\left(-\frac{1}{P}\right) + \frac{\mu_1}{\mu_1 + \mu_2} e^{\frac{\mu_1 + \mu_2}{\rho\mu_1}} \operatorname{Ei}\left(-\frac{\mu_1 + \mu_2}{\rho\mu_1}\right).$$
(7.24)

In order to find the μ_1 that satisfies the rate constraint of (7.2), we need to solve the following non-polynomial equation:

$$\frac{-e^{\frac{1}{\rho}}\mathrm{Ei}\left(-\frac{1}{\rho}\right) + (1-\mu_{1})e^{\frac{1}{\rho(1-\mu_{1})}}\mathrm{Ei}\left(-\frac{1}{\rho(1-\mu_{1})}\right)}{-e^{\frac{1}{\rho}}\mathrm{Ei}\left(-\frac{1}{\rho}\right) + \mu_{1}e^{\frac{1}{\rho\mu_{1}}}\mathrm{Ei}\left(-\frac{1}{\rho\mu_{1}}\right)} = \beta.$$
(7.25)

It does not seem that (7.25) has a closed form solution because it involves the exponential integral. We can numerically evaluate β versus μ_1 , as shown in Figure 7.2. Since $\mu_1 + \mu_2 = 1$, here we assume that μ_1 is varying between zero and one.

The generalization to a system with n > 2 users is straightforward: We simply need to expand the products in (7.21) into a summation of exponentials and then repeatedly use the exponential integral. Applying the rate constraints will lead to a non-polynomial system of equations with n-1 equalities and n-1 variables. Although it may be possible to solve numerically such a system of equations, it gives us little insight into the problem.

In order to find more explicit results, in the next section we simplify the system (7.25) by assuming that ρ is small (low SNR regime).

7.4.2 Case 2: Low SNR Regime

Assuming that the system is working in the regime of small ρ , the instantaneous rate can be approximated to first order as $\rho |h_i|^2$ (instead of $\log(1 + \rho |h_i|^2)$). It turns out that this leads to a system of polynomial equations, which can be theoretically dealt with using Groebner bases.⁸ Given a finite set of multivariate polynomials over a

⁸Groebner bases method was introduced by Bruno Buchberger in 1965 [130].



Figure 7.2: β versus μ_1 for a channel with n = 2.

field, a new set of polynomials with good properties can be found by an algorithm of Buchberger, called the Groebner basis, which can be used to find the solutions of the polynomial system. This method has been extensively studied, developed, and has been implemented on all major computer algebra systems.

In a channel with two users, the rates can be written as

$$R_1^w = \int_0^\infty x(1 - e^{-\frac{\mu_1 x}{\mu_2 \rho}}) \frac{e^{-x/\rho}}{\rho} dx = \rho(2 - \mu_1)\mu_1.$$
(7.26)

Similarly,

$$R_2^w = \rho(1 - \mu_1^2). \tag{7.27}$$

Therefore, the boundary of the rate region is characterized by (7.26) and (7.27). This parametric characterization can be made explicit by eliminating μ_1 from (7.26) and (7.27) as

$$\frac{1}{4} \left(1 - \frac{R_2^w}{\rho} + \frac{R_1^w}{\rho} \right)^2 = 1 - \frac{R_2^w}{\rho}.$$
(7.28)

Now, given the ratio of the rates and (7.28), we can easily obtain μ_1 such that the ratio of the rates will be equal to β .

This framework can be easily generalized to the case of more than two users. We omit the details and simply state that for n = 3, we may write the rates as

$$\begin{aligned} \frac{R_1^w}{\rho} &= 1 - \frac{\mu_2}{\mu_1 + \mu_3} - \frac{\mu_3}{\mu_1 + \mu_2} + \frac{\mu_2\mu_3}{\mu_2\mu_3 + \mu_1\mu_3 + \mu_1\mu_2},\\ \frac{R_2^w}{\rho} &= 1 - \frac{\mu_1}{\mu_2 + \mu_3} - \frac{\mu_3}{\mu_2 + \mu_1} + \frac{\mu_1\mu_3}{\mu_2\mu_3 + \mu_1\mu_3 + \mu_1\mu_2},\\ \frac{R_3^w}{\rho} &= 1 - \frac{\mu_1}{\mu_2 + \mu_3} - \frac{\mu_2}{\mu_3 + \mu_2} + \frac{\mu_1\mu_2}{\mu_2\mu_3 + \mu_1\mu_3 + \mu_1\mu_2},\\ 1 &= \mu_1 + \mu_2 + \mu_3, \end{aligned}$$

To find the explicit characterization of the rate region, we have to eliminate the μ_i s from the above set of polynomial equations. This can be done with the aid of Groebner bases using, say, Mathematica. However, the complexity of the algorithm becomes formidable, even for the case of n = 3.

On the other hand, it is possible to attempt to solve the above system of equations numerically. This, in principle, will allow us to map a set of rate constraints to a set of weights for the schedule. However, as mentioned earlier, this gives little insight and, moreover, it too can be quite complex for large n.

7.4.3 Throughput Loss

As mentioned earlier, it is clear that there is a price to pay in terms of throughput (sum-rate) to maintain the rate constraints. In this part, we numerically evaluate the throughput degradation due to imposing the rate constraint of β for a channel with two users. Assuming that the rate of the first user is β times the rate of the second user, Figure 7.3 shows the ratio of the throughput of the WO scheduling over the sum-rate capacity versus β .



Figure 7.3: Ratio of the throughput with rate constraints over the sum-rate capacity versus β for a channel with n = 2.

Clearly, when β equals one, the WO scheduling achieves the sum-rate capacity, and, therefore, throughput will be equal to the sum-rate capacity. As β increases, the throughput loss will be more. It is quite interesting to observe that even for very large β (e.g., close to 70), the throughput is above 80 percent of the sum-rate capacity. Therefore, the throughput does not seem to be too much affected by the differentiated rate scheduling. In the next section, we look into this throughput loss in the regime of large number of users.

7.5 Channels with Many Users

In Section 7.4 we observed that finding an explicit relationship between β_i s and the μ_i s in WO scheduling becomes very complicated, even for the case of n = 3.

Therefore, for the remainder of the chapter, we look into the regime of large number of users. We first characterize the optimal differentiated opportunistic beamforming scheme for maximizing the sum-rate and at the same time satisfy the rational rate constraints.

Then, we consider WO and TO scheduling schemes for multiple antenna GBCs and analyze the performance of superposition coding for single antenna systems for large number of users. It will turn out that having a large number of users will simplify the derivations and lead to explicit results.

7.5.1 Optimal Differentiated Opportunistic Beamforming Scheme

In this section we characterize the *optimal random opportunistic* beamforming scheme for providing different rates to different groups of users. By opportunistic, we mean that the transmitter based on its available knowledge about the channel realization, allocates a fraction of time to transmit each beam to each user. The goal is to maximize the sum-rate and at the same time satisfy the rational rate constraints of (7.2).

Let $\underline{X}^{(m)} = (\text{SINR}_{i,m}, 1 \leq i \leq n)$ be the vector of instantaneous SINRs for transmitted beam m of different users. We define a general opportunistic schedule with parameters $\tau_i(\underline{X}^{(m)})$ for i = 1, ..., n, where $\tau_i(\underline{X}^{(m)})$ is the fraction of the coherence interval that the transmitter assigns beam m to user i, given that the instantaneous SINRs for different receivers correspond to entries of $\underline{X}^{(m)}$. Clearly,

$$\sum_{i=1}^{n} \tau_i(\underline{X}^{(m)}) = 1, \qquad \tau_i(\underline{X}^{(m)}) \ge 0$$

As an example, for the opportunistic beamforming scheme of Section 7.3.2, we have that for each beam m

$$\tau_i(\underline{X}^{(m)}) = \mathbb{1}\left(i = \arg\max_j \underline{X}_j^{(m)}\right),$$

where $\mathbb{1}(a = b)$ is one if a = b and zero otherwise. It is not hard to persuade oneself that WO and TO beamforming are also special cases of the schedule defined here.

With this definition, the average rate assigned to user i is

$$R_i = M \mathbb{E}\left(\tau_i(\underline{X}^{(m)})\log(1+\underline{X}_i^{(m)})\right),$$

where the expectation is over the distribution of SINRs for a particular beam, say, m at different users, i.e., $\underline{X}^{(m)}$. Therefore, we can write Problem 7.1 in terms of $\tau_i(\cdot)$'s as follows

$$\max_{\tau_i(\underline{X}^{(m)})} M \sum_{i=1}^n \mathbb{E}\left(\tau_i(\underline{X}^{(m)}) \log(1 + \underline{X}_i^{(m)})\right),$$
(7.29)

subject to

$$E\left(\tau_{i}(\underline{X}^{(m)})\log(1+\underline{X}_{i}^{(m)})\right) = \beta_{i}E\left(\tau_{n}(\underline{X}^{(m)})\log(1+\underline{X}_{n}^{(m)})\right) \qquad i = 1, \dots, n-1$$

$$\tau_{i}(\underline{X}^{(m)}) \geq 0 \qquad \forall \underline{X}^{(m)} \in \mathbb{S}^{n}, \ i = 1, \dots, n$$

$$\sum_{i=1}^{n} \tau_{i}(\underline{X}^{(m)}) = 1 \qquad \forall \underline{X}^{(m)} \in \mathbb{S}^{n}, \qquad (7.30)$$

where without of loss of generality we have assumed that user n is in group \mathcal{G}_K and for user i in group k, β_i refers to the β_k defined in Section 7.2.

Looking at (7.29) and (7.30), we can see that the maximization problem is a linear program in $\tau_i(\cdot)$. The following theorem describes the optimal choice of functions $\tau_i(\cdot)$.

Theorem 7.6. Consider a MIMO Gaussian broadcast channel with the assumptions stated in Section 7.2. Then, the optimal differentiated opportunistic scheduling in the sense defined in (7.29) and (7.30) is given by

$$\tau_i(\underline{X}^{(m)}) = \mathbb{1}\left(i = \arg\max_j \gamma_j \log(1 + \underline{X}_j^{(m)})\right),\,$$

where $\gamma_i s$ are scalar chosen to satisfy the rational rate constraints of (7.30). In other

words, the optimal differentiated opportunistic scheduling is to assign beam m to the user with the highest "weighted" instantaneous rate in decoding information symbol carried on that beam.

Proof. We use dual form of the linear program in (7.29) and (7.30) to prove the theorem. The dual linear program can be written as

$$\min_{\eta_i(\underline{X}^{(m)}),\lambda_i,\,\zeta(\underline{X}^{(m)})} M \mathbb{E}\left(\zeta(\underline{X}^{(m)})\right)$$

subject to

$$\begin{aligned} \eta_i(\underline{X}^{(m)}) &\leq 0 \qquad \forall \underline{X}^{(m)} \in \mathbb{S}^n, \, i = 1, \dots, n \\ (1+\lambda_i)\log(1+\underline{X}_i^{(m)}) &= \eta_i(\underline{X}^{(m)}) + \zeta(\underline{X}^{(m)}) \qquad \forall \underline{X}^{(m)} \in \mathbb{S}^n, \, i = 1, \dots, n-1 \\ (1-\sum_{i=1}^{n-1}\lambda_i\beta_i)\log(1+\underline{X}_n^{(m)}) &= \eta_n(\underline{X}^{(m)}) + \zeta(\underline{X}^{(m)}), \end{aligned}$$

where λ_i , $\eta_i(\underline{X}^{(m)})$, and $\zeta(\underline{X}^{(m)})$ represent dual variables for the first, second, and third constraint given in (7.30). We can further simplify the above minimization problem to

$$\min_{\lambda_i,\,\zeta(\underline{X}^{(m)})} M \mathcal{E}\left(\zeta(\underline{X}^{(m)})\right) \tag{7.31}$$

subject to

$$(1+\lambda_i)\log(1+\underline{X}_i^{(m)}) \leq \zeta(\underline{X}^{(m)}) \quad \forall \underline{X}^{(m)} \in \mathbb{S}^n, i = 1, \dots, n-1$$
$$(1-\sum_{i=1}^{n-1}\beta_i\lambda_i)\log(1+\underline{X}_n^{(m)}) \leq \zeta(\underline{X}^{(m)}) \quad \forall \underline{X}^{(m)} \in \mathbb{S}^n.$$
(7.32)

Now, for a fixed choice of λ_i s, the minimizing $\zeta(\underline{X}^{(m)})$ is $\max_i \gamma_i \log(1 + \underline{X}_i^{(m)})$, where

we have defined

$$\gamma_{i} = \begin{cases} 1 + \lambda_{i} & 1 \le i \le n - 1, \\ 1 - \sum_{i=1}^{n-1} \beta_{i} \lambda_{i} & i = n. \end{cases}$$
(7.33)

Looking at the KKT conditions [131] at the optimal point $(\eta_i^{\star}(\cdot), \zeta^{\star}(\cdot), \tau_i^{\star}(\cdot), \lambda_i^{\star})$, we should have

$$\begin{split} \eta_i^{\star}(\underline{X}^{(m)})\tau_i^{\star}(\underline{X}^{(m)}) &= 0 \qquad \forall \, \underline{X}^{(m)} \in \mathbb{S}^n, \, 1 \leq i \leq n \\ \zeta^{\star}(\underline{X}^{(m)}) &= \max_j (\gamma_j^{\star} \log(1 + \underline{X}_j^{(m)})) \qquad \forall \, \underline{X}^{(m)} \in \mathbb{S}^n \\ \tau_i^{\star}(\underline{X}^{(m)}) &= \mathbbm{1} \left(i = \arg\max_j (\gamma_j^{\star} \log(1 + \underline{X}_j^{(m)})) \right), \qquad \forall \, \underline{X}^{(m)} \in \mathbb{S}^n, \, 1 \leq i \leq n \end{split}$$

where γ_j^{\star} is defined from λ_j^{\star} according to (7.33). Now, based on the above conditions, it is clear that the maximizing solution is the one that assigns each beam to the user with maximum $\gamma_j^{\star} \log(1 + \underline{X}_j^{(m)})$, and this completes the proof.

Remark 7.7. It is interesting to note that since $\log(1+x) \approx x$ for $x \ll 1$, in the low SNR regime, the optimal schedule becomes equivalent to WO beamforming.

It turned out that analyzing the above optimal scheme does not lead to explicit solution for $\gamma - j$'s. Therefore, we look at the performance of TO and WO beamforming next and devise explicit schemes for these beamforming schedules.

7.5.2 Time-Division Opportunistic (TO) Beamforming

The simplest scheme to give differentiated rates to different users is to assign different lengths of channels uses to different users, i.e., time-sharing. This should be done opportunistically to maximize the sum-rate. In particular, we divide each coherence interval into K slots of duration t_k each, $k = 1, \ldots, K$. During the k-th subinterval, the transmitter performs opportunistic beamforming to *only* the $\alpha_k n$ users in the k-th group. If α_k s are fixed and n grows, it is not hard to convince oneself that to satisfy the rational rate constraints, we must have

$$\frac{t_k}{T} = \frac{\alpha_k \beta_k}{\sum_{l=1}^K \alpha_l \beta_l}, \quad k = 1, \dots, K.$$
(7.34)

Intuitively, since each group has a size of order n, the sum-rate for each group scales like $M \log \log n$. Therefore, in order to satisfy the rate constraints, we need to only take into account the size of the group, i.e., α_i , and the rate ratio β_i . Therefore, using (7.34), we can easily show that

$$\lim_{n \to \infty} \frac{R^k}{R^K} = \beta_k, \qquad k = 1, \dots, K - 1.$$
 (7.35)

We can state the following result that quantifies the sum-rate loss due to the rate constraints and also the sum-optimality of the scheduling.

Theorem 7.8. Consider the fading MIMO Gaussian broadcast channel of Section 7.2. Let M, ρ , α_k , and β_k be fixed, and let the subintervals be chosen as (7.34). Then, the rational rate constraints are met and

$$\lim_{n \to \infty} \left(C_{sum} - C_{tdob} \right) = \Theta\left(\frac{1}{\log n}\right),\tag{7.36}$$

where C_{tdob} represents the sum-rate for the time-division opportunistic scheme.

Proof. That the rational rate constraints are met is fairly straight forward. As for the sum-rate we have,

$$C_{tdob} = \sum_{k=1}^{K} \frac{\alpha_k \beta_k}{\sum_{l=1}^{K} \alpha_l \beta_l} \left(M \log \log n \alpha_k + M \log \rho + o(1) \right)$$
$$= \sum_{k=1}^{K} \frac{\alpha_k \beta_k}{\sum_{l=1}^{K} \alpha_l \beta_l} \left(M \log \log n + \underbrace{\log \left(1 + \frac{\alpha_k}{\log n}\right)}_{=o(1)} + M \log \rho + o(1) \right)$$
$$= M \log \log n + M \log \rho + o(1).$$

259

$$C_{ob} - C_{tdob} = \int_0^\infty \underbrace{M \log(1+x) n f_s(x) \left(F_s(x)^{n-1} - \sum_{k=1}^K \frac{\alpha_k t_k}{T} F_s(x)^{\alpha_k n-1} \right)}_{h(x)} \mathrm{d} x.$$

Define

$$l^{-} = \rho(\log n - (M+3)\log\log n), \qquad l^{+} = \rho(\log n + (M+3)\log\log n).$$
(7.37)

We break the integral into the following three regions: $\mathcal{I}_1 = [0, l^-], \mathcal{I}_2 = [l^-, l^+]$ and $\mathcal{I}_3 = [l^+, \infty]$. It can be easily checked that

$$\left| \int_{\mathcal{I}_1} h(x) \mathrm{d} x \right| = O(\log \log n \int_{\mathcal{I}_1} Mn f_s(x) F_s(x)^{\min\{\alpha_k\}n-1} \mathrm{d} x)$$

$$= O(\log \log n F_s(x)^{\min\{\alpha_k\}n} |_0^{l^-})$$
(7.38)

$$= O\left(\log \log n \left(1 - \frac{(\log n)^{M+3}}{n(1+\rho \log n)^{M-1}}\right)^{\min\{\alpha_k\}n}\right) \\ = O(\log \log n \cdot e^{-(\log n)^4}).$$

As for the third region, we have

$$\begin{aligned} \left| \int_{\mathcal{I}_{3}} h(x) \mathrm{d} x \right| &\leq \int_{\mathcal{I}_{3}} Mn \log(1+x) f_{s}(x) \mathrm{d} x \\ &\leq \frac{1}{(\rho \log n)^{M-1}} \int_{\mathcal{I}_{3}} Mn \log(1+x) \frac{e^{-x/\rho}}{\rho} \mathrm{d} x \\ &= O\left(\frac{n}{(\rho \log n)^{M-1}} \left(-\log(1+x) e^{-\frac{x}{\rho}} |_{l^{+}}^{\infty} \right) - e^{\frac{1}{\rho}} \mathrm{Ei} \left(-\frac{l^{+}}{\rho} \right) \right), \end{aligned}$$
(7.39)

where $\operatorname{Ei}(-x) = -\int_x^\infty \frac{e^{-y}}{y} dy$ is the exponential integral. For large x we have

 $\operatorname{Ei}(-x) = \Theta(-\frac{e^{-x}}{x})$. Using this identity in the previous inequality gives

$$\left|\int_{\mathcal{I}_3} h(x) \mathrm{d} x\right| = O(\frac{\log \log n}{(\log n)^{2M+2}}).$$

Now we move on to the integral over \mathcal{I}_2 . Using integration by parts we have

$$\int_{\mathcal{I}_2} h(x) dx = \log(1+x) (F(x)^n - \sum_{k=1}^K \frac{t_k}{T} F(x)^{\alpha_k n}) |_{l^-}^{l^+} -M \int_{\mathcal{I}_2} \frac{F(x)^n - \sum_{k=1}^K \frac{t_k}{T} F(x)^{\alpha_k n}}{1+x} dx.$$

We can further simplify the integral to

$$\begin{split} \int_{\mathcal{I}_2} h(x) \mathrm{d} \, x &= \Theta(\frac{\log \log n}{(\log n)^{2M+2}}) \\ &\quad -\frac{M}{\rho \log n} \left(1 - O(\frac{\log \log n}{\log n})\right) \int_{l^-}^{l^+} \left(F(x)^n - \sum_{k=1}^K \frac{t_k}{T} F(x)^{\alpha_k n}\right) \mathrm{d} \, x. \end{split}$$

Next, we show that the last integral in the above equation is of order constant. Suppose $i = \arg \min_k \{\alpha_k\}$. Note that with this condition $\alpha_i \leq \frac{1}{2}$. It can be verified that

$$\begin{aligned} &\int_{l^{-}}^{l^{+}} \left(F(x)^{n} - \sum_{k=1}^{K} \frac{t_{k}}{T} F(x)^{\alpha_{k}n} \right) \mathrm{d} \, x = \Theta \left(\int_{l^{-}}^{l^{+}} \left(F(x)^{n} - F(x)^{\alpha_{i}n} \right) \mathrm{d} \, x \right) \end{aligned} \tag{7.40} \\ \stackrel{(a)}{=} &\Theta \left(\int_{-4\log\log n}^{(2M+2)\log\log n} \left(1 - \frac{e^{-y}}{n(1 + O(\frac{\log\log n}{\log n}))^{M-1}} \right)^{n} - \left(1 - \frac{e^{-y}}{n(1 + O(\frac{\log\log n}{\log n}))^{M-1}} \right)^{\alpha_{i}n} \mathrm{d} \, y \right) \\ \stackrel{(b)}{=} &\Theta \left(\int_{-4\log\log n}^{(2M+2)\log\log n} \exp\left(-\nu(n)e^{-y} \right) - \exp\left(-\nu(n)\alpha_{i}e^{-y} \right) \mathrm{d} \, y \right) \\ \stackrel{(c)}{=} &\Theta \left(\int_{\frac{1}{(\log n)^{2M+2}\nu(n)}}^{(\log n)^{4}\nu(n)} \frac{e^{-z} - e^{-\alpha_{i}z}}{z} \mathrm{d} \, z \right) \\ \stackrel{(d)}{=} &\Theta(\ln\alpha_{i} + \frac{\log\log n}{\log n}) = \Theta(1), \end{aligned}$$

where (a) follows by a change of variable $y = \frac{x}{\rho} - \log n + (M-1) \log \log n$, and (b)

holds because

$$(1 - \frac{e^{-y}}{n(1 + \Theta(\frac{\log \log n}{\log n}))^{M-1}})^n = \Theta\left((1 - O(\frac{(\log n)^4}{n}))\exp(-\nu(n)e^{-y})\right),$$

for large n, $\nu(n) = (1 + O(\frac{\log \log n}{\log n}))^{M-1}$, and for y in the range defined in the integral. (c) follows after change of variable of $z = \nu(n)e^{-y}$, and (d) is a direct consequence of the following asymptotic expansions for $-\text{Ei}(-x) = \int_x^{\infty} \frac{e^{-t}}{t} dt$ for small and large x, respectively [129]:

$$\operatorname{Ei}(-x) = \ln x + \gamma_0 + O(x), \qquad x \ll 1, \quad \operatorname{Ei}(-x) = -\frac{e^{-x}}{x}(1 + O(\frac{1}{x})) \qquad x \gg 1.$$

Putting Equations (7.38) to (7.40) together we get

$$(C_{ob} - C_{tdob}) = \Theta(\frac{1}{\log n}).$$

This Theorem implies that rate constraint does not drastically degrade the sumrate of the systems as long as the group sizes are order n and n is large. It also raises the question whether the difference between the sum-rate of TO beamforming and sum-rate capacity can be improved or not.

7.5.3 Weighted Opportunistic (WO) Beamforming

In the weighted opportunistic scheme we weigh the SINR of each user according to its group by μ_k , k = 1, ..., K. Then, during each coherence interval, the transmitter assigns the M random beams to the M users that have the largest *weighted* SINR. More specifically, let the SINR corresponding to beam m and user i be denoted by $x_{i,m}$. The distribution of $x_{i,m}$ was given in (7.11). With this notation, beam m is assigned to the user l_0 from group k_0 such that

$$(k_0, l_0) = \arg \max_{(k,i): i \in \mathcal{G}_k} \mu_k x_{i,m}.$$

Equivalently, one can see that beam m assigned to group k_0 such that

$$k_0 = \arg \max_{1 \le k \le K} \mu_k \max_{i \in \mathcal{G}_k} x_{i,m}.$$
(7.41)

In the WO beamforming scheme there are two questions to be answered: first, how to determine the weights such that the rational rate constraints are met (Here, unlike the TO case, the answer is not trivial), and second, what is the rate loss compared to the unconstrained sum-rate capacity of the broadcast channel itself?

The following theorems settle the aforementioned questions.

Theorem 7.9. Consider the fading MIMO Gaussian broadcast channel of Section7.2. Consider the WO beamforming scheme with

$$\mu_k = 1 + \frac{\log \beta_k}{\log n - (M - 1) \log \log n}.$$
(7.42)

Assuming, M, ρ , $\alpha_k s$ and $\beta_k s$ are fixed, we have

$$\lim_{n \to \infty} \frac{R^k}{R^K} = \beta_k, \quad k = 1, \dots, K.$$
(7.43)

Proof. Without loss of generality let us look at the average transmitted rate to the first user in the first group (denoted by R^1). Clearly,

$$R^{1} = M \int_{0}^{\infty} \log(1+x) f_{s}(x) \Pr\left(\mu_{1} x_{1,m} \ge \mu_{k} x_{i,m}, \quad \forall k, i \text{ s.t. } 1 \le k \le K, i \in \mathcal{G}_{k} \middle| x_{1,m} = x \right) \mathrm{d} x$$

Using the independence of the SINRs for different users, the probability of the event defined inside the above integral can be written as

$$\prod_{k=1}^{K} \prod_{i \in \mathcal{G}_k, i \neq 1} \Pr\left(x_{i,m} \le \frac{\mu_1}{\mu_k}x\right) = F_s(x)^{\alpha_1 n - 1} \prod_{k=2}^{K} F(\frac{\mu_1}{\mu_k}x)^{\alpha_k n}.$$

Accordingly, we have

$$R^{1} = M \int_{0}^{\infty} \underbrace{\log(1+x)f_{s}(x)F_{s}(x)^{\alpha_{1}n-1}\prod_{k=2}^{K}F_{s}(\frac{\mu_{1}}{\mu_{k}}x)^{\alpha_{k}n}}_{h_{1}(x)} \mathrm{d} x$$

We further split the above integral to three integrals over the intervals $\mathcal{I}_1 = [0, l^-]$, $\mathcal{I}_2 = [l^-, l^+]$, and $\mathcal{I}_3 = [l^+, \infty]$, where l^- and l^+ are defined in (7.37). The integral over the first region can be written as

$$\int_{\mathcal{I}_1} h_1(x) \mathrm{d} x \stackrel{(a)}{=} O(\log \log n \int_{\mathcal{I}_1} M f_s(x) F_s(x)^{\alpha_1 n - 1} \mathrm{d} x)$$
$$= O(\frac{\log \log n}{n} F_s(x)^{\alpha_1 n} |_0^{l^-})$$
$$= O(\log \log n \cdot e^{-(\log n)^4}),$$

where (a) follows from the fact that $0 \leq F_s(x) \leq 1$. Similarly, it can be shown that for the integral over the third region we have

$$\int_{\mathcal{I}_3} h_1(x) \mathrm{d} \, x = O(\frac{\log \log n}{(\log n)^{2M+2}}).$$

Hence, the main contribution is due to the integral over the second interval. Looking

at the behavior of $F_s(\frac{\mu_1}{\mu_k}x)^{\alpha_k n}$ for $x \in \mathcal{I}_2$ we have

$$F_{s}(\frac{\mu_{1}}{\mu_{k}}x) = F_{s}\left(\frac{\mu_{1}}{\mu_{k}}(y+\rho(\log n-(M-1)\log\log n))\right)$$

$$\stackrel{(a)}{=} F_{s}\left(\rho\left(y(1+O(\frac{1}{\log n}))+\log(\frac{\beta_{1}}{\beta_{k}})+\log n-(M-1)\log\log n\right)+O(\frac{1}{\log n})\right)$$
(7.44)

where (a) follows by defining $y = \frac{x}{\rho} - \log n + (M-1) \log \log n$, where $y \in \mathcal{I}'_2 = [(2M + 2) \log \log n, -4 \log \log n]$, and using the definition of μ_k 's in the theorem statement. Further simplification gives

$$\begin{pmatrix} F_s\left(\frac{\mu_1}{\mu_k}x\right) \end{pmatrix}^{\alpha_k n} \stackrel{(b)}{=} \left(1 - \frac{\beta_k}{\beta_1} \frac{e^{-y(1+O(\frac{1}{\log n}))}(1-O(\frac{\log\log n}{\log n}))}{n\rho^{M-1}} \right)^{\alpha_k n}$$

$$= \exp\left(\alpha_k n \log\left(1 - \frac{\beta_k}{\beta_1} \frac{e^{-y(1+O(\frac{1}{\log n}))}(1-O(\frac{\log\log n}{\log n}))}{n\rho^{M-1}}\right) \right)$$

$$= (1 - O(\frac{\log\log n}{\log n})) \exp\left(-\frac{\alpha_k \beta_k}{\beta_1 \rho^{M-1}} e^{-y}\right),$$

where (b) follows by substituting $F_s(x)$ in (7.44) with its expression from (7.12) and noting that for $y \in \mathcal{I}'_2$ we have $(1 + \frac{y}{\log n})^{1-M} = 1 - O(\frac{\log \log n}{\log n})$. The last equality is obtained by expanding the logarithm. Using similar arguments, it can be shown that for $y \in \mathcal{I}'_2$,

$$f_s(\rho y + \rho(\log n - (M-1)\log\log n)) = (\frac{1}{\rho^M n} - O(\frac{\log\log n}{n\log n}))e^{-y}.$$

Using the above asymptotic expressions for $F_s(\frac{\mu_1}{\mu_k}x)^{\alpha_k n}$ and $f_s(x)$, the integral over

 \mathcal{I}_2 can be written as

$$\begin{split} \int_{\mathcal{I}_2} h_1(x) \mathrm{d} x &= M \frac{\log \log \rho n}{n \rho^{M-1}} \left(1 + O(\frac{\log \log n}{\log n}) \right) \int_{\mathcal{I}'_2} e^{-y} \exp\left(-\frac{\sum_{k=1}^k \alpha_k \beta_k}{\beta_1 \rho^{M-1}} e^{-y}\right) \mathrm{d} y \\ &= M \frac{\log \log \rho n}{n \rho^{M-1}} \left(1 + O(\frac{\log \log n}{\log n}) \right) \int_{(\log n)^{-2M-2}}^{(\log n)^4} \exp\left(-\frac{\sum_{k=1}^k \alpha_k \beta_k}{\beta_1 \rho^{M-1}} z\right) \mathrm{d} z \\ &\stackrel{(a)}{=} M \frac{\log \log \rho n}{n \rho^{M-1}} \left(1 + O(\frac{\log \log n}{\log n}) \right) \frac{e^{-A_0 z}}{A_0} \Big|_{(\log n)^4}^{(\log n)^{-2M-2}} \\ &= M \frac{\log \log \rho n}{\rho^{M-1} A_0 n} (1 - O(\frac{\log \log n}{n \log n})) \\ &= M \frac{\beta_1}{\sum_{k=1}^K \beta_k \alpha_k} \frac{\log \log \rho n}{n} + O(\frac{\log \log n}{n \log n}), \end{split}$$

where $A_0 = -\frac{\sum_{k=1}^k \alpha_k \beta_k}{\beta_1 \rho^{M-1}}$ in (a) and $z = e^{-y}$.

Putting these integrals together we have

$$R^{1} = M \frac{\beta_{1}}{\sum_{k=1}^{K} \beta_{k} \alpha_{k}} \frac{\log \log \rho n}{n} + O(\frac{\log \log n}{n \log n}).$$

Similarly, we can check that for a user in group k the average transmitted rate is

$$R^k = M \frac{\beta_k}{\sum_{k=1}^K \beta_k \alpha_k} \frac{\log \log \rho n}{n} + O(\frac{\log \log n}{n \log n}),$$

and hence the rational constraints are readily obtained. Also notice that achievable sum-rate using WO beamforming is

$$C_{wob} = \sum_{k=1}^{K} \alpha_k n R^k = Mn \frac{\sum_{k=1}^{K} \beta_k \alpha_k}{\sum_{j=1}^{K} \beta_j \alpha_j} \frac{\log \log \rho n}{n} + O(\frac{\log \log n}{\log n}) = M \log \log \rho n + O(\frac{\log \log n}{\log n})$$

Comparing this with the sum-capacity we see that

$$\lim_{n \to \infty} (C_{wob} - C_{ob}) = 0.$$

Remark 7.10. Theorem 7.9 asserts that the average rates of users are quite sensitive to the change of $\mu_i s$. In order to further understand the impact of a change in $\mu_i s$ on the rates, we consider a two-group system. Following the methodology in the proof of Theorem 7.9, we can prove the following results. If

$$\frac{\mu_1}{\mu_2} = 1 - o\left(\frac{1}{\log n}\right)$$

then,

$$\lim_{n \to \infty} \frac{R^1}{R^2} = 1,$$

where R^k is the average rate provided to a user in group k. Moreover, if

$$\frac{\mu_1}{\mu_2} = c < 1$$

then,

$$\lim_{n \to \infty} \frac{R^1}{R^2} = 0,$$

where c is a constant independent of n.

The above Theorem also shows that, as in the case of TO beamforming, WO beamforming achieves the sum-rate of the unconstrained broadcast channel as $n \rightarrow \infty$. As a matter of fact, the simulation results of Section 7.6 suggest that throughput of the WO beamforming is much closer to the sum-capacity than the throughput of the TO beamforming.

In the next lemma we show that for a two group case, the difference between the throughput of WO and the sum-rate of opportunistic beamforming with no rational rate constraints scales at most inversely proportional to $\log n$. Carrying the proof over to more than two groups is a possible, however a cumbersome, task.

Lemma 7.11. Consider the setting of Theorem 7.9 and assume that there are two

groups present in the network. Let C_{wob} denote the sum of the rates obtained by the weighted opportunistic beamforming scheme. Then, for any group size $\alpha_i n$, and any β_i 's for i = 1, 2, the sum-rate loss of WO tends to zero with a convergence rate at least as fast as $\frac{1}{\log n}$, i.e.,

$$\lim_{n \to \infty} \left(C_{sum} - C_{wob} \right) = O\left(\frac{1}{\log n}\right).$$
(7.45)

Proof. For each beam m, let x_m and y_m denote the maximum SINR in groups \mathcal{G}_1 and \mathcal{G}_2 , respectively. Furthermore, assume that $\beta_1 \geq \beta_2$. We can write the loss in the throughput as

$$C_{ob} - C_{wob} = \sum_{m=1}^{M} \mathbb{E} \log \left(\frac{1 + x_m \mathbb{1}(x_m \ge y_m) + y_m \mathbb{1}(x_m < y_m)}{1 + x_m \mathbb{1}(x_m \ge \frac{\mu_2}{\mu_1} y_m) + y_m \mathbb{1}(x_m < \frac{\mu_2}{\mu_1} y_m)} \right).$$

The difference can be further simplified to get

$$C_{ob} - C_{wob} = \sum_{m=1}^{M} \mathbb{E} \left(\log(\frac{1+y_m}{1+x_m}) | y_m \ge x_m \ge \frac{\mu_2}{\mu_1} y_m \right).$$

Note that for $y_m \ge x_m \ge \frac{\mu_2}{\mu_1} y_m$ we have

$$\frac{1+y_m}{1+x_m} \le \frac{1+y_m}{1+\frac{\mu_2}{\mu_1}y_m} \le \frac{\mu_1}{\mu_2}.$$

Therefore, the sum-rate loss is upper-bounded as

$$C_{ob} - C_{wob} \le \sum_{m=1}^{M} \log(\frac{\mu_1}{\mu_2}) \cdot \Pr(y_m \ge x_m \ge \frac{\mu_2}{\mu_1} y_m).$$

Using the fact that $\Pr(y_m \ge x_m \ge \frac{\mu_2}{\mu_1}y_m) = \Theta(1)$ and $\frac{\mu_1}{\mu_2} = 1 + \Theta(\frac{1}{\log n})$, we get

$$C_{ob} - C_{wob} = O(\frac{1}{\log n}),$$

and this completes the proof. \blacksquare

In the next subsection we look into a scheme that employs superposition coding and clearly leads to the best throughput for single antenna systems as we actually work on the boundary of the capacity region. As the analysis becomes complicated, we just consider two groups and obtain a scheduling that maximizes the throughput while maintaining the rational rate constraints of (7.2). It should be mentioned that the ergodic capacity region of a broadcast channel with two users has been studied in [115]; here we look at a generalization of the result of [115] in which we have n users divided into two groups with different rate demands.

7.5.4 Superposition Coding for Single Antenna Broadcast Channels

In this section we analyze the performance of superposition coding for the case when there are only two groups of users $\mathcal{G}_1, \mathcal{G}_2$ with sizes $\alpha_1 n$ and $\alpha_2 n$. We assume that the average rate provided to a user in the first group is required to be $\beta > 1$ times the rate provided to a user in the second group.

In order to maximize the rate (sum-rate) while keeping the ratio of different group rates fixed and equal to β , we need to find the point on the boundary of the capacity region of the Gaussian broadcast channel with short-term power constraint ρ that satisfies the differentiated rate constraint. We know that every boundary point is the solution to a maximization problem of form

$$\max_{(R_1,\dots,R_n)\in\mathcal{C}_{BC}}\sum_{i=1}^n\mu_iR_i$$

for some positive values of μ_1, \ldots, μ_n . In our case, because of the symmetry among the users in each group, the values of μ_i s will be the same for the users in the same group. Therefore, we only need to characterize the boundary points that are the maximizing solution to the problem

$$\max_{(R_1,\dots,R_n)\in\mathcal{C}_{BC}}\mu_1(\sum_{i\in\mathcal{G}_1}R_i)+\mu_2(\sum_{i\in\mathcal{G}_2}R_i)$$

for $\mu_1, \mu_2 > 0$. The following lemma characterizes such boundary points. The proof of this lemma uses the duality of the broadcast channel and the multi-access channel for scalar channels explained in Section 7.3.

Lemma 7.12. Consider a scalar Gaussian broadcast system with the model described in Section 7.2. Consider the following optimization problem:

$$\max_{(R_1,\dots,R_n)\in\mathcal{C}_{BC}}\mu_1(\sum_{i\in\mathcal{G}_1}R_i) + \mu_2(\sum_{i\in\mathcal{G}_2}R_i),$$
(7.46)

where C_{BC} is the ergodic capacity region of broadcast channel with short-term power constraint ρ , and $\mu_1 \ge \mu_2$ are two positive numbers. Then, the solution of the above optimization problem is

$$\alpha_1 n R_i = E(\log(1+\rho x)|\mu_1 x \ge \mu_2 y) + E(\log(\frac{(\mu_1-\mu_2)y(1+\rho x)}{\mu_1(y-x)})|(x,y) \in \mathcal{R})$$
(7.47)

for $i \in \mathcal{G}_1$. Similarly, for $i \in \mathcal{G}_2$ we have

$$\alpha_2 n R_i = E(\log(1+\rho y)|\mu_1 x \le \mu_2 y) - E(\log(\frac{(\mu_1 - \mu_2)x(1+\rho y)}{\mu_2(y-x)})|(x,y) \in \mathcal{R}),$$
(7.48)

where $x = \max_{i \in \mathcal{G}_1} |h_i|^2$, $y = \max_{i \in \mathcal{G}_2} |h_i|^2$, and region \mathcal{R} is defined as

$$\mathcal{R} = \{ (x, y) \in R^+ \times R^+ | 0 \le \frac{\mu_2}{(\mu_1 - \mu_2)x} - \frac{\mu_1}{(\mu_1 - \mu_2)y} \le \rho \}.$$
(7.49)

Proof. The duality between the broadcast channel and the multi-access channel for the scalar case in [120] (and in (7.4)) states that

$$\mathcal{C}_{BC} = \bigcup_{\sum_{i} P_{i}(\cdot) = \rho} \mathcal{C}_{MAC}(P_{1}(\underline{h}), \dots, P_{n}(\underline{h})),$$

where $\underline{h} = (h_1, \ldots, h_n)$, $P_i(\underline{h})$ is the power allocation function of user *i*, and the union is over all the permissible power allocation functions. Furthermore,

$$\mathcal{C}_{MAC}(P_1(\underline{h}), \dots, P_n(\underline{h})) = \left\{ \underline{R} : \sum_{i \in \mathcal{S}} R_i \le \log(1 + \sum_{i \in \mathcal{S}} P_i(\underline{h}) |h_i|^2), \forall \mathcal{S} \subset \{1, \dots, n\} \right\}.$$

Using the above region, we can rewrite (7.46) as a maximization problem over all the power allocation functions and all the corresponding rate vectors in the capacity region of the dual multi-access channel. Based on this, it can be verified that the maximum of (7.46) occurs when we send only to the users with the best channel in each group. Therefore,

$$V_o = \mathbb{E} \max_{\substack{P_x, P_y \\ P_x + P_y = \rho}} (\mu_1 - \mu_2) \log(1 + P_x x) + \mu_2 \log(1 + P_x x + P_y y),$$



Figure 7.4: The decision region for power allocation in the superposition coding in two group case: If $(x, y) \in \mathcal{R}_1$, all the power is allocated to the best user of group one. If $(x, y) \in \mathcal{R}_2$, all the power is allocated to the best user of group two. If $(x, y) \in \mathcal{R}$, then power is split between the best users of the two groups as in (7.50).

where $x = \max_{i \in \mathcal{G}_1} |h_i|^2$ and $y = \max_{i \in \mathcal{G}_2} |h_i|^2$. Performing the maximization over P_x, P_y , we have one of the following possibilities:

- (a) If $\mu_1 x \ge \mu_2 y$, we assign all the power to the best user of the first group.
- (b) If $0 \leq \frac{\mu_2}{(\mu_1 \mu_2)x} \frac{\mu_1}{(\mu_1 \mu_2)y} \leq \rho$, then we split the power between the two best users in the two groups as

$$P_x = \frac{(\mu_1 - \mu_2)Pxy + \mu_1 x - \mu_2 y}{\mu_1 (y - x)x}, \ P_y = \rho - P_x.$$
(7.50)

(c) If $\frac{\mu_2}{(\mu_1-\mu_2)x} - \frac{\mu_1}{(\mu_1-\mu_2)y} > \rho$, all the power is assigned to the best user in \mathcal{G}_2 .

We have plotted the decision region for power allocation in the $(\frac{1}{x}, \frac{1}{y})$ region in Figure 7.5.4. In the weighted opportunistic scheduling, the power allocation policy would be to send to the best user in the first group if (x, y) is in \mathcal{R}_1 , and to send to the best user in the second group if (x, y) is in $\mathcal{R} \cup \mathcal{R}_2$.

The question that remains to be answered is to figure out how to choose μ_1 and μ_2 such that the rate constraint in (7.2) is satisfied. This is answered in the following
theorem.

Theorem 7.13. Suppose $\beta > 1$ is fixed, $\mu_1 = 1$, and $\mu_2 = 1 - \frac{1}{(\log n)^{\gamma}}$ where $\gamma = 1 + \frac{1}{\alpha_1\beta + \alpha_2}$ (i.e., $1 \le \gamma \le 2$); then

$$\lim_{n \to \infty} \frac{R^1}{R^2} = \beta. \tag{7.51}$$

Proof. Consider the first terms in the rate expressions for groups one and two in (7.47) and (7.48), respectively. Based on Remark 7.10, since $\frac{\mu_2}{\mu_1} = 1 - o(\frac{1}{\log n})$, it can be easily checked that these two terms are of order $\alpha_1 \log \log n$ and $\alpha_2 \log \log n$, respectively. Now we look at the second expectation in (7.47):

$$A^{\star} = E\left(\log(\frac{y(1+\rho x)}{((\log n)^{\gamma}-1)(y-x)})|(x,y) \in \mathcal{R}\right).$$

It can be verified that for $(x, y) \in \mathcal{R}$ the term inside the logarithm is greater than one. Hence, A^* is positive. Furthermore, similar to previous cases, the main contribution of the expectation comes from values of x and y around $\log n$. Therefore, we can simplify write A^* as

$$A^{\star} = E\left(\log(\frac{y(1+\rho x)}{((\log n)^{\gamma}-1)})|(x,y) \in \mathcal{R}\right) - \underbrace{E(\log(y-x)|(x,y) \in \mathcal{R})}_{B^{\star}}$$
$$= (2-\gamma)\log\log n(1-O(\frac{\log\log n}{\log n})) \cdot \Pr((x,y) \in \mathcal{R}) - B^{\star}$$

It can be shown that $Pr((x, y) \in \mathcal{R}) = \alpha_2 - o(1)$. In the following we show that the expectation term in the above equation is O(1). We have

$$B^{\star} = \Theta\left(\int_{l_1}^{l_2} \alpha_2 n e^{-y} (1 - e^{-y})^{\alpha_2 n - 1} \int_{\frac{y(1 - \frac{1}{(\log n)^{\gamma}})}{1 + \frac{\rho y}{(\log n)^{\gamma}}}}^{y(1 - \frac{1}{(\log n)^{\gamma}})} \log(y - x) \alpha_1 n e^{-x} (1 - e^{-x})^{\alpha_1 n - 1} \mathrm{d} x \mathrm{d} y\right),$$

where $l_1 = \log n - 4 \log \log n$, and $l_2 = \log n + 4 \log \log n$. Defining y - x = z and

 $w = y - \log n, B^*$ can be simplified to

$$B^{\star} = \Theta\left(\int_{-4\log\log n}^{4\log\log n} \alpha_2 e^{-w} (1 - \frac{e^{-w}}{n})^{\alpha_2 n - 1} \int_{(\log n)^{1 - \gamma}}^{(\log n)^{2 - \gamma}} \log(z) \alpha_1 e^z e^{-w} (1 - \frac{e^z e^{-w}}{n})^{\alpha_1 n - 1} \mathrm{d} z \mathrm{d} w\right)$$
$$= \Theta\left(\int_{-4\log\log n}^{4\log\log n} \alpha_2 e^{-w} \exp\left(-\alpha_2 e^{-w}\right) \int_{(\log n)^{1 - \gamma}}^{(\log n)^{2 - \gamma}} \log(z) \alpha_1 e^z e^{-w} \exp\left(-\alpha_1 e^z e^{-w}\right) \mathrm{d} z \mathrm{d} w\right).$$

Integration by part of the inner integral and further simplification gives

$$\begin{split} B^{\star} &= \Theta\left(\alpha_{2}(\gamma-1)\log\log n - \int_{-4\log\log n}^{4\log\log n} \alpha_{2}\frac{\exp\left(-\alpha_{2}e^{-w}\right)}{e^{w}} \int_{(\log n)^{1-\gamma}}^{(\log n)^{2-\gamma}} \frac{\exp\left(-\alpha_{1}e^{z}e^{-w}\right)}{z} \mathrm{d} \, z \mathrm{d} \, w\right) \\ &\stackrel{(a)}{=} \Theta\left(\alpha_{2}(\gamma-1)\log\log n - \alpha_{2} \int_{(\log n)^{1-\gamma}}^{(\log n)^{2-\gamma}} \frac{\exp\left(-\alpha_{1}e^{z}\frac{1}{(\log n)^{4}}\right)}{(\alpha_{2}+\alpha_{1}e^{z})z} \mathrm{d} \, z\right) \\ &\stackrel{(b)}{=} \Theta\left(\alpha_{2}(\gamma-1)\log\log n - \alpha_{2} \int_{(\log n)^{1-\gamma}}^{1} \frac{1}{(\alpha_{2}+(1-\alpha_{2})e^{z})z} \mathrm{d} \, z\right) \\ &\stackrel{(c)}{=} \Theta\left(\alpha_{2}(\gamma-1)\log\log n - \alpha_{2} \int_{(\log n)^{1-\gamma}}^{1} \frac{1}{z} \mathrm{d} \, z\right) \\ &= O(1), \end{split}$$

where (a) follows by integrating over w first. Equality (b) follows because in the interval $[1, (\log n)^{2-\gamma}]$, the integrand is upper-bounded by e^{-z} , and therefore the contribution of the integral in this interval is of order constant. Also, for $[(\log n)^{1-\gamma}, 1]$, the numerator of the integrand is 1 - o(1). Finally, (c) follows by verifying that the integral is increasing in α_2 and in both extremes, i.e. $\alpha_2 = 1$ and $\alpha_2 = 0$ the integral is $(\gamma - 1) \log \log n + O(1)$.

Putting the evaluated values of A^* and B^* together we get the following rate for a user in group \mathcal{G}_1 :

$$R^{1} = (1 + (2 - \gamma)\frac{\alpha_{2}}{\alpha_{1}})\frac{\log\log n}{n}(1 - O(\frac{\log\log n}{n\log n})).$$
(7.52)

Similarly, for a user in group \mathcal{G}_2 we have

$$R^{2} = (\gamma - 1) \frac{\log \log n}{n} (1 - O(\frac{\log \log n}{n \log n})).$$
(7.53)

Therefore, to meet the ratio constraints between the rates of users in different groups, we should have

$$\frac{(\gamma-1)}{(1+(2-\gamma)\frac{\alpha_2}{\alpha_1})} = \frac{1}{\beta},$$

or accordingly

$$\gamma = 1 + \frac{1}{\alpha_1 \beta + \alpha_2}$$

This completes the proof. \blacksquare

Finally, we look into the throughput loss due to the constraint of (7.2) using superposition coding. It is clear that the convergence rate for the superposition coding should be faster than or equal to TO and WO beamforming. In the next lemma, we provide a bound on the difference between the sum-capacity and the sumrate obtained by the scheduling discussed in this section.

Lemma 7.14. Suppose $\beta > 1$ is fixed and μ_1 , μ_2 are chosen as in Theorem 7.13. Then

$$\int_0^\infty n \log(1+\rho x) e^{-x} (1-e^{-x})^{n-1} - \sum_{i=1}^n R_i = O\left(\frac{1}{(\log n)^{2\gamma-1}}\right)$$

Proof. Here is the outline of the proof. Using (7.47), we can write the throughput under constraints of (7.2) as ⁹

$$E \log \left(1 + \rho x \mathbb{1}(x \ge \mu y) + \rho y \mathbb{1}(x < \mu y)\right) + E \log \left(\frac{(1 + \rho x)y}{(1 + \rho y)x} | (x, y) \in \mathcal{R}\right),$$

where x and y are defined in Lemma 7.12. Therefore, $\Delta(n)$, the difference of the

⁹In this proof we refer to μ_2 as μ .

sum-capacity and the throughput given above, can be written as

$$0 \le \Delta(n) = \mathcal{E} \log \left(\frac{1 + \rho x \mathbb{1}(x \ge y) + \rho y \mathbb{1}(x < y)}{1 + \rho x \mathbb{1}(x \ge \mu y) + \rho y \mathbb{1}(x < \mu y)} \right) - \mathcal{E} \log \left(\frac{(1 + \rho x)y}{(1 + \rho y)x} | (x, y) \in \mathcal{R} \right).$$

We can simplify the right hand side of the above equation to get

$$\Delta(n) = \mathcal{E} \log \left(\frac{1+\rho y}{1+\rho x} \middle| y \ge x \ge \mu y \right) + \mathcal{E} \log \left(\frac{(1+\rho y)x}{(1+\rho x)y} | (x,y) \in \mathcal{R} \right).$$

It can be easily checked that the second term is positive over region \mathcal{R} defined in (7.49). Therefore, we have

$$\Delta(n) \leq \operatorname{E} \log \left(\frac{1+\rho y}{1+\rho x} \middle| y \geq x \geq \mu y \right)$$

$$\leq \operatorname{E} \log \left(\frac{1+\rho y}{1+\rho \mu y} \middle| y \geq x \geq \mu y \right)$$

$$\leq \operatorname{E} \log \left(1 + \frac{\rho(1-\mu)y}{1+\rho \mu y} \middle| y \geq x \geq \mu y \right)$$

$$\leq \log(1 + \frac{1-\mu}{\mu}) \cdot \Pr(y \geq x \geq \mu y)$$

$$= -\log \mu \cdot \Pr(y \geq x \geq \mu y). \quad (7.54)$$

Now, using the techniques developed so far, it can be shown that

$$\Pr\left(y \ge x \ge \mu y\right) = O\left(\frac{1}{(\log n)^{\gamma-1}}\right).$$

Therefore, substituting μ with its value in (7.54), we have

$$\Delta(n) = O\left(-\log\left(1 - \frac{1}{(\log n)^{\gamma}}\right)\frac{1}{(\log n)^{\gamma-1}}\right) = O\left(\frac{1}{(\log n)^{2\gamma-1}}\right),$$

and this completes the proof. \blacksquare

276

7.6 Simulation Results

In this section we present simulation results for the three scheduling schemes studied in this chapter. We will show their performance in terms of satisfying the rational rate constraints and also the difference of their throughput with the sum-rate with no constraints present.

The first set of simulations are for M = 1, K = 2, and $\beta = 2$, i.e., one group requires twice the rate of the second group. We consider the groups to be of equal size. Figure 7.5 shows the sum of the transmitted rate for WO, TO, and SC as a function of the number of users. As expected, all show a log log n growth rate. In fact, the sum of the transmitted rates of WO and SC are quite close to the actual sum-rate capacity, signifying that the rate constraints do not lead to much of a rate hit on the throughput.

Figure 7.6 shows the ratio of the rates transmitted to the two groups as a function of the number of users for the WO and SC schedules. As we see, the rate of convergence of the SC to the desired ratio is slower than the WO beamforming. TO is not shown, as it clearly gives the correct ratio of $\beta = 2$.

The second set of simulations are for the same broadcast channel but with $\beta = 4$. The results are shown in Figure 7.7 and Figure 7.8.

Next we consider the performance for multiple antenna broadcast channels. For this set of simulations we consider a broadcast channel with two antennas at the transmitter, M = 2. We consider two groups of different size with $\alpha_2 = 2\alpha_1$, and different rate requirements as $\beta_2 = 2\beta_1$. Figure 7.9 shows the achievable throughput for TO and WO and compares it with the sum rate of opportunistic beamforming with no rate constraints. We have shown the ratio between the rates of users in different groups in Figure 7.10. As we see, WO schedule converges fast to the desired ratio. In Figure 7.11, we have shown the ratio of the rate for smaller size networks in



Figure 7.5: The sum of the transmitted rates for WO, TO, and SC, as well as the sum-rate capacity of the single antenna broadcast channel as a function of the number of users for a system with K = 2 and $\beta = 2$.



Figure 7.6: The ratio of the rates transmitted to the two groups of users as a function of the number of users for WO and SC for a system with $\beta = 2$.



Figure 7.7: The sum of the transmitted rates for WO, TO, and SC, as well as the sum-rate capacity of the single antenna broadcast channel as a function of the number of users for a system with K = 2 and $\beta = 4$.



Figure 7.8: The ratio of the rates transmitted to the two groups of users as a function of the number of users for WO and SC for a system with $\beta = 4$.



Figure 7.9: The sum of the transmitted rates for WO, TO, as well as the opportunistic for a broadcast channel with M = 2, K = 2, $\frac{\alpha_2}{\alpha_1} = 2$, and $\frac{\beta_1}{\beta_2} = \frac{1}{2}$ as a function of the number of users.

more detail. As we see, the ratio of the rates converges to the desired ratio even for small to moderate size networks. For instance, for a network of size n = 50, the ratio of the rates is 1.95 (only 0.05 off from the desired value).

Finally, the last set of simulation is done for a broadcast channel with two antennas at transmitter and three groups of users, i.e., K = 3 with equal size. The desired ratio is given as $\beta_1 = 1, \beta_2 = 2$, and $\beta_3 = 3$. In Figure 7.12 we have plotted the achievable sum-rate for TO and WO scheduling. In Figure 7.13 we have shown the achieved ratio of rates for different groups.

7.7 Conclusion

In this chapter we consider one of the design issues in the downlink of wireless cellular systems, namely differentiated quality of service provisioning. We consider a MIMO broadcast channel with fading, where users have different rate demands. In partic-



Figure 7.10: The ratio of the rates transmitted to the two groups of users as a function of the number of users for WO and TO for a system with M = 2, K = 2, $\frac{\alpha_2}{\alpha_1} = 2$, and $\frac{\beta_1}{\beta_2} = \frac{1}{2}$.



Figure 7.11: A closer look at the ratio of the rates transmitted to users in different groups for the example in Figure 7.10.



Figure 7.12: The sum of the transmitted rates for WO, TO, as well as the opportunistic for a broadcast channel with M = 2, K = 3, and $\beta_1 = 1$, $\beta_2 = 2$, and $\beta_3 = 3$ as a function of the number of users.



Figure 7.13: The ratio of the rates transmitted to the two groups of users as a function of the number of users for WO and TO for a system with M = 2, K = 3, and $\beta_1 = 1$, $\beta_2 = 2$, and $\beta_3 = 3$.

ular, we assume n homogenous users that are divided into K groups, each group of which requires the same rate, and where the ratio of the groups' rates are given. The transmitter would like to maximize the throughput (sum of the rates to all users) while maintaining the rational rate constraints. In general, this problem appears to be computationally intractable since the ergodic capacity region is described as the convex hull of (an infinite) set of rates. Furthermore, finding the exact capacity region requires complete knowledge of the channel states at the transmitter. Therefore, we are interested in *simple* schemes that require a *little* amount of information about the channels at the transmitter and operate close to the optimal capacity-achieving scheme. In particular, we propose three scheduling schemes to provide the rational rate constraints, namely weighted-opportunistic scheduling (WO), time-division opportunistic (TO), and superposition coding (SC) for single antenna systems. WO is a generalization of the opportunistic scheduling in which we transmit to only the user that has the largest "weighted" signal to noise plus interference ratio (SNIR). In TO, each group has its own time slot in which the transmitter chooses the user with the best SNIR from the corresponding group. Superposition coding is the scheme that achieves the information-theoretic capacity region. We first consider systems with n = 2 and n = 3, where each user requires a different rate. We focus on the achievable region by using the aforementioned WO. It turns out that determining the explicit relationship between the appropriate weights of the schedule and the desired ratios of the rates is analytically intractable even for the case of n = 3. For this reason, and also because most practical systems have many users, much of the chapter focuses on the asymptotic regime of large n, where explicit results can be found. For each scheduling we give an explicit scheme to guarantee the rational rate constraints. We also analyze the throughput loss due to the rate constraints for all three different schemes. In particular, we show that the throughput loss compared to the maximum throughput (i.e., the sum rate capacity without any rate constraints) tends to zero for

large n. Thus, there is not much of a penalty in providing different levels of service to different users. We also analyze the convergence rate of all the schemes and provide simulations supporting the theoretical analysis.

There are many directions in which this work can be extended. In this chapter we considered homogenous networks, i.e., networks where different users have the same quality of reception statistics. However, many practical network are heterogenous. Generalizing the schedules considered in this chapter to a heterogenous case is an important problem. Looking at other Quality of Service (QoS) requirements such as delay and reliability and also studying the existing trade-off between these measure are also very challenging and interesting problems.

Chapter 8

MIMO Gaussian Broadcast Channel with Estimation Error

8.1 Introduction

There has recently been a great deal of research on the capacity region of the MIMO Gaussian broadcast channels (GBC) (e.g., see [117],[132]-[137]). These channels are of practical importance since they can be used as a model for the downlink of cellular systems equipped with multiple antennas at the transmitter and the receiver side. In [118], the authors show that the entire capacity region is achieved by the dirty paper coding scheme first introduced in [119].

While dirty paper coding is the optimal transmission scheme, it is computationally expensive (although suboptimal schemes such as channel inversion or Tomlinson-Harashima precoding [151, 152, 153] give relatively close performance to the optimal scheme) and also requires the transmitter to have perfect knowledge of the channel state information (CSI) for all the users. As mentioned in the previous chapter, for a complete and optimal design of cellular systems, there are many practical issues that still need to be answered. Identifying the critical CSI needed at the transmitter and the amount feedback required for providing this CSI, finding computationally-efficient scheduling schemes, and providing different quality of service to different users in the network, are examples of these issues. In Chapter 7 we looked at one of these issues, namely differentiated quality of service provisioning in cellular systems. We proposed *simple* scheduling schemes based on random beamforming that requires *little* amount of side-information at the transmitter and provides different users with different rates.

In this chapter we look at another design issue mentioned above, namely the issue of the robustness of the capacity result with respect to error in channel state information. As mentioned in Chapter 7, the capacity of broadcast channels highly depends on the amount of channel state information in the transmitter (CSI). If perfect CSI is available at the transmitter, the throughput scales linearly with the number of transmit antennas (as the transmit power or the number of users increases). Comparing this result to a MIMO point-to-point system, we see that in the presence of perfect CSI, there is nothing to be lost by lack of cooperation among the receivers. On the other hand, if there is no CSI available at the transmitter, employing multiple antennas does not increase the throughput significantly. This is unlike the capacity result for MIMO point-to-point channels, where it is shown that even if the transmitter (or the receiver) does not know the channel, the capacity still scales with the number of antennas in the system [150, 109, 148, 140]. From a practical point of view, simple and effective scheduling schemes that are robust against noisy channel state information (and/or require partial knowledge of the channel) and also have a good performance are desirable [146]. There has been some progress on devising simple scheduling schemes that operate close to boundary points of the capacity region with limited feedback [121, 142, 145, 143]. However, the requirement of having accurate channel estimation is a strict constraint.

In this Chapter we consider the effect of channel estimation error on the capacity of MIMO Gaussian broadcast channels. We assume that the receivers have access to only an estimate (or noisy version) of their channels, and these estimates are fed back to the transmitter. We propose an achievable region based on the dirty paper coding scheme. This scheme is essentially similar to the one proposed for MIMO point-to-point and multi-access channels with uncertainty in channel measurements [111, 140]. We further show an interesting *duality* between the achievable rate region and the capacity of a multi-access channel where the noise covariance is dependent on the transmit power at different users. This duality is explored to show the effect of the estimation error on the sum-rate for large number of users and in the large power regime. It is shown that, for large number of users, as long as the estimation error is fixed with respect to the number of users, we achieve the same scaling law as if there was no estimation error. Of course, there is a loss due to the estimation error in the sum-rate which is obtained as a function of the covariance of the estimation error. However, in the large power regime, if the quality of our estimate does not increase with the transmit power, our achievable rate does not scale with the number of antennas. This is because in the large power, the system will be in the interferencedominated regime. As a matter of fact, in [126], the authors have shown that, unlike point-to-point systems, the sum-capacity of a broadcast channel with two users and two transmit antennas and with estimation error does not scale linearly with the number of antennas.

Based on the achievable rate region derived earlier, we analyze the performance of a training-based scheme for block fading models. We show that if the transmitter is willing to invest a fixed fraction of power in observing and training the channel, the sum-rate scales with the number of antennas at the transmitter. We also find the optimal training scheme that gives this linear scaling of the sum-rate in terms of the number of antennas.

The remainder of this chapter is organized as follows. After introducing the system model in the next section, we provide an achievable rate region for MIMO Gaussian broadcast channels with estimation error in Section 8.3. Section 8.4 looks at the optimal power allocation. In Section 8.5 we analyze the effect of the estimation error on the asymptotic behavior of the sum-rate for large number of users. Section 8.6 considers the achievable sum-rate based on training schemes and the conclusion comes in Section 8.7.

8.2 System Model

We consider a block fading MIMO Gaussian broadcast channel with channel estimation error. The transmitter employs M transmit antennas. We assume that there are n users in the system, each equipped with $r_i, i = 1, ..., n$ antennas. The channel matrix between the transmitter and user i is an $M \times r_i$ matrix and is denoted by H_i . A block fading model with coherence interval of length T is considered. We assume that the channel coefficients for each user are zero-mean jointly Gaussian random variables with covariance matrix $\operatorname{cov}(H_i) = \mathbb{E}(\operatorname{vec} H_i)(\operatorname{vec} H_i)^* = R_H$. The received signal at user i is given by

$$Y_i = H_i \underline{x} + \underline{n}_i,$$

where \underline{n}_i is additive white Gaussian noise with zero mean and identity covariance matrix. \underline{x} is the input vector with power constraint $E[\underline{x}^*\underline{x}] \leq P$.

In this chapter, the users and the transmitter do not have exact knowledge of the channel matrices. We assume that user *i* estimates its channel to \hat{H}_i . This estimate is fed back to the transmitter through a perfect channel (see Figure 8.1). The channel estimation error \tilde{H}_i which is equal to $H_i - \hat{H}_i$, is assumed to be uncorrelated from the estimate \hat{H}_i (i.e., MMSE estimation). The coordinates of \tilde{H}_i are assumed to be jointly Gaussian random variables with covariance matrices of form

$$\operatorname{cov}(\tilde{H}_i) = A_i^T \otimes K_i, \tag{8.1}$$

where A_i and K_i are positive semi-definite $M \times M$ and $r_i \times r_i$ matrices. The above



Figure 8.1: Multi-antenna Gaussian broadcast channel with channel estimation error.

covariance matrix models the possible correlation at the transmitter and the receiver side (see [144]). The capacity region of the aforementioned broadcast channel is known when H_i is available to the transmitter and to the *i*-th receiver for i = 1, ..., n [118]. Assuming the noise covariance matrix N_i for the *i*-th user, and under the transmit covariance matrix constraint, i.e., $E[\underline{x} \underline{x}^*] \preceq S$, the capacity region is given by dirty paper coding and can be written as

$$\mathcal{C}(S, \{N_i\}, \{H_i\}) = \operatorname{conv} \left\{ \bigcup_{\pi, \{B_i\}} \mathcal{R}(\pi, \{B_i\}, \{N_i\}, \{H_i\}) \right\},\$$

where the union is over all permutations on set $\{1, \ldots, n\}$ and all positive semi-definite covariance matrices B_1, \ldots, B_n such that $\sum_{i=1}^n B_i \leq S$ and

$$\mathcal{R}(\pi, \{B_i\}, \{N_i\}, \{H_i\}) = \left\{ (R_1, \dots, R_n) \middle| \\ 0 \le R_{\pi(i)} \le \log \frac{|N_{\pi(i)} + H_{\pi(i)}(\sum_{k=1}^i B_{\pi(k)}) H_{\pi(i)}^*|}{|N_{\pi(i)} + H_{\pi(i)}(\sum_{k=1}^{i-1} B_{\pi(k)}) H_{\pi(i)}^*|} \right\}$$

Finally, the capacity region of the broadcast channel with average total transmit

power constraint P, i.e., $tr(S) \leq P$, is given by the

$$\mathcal{C}(P, \{N_i\}, \{H_i\}) = \bigcup_{S: \text{tr}\,(S) \le P} \mathcal{C}(S, \{N_i\}, \{H_i\}).$$

In order to compute any point on the boundary of the capacity region, [132, 133] establish a duality between the capacity region of broadcast and multiple access channels under sum power constraints. This duality is considered in a more general scenario and based on the mini-max (and the Lagrangian) duality in [136, 137]. These results are very useful since the multi-access channel capacity region has a nice polymatroid structure [34] that makes it much easier to work with [103].

8.3 Inner Bound on the Capacity Region

In this section we give an inner bound on the capacity region of the MIMO Gaussian broadcast channel with estimation error. Here we assume that the transmitter and the receiver have access to an estimate of the channel rather than the actual channel. The uncertainty in the estimate is modeled as a Gaussian random variable. The results are based on the fact that the worst uncorrelated noise with given covariance matrix has Gaussian distribution. This was in fact used previously to obtain lower bounds on the capacity of MIMO point-to-point channels and multi-access channels in [111, 140].

Theorem 8.1. Consider a MIMO Gaussian broadcast channel described in Section 8.2 where the estimated channel for the *i*-th user is denoted by \hat{H}_i . We assume that \hat{H}_i is known to the transmitter and the corresponding user. Then, the capacity region of this broadcast channel includes the capacity region of a MIMO Gaussian broadcast channel with channel matrices \hat{H}_i and effective noise covariances $I + \operatorname{tr} (A_i(\sum_{l=1}^n B_l))K_i$ (as shown in Figure 8.2.) In other words, the capacity region



Figure 8.2: Inner bound on the capacity region with estimation error.

includes the following region

$$\operatorname{conv}\left\{\bigcup_{\substack{\pi,\{B_i\}\\\operatorname{tr}(\sum_i B_i)\leq P}} \mathcal{R}\left(\pi,\{B_i\},\{I+\operatorname{tr}(A_i(\sum_{l=1}^n B_l))K_i\},\{\hat{H}_i\}\right)\right\}$$

Proof. The proof follows a similar approach to the dirty paper coding scheme. Note that \hat{H}_i is the measurement (estimate) of the channel H_i , and \tilde{H}_i is the error in measurement. We can write the received signal at user i as

$$\underline{y}_{i} = \hat{H}_{i}\underline{x} + \underbrace{\tilde{H}_{i}\underline{x} + \underline{n}_{i}}_{\underline{y}_{i}}.$$
(8.2)

Without loss of generality, suppose $\pi(\cdot)$ is the identity permutation, i.e., $\pi(i) = i$ for all $i \in \{1, \ldots, n\}$. Let us consider a broadcast channel with no channel estimation error and with channel coefficients $\{\hat{H}_i\}$ and noise covariance matrices $\{N_{i,eq}\}$, where

$$N_{i,eq} = \sum_{l=1}^{n} \operatorname{tr} (B_l A_i) K_i + I, \qquad i = 1, \dots, n,$$
(8.3)

and B_1, \ldots, B_n are positive definite matrices with $\sum_{l=1}^n \operatorname{tr}(B_l) \leq P$. K_i and A_i are defined in (8.1). We will refer to this broadcast channel as BC_{eq}. The received signal

at user i in BC_{eq} is

$$\underline{y}_{i,eq} = \hat{H}_{i\underline{x}} + \underline{n}_{i,eq}, \tag{8.4}$$

where $\underline{n}_{i,eq}$ is zero-mean a Gaussian noise with covariance $N_{i,eq}$.

Using the dirty paper coding approach and the result of [138] concerning the capacity of memoryless channels with a random state known non-causally at the transmitter, we know that for BC_{eq} there exist Gaussian random vectors $\{\underline{x}_i, i = 1, ..., n\}$ and $\{\underline{u}_i, i = 1, ..., n\}$ with the following properties

- \underline{x}_i s for different *i*s are independent zero-mean Gaussian random vectors with covariance matrix B_i .
- \underline{u}_i is a Gaussian random vector that is a function of \underline{x}_j for $i \leq j \leq n$ and $\hat{H}_j, i \leq j \leq n$.
- The rate for user i in BC_{eq} can be written as

$$R_{i,eq} = I(\underline{u}_i; \underline{y}_{i,eq}) - I(\underline{u}_i; \underline{s}_i) = \log \frac{|N_{i,eq} + \hat{H}_i(\sum_{k=1}^i B_i)\hat{H}_i^*|}{|N_{i,eq} + \hat{H}_i(\sum_{k=1}^{i-1} B_i)\hat{H}_i^*|},$$
(8.5)

where $\underline{s}_i = \sum_{j=i+1}^n \underline{x}_j$ is the state known non-causally at the transmitter.

Now, looking back at the original broadcast channel with estimation error, we will use the same set of $\{x_i\}$ and $\{u_i\}$ for coding and show that the rates achieved by the coding scheme of [138] are lower-bounded by the rates achieved for channel BC_{eq}, i.e., <u> R_{eq} </u> shown in (8.5). The transmitter will send $\underline{x} = \sum_{i=1}^{n} \underline{x}_i$, where x_i is the signal intended for user *i*. It can be easily checked that \underline{v}_i in (8.2) and \underline{x}_j s are uncorrelated from each other, i.e.,

$$\Psi_{vx} = \mathbf{E} \, \underline{v}_i \underline{x}_i^* = 0, \qquad j = 1, \dots, n.$$

292

This suggests that \underline{v}_i and \underline{u}_i are also uncorrelated

$$\Psi_{vu} = \mathbf{E}\,\underline{v}_i\underline{u}_i^* = 0. \tag{8.6}$$

Furthermore, the covariance of \underline{v}_i in (8.2) can be written as

$$\operatorname{cov}(\underline{v}_{i}) = \operatorname{E} \underline{v}_{i} + \underline{v}_{i}^{*}$$

$$= I + \operatorname{E} \tilde{H}_{i} \underline{x} \underline{x}^{*} \tilde{H}_{i}^{*}$$

$$= I + \operatorname{E} \tilde{H}_{i} (\sum_{l=1}^{n} B_{l}) \tilde{H}_{i}^{*}$$

$$= I + \sum_{l=1}^{n} \operatorname{tr} (B_{l} A_{i}) K_{i}$$

$$= N_{i, eq}.$$

$$(8.7)$$

In other words, the covariance matrix of \underline{v}_i is equal to the covariance matrix of the Gaussian noise $\underline{n}_{i,eq}$ present in BC $_{eq}$.

For the covariance of \underline{y}_i in (8.2)we have

$$\operatorname{cov}\left(\underline{y}_{i}\right) = \operatorname{cov}\left(\underline{v}_{i}\right) + \hat{H}_{i}\left(\sum_{l=1}^{n} B_{l}\right)\hat{H}_{i}^{*} = \operatorname{cov}\left(\underline{y}_{i\,eq}\right).$$

Therefore, the covariance matrices of the received signals in the original broadcast channel and BC $_{eq}$ are equal.

Now, the rate achieved with coding of [138] is equal to

$$R_i = I(\underline{u}_i; \underline{y}_i) - I(\underline{u}_i; \underline{s}_i)$$
(8.8)

for each user i. Comparing (8.5) with (8.8), we see that only the first mutual information term is different for the two channels. The first term in (8.8) can be written

$$I(\underline{u}_i, \underline{y}_i) = h(\underline{u}_i) - h(\underline{u}_i | \underline{y}_i).$$

We have the following upper bound on $h(\underline{u}_i | \underline{y}_i)$ in terms of the covariance matrix $\operatorname{cov}(\underline{u}_i | \underline{y}_i) = \operatorname{E}_{|\underline{y}_i}(\underline{u}_i - \operatorname{E}_{|\underline{y}_i}\underline{u}_i)(\underline{u}_i - \operatorname{E}_{|\underline{y}_i}\underline{u}_i)^*$:

$$h(\underline{u}_i|\underline{y}_i) \le \log \det \pi e \operatorname{cov}\left(\underline{u}_i|\underline{y}_i\right),\tag{8.9}$$

since among all random vectors with the same covariance matrix, the one with a Gaussian distribution has the largest entropy.

The following lemma gives an important property of $\operatorname{cov}(\underline{u}_i|\underline{y}_i)$, the proof of which can be found in [147].

Lemma 8.2. Let $\underline{\hat{u}}_i = f(\underline{y}_i)$ be any estimate of \underline{u}_i given \underline{y}_i . Then we have

$$\operatorname{cov}(\underline{u}_i|\underline{y}_i) \le \operatorname{E}(\underline{u}_i - \underline{\hat{u}}_i)(\underline{u}_i - \underline{\hat{u}}_i)^*.$$

Substituting the Linear MMSE estimate $\underline{\hat{u}}_i = \Psi_{uy} \operatorname{cov}(\underline{y}_i) \underline{y}_i$ (Ψ_{uy} denotes the cross-covariance of \underline{u}_i and \underline{y}_i) in the above lemma yields

$$\operatorname{cov}\left(\underline{u}_{i}|y_{i}\right) \leq \operatorname{cov}\left(\underline{u}_{i}\right) - \Psi_{uy}\operatorname{cov}\left(y_{i}\right)\Psi_{yu}.$$
(8.10)

 Ψ_{uy} can be calculated as follows:

$$\Psi_{uy} = E \underline{u}_i \underline{y}_i^*$$

$$= E \underline{u}_i (\hat{H}_i \underline{x} + \underline{v}_i)^*$$

$$= \Psi_{ux} \hat{H}_i^* + \Psi_{uv}$$

$$= \Psi_{ux} \hat{H}_i^*$$

$$= E \underline{u}_i \underline{y}_{i, eq} = \Psi_{uy_{eq}}.$$

as

Therefore, the cross-covariance of \underline{u}_i and \underline{y}_i is equal to the cross-covariance of \underline{u}_i and $\underline{y}_{i,eq}$ in BC_{eq}. Combining (8.10) and (8.9) we have

$$\begin{split} h(\underline{u}_i | \underline{y}_i) &\leq \log \det \pi e \operatorname{cov} \left(\underline{u}_i | \underline{y}_i \right) \\ &\leq \log \det \pi e \left(\operatorname{cov} \left(\underline{u}_i \right) - \Psi_{uy} \operatorname{cov} \left(\underline{y}_i \right) \Psi_{yu} \right) \\ &\stackrel{(a)}{=} \log \det \pi e \left(\operatorname{cov} \left(\underline{u}_i \right) - \Psi_{uy_{eq}} \operatorname{cov} \left(\underline{y}_{i, eq} \right) \Psi_{yu_{eq}} \right) \\ &\stackrel{(b)}{=} h(\underline{u}_i | \underline{y}_{i, eq}), \end{split}$$

where in (a) we have used the facts that covariance matrices of y_i and $y_{i,eq}$ are the same and $\Psi_{uy} = \Psi_{uy_{eq}}$. Also, (b) is a consequence of the random vector \underline{u}_i given $\underline{y}_{i,eq}$ being a Gaussian and, therefore, MMSE estimate being the optimal estimate.

The above inequality suggests that

$$I(\underline{u}_i; \underline{y}_i) \ge I(\underline{u}_i; \underline{y}_{i, eq}),$$

and hence, from (8.8) and (8.5) we have

$$R_i \ge R_{i,eq}$$

for all users. This completes the proof. \blacksquare

Remark 8.3. Note that although we proved Theorem 8.1 for estimation error covariances of form $A_i^T \otimes K_i$, the theorem can be easily generalized for error covariance matrices of general form. In the general case, the equivalent noise covariance is given by (8.7), and the achievable set of rates is given in (8.5).

295

8.4 Optimal Power Allocation

In the previous section an achievable rate region for MIMO broadcast channels with estimation error was given. This region is based on dirty paper coding and is equal to the capacity region of a broadcast channel with noise covariances that depend on the covariance of the transmitted signal. It is well known that the dirty paper coding region is not convex in input covariance matrices, and finding the boundary points of the capacity region directly from the dirty paper coding regime is not computationally tractable. However, using the duality of the broadcast and multiple access channels [132, 133] and the mini-max duality introduced in [136, 137], it is possible to find the boundary points of the capacity region under some class of power constraints using convex optimization. In this section we consider finding the power allocation for any boundary point on the achievable rate region described in Theorem 8.1.

It is worth mentioning that since in our case the effective noise covariance matrices also depend on the input covariance matrices, the transformations used in [133] do not go through. As a matter of fact, the transformation used in [133] is valid only for a sum power constraint. For the presentation of this chapter we provide duality results in the following two cases.

For all users $A_i = I$:

It can be easily shown that any boundary point on the region described in Theorem 8.1 is achieved when $\sum_{i=1}^{n} \operatorname{tr} (B_i) = P$. Therefore, if $A_i = I$ for all the channels, the effective noise of (8.3) does not depend on B_i s anymore and is given by $I + PK_i$. In this case, one can use the duality of multiple access and broadcast channels with sum power constraints. Hence, the region of Theorem 8.1 is equal to the capacity region of a Gaussian Multiple access channel with sum power constraint P and channel coefficients $\hat{H}_i^*(I + PK_i)^{-\frac{1}{2}}$. Therefore, any point on the boundary can be computed



Figure 8.3: A dual representation of the inner bound of the capacity region with estimation error.

using convex optimization. We have summarized this result as follows.

Theorem 8.4. Consider the setting of Theorem 8.1. Further assume that the covariance matrix of the estimation error for *i*-th channel is of the form $\operatorname{cov}(\tilde{H}_i) = I \otimes K_i$. Then, the capacity region of the channel includes the capacity region of a multiple access channel with sum power constraint P and channel coefficients $\hat{H}_i^*(I + PK_i)^{-\frac{1}{2}}$.

MISO Broadcast with Estimation Error, $r_i = 1$:

In the rest of this chapter we consider the achievable rates for MISO broadcast channels with estimation error. For this case, we can state the achievable region based on the capacity region of a dual multiple access channel.

Theorem 8.5. Consider a MISO Gaussian broadcast channel with estimation error covariance $A_i \succ 0$ for user *i* and total transmit power constraint of *P*. Then, the capacity region includes the capacity region of a multiple access channel shown in Figure 8.3 with one antenna at each transmitter and *M* antennas at the receiver. The channel coefficient vector for transmitter *i* is \hat{H}_i^* . The noise covariance is I + $\sum_{i=1}^n P_i A_i$, where P_i is the transmit power for user *i*, with $\sum_i P_i \leq P$. *Proof.* We will use the approach taken in [136]. Instead of looking at the achievable rates, we will look at the feasibility of a set of SINR constraints in the broadcast and the dual multi-access scenario. Similar to [137], we consider beam-forming with dirty paper precoding. The transmitted signal in this case can be written as

$$x = \sum_{i} \underline{\phi}_{i} s_{i},$$

where $\underline{\phi}_i$ is the *i*-th beam that carries s_i , the information for user *i*. Without loss of generality we assume that $E |s_i|^2 = 1$. Looking back at the definition of B_i in Theorem 1, here we have assumed that $B_i = \underline{\phi}_i \underline{\phi}_i^*$. Next we consider the problem of minimizing total transmit power subject to a set of SINR constraints for broadcast channel

$$\begin{split} L^{BC} &= \min_{P, \underline{\phi}_i} \quad P \\ \text{subject to} \quad \frac{|h_i \underline{\phi}_i|^2}{\sum_{j>i} |h_i \underline{\phi}_j|^2 + 1 + \sum_{l=1}^n w_l^* A_i \underline{\phi}_l} \quad \ge \quad \gamma_i \\ &\sum_{i=1}^n \underline{\phi}_i^* \underline{\phi}_i \quad \le \quad P. \end{split}$$

Similarly, we can write the following problem for the dual multiple access channel. The SINRs shown below are achieved by using $\underline{\varphi}_i$ as a filter for *i*-th user information and using interference cancelation.

$$\begin{split} L^{MA} &= \min_{P_i \ge 0, \, \underline{\varphi}_i} \quad \sum_{i=1}^n P_i \\ \text{subject to} \quad \frac{P_i |h_i \underline{\varphi}_i|^2}{\sum_{j < i} |h_j \underline{\varphi}_i|^2 + \underline{\varphi}_i^* (\sum_{l=1}^n P_l A_l) \underline{\varphi}_i} \quad \ge \quad \gamma_i \end{split}$$

Following the steps of [137], we can show that both of the above problems have the

following dual

$$\max_{P_i \ge 0} \sum_{i=1}^{n} P_i$$

subject to
$$\sum_{j < i} P_j h_j^* h_j + \sum_{l=1}^{n} P_l A_l + I \succeq \frac{P_i}{\gamma_i} h_i^* h_i.$$

Furthermore, the strong duality holds, and the two problems have the same minimum power and are equivalent. Therefore, the achievable rate region of MISO broadcast channel is equivalent to the capacity region of a multiple access channel with Mantennas at the receiver, total power constraint P, and a noise covariance Q_{eq} that depends on the different users' transmit powers in the following form

$$Q_{\rm eq} = I + \sum_{l=1}^{n} P_l A_l,$$

and this proves Theorem 8.5. \blacksquare

Clearly, Theorem 8.5 implies that for a homogeneous system, where $A_i = A$ for all users, the capacity region of this channel includes the capacity region of a multiple access channel with total transmit power P and noise covariance matrix I + PA.

8.5 Scaling Laws of the Achievable Sum-rate

Using Theorem 8.5, we know that the following sum-rate is achievable for homogeneous MISO broadcast channels:

$$R_{sum} = \max_{\substack{P_i \ge 0\\\sum_{i=1}^{n} P_i \le P}} \log \frac{|I + PA + \sum_{i=1}^{n} P_i \hat{H}_i^* \hat{H}_i|}{|I + PA|}$$
(8.11)

This optimization problem is convex in the P_i s and can be therefore solved when n is not too large. The achievable ergodic sum-rate for fading channels is just the

expectation of R_{sum} over all channel realizations. Clearly, when *n* is large, computing the average sum-rate becomes computationally intensive. In what follows, we obtain the scaling law of the ergodic sum-rate for large number of users.

Defining $G_i = \hat{H}_i (I + PA)^{-\frac{1}{2}}$, the ergodic sum-rate is given by

$$R^{\star} = \mathcal{E}(R_{sum}) = \mathcal{E} \max_{P_i \ge 0, \sum_{i=1}^{n} P_i \le P} \log |I + \sum_{i=1}^{n} P_i G_i^* G_i|,$$
(8.12)

where the expectation is over G_i s for i = 1, ..., n. Here G_i s are independent Gaussian vectors with covariance matrix

$$E(G_i^*G_i) = (I + PA)^{-\frac{1}{2}}(R_H - A)(I + PA)^{-\frac{1}{2}}.$$
(8.13)

Note that (8.12) is in fact the ergodic sum-capacity of a MISO broadcast channel where channels have a Gaussian distribution with covariance matrix given in (8.13). The ergodic sum-capacity of MISO broadcast channels with spatial correlation in channel coefficients is analyzed for a large number of users in [139]. Assuming that R_H and A are fixed (in terms of n), one can use the result of [139] to state the following Theorem.

Theorem 8.6. Consider the setting of Theorem 8.5. Assume the channel covariance matrix is R_H and estimation error covariance is $A \leq R_H$. Then, as the number of users n goes to infinity, the achievable sum-rate scales like

$$R^{\star} = M \log \log n + M \log \frac{P}{M} + \log \det(R_H - A) - \log \det(I + PA) + o(1).$$
(8.14)

Theorem 8.6 suggests that as long as the estimation error covariance matrix is fixed in terms of n, one gets the same scaling as the case where the channel is known

perfectly at the receivers and the transmitter. In fact, the effect of estimation error shows up as a constant hit in the achievable rate.

At the end, we should remark that since for a homogeneous network, the equivalent noise in Theorem 8.5 is linear in the transmit power, in the high SNR regime (and for a fixed number of users), the achievable sum-rate will be of constant order (see also [126]).

8.6 Training

The results obtained so far are based on a given estimation error covariance. To estimate the channel, a training phase is often required. During this phase, some portion of the transmission interval and transmit power is used to send known training signals. In this section we consider training for block fading MISO broadcast channels with M transmit antennas, coherence interval of $T \ge M$, and total transmit power of P. We further assume that the channel coefficients are independent zero mean unit variance Gaussian random variables. We find the optimum amount of time and power that should be allocated for training to maximize our achievable sum-rate.

During the training phase, the transmitter sends T_{τ} training vectors with total transmit energy of $P_{\tau}T_{\tau}$. Let X_{τ} be the $M \times T_{\tau}$ matrix consisting of the training vectors. We have

$$\operatorname{tr}\left(X_{\tau}^{*}X_{\tau}\right) = P_{\tau}T_{\tau}.\tag{8.15}$$

The received signal at user i can be written as

$$y_{i,\tau} = h_i X_\tau + v_{i,\tau}$$

At the end of the training phase, each user finds the LMMSE estimate of its channel

and feeds it back to the transmitter. In order to obtain a meaningful estimate of h_i ,¹ we need at least as many measurements as unknowns, which implies that $T_{\tau} \geq M$. The estimate can be written as

$$\hat{h}_i = y_{i,\tau} (I + X_\tau^* X_\tau)^{-1} X_\tau^*.$$

Note that $y_{1,\tau}, \ldots, y_{n,\tau}$ are independent and identically distributed. The estimation error covariance for each user is

$$A_{\tau} = \operatorname{cov}(\tilde{h}_{i}) = I - X_{\tau}^{*}(I + X_{\tau}^{*}X_{\tau}X_{\tau})^{-1}X_{\tau}$$
$$= (I + X_{\tau}X_{\tau}^{*})^{-1}.$$

Let $T_d = T - T_{\tau}$ and $P_d T_d = PT - P_{\tau} T_{\tau}$. After the training phase, the transmitter starts sending data over the T_d time samples left and with total transmitter energy $P_d T_d$. Therefore, for a fixed P_{τ}, T_{τ} , using the result of Theorem 8.5, the following sum-rate is achievable:

$$R_{\tau} = \frac{T_d}{T} \mathbb{E} \max_{P_i \ge 0, \sum_i P_i \le P_d} \log \frac{|I + P_d A_{\tau} + \sum_{i=1}^n P_i \hat{h}_i^* \hat{h}_i|}{|I + P_d A_{\tau}|},$$
(8.16)

where h_i s are independent vectors whose elements are jointly Gaussian random variables with covariance matrix $I - A_{\tau}$ (which follows from the orthogonality principle). Now consider the eigenvalue decomposition of $X_{\tau}X_{\tau}^* = U\Omega U^*$, where U is unitary and Ω is diagonal. From (8.15) we have tr $(\Omega) \leq P_{\tau}T_{\tau}$. After some manipulation of (8.16) we can rewrite the achievable rate as

$$R_{\tau} = \frac{T_d}{T} \mathbb{E} \max_{\substack{P_i \ge 0\\\sum_i P_i \le P_d}} \log \frac{|I + (1 + P_d)\Omega^{-1} + \sum_{i=1}^n P_i g_i^* g_i|}{|I + (1 + P_d)\Omega^{-1}|}.$$
(8.17)

¹Throughout this section we use h_i rather than H_i to represent the channel vector for *i*-th user.

The g_i s are independent vectors whose elements are independent zero mean unit variance Gaussian random variables, and the expectation is over g_i . Now let us consider the case where Ω is a scaled version of identity. Using the trace constraint we have

$$\Omega = \frac{P_{\tau}T_{\tau}}{M}I$$

This Ω corresponds to the case where the training matrix X_{τ} is a multiple of a matrix with orthonormal columns. Using this Ω and simplifying (8.17), the following rate is achievable:

$$R_{\tau} = \frac{T_d}{T} \mathbb{E} \max_{P_i \ge 0, \sum_i P_i \le P_{\text{eff}}} \log |I + \sum_{i=1}^n P_i g_i^* g_i|,$$
(8.18)

where for each i, g_i is a vector of i.i.d zero-mean unit-variance Gaussian random variables. P_{eff} is the effective power and is given as

$$P_{\rm eff} = \frac{P_d P_\tau T_\tau}{P_\tau T_\tau + (1+P_d)M}$$

We can maximize the achievable lower bound of (8.18) over power and time allocated for training. Note that for a fixed T_{τ} (and T_d), the optimal power allocation is one that maximizes the effective transmit power P_{eff} . By maximizing P_{eff} over P_{τ} and P_d we get

$$P_{\text{eff}}^{\star}(T_d) = \frac{(PT)^2}{\left(\sqrt{(PT+T_d)M} + \sqrt{(M+PT)T_d}\right)^2}.$$
(8.19)

Also, the maximizing P_{τ} is given by

$$P_{\tau}^{*}(T_{d}) = \frac{PT\sqrt{(T_{d} + PT)M}}{(T - T_{d})(\sqrt{(PT + T_{d})M} + \sqrt{(M + PT)T_{d}})}.$$
(8.20)

In order to maximize the achievable rate over T_d we have to solve the following

optimization problem:

$$R^{\star} = \max_{T_d, 0 \le T_d \le T - M} \frac{T_d}{T} \mathbb{E}_{\{g_i\}} f(P_{\text{eff}}^{\star}(T_d)),$$
(8.21)

where $P_{\text{eff}}^{\star}(T_d)$ is given in (8.19), and f(x) is defined as

$$f(x) = \max_{p_i, \sum_i p_i \le 1} \log |I + x \sum_{i=1}^n p_i g_i^* g_i|$$

It is shown in the following lemma that the cost function in (8.21) is increasing in T_d . This suggests that the optimal T_d is T - M.

Lemma 8.7. The cost function of (8.21) is an increasing function in T_d .

Proof. The claim is that the following function is increasing in T_d :

$$C(T_d) = \frac{T_d}{T} \mathbb{E}_{\{g_i\}} f(P_{\text{eff}}^{\star}(T_d)),$$

where f(x) is defined in the equation right after (8.21). To show this, we differentiate with respect to T_d :

$$\frac{d}{dT_d}C(T_d) = \frac{1}{T} \mathbb{E}_{\{g_i\}} \left[f(P_{\text{eff}}^{\star}(T_d)) + T_d \frac{d}{dT_d} P_{\text{eff}}^{\star}(T_d) f'(P_{\text{eff}}^{\star}(T_d)) \right]$$

In order to show that $C(T_d)$ is increasing (or equivalently $\frac{d}{dT_d}C(T_d) > 0$), it suffices to show that the term inside the expectation is greater than zero for any value of $P_{\text{eff}}^{\star}(T_d)$. Using (8.19) after some manipulation we have

$$T_d \frac{d}{dT_d} P_{\text{eff}}^{\star}(T_d) = -P_{\text{eff}}^{\star}(T_d) \underbrace{\frac{\sqrt{T_d(M+PT)} + \sqrt{\frac{MT_d^2}{T_d+PT}}}{\sqrt{T_d(M+PT)} + \sqrt{M(PT+T_d)}}}_{z}.$$

It can be readily verified that z < 1. Therefore, it is enough to show that

$$f(x) - xf'(x) \ge 0,$$

or equivalently to show that $\frac{f(x)}{x}$ is decreasing. Note that $\frac{\log(1+ax)}{x}$ is decreasing in x. Suppose we write the f(x+y) in terms of the eigenvalues of the optimal matrix. Thus,

$$f(x+y) = \sum_{i=1}^{n} \log(1 + (x+y)\lambda_i)$$

$$\leq \sum_{i=1}^{n} \frac{x+y}{x} \log(1+x\lambda_i)$$

$$\leq \frac{x+y}{x} \log \det(I + x \sum_{i=1}^{n} p_i g_i^* g_i)$$

$$\leq \frac{x+y}{x} \max_{p_i \sum_{i=1}^{n} p_i \leq 1} \log \det(I + x \sum_{i=1}^{n} p_i g_i^* g_i)$$

$$= \frac{x+y}{x} f(x).$$

This completes the proof. \blacksquare

The next theorem summarizes the above arguments.

Theorem 8.8. Consider a block fading MISO broadcast channel with M transmit antennas, coherence interval of $T \ge M$, and total transmit power of P. Further, assume that the channel coefficients are independent zero-mean unit-variance Gaussian random variables. Then, the following sum-rate is achievable using training:

$$R^{\star} = \frac{T - M}{T} \mathbb{E}_{\{g_i\}} \max_{p_i, \sum_i p_i \le 1} \log |I + P^{\star}_{\text{eff}}(T - M) \sum_{i=1}^n p_i g^{\star}_i g_i|,$$
(8.22)

where $P_{\text{eff}}^{\star}(\cdot)$ is defined in (8.19). Furthermore, this rate is achieved by using orthogonal and fixed power training vectors over the first M time samples and transmitting The following Corollary gives further insights on the behavior of the sum-rate in different regimes.

Corollary 8.9. Consider the MISO broadcast channel model described in Theorem 8.8. Then, the achievable sum-rate

• for large P and fixed n scales like

$$R^{\star} = \min\{M, n\}\left(1 - \frac{M}{T}\right)\log P;$$

• for small P and fixed n scales like

$$R^{\star} = \frac{Tc\log e}{4M}P^2,$$

where c is the mean of the maximum of n i.i.d random variables with $\chi^2(2M)$ distribution; and

• for large number of users, n, and fixed P scales like

$$R^{\star} = M(1 - \frac{M}{T})\log(1 + P_{\text{eff}}^{\star}(T - M)\log n).$$

Corollary 8.9 shows that using training-based schemes, one can achieve the multiplexing gain of a MIMO point-to-point channel with M transmit and n receive antennas in the high SNR regime. However, the power invested in the training phase increases linearly with P (see (8.20) for large P). Also, the required feedback rate for sending the estimates to the transmitter should increase with P. Therefore, the above scheme suggests that as long as the transmitter is willing to invest a fixed fraction of the available power for learning the channel, good multiplexing gains can be achieved.

8.7 Conclusion

This chapter considers an important design issue in the downlink of cellular systems, that is the effect of channel estimation error on the capacity region of MIMO Gaussian broadcast channels. In practical systems, the mobile users and the base station (transmitter) have only an estimate of the channel available. For this case, an achievable rate region based on dirty paper coding is derived. It is further shown that for MISO case, this region is equivalent to the capacity region of a multi-access channel with noise covariance matrix that depends on the transmit power and the estimation error. A training-based scheme for block fading MISO Gaussian broadcast channels is analyzed, and the optimal length of training interval and the power used for training is derived. Designing practical schemes in the presence of channel estimation error is an important future work. Also, finding outer bounds on the capacity region of broadcast channels with estimation error is an interesting problem (see [126]).

Chapter 9 Future Work

Networks, and especially wireless networks, are changing the very fabric of our lives in different aspects. We are moving towards a *networked world* where everyone at any place and at all times is connected and can access and process information seamlessly. However, in order to reach the promised networked world, there are many issues regarding the capabilities and the design of wireless networks that should be addressed.

In this thesis, we analyze the performance limit of a few applications over special classes of wireless networks and look at a few design issues in cellular wireless networks. The results in the thesis have brought up a few interesting open problems.

• QoS Provisioning in Cellular Systems: Because of the highly heterogeneous nature of today's cellular systems, Quality of Service (QoS) provisioning is of great importance. In Chapter 7, we looked at this problem by modeling various QoS demands with differentiated rate requirements among users. However, in many applications rate requirement is not the main issue. Delay and reliability requirements can be strict and more important for delay-sensitive and robust applications. An important open problem is proposing a unified framework that integrates all different requirements and allows scheduling of heterogeneous networks with diverse demands.
In Chapter 8 we gave an inner bound on the capacity region of broadcast channels with channel estimation error. Finding outer bounds on the capacity region of broadcast channels with channel estimation error and with limited feedback from the users to the transmitter is also a challenging and important task.

• Optimization Theory View of Multi-terminal Information Theory: As mentioned earlier, multi-terminal Information Theory has fallen short of characterizing the absolute limits of communication in general networks. Part of the problem arises from the fact that the current standard tools for providing outer bounds on the achievable rates in a network setup are based on arguments developed for point-to-point communication systems. Usually, the outer bounds are based on information-theoretic cut-set bounds that assume full cooperation of users in each side of the cut. The multicast min-cut bound of Chapter 3 is an example of these bounds. In these bounds the fact that information is distributed in the network is neglected.

One major challenge is to provide a methodology that considers the distributed nature of information in networks and gives potentially tighter outer bounds on the set of achievable rates in a multi-terminal setup. One promising approach is applying optimization theory tools (such as duality and convex optimization) for providing outer bounds on the capacity region of multi-user and network setups. A special case of this method was proposed and used in Chapter 3. A first attempt could be finding equivalent forms of information theory techniques (such as cut-set bounding, Data Processing inequality, etc.) in the optimization domain.

• Analysis of Coding over Networks: The benefits of coding at intermediate nodes of the network is mostly investigated for the multicast scenario. In Chapter 3 we looked at the capacity region of broadcast problems over wireless erasure networks and provide inner and outer bounds. However, as we saw in Chapter 2, in a general setting destinations can demand different subsets of information messages, and there can be multiple copies of an information message available in the network. It is important to explore the benefits of coding for general network problems in wired and wireless networks.

The first step in analyzing the performance of coding over wireless networks is the modeling of the interference present in these networks. One possible approach is to investigate models that incorporate the interference in the network layer rather than the physical layer, which deserves further investigation. For instance, in the wireless erasure network model introduced in this chapter, a possible way to take into account the interference is through the erasure probabilities.

Another interesting topic in this area is identifying the critical side-information in a network setup. In Chapter 2 it was shown that by providing a certain sideinformation to the destination node we can achieve the min-cut upperbound. Presence of this side-information is critical for achieving the capacity. It would be interesting to find similar side-information for other types of networks.

• Communication, Control, and Computation in Wireless Ad-hoc Networks: Sensor and ad-hoc wireless networks are moving toward systems of interconnected devices (such as sensors, actuators, and controllers) that are capable of communicating among themselves and can perform computational tasks and estimation in a distributed manner.

In Chapter 5 of this thesis, we looked at estimation of a *single* dynamical process at a remote location that is connected to a sensor through an erasure wireless network. We characterize the optimal (minimum) steady-state error and its dependency on the parameters of the network. As a continuation of this work, the following important problem can be considered:

– What are optimal coding schemes for control and estimation of *multiple* dynamical systems in more general networks?

At a more general level, for a resource-efficient operation in these networks, a unified view of computation, communication, and control is required. Therefore, one fundamental challenge is to provide a theoretical framework that integrates various tasks and makes the analysis of different performance measures (and their trade-offs) possible. The work in Chapters 2, 5, and 6 is a first step towards this goal. The following important extensions of the results of these chapters merits further investigation:

- What is a practical model for analyzing power-efficient schemes that take into account the computation and communication power consumption simultaneously?
- How can one efficiently compute a function of the information available at different nodes in a distributed fashion in wireless networks?

Bibliography

- T. Rappaport, Wireless Communications: Principles and Practice, Prentice Hall, 2001.
- [2] C. E. Shannon, "A mathematical theory of communication," Bell Sys. Tech. J., vol. 27, pp. 379423, 1948.
- [3] L. Schulman, "Coding for interactive communication," *IEEE Trans. Info. Theory*, vol. 42, no. 6, pp. 1745–1756, 1996.
- [4] S. Tatikonda, Control under communication constraints, PhD thesis, MIT, 2000.
- [5] A. Sahai, "Evaluating channels for control: capacity reconsidered," Amer. Contr. Conf. Proc., 2000.
- [6] R. Gowaikar, A. F. Dana, R. Palanki, B. Hassibi, and M. Effros, "On the capacity of wireless erasure networks," *Proc. Intern. Symp. Info. Theory*, 2004.
- [7] A. F. Dana, R. Gowaikar, and B. Hassibi, "On the capacity region of broadcast over wireless erasure networks," Proc. 42nd Annual Allerton Conf. on Comm., Contr., Comp., Oct. 2004.
- [8] E. C. van der Meulen, "Three-terminal communication channels," Adv. Appl. Prob., vol. 3, 1971.
- [9] R. Ahlswede, N. Cai, S.-Y. R. Li, and R. W. Yeung, "Network information flow," *IEEE Trans. Info. Theory*, vol. 46, no. 4, pp. 1204–1216, 2000.

- [10] S.-Y. R. Li, R. W. Yeung, and N. Cai, "Linear network coding," IEEE Trans. Info. Theory, vol. 49, no. 2, pp. 371–381, 2003.
- [11] N. Cai and R. W. Yeung, "Secure network coding," Proc. of Intern. Symp. Info. Theory, 2002.
- [12] T. Ho, M. Médard, and R. Koetter, "An information theoretic view of network management," *Submitted to IEEE Trans. Info. Theory.*
- [13] T. M. Cover, "Broadcast channels," *IEEE Trans. Info. Theory*, vol. 18, no. 1, pp. 2–14, Jan. 1972.
- [14] T. M. Cover and A. A. El Gamal, "Capacity theorems for the relay channel," *IEEE Trans. Info. Theory*, vol. 25, pp. 572–584, Sep. 1979.
- [15] L.-L. Xie and P. R. Kumar, "An achievable rate for multiple level relay channel," *IEEE Trans. Info. Theory*, Submitted Nov. 2003.
- [16] G. Kramer, M. Gastpar, and P. Gupta, "Cooperative strategies and capacity theorems for relay networks," *IEEE Trans. Info. Theory*, Submitted Feb. 2004.
- [17] Y. Wu, P. A. Chou, and S.-Y. Kung, "Minimum-energy multicast in mobile ad hoc networks using network coding," *submitted to IEEE Trans. on Comm.*, Mar. 2004.
- [18] D. S. Lun, M. Médard, T. Ho., and R. Koetter, "Network coding with a cost criterion," Proc. Intern. Symp. Info. Theory and its Appl.(ISITA 2004), Oct. 2004.
- [19] Software Defined Radio Forum. [online] Available: http://www.sdrforum.org/.
- [20] M. Luby, "LT codes," Proc. 43rd Annual IEEE Symp. on Found. of Comp. Sci., Nov. 2002.

- [21] D. S. Lun, M. Médard, and M. Effros, "On coding for reliable communication over packet networks," Proc. 42nd Annual Allerton Conf. on Comm., Cont., Comp., Oct. 2004.
- [22] D. B. West, Introduction to graph theory, Prentice Hall, 1996.
- [23] R. W. Yeung, A first course in information theory, Kluwer Academic/Plenum Publishers, 2002.
- [24] V I. Levenshtein, "Binary codes capable of correcting deletions, insertion and reversals," *Soviet Physics-Doklady*, vol. 10, Feb. 1966.
- [25] J. D. Ullman, "On the capabilities of codes to correct synchronization errors," *IEEE Trans. Info. Theory*, vol. 13, Jan. 1967.
- [26] S. N. Diggavi and M. Grossglauser, "On transmission over deletion channels," Proc. 39th Annual Allerton Conf. on Comm., Cont., Comp., 2001.
- [27] R. Gowaikar, A. F. Dana, B. Hassibi, and M. Effros, "Practical schemes for wireless networks operation," *submitted to IEEE Trans. on Comm.*, 2004.
- [28] J. N. Laneman, D. N. C. Tse, and G. W. Wornell, "Cooperative diversity in wireless networks: efficient protocols and outage behavior," *IEEE Trans. Info. Theory*, Accepted for publication, Apr. 2004.
- [29] D. Slepian and J. K. Wolf, "Noiseless coding for correlated information sources," *IEEE Trans. Info. Theory*, Jul. 1973.
- [30] A. D. Wyner and J. Ziv, "The rate-distortion function for source coding with side-information at the receiver," *IEEE Trans. Info. Theory*, Jan. 1976.
- [31] A. F. Dana and B. Hassibi, "The capacity region of multiple input erasure broadcast channel," Proc. Intern. Symp. Info. Theory, Sep. 2005.

- [32] N. Ratnakar and G. Kramer, "The multicast capacity of deterministic relay networks with no interference," *IEEE Trans. Info. Theory*, vol. 52, no. 6, pp. 2425-2432, June 2006.
- [33] L. Song, R. W. Yeung, and N. Cai, "Zero-error network coding for acyclic networks," *IEEE Trans. Info. Theory*, vol. 49, no. 12, pp. 3129–3139, Dec. 2003.
- [34] J. Edmonds, "Submodular functions, matroids, and certain polyhedra," Combinatorial Optimization, 2001.
- [35] T. M. Cover,"Comment on broadcast channels," *IEEE Trans. Info. Theory*, vol.44, 1998.
- [36] A. A. Gamal," The Capacity of a class of broadcast channel," *IEEE Trans. Info. Theory*, vol. 25, 1979.
- [37] G. S. Poltyrev, "Carrying capacity for parallel broadcast channels with degraded components," *Probl. Peredachi. Inf.*, vol. 13, no. 2, pp. 23–35, 1977.
- [38] A. El Gamal,"Capacity of the product and sum of two reversely degraded broadcast channels," Prob. of Info. Transm., 1980.
- [39] S. Boucheron and M. R. Salamatian," About priority encoding transmission," *IEEE Trans. Info. Theory*, vol. 46, 2000.
- [40] P. Bergman, "Random coding theorem for broadcast channels with degraded components," *IEEE Trans. Info. Theory*, vol. 19, no. 3, pp. 197–207, Mar. 1973.
- [41] R. G. Gallager, "Capacity and coding for degraded broadcast channels," Probl. Pred. Inform., vol. 10, July-Sep. 1974.
- [42] M. Effros, M. Médard, T. Ho, S. Ray, D. Karger, and R. Koetter, "Linear network codes: a unified framework for source channel, and network coding," *invited paper to the DIMACS workshop on net. info. theory*, 2003.

- [43] K. Marton, "The capacity region of deterministic broadcast channels," IEEE Intern. Symp. Info. Theory, 1977.
- [44] K. Lih, "Majorization on finite partially ordered sets," Siam. Jour. Alg. Disc. Math., vol. 3, no. 4, Dec. 1982.
- [45] G. Kramer and S. A. Savari, "Edge-cut bounds on network coding rates," Jour. of Net. and Sys. Man., vol. 14, no. 1, pp. 49-67, Mar.2006.
- [46] R. W. Yeung, "A framework for linear information inequalities," *IEEE Trans. Info. Theory*, vol. 43, pp. 1924–1934, Nov. 1997.
- [47] Z. Zhang and R. W. Yeung, "On characterization of entropy function via information inequalities," *IEEE Trans. Info. Theory*, vol. 44, pp. 14401450, 1998.
- [48] M. S. Pinsker, "Capacity of noiseless broadcast channels," Probl. Inform. Transm., pp. 97-102, 1978.
- [49] K. Marton, "A coding theorem for the discrete memoryless broadcast channel," *IEEE Trans. Info. Theory*, vol. 25, no. 3, pp. 306311, 1979.
- [50] M. R. Aref, Information flow in relay networks, Ph.D. dissertation, Stanford Univ., Stanford, CA, Oct. 1980.
- [51] S. I. Gelfand and M. S. Pinsker, "Capacity of a broadcast channel with one deterministic component," *Probl. Pered. Inform.*, vol. 16, no. 1, pp. 24.34, March 1980.
- [52] Jr. L. R. Ford and D. R. Fulkerson, *Flows in networks*, Princeton University Press, Princeton, NJ, 1962.
- [53] T. M. Cover and J. A. Thomas, *Elements of information theory*. New York: Wiley, 1991.

- [54] R. Koetter and M. Médard, "An algebraic approach to network coding," *IEEE/ACM Trans. Net.*, vol. 11/5, pp. 782–795, Oct. 2003.
- [55] A. F. Dana, M. Sharif, B. Hassibi, and M. Effros, "Is broadcast plus multi-access optimal for Gaussian wireless networks with fading?," Proc. 37th Asilomar Conf. on Sig., Sys. and Comp.
- [56] A. F. Dana, R. Gowaikar, B. Hassibi, M. Effros, and M. Médard, "Should we break a wireless network into sub-networks?," *Proc. 41st Annual Allerton Conf.* on Comm., Cont., and Comp., 2003.
- [57] A. F. Dana, R. Gowaikar, R. Palanki, B. Hassibi, and M. Effros, "Capacity of wireless erasure networks," *IEEE Trans. Info. Theory*, vol. 52, no. 3, pp. 789–904 March 2006.
- [58] J. H. van Lint and R. M. Wilson, A Course in Combinatorics. Cambridge University Press, 2001.
- [59] J. S. Provan and M. O. Ball, "The complexity of counting cuts and of computing the probability that a graph is connected," *SIAM Jour. of Comp.*, vol. 12/4, pp. 384–393, 1983.
- [60] B. Schein and R. Gallager, "The gaussian parallel relay network," 2000.
- [61] L. G. Valiant, "The complexity of enumeration and reliability problems," SIAM Jour. of Comp., vol. 8, pp. 410–421, 1979.
- Τ. [62] H. L. Bodlaender Wolle, "A and note on the comof reliability problems," plexity network downloadable from http://www.cs.uu.nl/research/techreps/aut/thomasw.html, 2003.

- [63] G. Caire and D. Tuninetti, "The throughput of hybrid-arq protocols for the gaussian collision channel," *IEEE Trans. Info. Theory*, vol. 47, pp. 1971 – 1988, July 2001.
- [64] B. Warneke, M. Last, and K. S. J. Pister, "Smart Dust: Communicating with a cubic-millimeter computer," *IEEE Computer*, vol. 34, pp. 44-51, Jan. 2001.
- [65] Crossbow. [Online] Available: http://www.xbow.com/
- [66] D. Culler, D. Estrin, and M. Srivastava, "Overview of sensor networks," *IEEE Computer*, vol. 37, no. 8, pp. 41–49 Aug. 2004
- [67] I. Akyildiz, W. Su, Y. Sankarasubramaniam, and E. Cayirci, "A survey on wireless sensor networks," *IEEE Comm. Mag.*, vol. 40, no. 8, pp. 102-114, 2002.
- [68] L. Bushnell, "Special issue on networks and control," *IEEE Contr. Sys. Mag.*, vol. 21, no. 1, February 2001.
- [69] P. Antsaklis and J. Baillieul, "Special issue on networked control systems," *IEEE Trans. on Auto. Cont.*, vol. 49, no. 9, 2004.
- [70] A. Hassibi, S. P. Boyd, and J. P. How, "Control of asynchronous dynamical systems with rate constraints on events," in *Proc. IEEE Conf. Decision and Control*, Dec 1999, pp. 1345–1351.
- [71] P. J. Seiler, "Coordinated control of unmanned aerial vehicles," Ph.D. dissertation, University of California, Berkeley, 2002.
- [72] W. Zhang, M. S. Branicky, and S. M. Philips, "Stability of networked control systems," *IEEE Control System Magazine*, vol. 21, no. 1, pp. 84–89, Feb 2001.
- [73] Q. Ling and M. D. Lemmon, "Power spectral analysis of netwroked control systems with data droputs," *IEEE Transactions on Automatic control*, vol. 49, no. 6, pp. 955–960, June 2004.

- [74] J. Nilsson, "Real-time control systems with delays," Ph.D. dissertation, Department of Automatic Control, Lund Institute of Technology, 1998.
- [75] C. N. Hadjicostis and R. Touri, "Feedback control utilizing packet dropping network links," in *Proc. of the IEEE Conference on Decision and Control*, 2002.
- [76] B. Azimi-Sadjadi, "Stability of networked control systems in the presence of packet losses," in *Proceedings of 2003 IEEE Conference on Decision and Control*, Dec 2003.
- [77] B. Sinopoli, L. Schenato, M. Franceschetti, K. Poolla, and S. S. Sastry, "Time varying optimal control with packet losses," in 43rd IEEE Conference on Decision and Control (CDC), vol. 2, Bahamas, 2004, pp. 1938–1943.
- [78] O. C. Imer, S. Yuksel, and T. Basar, "Optimal control of dynamical systems over unreliable communication links," in NOLCOS 2004, Stuttgart, Germany, 2004.
- [79] B. Sinopoli, L. Schenato, M. Franceschetti, K. Poolla, M. Jordan, and S. Sastry, "Kalman filtering with intermittent observations," *IEEE Transactions on Automatic Control*, vol. 49, no. 9, pp. 1453–1464, Sep 2004.
- [80] V. Gupta, D. Spanos, B. Hassibi, and R. M. Murray, "Optimal LQG control across a packet-dropping link," Systems and Controls Letters, 2006, accepted.
- [81] J. Hespanha, P. Naghshtabrizi, and Y. Xu, "Networked control systems: Analysis and design," July 2005, submitted.
- [82] S. Tatikonda, "Some scaling properties of large scale distributed control systems," in *Proceedings of the 42nd IEEE Conference on Decision and Control*, vol. 3, 2003, pp. 3142–3147.

- [83] C. Robinson and P. R. Kumar, "Control over networks of unreliable links: Location of controllers, control laws and bounds on performance," in 45th IEEE Conference on Decision and Control, 2006, submitted.
- [84] B. Hassibi, A. H. Sayed, and T. Kailath, Indefinite-Quadratic estimation and control: a unified approach to H² and H[∞] theories, SIAM studies in applied and numerical mathematics, 1999.
- [85] W. J. Cook, W. H. Cunningham, W. R. Pulleyblank, and A. Schrijver, Combinatorial Optimization. John Wiley and Sons, New York, 1998.
- [86] K. Dohmen, "Inclusion-exclusion and network reliability," The Electronic Journal of Combinatorics, vol. 5, no. 1, p. Research paper 36, 1998.
- [87] T. Kailath, A. H. Sayed, and B. Hassibi, *Linear Estimation*. Prentice Hall, 2000.
- [88] M. Mohri, "Semiring frameworks and algorithms for shortest-distance problems," Journal of Automata, Languages and Combinatorics, pp. 321–350, 2002.
- [89] T. H. Cormen, C. E. Leiserson, R. L. Rivest, and C. Stein, Introduction to Algorithms. MIT Press and McGraw-Hill, 1990.
- [90] V. Gupta, A. F. Dana, J. P. Hespanha, and R. Murray, "Data transmission over wireless networks for estimation and control," submitted to *IEEE Trans.* on Auto. Cont., July 2006.
- [91] M. Gastpar and M. Vetterli, "On the asymptotic capacity of Gaussian relay networks," Proc. of '02 Intern. Symp. on Inform. Theory, p. 195, 2002.
- [92] P. Gupta and P.R. Kumar, "The capacity of wireless networks," *IEEE Trans. Inform. Theory*, vol. 46, no. 2, pp. 388–404, 2000.

- [93] P. Gupta and P. R. Kumar, "Towards an information theory of large networks: an achievable rate region," *IEEE Trans. on Inform. Theory*, pp. 1877–1894, Aug. 2003.
- [94] L-L. Xie and P. R. Kumar, "A network information theory for wireless communication: scaling laws and optimal operation," *IEEE Trans. on Inform. Theory*, pp. 748–767, May 2004.
- [95] O. Lévêque and E. Telatar, "Information theoretic upper bounds on the capacity of large extended ad hoc wireless networks," *IEEE Trans. on Inform. Theory*, pp. 858–865, Mar. 2005.
- [96] M. Grossglauser and D. Tse, "Mobility increases the capacity of ad-hoc wireless networks," *IEEE/ACM Trans. on Networking*, vol. 10, no. 4, pp. 477–486, Aug. 2002.
- [97] S. D. Sevetto, "On the feasibility of large-scale wireless sensor networks," Proc.
 40th Annual Allerton Conf. Comm., Cont., and Comp., Oct. 2002.
- [98] A. Scaglione and S. D. Servetto, "On the interdependence of routing and data compression in multi-hop sensor networks," Proc. 8th ACM Intern. Conf. on Mob. Comp. and Net. (MobiCom), Sept. 2002.
- [99] D. Marco, E. J. Duarte-Melo, M. Liu, and D. L. Neuhoff, "On the many-to-one transport capacity of a dense wireless sensor network and the compressibility of its data," *Info. Process. in Sens. Net. (IPSN '03)*, Apr. 2003.
- [100] A. Ephremides, "Energy concerns in wireless networks," *IEEE Wireless Comm.*, pp. 48–59, Aug. 2002.
- [101] S. Verdú, "Spectral efficiency in the wideband regime," IEEE Trans. Inform. Theory, vol. 48, pp. 1319–1343, 2002.

- [102] R. Knopp and P. A. Humblet, "Information capacity and power control in single-cell multiuser communications," in Proc. IEEE Inter. Conf. Comm., vol. 1, pp. 331-335, 1995.
- [103] D. N. C Tse and S. V. Hanly, "Multi-access fading channels- part I: polymatroid structure, optimal resource allocation and throughput capacity," *IEEE Trans.* on Inform. Theory, vol. 44, Issue 7, pp. 2796–2815, Nov. 1998.
- [104] W. R. Heinzelman, A. Chandrakasan, and H. Balakrishnan, "Energy-efficient communication protocol for wireless microsensor networks," *Proc. IEEE Intern. Conf. Sys. Sci.*, Jan. 2000.
- [105] B. Hassibi and A. Dana, "On the power efficiency of sensory and ad-hoc wireless networks," Proc. of the '02 Asil. Conf. on Sig., Sys., and Comp., Nov. 2002.
- [106] V. Rodoplu and T.H. Meng, "Minimum energy mobile wireless networks," *IEEE Journ. Sel. Area in Comm.*, vol. 17, no. 8, pp. 1333–1344, Aug. 1999.
- [107] J. Gomez, A.T. Campbell, M. Naghshineh, and C. Bisdikian, "Power-aware routing in wireless packet networks," *IEEE Intern. Workshop on Mobile Mult. Comm.*, pp. 380–383, Nov. 1999.
- [108] A. Lozano, A. Tulino, and S. Verdú, "Multi-antenna capacity in low-power regime," *IEEE Trans. Inform. Theory*, pp. 2527–2544, Oct. 2003.
- [109] E. Telatar, "Capacity of multi-antenna Gaussian channel," European Trans. Telecommunications, vol. 10, pp. 585–595, Nov. 1999.
- [110] M. Gastpar and M. Vetterli, "On the capacity of wireless networks: The relay case," Proc. IEEE Infocom 2002, June 2002.

- [111] Muriel Médard, "The effect upon channel capacity in wireless communications of perfect and imperfect knowledge of the channel," *IEEE Trans. Inform. Theory*, vol. 46, no. 3, pp. 933–946, May 2000.
- [112] R. A. Horn and C. R. Johnson, *Matrix analysis*, Cambridge University Press, New York, 1985.
- [113] A. Dana and B. Hassibi, "On the Power Efficiency of Sensory and Ad Hoc Wireless Networks," *IEEE Trans. Info. Theory*, vol.52, no. 7, pp. 2890–2914, July 2006.
- [114] A. F. Dana and B. Hassibi, "Power bandwidth trade-off for sensory and adhoc wireless networks," *IEEE International Symposium on Information The*ory,2004, July 2004, pp. 470.
- [115] L. Li and A. Goldsmith, "Capacity and optimal resource allocation for fading broadcast channels. I. ergodic capacity," *IEEE Trans. Info. Theory*, vol. 47, no. 3, pp. 1083–1102, 2001
- [116] H. Vishwanathan, S. Venkatesan, and H. Huang, "Downlink capcity evaluation of cellular networks with known interference cancellation," *IEEE Jour. Selec. Areas. Commu.*, vol. 21, no. 5, pp. 802–811, June 2003.
- [117] G. Caire and S. Shamai (Shitz), "On the achievable throughput of a multiantenna Gaussian broadcast channel," *IEEE Trans. on Info. Theory*, vol. 49, no. 7, pp. 16911706, Jul. 2003.
- [118] H. Weingarten, Y. Steinberg, and S. Shamai, "The capacity region of the gaussian MIMO broadcast channel," in *Proc. of IEEE ISIT*, pp. 174, July 2004.
- [119] M.Costa, "Writing on dirty paper", *IEEE Trans. Inform. Theory*, vol. IT-29, pp. 439-441, May 1983.

- [120] N. Jindal, S. Vishwanath, and A. Goldsmith, "On the duality of Gaussian multiple-access and broadcast channels" *IEEE Trans. Info. Theory*, vol. 50, no. 5, pp. 768–783, 2004.
- [121] M. Sharif and B. Hassibi, "On the capacity of MIMO broadcast channels with partial side information," *IEEE Trans. Info. Theory*, vol. 51, no. 2, pp. 506–522, 2005.
- [122] N. Jindal and A. Goldsmith, "Capacity and optimal power allocation for fading broadcast channels with minimum rates, *IEEE Trans. Info. Theory*, vol. 49, no. 11, pp. 2895–2909, 2003.
- [123] M. Sharif, A. Dana and B. Hassibi, "Differentiated rate scheduling for Gaussian broadcast channels", Proceedings of the 2005 IEEE International Symposium on Information Theory
- [124] N. Jindal, "High SNR Analysis of MIMO Broadcast Channels," *IEEE Intern. Symp. on Info. Theory*, Sept. 2005.
- [125] M. Sharif and B. Hassibi, "Scaling laws of sum rate using time-sharing, DPC, and beamforming for MIMO broadcast channels", *Information Theory*, 2004. *ISIT 2004. Proceedings. International Symposium on*, 27 June-2 July 2004 Page(s):175
- [126] A. Lapidoth, S. Shamai and M.A. Wigger, "On the capacity of fading MIMO broadcast channels with imperfect transmitter side-information", Proceedings of the 43rd Allerton Conference on Communication, Control and Computing, September 2005
- [127] P. Viswanath, D. N. Tse, and R. Laroia, "Opportunistic beamforming using dumb antennas," *IEEE Trans. Inform.*, vol. 48, no. 6, pp. 1277–1294, June 2002.

- [128] H. A. David, Order Statistics, New York, Wiley, 1970.
- [129] I. S. Gradshteyn and I. M. Rsyzhik, Table of Integrals, Series, and Products, Academic Press Inc., London, 1965.
- [130] B. Sturmfels, Grobner Bases and Convex Polytopes, American Mathematical Society, Providence, R. I., 1996.
- [131] S. Boyd and L. Vanderberghe, Convex Optimization, Cambridge University Press, 2003.
- [132] P. Viswanath and D. N. Tse, "Sum capacity of the vector Gaussian broadcast channel and downlink-uplink duality," *IEEE Trans. Inform.*, vol. 49, no. 8, pp. 1912–1921, Aug. 2003.
- [133] S. Vishwanath, N. Jindal, and A. Goldsmith, "Duality, achievable rates and sum rate capacity of Gaussian MIMO broadcast channel," *IEEE Trans. Inform.*, vol. 49, no. 10, pp. 2658–2668, Oct. 2002.
- [134] W. Yu, and J. Cioffi, "Sum Capacity of Gaussian Vector Broadcast Channels," *IEEE Trans. on Info. Theory*, vol. 50, no. 9, pp.1875-1892, Sept. 2004.
- [135] H. Weingarten, Y. Steinberg, and S. Shamai (Shitz), "The capacity region of the Gaussian MIMO broadcast channel," *Proceedings of Conference on Information Sciences and Systems*, Mar. 2004.
- [136] W. Yu, T. Lan, "Transmitter Optimization for the Multi-Antenna Downlink with Per-Antenna Power Constraints," submitted to the IEEE Trans. on Signal Processing, Dec. 2005
- [137] W. Yu, "Uplink-Downlink Duality via Minimax Duality," IEEE Trans. on Info. Theory, vol. 52, no. 2, Feb. 2006.

- [138] S. Gelfand and M. Pinsker, "Coding for channel with random parameters," Probl, Control Info. Theory, vol.9, no. 1, pp. 19-31, Dec. 1980.
- [139] T. Y. Al-Naffouri, M. Sharif, and B. Hassibi, "How much does transmit correlation affect the sum-rate of MIMO downlink channels?," *submitted to IEEE Trans. on Wireless Comm*, 2006.
- [140] B. Hassibi and B.M. Hochwald, "How much training is needed in a multipleantenna wireless link?," *IEEE Trans. on Info. Theory*, vol.49, no.4, Apr. 2003, pp. 951-964.
- [141] A. F. Dana, M. Sharif, and B. Hassibi, "On the Capacity Region of Multi-Antenna Gaussian Broadcast Channels with Estimation Error," in preparation for submission.
- [142] M. Sharif, A. F. Dana, A. Vakili, and B. Hassibi, "Differentiated rate scheduling for the downlink of cellular system," to be submitted to IEEE Trans. on Comm, 2006.
- [143] B. Hochwald, B. Peel, and A. L. Swindlehurst, "A Vector perturbation technique for near capacity multiantenna multiuser communication-Part II: Perturbation," *IEEE Trans. Comm.*, no. 3, pp. 537–545, March 2005.
- [144] S. A. Jafar and A. J. Goldsmith, "Multiple-Antenna capacity in correlated Rayleigh fading with channel covariance information", *IEEE Transactions on Wireless Communications*, vol. 4, no. 3, pp. 990-997, May 2005.
- [145] T. Yoo and A. Goldsmith, "On the optimality of multi-antenna broadcast scheduling using zero-forcing beamforming" To appear: IEEE JSAC Special Issue on 4G Wireless Systems.

- [146] M. Kobayashi, G. Caire, and D. Gesbert, "Opportunistic beamforming vs. space time coding in a queued downlink," 14th IST Mobile and Wireless Communications Summit, June 19-23, 2005, Dresden, Germany.
- [147] T. Söderström and P. Stoica, System Identification. London: Prentice Hall, 1989.
- [148] L. Zheng and D. N. Tse, "Communications on the Grassman manifold: a geometric approach to the noncoherent multiple-antenna channel," *IEEE Trans. Info.*, vol. 48, no. 2, pp. 359–384, Feb. 2002.
- [149] B. Hassibi and T. L. Marzetta, "Multiple-antennas and isotropically random unitary inputs: the received signal density in closed form," *IEEE Trans. Info.*, vol. 48, no. 6, pp. 1473–1484, June 2002.
- [150] G. J. Foschini and M. J. Gans, "On limits of wireless communications in a fading environment when using multiple antennas," Wireless Personal Communications, vol. 6, pp. 311–335, Mar. 1998.
- [151] W. Yu and J. M. Cioffi, "Trellis precoding for the broadcast channel," in Proc. IEEE Glob. Comm. Conf., 2001.
- [152] R. Zamir, S. Shamai, and U. Erez, "Nested linear/lattice codes for structured multiterminal binning," *IEEE Trans. Info.*, vol. 48, no. 6, pp. 1250–1277, June 2002.
- [153] C. B. Peel, B. Hochwald, and A. L. Swindlehurst, "A vector perturbation technique for near capacity multi-antenna multi-user communication- Part I: Channel inversion and regularization," *IEEE Trans. Commu.*, vol. 53, no. 1, pp. 195–202, Jan. 2005.
- [154] G. Caire, G. Taricco, and E. Biglieri, "Optimum power allocation over fading channels," *IEEE Trans. Info.*, vol. 45, no. 5, pp. 1468–1489, July 1999.

[155] A. Ephremides and B. Hajek, "Information theory and communication networks: an unconsummated union," *IEEE Trans. Info.*, vol. 44, no. 10, pp. 2416–2434, Oct. 1998.